BEFORE THE HON'BLE PUNJAB STATE ELECTRICITY REGULATORY COMMISSION, CHANDIGARH

REVIEW PETITION NO. OF 2020

IN THE MATTER OF:

GVK Power (Goindwal Sahib) Limited

Petitioner

...

...

Versus

Punjab State Power Corporation Limited

Respondent

INDEX

S.No.	Particulars	Page Nos.
1.	Petition under Section 94(1)(f) of the Electricity Act, 2003 read with Regulation 64 of Punjab State Electricity Regulatory Commission (Conduct of Business) Regulations, 2005 seeking review of order dated 05.08.2020 in Petition No. 69 of 2017 and for Clarification of order dated 05.08.2020 along with supporting affidavit	1 - 21
2.	Annexure P – 1: Certified copy of the Order dated 05.08.2020 passed by this Hon'ble Commission in Petition No. 69 of 2017	22 – 58
3.	Annexure P – 2: Copy of the Amended and Restated PPA dated 26.05.2009 between GVK and PSPCL	59 – 212
4.	Annexure P – 3: Copy of the Order dated 07.09.2020 passed by this Hon'ble Commission in Petition No. 32 of 2019	213 – 275
5.	Annexure P – 4: Copy of the Order dated 17.09.2020 passed by this Hon'ble Commission in Petition No. 34 of 2019	276 - 338
6.	Annexure P – 5: Proof of payment in terms of Punjab State Electricity Regulatory Commission (Fee) Regulations 2005	339

GVK Power (Goindwal Sahib) Limited Petitioner

Through

S

J. Sagar Associates Advocates for the Petitioner B-303, 3rd Floor, Ansal Plaza Hudco Place, August Kranti Marg New Delhi – 110 049

Date: 21.10.2020 Place: Hyderabad

BEFORE THE HON'BLE PUNJAB STATE ELECTRICITY

REGULATORY COMMISSION, CHANDIGARH

REVIEW PETITION NO. OF 2020

IN THE MATTER OF:

GVK Power (Goindwal Sahib) Limited Plot No.10, Paigah House, Sardar Patel Road, Secunderabad – 500003

...Petitioner

Versus

Punjab State Power Corporation Limited The Mall, Patiala (Punjab)

...Respondent

PETITION UNDER SECTION 94(1)(f) OF THE ELECTRICITY ACT, 2003 READ WITH REGULATION 64 OF PUNJAB STATE ELECTRICITY REGULATORY COMMISSION (CONDUCT OF BUSINESS) REGULATIONS, 2005 SEEKING REVIEW OF ORDER DATED 05.08.2020 IN PETITION NO. 69 OF 2017 AND FOR CLARIFICATION OF ORDER DATED 05.08.2020

Most respectfully showeth:

I. CONSPECTUS

1. The Petitioner, GVK Power (Goindwal Sahib) Ltd. ("**Petitioner**" / "**GVK**") has filed the present review petition under Section 94(1)(f) of the Electricity Act, 2003 ("**Electricity Act**") read with Regulation 64 of Punjab State Electricity Regulatory Commission (Conduct of Business) Regulations, 2005 ("**PSERC CBR 2005**") seeking review of Order dated 05.08.2020 passed by this Hon'ble Commission in Petition No. 69 of 2017 ("**Order under Review**"). A certified copy of the Order dated 05.08.2020 passed by this Hon'ble Commission in Petition No. 69 of 2017 is annexed hereto and marked as **Annexure P-1**.

2. By way of the Order under Review, this Hon'ble Commission has *inter-alia* provisionally approved the Annual Fixed Charges ("**AFC**") for FY 2017-18, FY 2018-19 and FY 2019-20 ("**Control Period**").

3. The present petition has been filed seeking review of the Order dated 05.08.2020 on account of error apparent on the face of the record and for clarification, as under:

(i) The Interest & Finance Charges towards Long Term Loans have been

approved on actuals whereas the same ought to have been allowed on normative basis for the Control Period.

- (ii) The components of O&M Expenses have been wrongly computed as under:
 - (a) Employee Cost: Escalation of 4.86 % applied to GVK's employee cost for FY 2017-18 to determine the Employee Cost for FY 2018-19 instead of the actual expenditure incurred by GVK as per the audited accounts towards employees cost for FY 2018-19.
 - (b) R&M and A&G Expenses:
 - The basis for computation of baseline values may be clarified.
 - The provisional baseline values considered for computing R&M and A&G Expenses is erroneous as incorrect comparison has been made between GVK's Project to PSPCL's generating stations for the purpose of industry benchmarks. Such comparison is erroneous since accounts and accounting policies followed by GVK and PSPCL are different. Without such analysis, GVK's Project and PSPCL's generating stations cannot be compared.
 - (c) K-Factor: There is an error in the computation of "K-Factor" since baseline O&M Expenses for the Control Period have been wrongly computed.
- (iii) Interest on Working Capital
 - (a) There is an error in the computation on working capital in as much as the actual fuel charges paid by PSPCL to GVK have been considered, whereas working capital is to be computed on normative basis in terms of Regulation 34 read with Regulation 25.1 of the PSERC Tariff Regulations 2014.
 - (b) Since the component of Working Capital viz. Fuel Cost for two months has been erroneously determined, there is an error in the computation of 'Receivables equivalent to two months'.
- (iv) Depreciation: There is an error in the computation of depreciation, inasmuch a double adjustment on account of free hold land in computing the depreciation has been done. While arriving at the weighted average rate of depreciation as 4.77%, the rate of depreciation on free hold land was considered as 0%, However, in computing the depreciation, the value of land from the Average Gross Fixed Asset ("GFA") has been deducted, which has led to a double deduction. The depreciation allowed ought to be Rs 141.56

Crores.

II. DESCRIPTION OF PARTIES

4. The Petitioner, GVK Power (Goindwal Sahib) Limited, a company incorporated under the provisions of the Companies Act, 1956 on 04.12.1997. The registered office of the Appellant is Paigah House, 156-159 Sardar Patel Road, Secunderabad, 540003.

5. The Respondent, Punjab State Power Corporation Ltd ("**PSPCL**") is a distribution licensee, in terms of Section 2(17) of the Electricity Act, operating in the State of Punjab having its registered office The Mall, Patiala (Punjab).

III. FACTUAL BACKGROUND

6. On 26.05.2009, GVK executed the Amended and Restated Power Purchase Agreement ("*PPA*") with erstwhile Punjab State Electricity Board (now PSPCL), for supply of Contract Capacity (540 MW) from the Project. A copy of the Amended and Restated PPA dated 26.05.2009 between GVK and PSPCL is annexed hereto and marked as **Annexure P-2**.

7. On 16.04.2016, the Commercial Operation Date ("**COD**") of the Project was attained.

8. On 15.09.2017, GVK filed Petition No. 54 of 2017 before this Hon'ble Commission seeking approval of completed capital cost and determination of tariff for FY 2016-17.

9. On 29.12.2017, GVK filed Petition No. 69 of 2017 for determining tariff during the Control Period (2017-18 to 2019-2020) in terms of the PSERC (Terms and Conditions of Tariff) Regulations, 2014 ("**PSERC Tariff Regulations 2014**").

10. On 26.11.2019, GVK filed Petition No. 32 of 2019 for True up of Tariff for FY 2016-17 in terms of the Punjab State Electricity Regulatory Commission (Terms and Conditions for Determination of Tariff) Regulations, 2005 ("**PSERC Tariff Regulations 2005**").

11. On 23.12.2019, GVK filed Petition No. 34 of 2019 for Truing up of Tariff for FY 2017-18 and FY 2018-19 in terms of PSERC Tariff Regulations 2014.

12. On 17.01.2020, this Hon'ble Commission passed the final order in Petition No. 54 of 2017 and determined: -

(a) Completed capital cost of the Project as Rs 3058.37; and

(b) Annual Fixed Charges ("**AFC**") for FY 2016-17.

13. On 23.01.2020, GVK filed Appeal No. 41 of 2020 before the Hon'ble Appellate Tribunal for Electricity ("**Hon'ble Tribunal**") challenging the Order dated 17.01.2020 passed by this Hon'ble Commission in Petition No. 54 of 2017. The said appeal is pending adjudication before the Hon'ble Tribunal. The submissions in the present petition are without prejudice to GVK's rights and contentions in Appeal No. 41 of 2020.

14. On 05.08.2020, this Hon'ble Commission passed Order in Petition No. 69 of 2017.

15. On 07.09.2020, this Hon'ble Commission passed Order in Petition No. 32 of 2019 approving the AFC for FY 2016-17. A copy of the Order dated 07.09.2020 passed by this Hon'ble Commission in Petition No. 32 of 2019 is annexed hereto and marked as **Annexure P-3**.

16. On 17.09.2020, this Hon'ble Commission passed Order in Petition No. 34 of 2019 approving the AFC for FY 2017-18 and FY 2018-19. A copy of the Order dated 17.09.2020 passed by this Hon'ble Commission in Petition No. 34 of 2019 is annexed hereto and marked as **Annexure P-4**.

17. GVK reserves its right to challenge the above-mentioned orders in accordance with law. It is respectfully submitted that there are errors apparent on the face of the record in Order dated 05.08.2020 and the same is s contrary to the principles enshrined in the PSERC Tariff Regulations 2014. Accordingly, GVK has filed the present petition seeking review of Order dated 05.08.2020 as detailed in the subsequent paragraphs.

IV. LEGAL FRAMEWORK FOR REVIEW

18. The present petition has been filed in terms of Section 94(1)(f) of the Electricity Act read with Regulation 64 of the PSERC CBR 2005. Regulation 64 of PSERC CBR 2005 provides as under:

"64. Review of the decisions, directions and orders:-

(1) Any person aggrieved by a decision or order of the Commission, from which no appeal is preferred or allowed, and who, from the discovery of new and important matter or evidence which, after the exercise of due diligence, was not within his knowledge or could not be produced by him at the time when the decision/order was passed by the Commission or on account of some mistake or error apparent on the face of record, or for any other sufficient reason, may apply for review of such order within 60 days of the date of decision/ order of the Commission..."

19. In terms of per Order 47, Rule 1 of the Code of Civil Procedure, 1908 ("CPC"),

the power of review can be exercised on any of the following grounds:

- Discovery of new and important matter or evidence which, after the exercise of due diligence, was not within knowledge of the petitioner or could not be produced by him;
- (b) Mistake or error apparent on the face of the record;
- (c) Any other sufficient reason.
- 20. The Hon'ble Supreme Court in the following cases, with respect to exercise of review jurisdiction has observed as under:

(a) BCCI v. Netaji Cricket Club & Ors : (2005) 4 SCC 741

"89. Order 47, Rule 1 of the Code provides for filing an application for review. Such an application for review would be maintainable not only upon discovery of a new and important piece of evidence or when there exists an error apparent on the face of the record but also if the same is necessitated on account of some mistake or for any other sufficient reason.

90. Thus, a mistake on the part of the court which would include a mistake in the nature of the undertaking may also call for a review of the order. An application for review would also be maintainable if there exists sufficient reason therefore. What would constitute sufficient reason would depend on the facts and circumstances of the case. The words 'sufficient reason' in Order 47, Rule 1 of the Code is wide enough to include a misconception of fact or law by a court or even an Advocate. An application for review may be necessitated by way of invoking the doctrine "actus curiae neminem gravabit".'

(b) Thungabhadra Industries Ltd Vs. Govt of Andhra Pradesh : AIR 1964 SCC 1372

"11. We do not consider that this furnishes a suitable occasion for dealing with this difference exhaustively or in any great detail, but it would suffice for us to say that where without any elaborate argument one could point to the error and say here is a substantial point of law which stares one in the face, and there could reasonably be no two opinions entertained about it, a clear case of error apparent on the face of the record would be made out..."

21. Additionally, on certain issues, as detailed hereunder, GVK is seeking

issuance of clarifications from the Order under Review by this Hon'ble Commission.

V. GROUNDS WITH LEGAL SUBMISSIONS

22. It is submitted that there are errors apparent on the face of record in the Order under Review with respect to the calculation of the following components for determination of tariff for FY 2017-18 to FY 2019-20, which have been computed contrary to the PSERC Tariff Regulations 2014:

- (a) Interest on Loan
- (b) O&M Expenses
- (c) Interest on working capital

(d) Depreciation

It is submitted that there is error apparent on the face of the record since certain elements of tariff have been computed contrary to the PSERC Tariff Regulations 2014.

Re. Interest on Loan

23. It is submitted that in terms of Regulation 24 of the PSERC Tariff Regulations 2014, interest ought to be computed taking into account the opening gross normative loan on the completed capital cost approved this Hon'ble Commission and the actual rate of interest paid by GVK. GVK had computed the interest taking into account repayment towards outstanding loan amounts and applicable interest rates in line with the PSERC Tariff Regulations, 2014. The computation of interest on term loans is based on the following:

- (a) The opening gross normative loan on the Completed Capital Cost as approved by this Hon'ble Commission.
- (b) The rate of interest has been considered at the actual applicable interest rate which was at 13.22 % p.a., for FY 2017-18 and 13.22 % p.a. for FY 2018-19.
- (c) The repayment for the period FY 2017-18 to FY 2018-19 has been considered equal to the depreciation allowed for that year.

24. However, in the Order under Review, interest charges have been allowed at a much lower rate, as under:

"6.7 ... The interest on long term loan is calculated as under:

				(Rs. Crore)
Sr. No.	Particulars	FY 2017-18	FY 2018-19	FY 2019- 20
1.	Opening balance of loan	1999.59	1866.15	1723.94
2.	Add: Receipt of loan during the year	9.03	0.59	0.49
З.	Less: Repayment of loan during the year equivalent to the depreciation determined in Table no.19	142.47	142.80	142.83
4.	Closing balance of loan	1866.15	1723.94	1581.60
5.	Average Loan	1932.87	1795.05	1652.77
6.	Rate of interest	12.84%	12.84%	12.84%
7.	Interest Charges	248.18	230.48	212.22

Table No 23: Interest & Finance charges on Long Term Loans determined by the Commission for FY 2017-18 to FY 2019-20

6.8 As per the Annual Audited Accounts of GVK for FY 2017-18 and FY 2018-19 interest has been worked out to Rs. 573.57 Crore and Rs.636.29 Crore respectively on average Loans of Rs. 3567.86 Crore and Rs 3758.93 Crore for FY 2017-18 and FY 2018-19 respectively. 6.9 GVK has a interest liability of Rs. 313.91 Crore and Rs. 852.24 Crore towards interest accrued for FY 2017-18 and FY 2018-19 respectively. The interest actually paid on long term loans is worked out as under:

 Table No.24: Interest actually paid by GVK for FY 2017-18 and FY 2018-19

 (Def Grame)

				(Rs. Crore)
Sr. No		FY 2016-17	FY2017-18	FY 2018-19
1.	Interest charges as per annual audited accounts	448.37*	573.57	636.29
2.	Interest paid	221.95	259.66	97.96
З.	Closing balance of interest due but not paid as per annual audited accounts	226.42	313.91	852.24

* In the Annual Audited Accounts of GVK for FY 2016-17 interest charges were shown as Rs 451.91 Crore. But in the Annual Audited Accounts of FY 2017-18, the previous year figures of interest charges for FY 2016-17 has been rearranged/regrouped as Rs. 448.37 Crore and Rs 3.54 (451.91 - 448.37) Crore has been shown as Other Finance Charges. The necessary adjustments of Interests will be considered during the True up of FY 2016-17.

6.10 The Commission determined interest charges for FY 2017-18, FY 2018-19 and FY 2019-20 as under:

Table No. 25: Interest charges allowed by the Commission for FY 2017-18 to FY 2019-20

				(Rs. Crore)
Sr. No	Particulars	FY 2017- 18	FY 2018-19	FY 2019- 20
1	Interest determined as per table 23	248.18	230.48	212.22
2	Interest actually paid by GVK as per table 24	259.66	97.96	0.00
3	Interest Allowed	248.18	97.96	0.00

The balance amount of interest i.e. Rs.132.52 (230.48- 97.96) Crore and Rs. 212.22 Crore of FY 2017-18 and FY 2018-19 respectively will be considered in the year in which they will actually be paid by GVK."

25. It is submitted that this Hon'ble Commission in Order dated 17.01.2020 has worked out average long-term loan as Rs. 2070.22 Crores based on normative parameters in terms of PSERC Tariff Regulations 2005. It is submitted that once normative approach has been adopted by the Hon'ble Commission, there cannot be any further adjustments on the basis of actual expenses on account of it being less. This position of law has been upheld by the Hon'ble Tribunal in *Haryana Power Generation Co. Ltd. v. Haryana Electricity Regulatory Commission*, Judgment dated 31.07.2009 in Appeal No. 42 & 43 of 2008. The relevant portion is reproduced hereunder:

"34. ... In our opinion, once the State Commission adopts normative approach, it is neither in the interest of the long term development of the electricity industry in the State nor is a fair play to the appellant to deny

the benefits of the normative approach to the appellant. The very purpose of normative approach is that the parties are informed of the benchmarks beforehand and that if they are in a position to better the benchmarks, they are entitled to the benefits unless there is some unhealthy practice adopted by them. In the case before us, if the appellant is able to raise resources below the benchmark rates, it indicates efficiency on the part of the appellant for which it should be allowed benefit in terms of the norms. Otherwise, the purpose of normative approach would get defeated and the appellant may not remain adequately motivated to work with the desired efficiency. It is true that the consumers should not be burdened with unnecessary costs, but the same is equally applicable to the appellant when it is denied recovery of costs incurred by it if the same is not in line with the norms."

Evidently, as the average long-term loan has been calculated on normative basis, the interest paid by GVK ought to be considered on normative basis. Accordingly, computation of interest on actuals instead of normative loan amount as claimed by GVK, is contrary to the PSERC Tariff Regulations 2014, as well as the settled position of law as laid down by the Hon'ble Tribunal. The interest on loan approved in the Order under Review is based on actuals, the same is an error apparent as the same ought to have been considered and allowed on normative parameters.

26. It is submitted that this Hon'ble Commission has allowed interest on loan amounts computed on normative average loan and not on actuals in Order dated 03.09.2019 passed by this Hon'ble Commission in *Everest Power Pvt. Ltd. v. Punjab State Power Corporation Ltd. & Anr.*, Petition No. 23 of 2017, as under:

"5.9 The Closing loan balance of Rs. 395.84 crore was determined by the Commission in True-Up of FY 2016-17, which is considered as the opening loan balance for FY 2017-18. Asset addition of Rs.2.65 crore for FY 2017-18, Rs.1.51 crore for FY 2018-19 and Rs.4.31 crore for FY 2019- 20 has been approved in this Order. 70% of asset addition has been considered to be sourced from debt i.e. Rs.1.86 crore(2.65*70%) for FY 2017-18, Rs.1.06 crore(1.51*70%) for FY 2018-19 and Rs.3.02 crore(4.31*70%) for FY 2019-20 as normative loan. Repayment of loan equal to depreciation allowed has been considered on normative basis as per Regulation 24.3 of PSERC MYT Regulations, 2014. The Petitioner claimed the weighted average rate of interest 13.35% for FY 2017-18. The weighted average rate of interest 13.18% and @13.24% for FY 2018-19 and FY 2019-20 respectively. The interest on long term loans is calculated in the Table-2.18.

	Particulars	FY 2017-18	FY 2018-19	FY 2019-20
1.	Opening balance of loan	395.84	356.43	316.11
2.	(+)Loan addition during year	1.86	1.06	3.02
З.	(-): Repayment(normative) during the year	41.27	41.38	41.53
4.	Closing balance of loan	356.43	316.11	277.60
5.	Average Loan	376.14	336.27	296.86
6.	Weighted Average Rate interest on Loan (%)	13.35%	13.18%	13.24%

7.	Interest on Loan	50.21	44.32	39.30
----	------------------	-------	-------	-------

27. It is evident from the Order under Review that the Hon'ble Commission has taken the opening loan balance as Rs 1999.59 Crore for FY 2017-18, Rs 1866.16 for FY 2018-18 which is the normative loan amount determined by this Hon'ble Commission in Petition No. 54 of 2017. Once this Hon'ble Commission has computed a normative outstanding loan amount, the interest payable ought to also be considered on normative basis and allowed to GVK.

28. Accordingly, the interest (excluding finance charges) on long term loans that ought to be allowed by this Hon'ble Commission is as under:

(Rs. In Crores)

2017-18	2018-19	2019-20
255.53	236.81	217.50

The same is in line with the interest on long terms loan claimed by GVK in Petition No. 34 of 2019 (True Up for FY 2017-18 and FY 2018-19) and Petition No. 33 of 2020 (Annual Performance Review for FY 2019-20).

Re. Operation and Maintenance Expenses

29. In terms of Regulation 26 of the PSERC Tariff Regulations 2014, O&M Expenses are computed on the basis of (a) Employee Costs and (b) Repair and Maintenance Costs, Administrative and General Costs. Further, the baseline values for determination of O&M Costs are to be determined by this Hon'ble Commission in terms of Regulation 8 of the PSERC Tariff Regulations 2014.

30. Regulation 8 of the PSERC Tariff Regulations 2014, provides that baseline values for the control period shall be determined by the Hon'ble Commission *inter alia* based on:

- (a) Figures approved by the Hon'ble Commission in the past;
- (b) Latest audited accounts;
- (c) Estimates of the expected figures for the relevant year;
- (d) Industry standards and benchmarks.

31. It is submitted in the Order under Review, baselines values have been computed erroneously and contrary to Regulation 8 of the PSERC Tariff Regulations 2014 since two different set of baseline values have been used for calculating the Employee Expenses and R&M and A&G Expenses. For the purposes of calculating the Employee Cost, this Hon'ble Commission has considered the actual employee

cost for FY 2017-18 as the provisional baseline, whereas for calculating the R&M and A&G Expenses the baseline values have been determined based on audited accounts of GVK for FY 2016-17, FY 2017-18 and FY 2018-19 along with the industry benchmark. However, it is unclear as to the basis for determining the baseline cost.

A. Employee Costs

32. The findings of this Hon'ble Commission towards Employee Expenses in the Order under Review are as under:

"3.10 The actual "Other Employee Cost" as per annual audited accounts for FY 2017-18 is Rs. 9.26 Crore. Accordingly, the Commission considers Rs. 9.26 Crore as base "Other Employee Cost" for FY2017-18.

3.11 The Commission has considered the actual "other employee cost" for FY 2017-18 as the provisional base line for determining the other employee cost. Since the plant has achieved only 4% PLF during FY 2016-17 and has run for 349 days in that year, therefore taking FY 2016-17 as the baseline year is not feasible. The Employee Cost is to be determined as per Regulation 26.1 of PSERC MYT Regulations, 2014 (as amended from time to time). Relevant sections of Regulation 26 of MYT Regulations, 2014 are given in Para 3.7.

[...]

3.13 The Commission considers escalation of 4.86% and 4.60% for determining other employee cost for FY 2018-19 and FY 2019-20 respectively. The other employee cost provisionally approved by the Commission for 1st Control Period is as follows:

 Table No.10: Employee Expenses of GVK approved provisionally by the

 Commission for 1st Control Period (Rs. Crore)

Sr. No	Particulars	FY 2017-18	FY 2018-19	FY 2019-20
1.	Other Employee Cost of Previous year	9.26	9.26	9.71
2.	Escalation Factor		4.86%	4.60%
З.	Other Employee Cost	9.26	9.71	10.16
4.	Terminal Benefits	0.73	0.99	1.25
5.	Total Employee Cost	9.99	10.70	11.41

33. It is submitted that after having considered actual employee cost for FY 2017-18 as the baseline value, escalation of 4.86% and 4.60% has been considered for determining employee cost for FY 2018-19 and FY 2019-20 respectively. Such a computation constitutes an error apparent since:

(a) Using FY 2017-18 as baseline does not reflect the increase in number of employees in FY 2018-19 due to increase in plant operations. The same is evident from the fact that the PLF of the plant increased from 32% in FY 2017-18 to 52% in FY 2018-19

- (b) In terms of Regulation 8.1(b) of the PSERC Regulations 2014, the latest audited accounts of GVK ought to be considered. The audited accounts for FY 2018-19 were placed on record by GVK in Petition No. 34 of 2019 and the same were available with the Hon'ble Commission at the time of passing of the present Order under Review.
- (c) This Hon'ble Commission has itself observed in the Order under Review that using FY 2016-17 for baseline value would not be feasible as only 4% PLF was achieved as under:

"3.11 The Commission has considered the actual 'other employee cost' for FY 2017-18 as the provisional base line for determining the other employee cost. Since the plant has achieved only 4% PLF during FY 2016-17 and has run for 349 days in that year, therefore taking FY 2016-17 as the baseline year is not feasible. The Employee Cost is to be determined as per Regulation 26.1 of PSERC MYT Regulations, 2014 (as amended from time to time)..."

Similarly, using employee cost incurred by GVK in FY 2017-18 as a baseline to compute the employee cost for FY 2018-19 would not be correct since employee cost of FY 2018-19 does not reflect the increase in the number of employees.

34. As per the audited accounts, GVK has incurred the Employee Cost of Rs. 9.99 Crores for FY 2017-18 and Rs. 12.67 Crore for FY 2018-19. The increase in cost is due to increase in the number of employees in FY 2018-19 from 117 to 143 due to the increase in operation of the Project as compared to previous years.

35. It is submitted GVK is required to pay competitive remuneration to employees in order to match industry standards. Regulation 8.1(b) of PSERC Tariff Regulations 2014 provides that approval of employee costs shall be based in industry benchmarks. However, the increase in employee cost at the rate of 4.86% and 4.60% is not reflective of the industry standard. It appears that this Hon'ble Commission has considered the employee cost of PSPCL's generating stations as industry standard for computing the employee cost of GVK.

36. In this regard, it is pertinent to note that in terms of Note 6 to Regulation 26 of PSERC Tariff Regulations 2014, an exceptional increase in employee cost on account of pay revision etc. would be considered separately by the Hon'ble Commission. Pertinently, GVK is a private company and pay revisions, such as in government organisations / companies, are not applicable to it. Thus, GVK has to provide regular increments to employees, which in the year 2017-18 and FY 2018-19 was 15% Therefore, it is prayed that the Hon'ble Commission may allow

Employee Cost for FY 2017-18 and FY 2018-19 on actuals and the Order under Review may be modified in line with the aforesaid submissions.

37. Accordingly, the employee expenses as part of the O&M Expenses that ought to be allowed by this Hon'ble Commission based the audited accounts for FY 2018-9 and for FY 2019-20, are as under:

		Rs. In Crores
2018-19	2019-20	
12.67	15.12	

B. Repair & Maintenance and A&G Expenses

38. It is submitted that this Hon'ble Commission has considered the R&M and A&G expenses based on audited accounts of GVK for FY 2016-17 to FY 2018-19 along with industry benchmark for determining the baseline values, as under:

"3.21 The Commission has considered the R&M and A&G expenses based on audited accounts of GVK for FY 2016-17,FY 2017-18 & FY 2018-19 and the industry benchmark for determining the baseline values of R&M and A&G expenses FY 2017-18 as under:

Table No. 14: Determination of Baseline value of the R&M and A&Gexpenses based on the Annual Audited Accounts for FY 2016-17, 2017-18and FY 2018-19(Bs. Crore)

				(Rs. Crore)		
Sr. No	Particulars	FY 2016- 17	FY 2017- 18	FY 2018- 19	Provisional Base line value for FY 2017-18	
1	A&G Expenses					
1.	Insurance	2.68	6.02	2.84	2.76	
2.	Rent	0.03	0.03	0.06	0.05	
З.	Rates & Taxes	0.51	3.63	1.12	0.82	
4.	Legal & Professional Charges	3.41	4.67	9.34	3.00	
5.	Auditor's Remunerations: Statutory Audit	0.02	0.03	0.07	0.03	
	Tax Audit	0	0.02	0	0.02	
	Other Services	0	0.01	0.01	0.01	
6.	Provision for Diminution in value of Investment	0.28	0.12	0.00	0.00	
7.	Communication cost	0.22	0.21	0.08	0.08	
8.	Travelling expenses	0.28	0.75	0.98	0.63	
9.	Miscellaneous expenses	1.54	5.69	2.12	2.12	
10.	Inventory Written off	0	0.53	0.00	0.00	
11.	Contract Manpower	5.15	13.48	16.63	8.00	
12	Ash Handling Charges	0.77	3.12	3.10	0.00	
13.	Water drawl charges	0	1.21	0.00	0.00	
14.	Power & Fuel	6.93	13.33	4.36	4.36	
15.	Total	21.82	52.85	40.71	21.88	

11	R & M Expenses				
1.	Consumption of Stores & Spares	1.19	10.07	11.64	4.00
2.	Repair: Buildings	0	0.27	0.29	0.27
З.	Repair: Plant & Machinery	0.12	1.17	1.48	1.17
4.	Repair: Other Assets	0.29	0.6	1.06	0.60
5.	Total	1.60	12.11	14.47	6.04
	R&M and A&G Expenses	23.42	64.96	55.18	27.92

Table No. 16: R&M and A&G determined by the Commission for the 1stControl Period (FY 2017-18 to FY 2019-20)

				s. Crore)
Sr. No	Particulars	FY 2017-18	FY 2018-19	FY2019-20
1	Opening GFA	3058.37	3071.27	3072.11
2	Addition during the year	12.90	0.84	0.70
3	Closing GFA	3071.27	3072.11	3072.81
4	Average GFA	3064.82	3071.69	3072.46
5	K factor	0.909%	0.909%	0.909%
6	WPI Index (as per para 3.12)		4.28%	1.66%
7	K factor inflation adjusted (5x6)		0.00948	0.00924
8	R&M and A&G expenses after WPI increase (4x7)	27.86	29.12	28.39
9	Audit Fee	0.06	0.08	0.08
10	Total R&M and A&G expenses	27.92	29.20	28.47

39. Evidently, provisional baseline values have been computed comparing A&G and R&M expenses of PSPCL's generating stations with that of GVK's Project. It is submitted that such a comparison constitutes an error apparent since:

- (a) No detailed analysis or break up of R&M and A&G Costs incurred by PSPCL's Plants have been provided. In such a scenario it would not be appropriate to treat the same as an industry benchmark for computation of baseline values in terms of Regulation 8 of the PSERC Tariff Regulations 2014.
- (b) In order to determine the baseline values, a detailed comparative analysis of PSPCL's and GVK's accounting policies ought to be done. In the absence of such an analysis, comparison of two differently situated power plants is erroneous and without basis.

40. In any event, comparing PSPCL's generating station and GVK's Project on the basis of actual expenditure incurred towards R&M & A&G expense is misleading and any comparison should be based total O & M cost including employee cost, as GVK runs the plant through the support of outsourced manpower, which is cost effective and operationally efficient as it does not increase the fixed cost of the plant. Moreover, since the employees are outsourced, GVK has flexibility to scale up or scale down the cost based on the operation requirements. However in accounting

since these outsourced manpower are not employees of the company it is accounted under the account head contract manpower or legal and professional expenses, which comes under Administrative and General Expenses. In the event such a comparison between GVK and PSPCL were to be carried out, the ratio of the O&M Expenses as a percentage of the operating revenue of PSPCL's generating station would accurately reflect the industry standards and benchmarks in the State of Punjab. Such comparison for FY 2017-18 and 2018-19 is given hereunder:

Particulars	FY 2017-18			FY 2018-19		
	GGSSTP	GHTP	GVK	GGSSTP	GHTP	GVK
ARR	1,161.55	1,337.38	1,504.50	1,036.95	1,215.40	1,809.60
Employee Cost	306.18	125.01	9.99	321.06	140.22	12.67
R&M and A&G						
Expenses	61.06	48.59	56.99	62.58	49.80	52.94
0& M Expenses	367.24	173.60	66.98	383.64	190.02	65.61
% of O & M expenses on ARR	31.62%	12.98%	4.45%	37.00%	15.63%	3.63%

0 & M Expenses comparison for FY 2017-18 and FY 2018-19

Note: Figures for O&M Expenses of PSPCL's Plants (GGSSTP and GHTP) have been taken from Tariff Order dated 27.05.2019 in Petition No. 2 of 2019 (for FY 2017-18) and Tariff Order dated 01.06.2020 in Petition No. 30 of 2019 (for FY 2018-19) Source : . <u>http://pserc.gov.in/pages/tariff-orders.html</u>

It is evident from the above that GVK's expenses as compared to PSPCL are significantly lower. Therefore, O&M Expenses of GVK for FY 2017-18 on actuals ought to be taken as baseline figures for arriving at K factor.

41. It is submitted that since there is an error in the computation of the baseline values for computation of R&M and A&G Expenses therefore the computation of K-Factor is also erroneous.

Re. Interest on Working Capital

42. In terms of Regulation 34 of the PSERC Tariff Regulations 2014, the components of working capital for calculating the interest of working capital, as under:

- Fuel Cost for 2 months corresponding to the normative annual plant availability factor;
- (b) Operation and maintenance Expenses for 1 month;
- (c) Maintenance spares @ 15% of the O&M Expenses;

(d) Receivables equivalent to 2 months of fixed and variable charges for sale of electricity calculated on the normative plant availability factor.

43. It is an admitted position that interest on working capital is payable on normative basis notwithstanding that the generating company has not taken working capital loan from any outside agency or has exceeded the working capital loan amount worked out on normative basis. In this regard, it is submitted that as the PSERC Tariff Regulations 2014 specifically provide that the interest on working capital is to be determined based on normative working capital. Once normative approach has been used to arrive at completed capital cost of the Project, then the same ought to continue.

44. However, by way of the Order under Review, the components of working capital have been limited to actuals instead of the normative parameters prescribed under Regulation 34 of PSERC Tariff Regulations, 2014, as under:

				(KS. Crore)
Sr. No.	Particulars	FY 2017-	FY 2018-	FY2019-
1.	Fuel Cost for two months	79.18	140.68	73.74
2.	Maintenance spares @15% of O&M	5.69	5.98	5.98
З.	O&M Expenses for one month	3.16	3.32	3.32
4.	Receivables for two months	180.12	242.19	132.42
5.	Total Working Capital	268.15	392.17	215.46
6.	Rate of Interest (%)	12.25%	12.25%	12.25%
7.	Interest on Working Capital	32.85	48.04	26.39

"[...] Table No.27: Interest on Working Capital approved by the Commission for the 1st Control Period (FY 2017-18 to FY 2019-20) (Rs. Crore)

45. It is submitted that the Hon'ble Tribunal in the following judgments has held that when the applicable Tariff Regulations provide that working capital is to be assessed on normative basis then the interest on working capital also has to be determined on normative working capital calculated in accordance with the Regulations, as under:

- (a) *Reliance Infrastructure Ltd. v. Maharashtra Electricity Regulatory Commission* : 2009 ELR (APTEL) 0672 (Para 11 – 13)
- (b) Reliance Infrastructure Ltd. v. Maharashtra Electricity Regulatory Commission, Appeal No. 117 of 2008 (Judgment dated 28.08.2009) (Para 15 – 16)

- (c) DPSC Limited v. West Bengal Electricity Regulatory Commission, Appeal No. 67 of 2009 (Judgment dated 06.09.2011) (Para 9.10 – 9.11)
- (d) Gujarat Urja Vikas Nigam Ltd. v. Gujarat Electricity Regulatory Commission & Anr., Appeal No. 37 of 2014 (Judgment dated 03.03.2015) (Para No. 31).

46. Accordingly, it is submitted that there is an error apparent on the face of the record inasmuch while computing Working Capital in the Order under Review, working capital has been considered based on actual expenditures instead of normative parameters. In this regard, it is noteworthy that:

- (a) Fuel Cost for two months: Regulation 34.1(a)(i) provides that fuel cost for 2 months shall correspond to the **normative annual plant availability factor**. However, GVK's claim has been limited to two months of the Energy Charges paid by PSPCL. Such an artificial capping, not only impacted Fuel Cost but also the component of 'Receivables for 2 months' as provided under Regulation 34.1(a)(iv).
- (b) O&M Expenses for one month and Maintenance Spares: As stated hereinabove, GVK has prayed for re-computation of O&M expenses in line with the PSERC Tariff Regulations 2014, accordingly, any change in O&M expenses will also impact the component of working capital.

47. Accordingly, the Interest on Working Capital that ought to be allowed by this Hon'ble Commission is as under:

S. No.	Particulars	FY 2017- 18	FY 2018- 19	FY 2019- 20
1	Fuel Cost – Primary Fuel & Secondary Fuel (for 2 months)	250.75	301.60	272.93
2	O&M expenses for one month	5.58	5.47	7.97
3	Maintenance spares (15% of O&M expenses)	10.05	9.84	383.62
4	Receivables (2 months of Fixed and Variable Cost based on Normative Annual Plant Availability Factor)	365.61	415.22	14.35
5	Total working capital	631.99	732.13	678.87
6	Rate of interest (p.a.)	12.25%	12.25%	12.25%
7	Interest on working capital	77.42	89.69	83.16

Re. Depreciation

48. Regulation 21 of the PSERC Tariff Regulations 2014 provides for calculation of Depreciation in respect of coal based thermal generating plants. This Hon'ble Commission by way of the Order under Review has calculated the depreciation as under:

			(R	ls. Crore)
Sr. No.	Particulars	FY 2017-18	FY 2018-19	FY2019-20
1.	Opening GFA	3058.37	3071.27	3072.11
2.	Addition during the year	12.90	0.84	0.70
З.	Closing GFA	3071.27	3072.11	3072.81
4.	Average GFA	3064.82	3071.69	3072.46
5.	Land & Land rights	96.75	96.75	96.75
6.	Average GFA(Net of Land & Land rights)	2968.07	2974.94	2975.71
7.	Rate of depreciation	4.80%	4.80%	4.80%
8.	Depreciation	142.47	142.80	142.83

"Table No. 19: Gross Fixed Assets (net of Land and land Rights) determined by the Commission for FY 2017-18 to FY 2019-20

49. However, it is submitted that there is a calculation error in computing the depreciation. It is submitted that the weighted average rate of depreciation that ought to be considered is 4.77%, which is worked out by considering depreciation of Free hold land as 0%.

Particulars	Asset Value	Rate of
		Depreciation (%)
Free hold Land	96.75	0.00
Plant & Equipment	2447.66	5.28
Capital Spares	12.90	5.28
Buildings	486.29	3.34
Computers	0.63	15.00
Office Equipment	1.12	6.33
Furniture & Fixtures	1.23	6.33
Vehicles	0.63	9.50
Computer Software	0.12	15.00
Right to use Railway	23.94	0.00
Line	23.74	0.00
Total	3071.27	4.77

50. Accordingly, the depreciation as calculated by GVK and which ought to be allowed by this Hon'ble Commission, is as under:

Particulars	Rs Crs
Opening Capital Cost	3058.37
Less :- Undischarged Liability	-
Opening Capital Cost excluding undischarged liability	3058.37
Add: - Additional Capitalization during the	12.90

year	
Less: - Undischarged liabilities included in above capitalization	-
Add: - Liabilities discharged during the year	-
Closing Capital Cost	3017.27
Average Capital Cost	3064.82
Free Hold Land	96.75
Rate of Depreciation	4.77%
Depreciation (as per Company) (Average Capital Cost * Rate of Depreciation)	146.17
Depreciation (as per Company) ((Average Capital Cost- Freehold Land) * Rate of Depreciation))	141.56

51. It is submitted that once the weighted average rate has been arrived at by considering the Freehold land at 0% depreciation, there should not be further deductions of Freehold land from the Average Capital Cost. Therefore, there has been a double deduction of the value of land in the computation of depreciation and the same constitutes an error apparent and mistake in the calculation of depreciation.

52. In these circumstances, it is most respectfully submitted that the facts and circumstances of the present case warrants intervention of this Hon'ble Commission by exercising its power to review its Order dated 05.08.2020 and grant relief to GVK as sought herein. Non-intervention by this Hon'ble Commission would cause substantial harm/ prejudice to GVK and the same would not be in line with the principles of '*Actus Curiae Neminen Gravabit*' i.e. no person should be prejudiced by the act of courts.

53. It is submitted that the above facts and findings clearly demonstrate the errors apparent on the face of the record. The present Review Petition is being filed bona fide and in the interest of justice. No appeal has been preferred by GVK challenging the Order Under Review.

54. GVK craves leave of this Hon'ble Commission to produce such additional material as may be considered necessary for proper adjudication of this Review Petition and to refer to rely upon its submissions/ documents as made/ filed in the Original Petition.

55. The certified copy of the Order under Review was received by GVK on 21.08.2020. As per Regulation 64 of the PSERC CBR 2005, a review petition may be filed within 60 days. Hence the present petition is within the prescribed limitation period.

56. GVK has paid the requisite fee of Rs. 10,000/- as per the Punjab State Electricity Regulatory Commission (Fee) Regulations, 2005, on 05.10.2020. A copy of the transaction statement of the fee paid to this Hon'ble Commission is annexed hereto and marked as **Annexure P-5**.

PRAYER

57. In view of the above and in the interest of justice, it is most respectfully prayed that this Hon'ble Commission may be pleased to:-

- (a) Review/modify the Order dated 05.08.2020 in Petition No. 69 of 2017 in terms of the submissions made paragraph 18 to 53 in the present Review Petition; and
- (b) Pass such other order(s) as this Hon'ble Commission may deem fit and proper in the facts and circumstances of the case.

GVK Power (Goindwal Sahib) Ltd.

Through:

June 1

J. Sagar Associates Advocates for the Petitioner B-303, 3rd Floor, Ansal Plaza, Hudco Place, August Kranti Marg New Delhi -110049

Place: Hyderabad Date: 21.10.2020

BEFORE THE PUNJAB STATE ELECTRICITY REGULATORY COMMISSION AT CHANDIGARH

PETITION NO. OF 2020

IN THE MATTER OF:

Petition on behalf of Petitioner / GVK Power (Goindwal Sahib) Ltd. under Section 94(1)(f) of the Electricity Act, 2003 read with Regulation 64 of Punjab State Electricity Regulatory Commission (Conduct of Business) Regulations, 2005 seeking review of Order dated 05.08.2020 passed in Petition No. 69 of 2017

IN THE MATTER OF:

led by Gov

GVK Power (Goindwal Sahib) Limited Paigarh House, 156 – 159, Sardar Patel Road, Secunderabad – 540 003

...Petitioner

Versus

Punjab State Power Corporation Limited The Mall, Patiala (Punjab)

...Respondent

AFFIDAVIT

I, P. Rama Mohana Rao, son of P. Krishna Murthy, aged about 61 years, resident of E-308, First Lane, Czech Colony, Sanath Nagar, Hyderabad - 500018, working as Assistant Vice President (Finance) with the Petitioner Company, do solemnly affirm and state as follows:-

1. I say that I am duly authorized and competent to affirm this Affidavit for and on behalf of the GVK Power (Goindwal Sahib) Ltd. and I am acquainted with the facts and circumstances of the present case. I state that I have read and understood the contents of the accompanying Petition.

2. I state that the facts stated in the accompanying Petition are true and correct to the best of my knowledge based on the records maintained by the Petitioner and that the legal submissions made therein are based upon information received by me and believed to be true. The present Petition has been drafted pursuant to my instructions and its contents are true and correct.

I state that the Annexures, if any, annexed to the Petitioner are true copies of prespective originals.

For GVK POWER (GOINDWAL SAHIB) LTD.

Authorised Signatory

4. I say that no similar petition, writ petition, suit or appeal regarding the matter in respect of which the present Petition has been preferred or is pending before any Court or any other authority.

For GVK POWER (GOINDWAL SAHIB) LTD.

() ∧ √ Authorised Signatory

DEPONENT

VERIFICATION

.

I, the deponent above named, do hereby verify that the contents of my above affidavit are true and correct, no part of it is false and nothing material has been concealed therefrom.

Verified at on this day of , 2020. For GVK POWER (GOINDWAL SAHIB) LTD. QAN Authorised Signatory ATTESTED 21/10/2020 M DEPONENT Mohd Hameed Khan AF B.A.B.Ed., LL.B ADVOCATE & NOTARY H.No. 1-8-495/19/C, Vikar Nagar, Near Prakash Nagar Begumpet, Secunderabad, Hyd Cell: 9391025702ed by Go

ANNEXURE P-1

22

PUNJAB STATE ELECTRICITY REGULATORY COMMISSION Site No.3, Block B, Sector 18- A Chandigarh

Petition No. 69 of 2017

In the matter of : Petition for determination of Tariff for the Control Period (FY 2017-18 to 2019-20) under Section 62 and 86 of the Electricity Act, 2003 read with (a) Punjab State Electricity Regulatory Commission (Terms & Conditions of Determination of Generation, Transmission, Wheeling and Retail Supply Tariff) Regulations, 2014 (b) Amended and Restated Power Purchase Agreement dated 26.05.2009 executed between Petitioner (Goindwal Sahib) Limited and Punjab State Electricity Board).

AND

In the matter of: GVK Power (Goindwal Sahib) Limited, Paigah House, 156-159, Sardar Patel Road, Secunderabad-540003.

Petitioner

Versus Punjab State Power Corporation Limited, The Mall, Patiala

Respondent

To,

- Chief Engineer, ARR/TR, Punjab State Power Corporation Limited, Shed No. F-4, Shakti Vihar, Patiala.
- 2 GVK Power (Goindwal Sahib) Limited Limited) Paigah House, 156-159, Sardar Patel Road, Secunderabad-540003.

No. PSERC/Reg. 1 3759

Dated: 21.08.2020

A copy of the Order dated 05.08.2020 passed by the Commission in Petition No. 69 of 2017 is enclosed herewith.

y. Registrar

DA/As above

Petition No. 6902017

PUNJAB STATE ELECTRICITY REGULATORY COMMISSION 20 SITE NO. 3, BLOCK B, SECTOR 18-A MADHYA MARG, CHANDIGARH

Petition No. 69 of 2017 Date of Order: 05.08.2020

Petition for determination of Tariff for the Control Period (FY 2017-18 to 2019-20) under Section 62 and 86 of the Electricity Act, 2003 read with (a) Punjab State Electricity Regulatory Commission (Terms & Conditions for Determination of Generation, Transmission, Wheeling and Retail Supply Tariff) Regulations, 2014; and (b) Amended and Restated Power Purchase Agreement dated 26.05.2009 executed between Petitioner (Goindwal Sahib) Limited and Punjab State Electricity Board).

AND کې In the matter of: GVK Power (Goindwal Sahib) Limited, Paigah House, 156-159, Sardar Patel Road, Secunderabad-540003.

.. Petitioner

Versus Punjab State Power Corporation Limited, The Mall, Patiala (Punjab)

.. Respondent

Present:	Ms. Kusumjit Sidhu, Chairperson
	Sh. S.S. Sarna, Member
	Ms. Anjuli Chandra, Member

ORDER

- 1.1 GVK Power (Goindwal Sahib) Limited (GVK) has developed a 2 X 270 MW Coal Based Thermal Power Project at Goindwal Sahib Village, Tarn Taran District, Punjab and entered into an Amended and Restated Power Purchase Agreement (PPA) on 26.05.2009 with Punjab State Power Corporation Limited (PSPCL) for the supply of power from the said project. The present petition has been filed for determination of Tariff for the Control Period, FY 2017-18 to 2019-20. GVK has prayed to:
 - a) Condone the delay in filing of the instant petition.
 - b) Admit the present petition.
 - Allow the Annual Fixed Cost for the Control Period, Rs. 1053.81
 Crore for FY 2017-18, Rs. 1041.75 Crore for FY 2018-19 and
 Rs. 1023.87 Crore for FY 2019-20.
 - d) Allow the annual Energy Charge for the Control Period Rs. 1298.35 Crore for EY 2017-18, Rs. 1298.87 for FY 2018-19 and Rs. 1363.55 Crore for FY 2019-20.



Petition No. 69 of 2017

- The delay was condoned and the petition was admitted vide order date 241.2 25.01.2018 directing PSPCL to file its reply to the petition, GVK was directed to issue a public notice inviting suggestions/objections from the public. The public notice could not be published as per the timelines given in the order dated 25.01.2018 and GVK was directed vide order dated 12.03.2018 to publish the public notice inviting suggestions/objections from the public and the petition was fixed for hearing as well as public hearing on 25.04.2018. The public notice was published on 15.03.2018 in Daily Post (English), Ajit (Punjabi) and in Ajit Samachar (Hindi) on 16.03.2018. The petition was taken up for hearing on 25.04.2018 wherein the Id. Counsel for PSPCL stated that the petitioner has filed petition no. 54 of 2017 for determination of tariff for FY 2016-17 and the decision of the Commission in that petition will have a bearing in this petition. PSPCL was directed to file a short reply by way of an affidavit and GVK was directed to file on affidavit the details of actual expenses incurred in FY 2017-18.
 - 1.3 PSPCL filed its reply vide memo no. 6923, dated 21.05.2018 submitting that the details inter alia regarding approved capital cost, primary and secondary fuel cost and actual financial statements of the petitioner company up till FY 2017-18, are necessary details for the purpose of adjudication of the present petition without which the process of tariff determination under the present petition cannot commence. GVK had filed petition no. 70 of 2017 for approval of Capital Investment Plan and Business Plan for MYT control period in which the Commission vide order dated 01.10.2019 ordered that GVK is required to submit the actual Capital Expenditure alongwith audited accounts of FY 2017-18 and 2018-19 and accordingly ordered that GVK is required to submit a revised petition for determination of AFC etc.
 - 1.4 The Commission, vide order dated 07.02.2020, directed GVK to provide details as per annexures enclosed with the order on monthly basis for the Control Period FY 2017-18, 2018-19, and 2019-20 for working fuel cost/energy charges . PSPCL was directed to provide details of the payments made to GVK on monthly basis with regard to energy charges including details of fuel cost of three years of the control period FY 2017-18,2018-19 and 2019-20 clearly segregating the payments made to GVK on the basis of weighted average cost of the coal to PSPCL's generating stations and



25

payments on the basis of Shakti Coal. In reference to the order dated 07.02.2020, GVK submitted information vide affidavit dated 11.03.2020 and PSPCL submitted information vide memo no. 5291 dated 11.03.2020. Vide order dated 13.03.2020 GVK and PSPCL were directed to file their respective reply to the said information by 16.03.2020. PSPCL has submitted vide memo no. 5321, dated 16.03.2020 that the details submitted by GVK have been tallied and are in order.

1.5 The petitioner prayed that pending final adjudication of petition No. 69 of 2017, the tariff determined by the Commission vide order dated 17.01.2020 in petition No. 54 of 2017 be continued. The Commission vide Order dated 20.03.2020 decided that the tariff determined for FY 2016-17 in petition No. 54 of 2017 shall be applicable provisionally for MYT Control Period FY 2017-18 to FY 2019-20 subject to the final Order of the Commission in petition No. 69 of 2017. During the hearing on 24.06.2020 the Ld. Counsel for the parties requested to submit their comprehensive written submissions and the Commission as a last opportunity allowed the same to be filed by 01.07.2020. After hearing the parties order was reserved vide Order dated 26.06.2020 and GVK submitted its additional affidavit vide memo No. 5523 dated 26.06.2020.

2.0 Capital Cost for FY 2017-18 to FY 2019-20

Õ

2.1 GVK has submitted that it has claimed completed capital cost of Rs. 4441 crore as on the Commercial Operation Date i.e. 16.04.2016 for determination of tariff for FY 2016-17 to FY 2018-19. The summary of additional capital expenditure as claimed by GVK for FY 2017-18 to FY 2019-20 is as follows:

Sr. No.	Head	FY 2017-18	FY 2018-19	FY 2019-20	Total
1	Buildings	57	0	0	57
2	Plant & Machinery - Initial capital spares - Balance BOP works -Balance BTG works	66 53 4	0 0 0	0 0 0	66 53 4
3	Topping of Roads	50	0	0	50
4	Total :	230	0	0	230

Table No. 1: Additional capitalisation during the 1st Control Period (Rs. Crore)



Petition No. 69 of 2017 26

2.2 GVK requested the Commission to consider the above amount of Rs.230 Crore towards additional capitalization under various heads and approve the Capital Investment Plan including the spill-over of Capital expenditure after the date of Commercial Operation Date ("COD"). GVK also submitted that the above estimates are indicative only and actual expenses may vary. GVK further requested the Commission that in the event that it is unable to incur the said additional capital cost of Rs. 230 Crore in FY 2017-18, the same or the balance amount be allowed to be incurred in FY 2018-19.

Commission's Analysis

- 2.3 The Commission vide its Order dated 17.01.2020 in petition no. 54 of 2017 has approved the capital cost of the project after prudence check as Rs. 3058.37 Crore as on the date of commissioning of the GVK Power Plant i.e. 16.04.2016 (GVK's appeal against this order is pending in APTEL).
- 2.4 The Commission has provisionally approved Capital Investment of Rs 14.44 Crore vide its Order dated 30.7.2020 in Petition No. 70 of 2017 on the basis of actual expenditure incurred during the Control Period as reflected in their Annual Audited Accounts submitted in the said petition as under:

Table No. 2: Capital Investment provisionally approved by the Commission during each year of the 1st Control Period

Sr. No.	Particulars	FY 2017-18	FY 2018-19	FY 2019-20	Rs. Crore) Total
1.	BTG Spares	, 12.90	Nil	Nil	12.90
2.	Coal Testing Laboratory	 Nil 	0.84	0.70	1.54
3.	Total	12.90	0.84	0.70	14.44

2.5 GVK has not claimed any additional capital expenditure from 17.04.2016 to 31.03.2017. Therefore, considering the capital cost of Rs. 3058.37 Crore as on 31.03.2017 and the additional capital expenditure of Rs. 12.90 Crore during FY 2017-18, the total provisional capital cost as on 31.03.2018 works out Rs. 3071.27 crore for FY 2017-18.

Similarly, considering the capital cost of the project as Rs. 3071.27 crore as on 01.04.2018 and additional capital expenditure of Rs. 0.84 crore during FY



27

2018-19, the total provisional capital cost as on 31.03.2019 works out Rs. 3072.11 crore for FY 2018-19.

Further, considering the capital cost of Rs. 3072.11 crore as on 01.04.2019 and additional capital expenditure of Rs. 0.70 crore during FY 2019-20, the total provisional capital cost as on 31.03.2020 works out to Rs. 3072.81 crore.

Accordingly, the Commission approves the following Capital cost for FY 2017-18 to FY 2019-20:

Table No. 3: Capital cost provisionally approved by the Commission for FY 2017-18 to FY 2019-20

			ίλ.	(Rs. Crore)
0.11.	Head	FY 2017-18	FY 2018-19	FY 2019-20
Sr.No.		3058.37	3071.27	3072.11
1	Capital cost as on 1st April of the year	12.90	0.84	0.70
2.	Additional capital expenditure during the	12.00		
	year Capital cost as on 31 st March of the year	3071.27	3072.11	3072.81
3.	Capital cost as on or materies die j			

The Commission provisionally allows additional capitalization of Rs. 12.90 Crore, Rs. 0.84 Crore and Rs. 0.70 Crore for FY 2017-18, FY2018-19 and FY 2019-20 respectively equal to the capital expenditure allowed provisionally for FY 2017-18, FY2018-19 and FY 2019-20.

3.0 OPERATION & MAINTENANCE EXPENSES:

GVK's Submission

C)

0

3.1 GVK submitted that Regulation 26 of the PSERC Tariff Regulations, 2014 along with its subsequent amendment vide 03.02.2016 provide for the Operation & Maintenance expenses. Further, Regulation 8.1 and 8.2 of PSERC MYT Regulations, 2014 specifies the baseline values. The baseline values for the Control Period are to be determined by this Commission based on estimates of the expected figures for the relevant year, industry bench mark / norms and other factors etc. However, since, no baseline values were approved by the Commission, GVK has adopted the baseline values for the O&M expenses as provided in the CERC Regulations.

Employee Cost:

3.2 GVK submitted that the employee cost includes salaries payable to employees, allowances and other terminal benefits.



Petition No. 69 of 2017

3.3 GVK further submitted that the escalation Index has been computed in terms of the PSERC MYT Regulations, 2014 considering the WPI Index (available till March, 2017) and CPI Index (available till March, 2017), as under:

Period	FY 2015-16	FY 2016-17	Increase / Decrease
CPI Index (April-March)	265.00	275.92	4.119 %
WPI Index (April-March)	176.675	183.20	3.693 %
CPI: WPI Index (50:50)			3,906 %

Table No. 4: Computation of Escalation Index

- 3.4 GVK submitted that the Inflation factor for the Control Period as per PSERC Tariff Regulations, 2014 has been computed as 3.906%. GVK further submitted that the inflation factor is on the lower side considering the average rise in salaries and other expenses. In order to retain employees with experience, GVK has to pay competitive remuneration to employees in order to match industry standards. Considering the fact that Employee Expenses is the most critical component in the overall operational expenditure, a 3.906% hike in Employee Expenses is insufficient to maintain salaries even at industry average for the Control Period.
- 3.5 GVK has estimated the Employee Cost for FY 2017-18 of Rs.15.58 Crore with an annual escalation of 15% in Employee Cost for the next two year of the Control Period. Accordingly, the Employee Cost for the Control Period is as under: -

	*		<u>(r</u>	(s. Grore)
~		Control p	period (Proje	ections)
Sr.		FY	FY	FY
No.	Particulars	2017-18	2018-19	2019-20
1.	Salaries and Wages	11.75	13.51	15.54
2.	Staff Welfare expenses	1.25	1.44	1.65
3.	Apprentice & Other Training	0.62	0.72	0.83
	Expenses			
4.	Sub Total (A)	13.62	15.67	18,02
	Contribution to Terminal Benefits			
1.	Contributions to Provident and	0.94	1.08	1.24
	other funds			
2.	Gratuity	0.39	0.45	0.52
З.	E.L. Encashment	0.62	0.72	0.83
4.	Sub Total (B)	1.95	2.25	2.59
	Total Employee cost (A+B)	15.58	17.92	20.61

Table No 5: Employee Cost claimed by GVK for FY 2017-18 to FY 2019-20



6

Commission's Analysis

- 3.6 The baseline values of O&M expenses are to be determined as per Regulations 8(1) of PSERC MYT Regulation-2014, which states as under:
 "8.1 Baseline values
 - a) The baseline values for the control period shall be determined by Commission and the projections for the control period shall be based on these figures.
 - b) The baseline values shall be inter-alia based on figures approved by the Commission in the past, latest audited accounts, estimates of the expected figures for the relevant year, industry benchmarks/norms and other factors considered appropriate by the Commission."
- 3.7 Regulation 26 of PSERC (Terms and Conditions of Gen. Transmission, wheeling and Retail Supply Tariff) Regulations, 2014 (Amended vide No. PSERC/Reg./111 dated 03.02.2016) provides for determination of O&M expenses and note-9 thereunder provides for the method for determination of unfunded liability. The Commission determines O&M expenses as per Regulation 26 of PSERC MYT Regulations, 2014, as under:

"26.1. The O&M expenses for the nth year of the Control Period shall be approved based on the formula shown below:

 $O&M_n = (R&M_n + EMP_n + A&G_n) \times (1-X_n)$ Where,

- R&M_n Repair and Maintenance Costs of the Applicant for the nth year;
- $EMP_n Employee$ Cost of the Applicant for the nth year;
- A&G_n Administrative and General Costs of the Applicant for the nth year;

The above components shall be computed in the manner specified below:

(i) $R\&M_n + A\&G_n = K^*GFA^*(WPI_n / WPI_n - 1)$

Where,

64

- " 'K' is a constant (expressed in %) governing the relationship between R&M and A&G expenses and Gross Fixed Assets (GFA) for the nth year. The value of "K" will be specified by the Commission in the MYT order.
- 'GFA' is the average value of the Gross Fixed Assets of the nth year.
- 'WPIn' means the average (on monthly basis) of Wholesale



Price Index (all commodifies) over the year for the nth year.

(ii) $EMP_n = (EMP_n-1)^*(INDEX_n / INDEX_{n-1})$

- INDEX_n Inflation Factor to be used for indexing the Employee Cost.
- This will be a combination of the Consumer Price Index (CPI) and the Wholesale Price Index (WPI) of the year and shall be calculated as under:-

$INDEX_{n} = 0.50^{\circ}CPI_{n} + 0.50^{\circ}WPI_{n}$

'WPI_n' means the average rate (on monthly basis) of Wholesale Price Index (all commodities) over the year for the nth year.

'CPI_n' means the average rate (on monthly basis) of Consumer Price Index (Industrial workers) over the year for the nth year.

.

......

.

.

.

(iii) X_n is an efficiency factor for $_n$ th year

The value of X_n shall be determined by the Commission in its first MYT order for the Control Period."

Employee Cost

3.8 The employee cost is considered in two parts -Terminal benefits and other employee cost. In the Petition no 54 of 2017, the Commission has provisionally allowed Employee Cost as per Annual Audited Accounts for FY 2016-17 as under:

			(Rs. Crore)
	Actual Cost as	Claimed by	Approved by
Particulars	per audited	GVK	the
	annual accounts	(Projections)	Commission
Terminal Benefits	0.65	1.80	0.65
Other Employee cost	8.01	12.58	8.01
Total	8.66	14.38	8.66
	Terminal Benefits Other Employee cost	Particularsper audited annual accountsTerminal Benefits0.65Other Employee cost8.01	Particularsper audited annual accountsGVK (Projections)Terminal Benefits0.651.80Other Employee cost8.0112.58

Table No.6: Employee Cost of FY 2016-17 provisionally allowed

Employee Cost for FY 2017-18, FY 2018-19 and FY 2019-20

3.9 As per Annual Audited Accounts for FY 2017-18 and FY 2018-19, the actual Terminal benefits paid are Rs 0.73 Crore and Rs 0.99 Crore respectively. The Commission provisionally allows the actual Terminal benefit of Rs 0.73 Crore



and Rs. 0.99 Crore for FY 2017-18 and FY 2018-19 respectively. In view of the increasing trend in the actual Terminal benefits in FY 2017-18 and FY 2018-19, over their previous years, terminal benefits for FY 2019-20 have been provisionally worked out to Rs. 1.25 Crore. Accordingly, terminal benefits provisionally allowed by the Commission for 1st Control Period is as under:

Table No.7: Terminal Benefits provisionally allowed by the Commission for the 1st Control Period (FY 2017-18 to FY 2019-20)

[·	(Rs. Crore)
Sr. No	· Particulars	FY 2017-18		
1	Terminal benefits	0.73	0.99 -,	1.25

3.10 The actual 'Other Employee Cost' as per annual audited accounts for FY
 2017-18 is Rs. 9.26 Crore. Accordingly, the Commission considers Rs. 9.26
 Crore as base 'Other Employee Cost' for FY2017-18.

C Z

(13

- 3.11 The Commission has considered the actual 'other employee cost' for FY 2017-18 as the provisional base line for determining the other employee cost. Since the plant has achieved only 4% PLF during FY 2016-17 and has run for 349 days in that year, therefore taking FY 2016-17 as the baseline year is not feasible. The Employee Cost is to be determined as per Regulation 26.1 of PSERC MYT Regulations, 2014 (as amended from time to time). Relevant sections of Regulation 26 of MYT Regulations, 2014 are given in Para 3.7.
- 3.12 Accordingly, the INDEXn for FY 2018-19 and FY 2019-20 has been calculated as under:

Table No. 8: WPI and CPI Increase considered for FY 2018-19

Sr. No.	Particulars	FY 2017-18	FY 2018-19	Increase (%)
1	CPI	284.42	299.92	5.45
2	WPI	114.88	119.79	4.28

INDEX n/INDEX n-1 = (0.5*5.45) + (0.5*4.28) = 4.86%

Table No. 9: WPI and CPI Increase considered for FY 2019-20

Sr. No.		FY 2018-19		Increase (%)
1	CPI	299.92	322.50	7.53
2	WPI	119.79	121.78	1.66

INDEX n/INDEX n-1 = (0.5*7.53) + (0.5*1.66) = 4.5963%



Petition No. 69 of 20**372**

 $\xi_{1,j}$

3.13 The Commission considers escalation of 4.86% and 4.60% for determining other employee cost for FY 2018-19 and FY 2019-20 respectively. The other employee cost provisionally approved by the Commission for 1st Control Period is as follows:

Table No.10: Employed	Expenses of GVK approved provisionally by	
the Comm	ission for 1 st Control Period (Rs. Crore)	

Sr. No		FY 2017-18		
1.	Other Employee Cost of Previous year		1	F1 2019-20
- 1	Escalation Factor		4.86%	9.71
	Other Employee Cost	9.26	9.71	4.60%
4.	Terminal Benefits	0.73	0,99	1.10
	Total Employee Cost	9 9 9	10.70	11.41

Repair and Maintenance and A&G Expenses GVK's Submission

3.14 GVK submitted that the R&M and A&G expenses are linked to the K factor and WPI Index, "K" being the constant (expressed in percentage) governing the relationship between R&M and A&G expenses and Gross Fixed Assets. For computing the K factor, GVK has estimated the R&M expenses and A&G expenses by deducting the projected Employee Cost for FY 2017-18 from the total O&M expenses as per the CERC Regulations for FY 2017-18. The details are as under: -

Table No.11: Computation of	of K for the 1 st	Control period (Re	(roro)

С В I		
Sr. No.	Particulars	FY 2017-18
1.	Total O&M expenses for the FY 2017-18 as per CERC Regulations	154.98
2.	Less : Employee Cost for the FY 2017-18	15.58
3.	R&M and A&G costs	139.40
4.	Opening GFA	4441
5.	Closing GFA	4671
6.	Average GFA	4556
7.	R&M and A&G Exp. as % of GFA	3.06 %
8	K factor	3.06 %

3.15 In view of the above, GVK prayed the Commission to approve the K Factor as mentioned above. GVK has considered the said K factor (as 3.06%) for FY 2017-18.



3.16 GVK submitted that the increase in WPI Index works out to 3.69% (available actual till March, 2017) as per the PSERC Tariff Regulations, 2014. Hence, for the purpose of R&M and A&G Expenses, GVK has considered the escalation Index of 3.69% (i.e., average of increase in WPI from FY 2015-16 to FY 2016-17 as per latest data available). Accordingly, GVK has projected combined R&M and A&G expenses as under:

Sr No	Particulars	FY2017-18	FY2018-19	FY2019-20
1	Opening FGA	4441	4671	4671
2	Additional Capitalization	230	·, 0	0
3	Closing GFA	4671	4671	4671
4	Average GFA	4556	4671	4671
5	K Factor (adjusted for WPI index)	3.06%	3.17%	3.29%
6	Escalation factor	3.693%	3.693%	3.693%
7	R&M and A&G Expenses	139.40	148.19	153.67
8.	Employee Cost	15.58	17.92	20.61
9.	O&M Expenses	154.98	166.11	174.28

Table No. 12: O&M Expenses projected by GVK (Rs. Crore)

3.17 GVK submitted that since this is the first Control Period of operation of the Project, it has considered efficiency factor as nil while computing the O&M expenses. GVK prayed the Commission to approve the same. Accordingly, GVK prayed the Commission to determine the K factor and allow the O&M expenses as detailed above in line with the provisions of PSERC Tariff Regulations, 2014.

Commissions' Analysis

1

Ű.

- 3.18 R&M and A&G expenses are part of O&M expenses. The Commission has provisionally approved R&M and A&G expenses as Rs. 23.14 Crore (from 17.4.2016 to 31.3.2017) for FY 2016-17 in Petition no 54 of 2017. The opening value of approved Gross Fixed Assets as on 1.04.2017 is Rs. 3058.37 Crore as determined in the para 25 of Petition no 54 of 2017.
- 3.19 GVK in their annual audited accounts for FY 2016-17, FY 2017-18 and FY 2018-19 have shown other expenses including R&M and A&G expenses as Rs. 23.42 Crore, Rs.232.63 Crore and Rs.130.67 Crore respectively. It has been observed that expenses relating to R&M and A&G are Rs. 23.42 Crore, Rs. 64.96 Crore and Rs 55.18 Crore for FY 2016-17, FY 2017-18 and FY 2018-19 respectively. On scruting of FY 2016-17, FY 2017-18 and FY



Petition No. 69 of 2017

expenses are abnormally high for FY 2017-18 as compared to FY 2016-17 and FY 2018-19. These are:

- Insurance: GVK has incurred expenses of Rs. 6.02 Crore in FY 2017-18 as compared to Rs. 2.68 Crore, and Rs. 2.84 Crore on Insurance in FY 2016-17 and FY 2018-19 respectively.
- Rates and Taxes: GVK has incurred expenses of Rs. 3.63 Crore in FY 2017-18 as compared to Rs. 0.51 Crore and Rs. 1.12 Crore on Rates and taxes in FY 2016-17 and FY 2018-19 respectively.
- iii) Power and Fuel: GVK has incurred expenses of Rs. 13.33 Crore in FY 2017-18 as compared to Rs. 6.93 Crore and Rs. 4.36 Crore on power and fuel in FY 2016-17 and FY 2018-19 respectively.
- iv) Provision for Diminution in value of Investment: In the order dated 17.1.2020 in Petition 54 of 2017,the Commission has noted that GVK had diverted the funds meant for capital expenditure out of the loan taken from financial institutions by investment in Mutual Funds. The Commission has neither considered interest paid on loans used for such other investments nor income earned from such other business/investment as part of the capital expenditure.
- v) Miscellaneous Expenses: GVK has incurred expenditure of Rs. 5.69
 Crore in FY 2017-18 as compared to Rs. 1.54 Crore and Rs. 2.12
 Crore on miscellaneous expenses in FY 2016-17 and FY 2018-19
 respectively.
- vi) Legal and Professional charges: GVK has incurred legal and professional charges of Rs 4.67 Crore and Rs.9.34 Crore during FY 2017-18 and FY 2018-19 respectively as compared to Rs 3.41 Crore in FY 2016-17.
- vii) GVK has achieved PLF of approximate 4%, 32% and 52% during FY 2016-17, FY 2017-18 and Rs 2018-19 respectively. In this regard the following has been observed:
 - Contract Manpower: GVK has shown expenses on contract manpower as Rs 5.15 Crore, Rs. 13.48 Crore and Rs 16.63 Crore for FY 2016-17, FY 2017-18 and FY 2018-19 respectively as against employee cost at Rs 8.66 Crore, Rs 9.99 Crore and



12

í.

Rs 12.67 Crore for FY 2016-17, FY 2017-18 and Rs 2018-19 respectively.

- Consumption of stores: GVK has incurred Rs 10.07 Crore on consumption of stores and spares in FY 2017-18 as against Rs.
 1.19 Crore in FY 2016-17 which is abnormally high considering the fact that the plant is new and has only been operational since 16.4.2016.
- Ash Handling Charges: GVK has claimed ash handling charges of Rs 0.77 Crore, Rs.3.12 Crore and Rs.3.10 Crore for FY 2016-17, FY 2017-18 and FY 2018-19 respectively, but has not shown any income from sale of ash. These charges cannot be allowed considering labour, power and water charges are already being allowed separately.
- 3.20 The Commission observes that, the A&G and R&M expenses of GHTP and GGSSTP (Thermal Plants) of PSPCL are lower than GVK inspite of being older and of higher capacity as shown in the table below:

Table No. 13: Comparison of A&G and R&M expenses of PSPCL ownedThermal Plants and GVK for FY 2017-18 and FY 2018-19

(Rs. Crore)

Sr.	Particulars	FY 2017-18			FY 2018-19		
No.		GHTP	GGSSTP	GVK	GHTP	GGSSTP	GVK
1	Capacity (MW)	920	840	540	920	840	540
2	A&G and R&M expenses	48.59	61.06	64.96	49,80	62.58	55.18

3.21 The Commission has considered the R&M and A&G expenses based on audited accounts of GVK for FY 2016-17,FY 2017-18 & FY 2018-19 and the industry benchmark for determining the baseline values of R&M and A&G expenses FY 2017-18 as under:


Table No. 14: Determination of Baseline value of the R&M and A&G exp36 es based on the Annual Audited Accounts for FY 2016-17, 2017-18 and FY 2018-19

(Rs. Crore)

k kj

THE OF THE STREET

			· } · · · · · · · · · · · · · · · · · ·		(Rs. Cror
Sr. No	Particulars	FY 2016-17	FY 2017-18	FY 2018-19	Provisiona Base line value for
1	A&G Expenses				FY 2017-18
1.	Insurance	2.68	6.02	x 0 (°	
<u>2.</u> 3.	Rent	0.03	0.02	2.84	2.76
<u> </u>	Rates & Taxes	0.51	3.63	1.12	0.05
4.	Legal & Professional Charges	3.41	4.67	9.34	0.82
5.	Auditor's Remunerations: Statutory Audit	0.02	0.03	0.07	3.00
	Tax Audit	0	0.03	<u>0.07</u>	0.03
6,	Other Services Provision for Diminution in value of	0	0.01	0.01	0.02
7.	Investment Communication cost	0.28	0.12	0.00	0.00
8.	Travelling expenses	0.22	0.21	0.08	0.08
9,	Miscellaneous expenses	0.28	0.75	0.98	0.63
10,	Inventory Written off	1.54	5.69	2.12	2.12
11,	Contract Manpower	0	0.53	0.00	0.00
12	Ash Handling Charges	5.15	13.48	16.63	8.00
13.	Water drawl charges	0.77	3.12	3.10	0.00
14.	Power & Fuel	0	1.21	0.00	0.00
5.	Total	6.93	13.33	4.36	4.36
11	R & M Expenses	21.82	52.85	40.71	21.88
1.	Consumption of Stores & Spares	· · ·	*		
2.	Repair: Buildings	1.19	10.07	11.64	4.00
3.	Repair: Plant & Machinery	0 10	0.27	0.29	0.27
1.	Repair: Other Assets	0.12	1.17	1.48	1.17
;.	Total i	0.29	0.6	1.06	0.60
	R&M and A&G Expenses	1.60 23.42	12.11	14.47	6.04
		~V.74	64.96	55.18	27.92

3.22 The Commission has provisionally approved investment/expenditure of Rs. 12.90 Crore, Rs. 0.84 Crore and Rs. 0.70 Crore for FY 2017-18, FY 2018-19 and FY 2019-20 respectively as per the Capital Investment Plan. Capitalization of assets is considered as an addition to Gross Fixed Assets based on the nature of the capital expenditure. The addition of capitalization/GFA during FY 2017-18, FY 2018-19 and FY 2019-20 is being



14

considered provisionally at Rs. 12.90 Crore, Rs. 0.84 Crore and Rs. 0.70 Crore respectively.

3.23 The Commission considers base R&M and A&G expenses as Rs 27.92 Crore for FY 2017-18 and determines the K factor for the 1st Control Period by as under:

Table No.15: R&M and A&G expenses based on K factor and indexation for FY2017-18 to FY 2019-20 determined by the Commission

(Rs. Crore)

Sr. No.	Particulars	Amount
1.	Opening GFA as on 1.4.2017	3058.37
2	Addition during the year	× 12.90
3.	Closing GFA as on 31.3.2018	3071.27
4.	Average GFA	3064.82
5.	R&M and A&G expenses	27.92
6	Less: Audit Fee	0.06
7	Base R&M and A& G expenses	27.86
8.	K factor	0.909%

3.24 Taking the K factor determined above, the Commission determines the R&M and A&G expenses for FY 2017-18 to FY 2019-20

Table No. 16: R&M and A&G determined by the Commission for the 1st ControlPeriod (FY 2017-18 to FY 2019-20)

(Rs. Crore)

C. N.	FN. 1 1	ENCODIE 40	EX 0010 10	510.10
Sr. No	Particulars	FY 2017-18	FY 2018-19	FY2019-20
1	Opening GFA	3058.37	3071.27	3072.11
2	Addition during the year	12.90	0.84	0.70
3	Closing GFA	3071.27	3072.11	3072.81
4	Average GFA	3064.82	3071.69	3072.46
5	K factor	0.909%	0.909%	0.909%
6	WPI Index (as per para 3.12)		4.28%	1.66%
7	K factor inflation adjusted (5x6)		0.00948	0.00924
8	R&M and A&G expenses after WPI increase (4x7)	27.86	29.12	28.39
9	Audit Fee	0.06	0.08	0.08
10	Total R&M and A&G expenses	27.92	29.20	28.47

^{3.25} The O&M expenses consisting of employee cost and R&M and A&G expenses as determined in para 3.13 and 3.24 above are approved as under:



i						(RS. Crore)
	Sr. No		FY		FY 2018-19	FY 2019-20
	1	Employee Cost		9.99	10.70	11 4 1
	2	R&M and A&G Expenses		27.92	29.20	28.47
	3	O&M Expenses		37.91	39.90	39.88

Table No.17: O&M expenses for FY 2017-18 to FY 2019-20

4.0 Depreciation

GVK's Submission:

- 4.1 GVK submitted that Regulation 21 of the PSERC Tariff Regulations, 2014, as amended by Notification dated 03.02.2016, provides for calculation of Depreciation in respect of Coal Based Thermal Generating Plants'.
- 4.2 Depreciation for Generation Assets has been calculated annually as per the straight line method over the useful life of the Asset at the rate of depreciation specified by the Central Electricity Regulatory Commission ("CERC") from time to time. Accordingly, GVK has computed the depreciation on the Gross Fixed Assets excluding land. The depreciation charges for the Control Period are given in the following table:

 Table No. 18: Depreciation Charges projected by GVK for the 1st Control Period

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

 .

<u> </u>			• •	
Sr.No	Particulars	FY 2017-18	FY 2018-19	FY2019-20
1.	Opening Gross Fixed Assets (Excluding Land Cost)	4259	4489	4489
2.	Additional Capitalization (Excluding Land Cost)	230	0	0
	Closing Gross Fixed Assets	4489	4489	4489
4.	Average Gross Fixed Assets	4374	4489	4489
	Rate of Depreciation	4.95%	4.93%	4.93%
6,	Depreciation Charges	216.31	221.31	221.31

4.3 Accordingly, GVK prayed to the Commission to allow Depreciation as detailed above in line with the provisions of the PSERC Tariff Regulations, 2014.

Commission's Analysis

4.4 As regards the Depreciation, Regulation 21 of PSERC MYT Regulations,2014 has been amended vide notification dated 03.02.2016 as under:

"21.1. The value base for the purpose of depreciation shall be the capital cost of the assets admitted by the Commission:



Provided that land other than the land held under lease and the land for reservoir in case of hydro generating station shall not be a depreciable asset and its cost shall be excluded from the capital cost while computing depreciable value of the asset;

Provided further that depreciation shall be calculated after deduction of consumer contributions, capital subsidies/ Government grants.

21.2. The cost of the asset shall include additional capitalization.

21.3. The cost shall include foreign currency funding converted to equivalent rupees at the exchange rate prevalent on the date when foreign currency shall actually be availed but not later than the date of commercial operation.

21.4. Depreciation for generation and transmission assets shall be calculated annually as per straight line method over the useful life of the asset at the rate of depreciation specified by the Central Electricity Regulatory Commission from time to time.

C 🌡

21.5. Depreciation for distribution assets and other assets not specified by CERC shall be at the rates notified by the Commission:

Provided that the total depreciation during the life of the asset shall not exceed 90% of the original cost;

Provided that the remaining depreciable value as on 31st March of the year closing after a period of 12 years from date of commercial operation/ put in use of the asset shall be spread over the balance useful life of the assets;

Provided further that in case of hydro generating stations, the salvage value shall be as provided in the agreement signed by the developers with the State Government for creation of the site.

21.6. Depreciation shall be chargeable from the first year of commercial operation/asset is put in use. In case of commercial operation of the asset/put in use of asset for part of the year, depreciation shall be charged on pro rata basis."

4.5 GVK in Petition no 34 of 2019 has claimed rate of depreciation as 4.80% for true up of FY 2017-18 and FY 2018-19 on the basis of Annual Audited Accounts. The Commission determines Gross Fixed Assets (net of land and land rights) and considers rate of depreciation as 4.80% for depreciation as under:



Petition No. 69 of 20/7

Table No. 19: Gross Fixed Assets (net of Land and land Rights) determined by the Commission for FY 2017-18 to FY 2019-20

				(Rs. Crore)
Sr. No.	Particulars	FY 2017-18	FY 2018-19	FY2019-20
1.	Opening GFA	3058.37	3071.27	3072.11
2.	Addition during the year	12.90	0.84	0.70
З.	Closing GFA	3071.27	3072.11	3072.81
4.	Average GFA	3064.82	3071.69	3072.46
5.	Land & Land rights	96.75	96.75	96.75
6.	Average GFA(Net of Land & Land rights)	2968.07	2974.94	2975.71
7.	Rate of depreciation	4.80%	4.80%	4.80%
8.	Depreciation	142.47	142.80	142.83

Accordingly, the Commission approves depreciation charges of Rs.142.47 Crore, Rs.142.80 Crore and Rs. 142.83 Crore for FY 2017-18, FY 2018-19 and FY 2019-20 respectively.

5.0 Return on Equity

GVK's Submission

- 5.1 GVK submitted that Regulations 19 & 20 of PSERC Tariff Regulations, 2014 provides for Debt Equity Ratio and Return on Equity.
- 5.2 GVK further submitted that it has infused an amount of Rs. 1,251 Crore as Equity. Accordingly, the Return on Equity has been calculated on the said amount for the present Control Period. In terms of the PSERC Tariff Regulations, 2014, the total equity to be considered for the determination of tariff and Return on Equity at the rate of 15.50% for each of the year during the Control Period as under:

				(Ks. Crore
Sr. No	Particulars	FY2017-18	FY 2018-19	FY2019-20
1.	Opening Capital Cost	4441	4773	4773
2.	Add: Margin money for working capital	102	0	0
3.	Additional Capitalization	230	0	0
4.	Closing Capital Cost	4773	4773	4773
5.	Equity (26.20%)	1251	1251	1251
6.	Rate of Return on Equity	15.50%	15.50%	15.50%
7.	Return on Equity	193.91	193.91	193.91

Table No. 20: Return on Equity projected by GVK

^{5.3} In view of the above, GVK prayed to allow the Return on Equity at the rate of 15.50% in accordance with the RSERC MXT Regulations, 2014.



N.

新設

Commission's Analysis:

5.4 Regulation 20 of PSERC MYT Regulations, 2014 provides for recovery of Return on Equity which is reproduced hereunder:

"20. RETURN ON EQUITY

Return on Equity shall be computed at the rate of 15.5% on the paid up equity capital determined in accordance with regulation 19:

Provided that assets funded by consumer contributions, capital subsidies/Govt. grants shall not form part of the capital base for the purpose of calculation of Return on Equity."

5.5 Regulation 19 of PSERC MYT Regulations, 2014 provides for Debt-Equity Ratio which is reproduced hereunder:

"19. DEBT EQUITY RATIO

19.1. Existing Projects - In case of the capital expenditure projects having Commercial Operation Date prior to the effective date, the debt-equity ratio shall be as allowed by the Commission for determination of tariff for the period prior to the effective date:

Provided that the Commission shall not consider the increase in equity as a result of revaluation of assets (including land) for the purpose of computing return on equity.

19.2. New Projects – For capital expenditure projects declared under commercial operation on or after the effective date:

a. A Normative debt-equity ratio of 70:30 shall be considered for the purpose of determination of Tariff;

b. In case the actual equity employed is in excess of 30%, the amount of equity for the purpose of tariff determination shall be limited to 30%, and the balance amount shall be considered as normative loan;

c. In case, the actual equity employed is less than 30%, the actual debtequity ratio shall be considered;

d. The premium, if any raised by the Applicant while issuing share capital and investment of internal accruals created out of free reserve, shall also be reckoned as paid up capital for the purpose of computing return on



equity subject to the normative debt-equity ratio of 70:30, provided such premium amount and

internal accruals are actually utilized for meeting capital expenditure of the Applicant's business.

19.3. Renovation and Modernization: Any approved capital expenditure incurred on Renovation and Modernization including the approval in the Capital Investment plan shall be considered to be financed at normative debt-equity ratio of 70:30. If the actual equity employed is less than 30% then the actual debt equity ratio shall be considered."

5.6

The equity approved by the Commission in its Order dated 17.01.2020 in Petition no 54 of 2017 is Rs. 917.51 Crore as on 31.03.2017 which is considered as opening balance for FY 2017-18. As per Regulation 20 of PSERC MYT Regulations, 2014, Debt Equity ratio of 70:30 has to be considered. The Commission is provisionally considering this debt equity ratio for working out the equity in the control period. Equity of Rs. 3.87 (30% of Rs.12.90 Crore of assets addition of FY 2017-18) Crore, Rs 0.25 (30% of Rs.0.84 Crore of asset addition of FY 2018-19) Crore and Rs 0.21 (30% of Rs.0.70 Crore of asset addition of FY 2019-20) Crore have been considered for determining return on equity for FY 2017-18,FY 2018-19 and FY 2019 -20 respectively. However, this shall be reviewed based on actual equity infusion in the Capital investment.

5.7 Thus, Return on Equity @ 15.50% on the average paid up equity capital is worked out as under:

Table No. 21: Return on Equity for FY 2017-18 to FY 2019-20 determined by the Commission

(Rs. Crore)

Ň.

Sr. No	Particulars	FY 2017-18	FY 2018-19	FY2019-20
1.	Opening Equity for the year	917.51	921.38	921.63
2.	Addition of Equity during the year	3.87	0.25	0.21
3.	Closing Equity for the year	921.38	921.63	921.84
4.	Average Equity for the year	919.45	921.51	921.74
5.	Rate of Return on Equity (%)	15.50%	15.50%	15.50%
6.	Return on Equity	142.51	142.83	142.87

Accordingly, the Commission approves Return on Equity of Rs. 142.51 Crore, Rs. 142.83 Crore and Rs. 142-87 Crore for FY 2017-18, FY 2018-19 and FY 2019-20 respectively.



6.0 INTEREST AND FINANCE CHARGES ON LONG TERM LOAN CAPITAL:

GVK's Submissions

Č

- 6.1 GVK submitted that Regulation 24 of the PSERC Tariff Regulations, 2014 provides for Interest and Finance Charges on Loan Capital.
- 6.2 GVK further submitted that the interest payable by it towards Long Term Loans has been calculated on the outstanding loan amounts and prevailing interest rates on the said amounts on the basis of the Completed Capital Cost of the Project as determined by this Commission. In furtherance of the same, GVK has considered the outstanding loan amounts as on 31.03.2017 for the opening balance for FY 2017-18. The loans to be availed during the Control Period, if any, have also been considered in the present Petition. The interest expenses have been computed taking into account repayment towards outstanding loan amounts and applicable interest rates in line with the PSERC Tariff Regulations, 2014.
- 6.3 GVK further submitted that in terms of the PSERC Tariff Regulations, 2014, the computation of interest on term loans is based on the following:
 - (a) The opening gross normative loan on the Completed Capital Cost as approved by the Commission.
 - (b) The weighted average rate of interest has been worked on the basis of the actual payment made which works out to 12.84 % p.a. for FY 2017-18, 12.84 % p.a. for FY 2018-19 and 12.84 % p.a. for FY 2019-20. The State Bank of India advance rate as on 01.04.2016 was 14.05% p.a.
 - (c) The repayment for the Control Period i.e., FY 2017-18 to FY 2019-20 has been considered equal to the depreciation allowed for that year.
- 6.4 The calculation of interest on long term loan is as given below:

Table No. 22: Interest on Long Term Loans projected by GVK for the 1 st
Control Period (FY 2017-18 to FY 2019-20)

<u>Sr. No</u>	Particulars	FY 2017-18	FY2018-19	FY2019-20
1.	Opening Capital Cost	4543	4773	4773
2.	Additional Capitalization	230	· 0	0
3.	Closing Capital Cost	4773	4773	4773



Sr. No	Particulars	FY 2017-18	FY2018-19	FY2019-20
4.	Gross Normative Loan on Opening Capital Cost (73.80%) (A)	3522	3522	3522
5.	Less: Cumulative Repayment (B)	202.59	418.90	640.25
6.	Net Loan Opening(A- B)=C	3319.41	3103.10	2881.75
7.	Less: Repayment during the year (D)	216.31	221.34	221.34
8.	Addition due to additional Capitalization during the year (E)	0	0	0
9.	Closing Loan Balance of Year (C-D+E)	3103.10	2881,75	2660.41
10.	Average Loan	3211.25	2992.42	2771.08
11.	Weighted Average Rate of Interest on Loan (p.a.)	12.84%	12.84%	12.84%
12.	Interest on Loan	412.26	384.16	355,75

6.5 Accordingly, GVK has prayed to allow the interest on Loan Capital as detailed above in accordance with the provisions of the PSERC Tariff Regulations, 2014.

Commission's Analysis

6.6 Regulation 24 of PSERC MYT Regulations, 2014 provides for Interest on Loan Capital, which is reproduced hereunder:

"24. INTEREST ON LOAN CAPITAL

24.1. For existing loan capital, interest and finance charges on loan capital shall be computed on the outstanding loans, duly taking into account the actual rate of interest and the schedule of repayment as per the terms and conditions of relevant agreements. The rate of interest shall be the actual rate of interest paid/payable (other than working capital loans) on loans by the licensee or the State Bank of India Advance Rate as on April 1 of the relevant year, whichever is less.

24.2. Interest and finance charges on the actual loan capital for new investments shall be computed on the loans, duly taking into account the actual rate of interest and the schedule of repayment as per the terms and conditions of relevant agreements. The rate of interest shall



6 A

be the actual rate of interest paid/payable (other than working capital loans) on loans by the licensee or the State Bank of India Advance Rate as on April 1 of the relevant year, whichever is less.

24.3. The repayment for each year of the tariff period shall be deemed to be equal to the depreciation allowed for the corresponding year. In case of de-capitalisation of assets, the repayment shall be adjusted by taking into account cumulative depreciation made to the extent of de-capitalisation.

24.4. The Commission shall allow obligatory taxes on interest, finance charges (including guarantee fee payable to the Government) and any exchange rate difference arising from foreign currency borrowings, as finance cost.

24.5. The interest on excess equity treated as loan shall be serviced at the weighted average interest rate of actual loan taken from the lenders."

6.7The Closing loan balance of Rs.1999.59 Crore as on 31.3.2017 was provisionally determined by the Commission in FY 2016-17 in Petition no 54 of 2017, which is considered as the opening loan balance for FY 2017-18. Asset addition of Rs.12.90 Crore for FY 2017-18, Rs.0.84 Crore for FY 2018-19 and 0.70 Crore for FY 2019-20 has been approved in Petition no 70 of 2017. 70% of asset addition has been considered to be sourced from debt i.e. Rs.9.03 (12.90*70%) Crore for FY 2017-18, Rs.0.59 (0.84*70%) Crore for FY 2018-19 and Rs.0.49 (0.70,70%) Crore as normative loan. Repayment of loan has been considered equal to depreciation allowed as per Regulation 24.3 of PSERC MYT Regulations, 2014. GVK has claimed the weighted average rate of interest 12.84% for 1st Control period which has been considered by the Commission for calculating interest on long term loan. The State Bank of India advance rate as on 1.04.2017 and 1.04.2018 was 13.85% and 13.45% respectively and since rates at which loan has been taken by GVK are less than SBI advance rate, the same have been considered to calculate the interest. The interest on long term loans is calculated as under:

)

(🌔



Table No 23: Interest & Finance charges on Long Term Loans determined bythe Commission for FY 2017-18 to FY 2019-20

Sr.	Particulars	······	1	(Rs. Crore
No. 1.	and a state of a state	FY 2017-18	FY 2018-19	
2.	Opening balance of loan Add: Receipt of loan during	1999.59	1866.15	1723.94
3.	<u>i ne year</u>	9.03	0.59	0.49
	Less: Repayment of loan during the year equivalent to the depreciation determined in Table no.19	142.47	142.80	142.83
<i>.</i>	Closing balance of loan Average Loan	1866.15	1723.94	1581.60
5.	Rate of interest	1932.87 12.84%	1795.05 12.84%	1652.77
•	Interest Charges	248.18	000	12.84%

6,8

As per the Annual Audited Accounts of GVK for FY 2017-18 and FY 2018-19 interest has been worked out to Rs. 573.57 Crore and Rs.636.29 Crore respectively on average Loans of Rs. 3567.86 Crore and Rs 3758.93 Crore for FY 2017-18 and FY 2018-19 respectively.

6.9 GVK has a interest liability of Rs. 313.91 Crore and Rs. 852.24 Crore towards interest accrued for FY 2017-18 and FY 2018-19 respectively. The interest actually paid on long term loans is worked out as under:

Table No.24: Interest actually paid by GVK for FY 2017-18 and FY 2018-.19

Sr. No	Farticulars	EV 2016 47	(Rs. Crore)
1.	audited accounts	448.37*	<u>573.57</u>	FY 2018-19 636.29
	Interest paid Closing balance of interest due but not paid as per annual audited accounts	221.95 226.42	259.66 313.91	97.96 852.24

* In the Annual Audited Accounts of GVK for FY 2016-17 interest charges were shown as Rs 451.91 Crore. But in the Annual Audited Accounts of FY 2017-18, the previous year figures of interest charges for FY 2016-17 has been rearranged/regrouped as Rs. 448.37 Crore and Rs 3.54 (451.91 - 448.37) Crore has been shown as Other Finance Charges. The necessary adjustments of Interest charges and Finance charges will be considered during the True up of FY 2016-17.

6.10 The Commission determined interest charges for FY 2017-18, FY 2018-19 and FY 2019-20 as under:



Sr. No	Particulars	FY 2017-18	FY 2018-19	FY 2019-20
1	Interest determined as per table 23	248.18	230.48	212.22
2	Interest actually paid by GVK as per table 24	259.66	97.96	0.00
3.	Interest Allowed	248.18	97. 96	0.00

Table No. 25: Interest charges allowed by the Commission for FY 2017-18 to FY 2019-20

The balance amount of interest i.e. Rs.132.52 (230.48- 97.96) Crore and Rs. 212.22 Crore of FY 2017-18 and FY 2018-19 respectively will be considered in the year in which they will actually be paid by GVK.

6.11 As per the Annual Audited Accounts for FY 2017-18 and FY 2018-19, finance charges amount to Rs. 3.44 Crore and 0.28 Crore on the average loan amount of Rs. 3567.86 Crore and Rs.3758.93 Crore respectively has been shown. Finance charges proportionately works out as Rs. 1.86 Crore and Rs 0.13 Crore on average loan of Rs 1930.65 Crore and Rs.1788.66 Crore for FY 2017-18 and FY 2018-19 respectively and the same are allowed. The Commission considers Finance charges for FY 2019-20 equivalent to the Finance Charges allowed for FY 2018-19 i.e. Rs. 0.13 Crore. This shall be reviewed when the annual audited accounts for FY 2019-20 are available. Thus, total interest and finance charges for FY 2017-18, FY 2018-19 and FY 2019-20 work out to Rs. 250.04 (248.18+1.86) Crore, Rs. 98.09 (97.96+ 0.13) Crore and Rs. 0.13 (0.00+0.13) Crore respectively.

Accordingly, the Commission allows interest and finance charges of Rs. 250.04 Crore, Rs. 98.09 Crore and Rs. 0.13 Crore for FY 2017-18, FY 2018-19 and FY 2019-20 respectively.

7.0 INTEREST ON WORKING CAPITAL:

GVK's Submissions

(1)

- 7.1 GVK submitted that Regulation 34 of the PSERC Tariff Regulations, 2014 provides for components of Interest on Working Capital in respect of Coal Based Thermal Generating Plants.
- 7.2 GVK further submitted that the weighted average rate of interest is computed at 12.25% p.a., for FY 2017-18, 12,25% p.a., for FY 2018-19 and 12.25%



p.a., for FY 2019-20. The State Bank of India advance rate as on 01.04.2016 was 14.05% p.a.

7.3 GVK has calculated the interest on working capital for MYT Control Period (i.e. FY 2017-18 to FY 2019-20) as per PSERC Tariff Regulations, 2014. Interest on working capital is projected for the Control Period by applying the rates as mentioned above on the components of Working Capital as given in the table below:

Table No. 26: Interest on Working Capital projected by GVK for FY2017-18 to FY 2019-20

·				(Rs. Crore)
Sr.	Particulars	FY 2017-18	FY 2018-19	FY 2019-20
No.			×.	
1.	Fuel Cost – Primary Fuel &	216.39	216.48	227.26
	Secondary Fuel (for 2 months)			the for 1 to a first by
2.	O&M expenses for one month	12.92	13.84	14.52
3.	Maintenance spares (15% of	1.94	2.08	2.18
· · · · · · · · · · · · · · · · · · ·	O&M expenses)			
4.	Receivables (2 months of Fixed	392.03	390.10	397.90
	and Variable Cost based on			
	Normative Annual Plant			
	Availability Factor)			
5.	Total working capital	623.27	622.50	641.86
6.	Rate of interest (p.a.)	12.25%	12.25%	12.25%
7.	Interest on working capital	76.35	76.26	78.63

- 7.4 Accordingly, GVK prayed the Commission to allow the interest on Working Capital as detailed above in line with the provisions of the PSERC Tariff Regulations, 2014.
 - Commission's Analysis: '
- 7.5 Regulation 34 of PSERC MYT Regulations, 2014 provides for Interest on working Capital which is reproduced hereunder:

"34.1. Components of Working Capital

- a. Coal-based Thermal Generating Plants: The Working Capital shall cover the following:
- *i.* Fuel Cost for 2 months corresponding to the normative annual plant availability factor;
- ii. Operation and maintenance (O&M) Expenses for 1 month;
- iii. Maintenance spares @ 15% of the O&M expenses;
- iv. Receivables equivalent to two (2) months of fixed and variable charges for sale of electricity calculated for the normative annual plant



26

availability factor.

b.

C."

7.6

Rate of Interest on Working Capital shall be as per Regulation 25.1(Amended vide No. PSERC/Reg./111 dated 03.02.2016). which is reproduced hereunder:

25.1 The rate of interest on working capital shall be equal to the weighted average rate of interest paid/ payable on loans by the licensee/generating company/SLDC or the State Bank of India Advance Rate as on April 1 of the relevant year, whichever is less. The interest on working capital shall be payable on normative basis notwithstanding that the licensee/generating company/SLDC has not taken working capital loan from any outside agency or has exceeded the working capital loan amount worked out on the normative figures.

- 7.7 As per PSERC Regulations, the rate of interest on working capital shall be equal to the weighted average rate of interest paid/ payable on loans by the generating company or the State Bank of India Advance rate as on April 1 of the relevant year, whichever is less. The interest on working capital is payable on normative basis notwithstanding that the generating company has not taken working capital loan from any outside agency or has exceeded the working capital loan amount worked out on normative basis.
- 7.8 The Weighted Average Rate of Interest has been considered for FY 2017-18 to FY 2019-20 @12.25% ås claimed by GVK in this petition. The State Bank of India advance rate as on 1.04.2017 and 1.04.2018 was 13.85% and 13.45% respectively.
- 7.9 Interest on working capital for MYT Control Period has been calculated as per PSERC MYT Regulations 2014. Interest on Working capital is projected for control period from FY 2017-18 to FY 2019-20 by applying the rate of interest of 12.25 % p.a. on components of Working capital i.e. (maintenance spares @ 15% of O&M expenses, O&M expenses for one month and Receivables @ 2 month Annual Fixed Cost) as given in table below.



	2017-10 LO F	1 ZUI9-ZUJ	(RS. Crore)
Particulars	FY 2017-18	FY 2018-19	FY2019-20
Fuel Cost for two months	79.18	140.68	73.74
Maintenance spares @15% of O&M	5.69	5.98	5.98
O&M Expenses for one month	3.16	3.32	3.32
Receivables for two months	180.12	242.19	132.42
Total Working Capital	268.15	392.17	215.46
Rate of Interest (%)	12.25%	12.25%	12.25%
Interest on Working Capital	32.85	48.04	26.39
	Particulars Fuel Cost for two months Maintenance spares @15% of O&M O&M Expenses for one month Receivables for two months Total Working Capital Rate of Interest (%)	ParticularsFY 2017-18Fuel Cost for two months79.18Maintenance spares @15% of O&M5.69O&M3.16Receivables for one month3.16Receivables for two months180.12Total Working Capital268.15Rate of Interest (%)12.25%	Fuel Cost for two months 79.18 140.68 Maintenance spares @15% of O&M 5.69 5.98 O&M Expenses for one month 3.16 3.32 Receivables for two months 180.12 242.19 Total Working Capital 268.15 392.17 Rate of Interest (%) 12.25% 12.25%

Table No.27: Interest on Working Capital approved by the Commission for the 1st Control Period (EY 2017-18 to EY 2019-20) (Rs. Crore)

Thus, the Commission approves working capital requirement of Rs. 268.15 Crore for FY 2017-18, Rs. 392.17 Crore for FY 2018-19 and Rs. 215.46 Crore for FY 2019-20 and interest thereon of Rs. 32.85 Crore, Rs. 48.04 Crore and Rs. 26.39 Crore for FY 2017-18, FY 2018-19 and FY 2019-20 respectively.

8.0 STATUTORY LEVIES AND TAXES

8.1 GVK submitted that Regulation 23 of the PSERC Tariff Regulations, 2014 provides for Income Tax which is reproduced hereunder:

"23.1 Obligatory taxes, if any, on the income of the generating company or the licensee or the SLDC from its core/licensed business shall be computed as an expense and shall be recovered from the customers/consumers:

Provided that tax on any income other than the core/licensed business shall not constitute a pass through component in tariff and tax on such other income shall be payable by the generating company or the licensee or the SLDC.

23.2 Tax on income, if actually liable to be paid, shall be limited to tax on return on equity allowed, excluding incentives.

23.3 Tax on income shall be considered at income tax rate including surcharge, cess etc. as applicable during the relevant year in accordance with the provisions of Income Tax Act, 1961 duly amended from time to time.

23.4 The benefits of tax holiday and the credit for carrying forward losses applicable as per the provisions of the Income Tax Act, 1961 shall be fully passed on to the customers/consumers.

23.5 The penalty, if any, arising on account of delay in deposit of tax or short deposit of tax amount shall not be claimed by the generating company or the licensee or the SLDC, as the case may be."

8.2 GVK submitted that it is entitled at antuals, as pass through, any cess, duty,



tax, government levy and royalty etc., payable by the Petitioner for generation and supply of power to the Respondent from time to time.

Petition No. 69 of 2017

51

Commission's Analysis

- 8.3 The benefits of any tax Holiday has to be passed on to the consumer/customer as per PSERC MYT Regulations. GVK has not submitted the details of tax holiday availed. Actual tax liability, after taking into account tax holidays etc., will be considered during the True up of respective years. The Commission directs that statutory levies including income tax with requisite documents may be claimed as per Regulations 23 of PSERC MYT Regulations, 2014 during the true-up of the respective years.
 - 9.0 Non-Tariff Income

Ò

 $\langle c \rangle$

9.1 GVK has not claimed any non tariff income for FY 2017-19 to 2019-20.

Commission's Analysis

- 9.2 The Non-Tariff Income has to be determined as per Regulation-28 of PSERC MYT Regulations-2014(amended from time to time).
- The Commission notes that Audited Annual Accounts of GVK for FY 2017-18 9.3 and FY 2018-19 has shown "Other income" as Rs.0.61 Crore and Rs 0.27 Crore respectively which includes income from bank deposits Rs 0.47 Crore and Rs 0.22 Crore respectively. As per para 20.3.4 of the Commission order dated 17.1.2020 in Petition no 54 of 2017, GVK had diverted the funds meant for capital expenditure out of the loan taken from financial institutions by investment in Mutual Funds. The Commission has neither considered interest paid on loans used for such other investments nor income earned from such other business/investment as part of the capital expenditure. Other income includes non operating income of Rs. 0.14 Crore and Rs 0.05 Crore for FY 2017-18 and FY 2018-19 respectively on account of credit balance written off and sale of scrap. The Commission notes that Rs.3.12 Crore and Rs 3.10 Crore have been booked under Ash Handling charges during FY 2017-18 and FY 2018-19 respectively but no income from Ash has been booked during FY 2017-18 and FY 2019-20 which will be considered during True up of FY 2017-18 and FY 2018-19. Accordingly, the Commissions determines Non-Tariff Income as Rs. 0.14 Crore and Rs 0 05 Grore as per Audited Annual Accounts



Petition No. 69 of 20%

of FY2017-18 and FY 2018-19 respectively and Rs 0.05 Crore has been considered for FY 2019-20 on the basis of FY 2018-19.

Accordingly, the Commission approves Non-Tariff Income as Rs. 0.14 Crore, Rs. 0.05 Crore and Rs. 0.05 Crore for FY 2017-18, FY 2018-19 and FY 2019-20 respectively.

10.0 Annual Fixed Charges (AFC) for FY 2017-18 to FY 2019-20

10.1 The Annual Fixed Charges from FY 2017-18 to FY 2019-20, as projected by GVK and approved by the Commission is summarized in the following table:-Table No.28: AFC provisionally approved by the Commission for

FY 2017-18 to 2019-20

(Rs. Crore)

Sr. Particulars		Sub	Submitted by GVK			Approved by the Commission		
No.		FY	FY	FΥ	FY	FY	FY	
		2017-18	2018-19	2019-20	2017-18	2018-19	2019-20	
1.	O&M Expenses	154.98	166.11	174.28	37.91	39.90	39.88	
2.	Depreciation	216.31	221.31	221.31	142.47	142.80	142.83	
3.	Interest charges	412.26	384.16	355.75	250.04	98.09	0.13	
4.	Return on Equity	193.91	193.91	193.91	142.51	142.83	142.87	
5.	Interest , on Working Capital	76.35	76.26	78.63	32.85	48.04	26.39	
6.	Income tax	0.00	0.00	0.00	0.00	0.00	0.00	
7.	Total Expenses	1,053.81	1,041.75	1,023.88	605.78	471.66	352.10	
8,	Less: Non-Tariff Income	0.00'	. 0.00	0.00	0.14	0.05	0.05	
9.	Annual Fixed Charges	1,053.81	1,041.75	1,023.88	605.64	471.61	352.05	

- 10.2 GVK shall be entitled for payment of capacity charges in accordance with Regulation 30 of the Central Electricity Regulatory Commission (Terms and Conditions of Tariff) Regulations, 2014 (as amended) where the same is not specified in the PSERC Tariff Regulations.
- 11.0 Energy charges for FY 2017-18 to FY 2019-20

GVK's Submission



11.1 GVK has submitted the monthly Declared Capacity and Scheduled Generation data for FY 2017-18, FY 2018-19 and FY 2019-20 (upto December, 2019) including the details of fuel cost vide submissions received on 12.03.2020, in response to Commission's interim Order dated 07.02.2020. The Declared Capacity and Scheduled Generation data as submitted by GVK is as under:

Table No. 29: Declared Capacity and Scheduled Generation data as submitted by GVK for FY 2017-18 to FY 2019-20 (upto Dec,2019)

[FY	
	Sr		FY	ĘΥ	2019-20	
	No.	Description	2017-18	2018-19	(Upto Dec.	
	100.			•	2019)	
	1.	Declared Capacity (MU)	1814.34	2850.62	3201.95	
	2.	Scheduled Generation (MU)	1387.12	2203.24	1129,19	ļ

PSPCL has not submitted any comments/reply on the same.

- 11.2 GVK in the present petition has submitted the details of components of the Energy Charge as hereunder:
 - i) Landed Cost of Primary Fuel:

In terms of the Coal Judgment and the Cancellation Order, the allocation of the Captive Coal Blocks to the Petitioner was cancelled. On 11.05.2015, the Petitioner filed the Petition No. 33 of 2015 seeking relief on account of Change in Law and Force Majeure events being the cancellation of the Petitioner's Captive Coal Blocks pursuant to the Coal Judgment and the Cancellation Order. On 12.08.2015, the Commission passed an Order and directed that the disputes raised in the Petition No. 33 of 2015 and Petition No. 65 of 2013 shall be referred to Arbitration. On 02.09.2015, the Commission passed an Order and Petition No. 65 of 2013. During the pendency of the arbitration proceedings, the Commission passed Order dated 01.02.2016 in Petition 33 of 2015 and Petition Alternate sources.

In compliance of the Commission's Order dated 01.02.2016 in Petition 33 of 2015 and Petition 55 of 2016, GVK was successful in securing



long term coal linkage under the SHAKTI Scheme for the Project to the extent of 1.706 MTPA from February 2018 onwards. However, the quantity of coal allocated was sufficient to achieve only 63% PLF. The balance coal was being sourced from other sources, including through e-auction and imported coal, to achieve target availability. GVK submitted that road transportation charges, handling charges related to loading the coal in to the trucks, unloading of the same at the railway siding and loading the same to the railway wagons, coal sampling cost for an analysis to obtain the GCV of coal received and the railway freight, all form part of the Landed Cost of Coal. The coal washery charges are not considered in the present landed cost of coal as given in this petition. However, the coal washery charges, if any, paid by GVK at a future date as charged by the Coal mining company or to abide by the Environmental Norms, the same shall form part of the landed cost of the Fuel. For computation of the Variable Charge, the Petitioner has considered an escalation of 5% p.a. on the price of the Primary Fuel for the Control Period.

ii) Landed cost of secondary fuel:

The Secondary fuel for the Plant is LDO / HFO. This Secondary Fuel is being procured from PSU Oil Marketing companies. This Secondary Fuel is delivered at site and the weighted average cost incurred on the same may be allowed by the Commission. The energy charge in Rupees per kWh is determined to three decimal places as per the following formulae prescribed in Regulation 39.4 of the PSERC Tariff Regulations, 2014.

iii) Normative Auxiliary energy consumption:

As per Regulation 36 of the PSERC Tariff Regulations, 2014, the norms for performance parameters, which includes normative auxiliary energy consumption for Coal-based generating stations, shall be as per the CERC norms. Accordingly, as per Regulation 36(E)(a) of the CERC Tariff Regulations, 2014, auxiliary consumption is 8.5% for a unit of 200 MW series. Further, for thermal generating stations with induced draft cooling to the target additional 0.5% is allowed. Thus,



normative auxiliary consumption has been considered at 9% for the computation of energy charge.

iv) Fuel Transit & Handling Losses:

In accordance with Regulation 40 of the PSERC Tariff Regulations 2014, the normative transit and handling loss to be taken @ 1% or actual whichever is lower and the same are considered @ 1% while computing the energy charge.

v) Station Heat Rate:

As per Regulation 36 of the PSERC Tariff Regulations, 2014, the norms for performance parameters, which includes Station Heat Rate for Coal-based generating stations, the same shall be as per the CERC norms. Regulation 36(C)(b) of the CERC Tariff Regulations 2014, specifies that for tariff calculation, the design station heat rate is to be multiplied by a factor of 1.045 for a new coal based thermal power plant. The Guaranteed Heat Rate for the power plant under the EPC contract is 2221 kcal/kWh. Accordingly, the Station Heat Rate of the Petitioner's plant for tariff calculations comes out to be 2321 kcal/kWh, which has been considered for Energy Charge calculation.

Accordingly, GVK has computed the projected Energy Charges for the Control Period in the petition as under:

Table No. 30: Projected Energy Charges submitted by GVK for FY 2017-18

Sr.No.	Particulars	FY 2017-18	FY2018-19	FY2019-20
	Variable Charge per Unit (Rs./ kWh)	3.63	3.64	3.82
2.	Energy sold (MU)	3573	3573	3573
3.	Annual Energy Charge	1298.35	1298.87	1363.55
	(Rs. Crore)	<u> </u>		<u></u>

11.3 GVK vide affidavit dated 11.03.2020, in compliance to the Commission's Interim Order dated 07.02.2020 submitted the details of fuel cost/energy charges and further reiterated its submissions regarding consideration of normative auxiliary consumption @9% for the computation of energy charges. With reference to fuel transit and handling losses, GVK submitted that normative transit and handling loss are considered in accordance with



Regulation 40 of the PSERC Tariff Regulations 2014 and Regulation 30(8) of the CERC Tariff Regulations. Further, GVK submitted that SHR as per the Regulations works out to be 2321 kcal/kWh for FY 2017-18 and FY 2018-19 and 2332 kcal/kWh for FY 2019-20. The fuel cost details furnished by GVK are as under:

Sr.			Actual	Actual	(F	Y 20 19-20)	
No.	Item	Unit	FY 2017-18	FY 2018-19	Actual till Dec 2019 ,	Projection Jan. 2020 to March 2020	Total
1	Gross Generation	MU	1536	2436	1244.68	117,936	1362.62
2.	Auxiliary Consumption	%	9.72%	9.55%	9.28%	9,28%	9.28%
3.	Net Generation	MU	1387	2203	1129	107	1236
4.	Station Heat Rate	kcal/kWh	2392	2348	2350	2350	2350
5.	Calorific Value of Coal	kcal/Kg	3730	3396	3349	3349	3349
6.	Coal Transit Loss	%	4.20	2.43	1,60	1,60	1.6
7.	Total Coal Consumption	Tonne	1019703	1660491	851208	82758	933966
8.	Total Oil Consumption	KL	2055	1596	1018.39	96.49	114.88
9.	Specific Oil Consumption	ml/kWh	1.34	0.66	0,82	0.82	0.82
10.	Calorific Value of Oil	kcal/Litre	10368	10396	10387	10387	10387
11.	Price of Coal	Rs./Tonne	6120	6304	5743	5743	5743
12.	Price of Oil	Rs/ KL	34936	45658	45078	45078	45078
13.	Total Coal Cost	Rs. crore	651.44	1072.85	496.79	48.30	545.09
14.	Total Cost of Oil	Rs. crore ,	7.18	7.29	4.59	0.43	5.02
15.	Total Fuel Cost	Rs. crore	658.62	1080.14	501.38	48.73	550.11

Table No.31: Fuel cost data furnished by GVK in affidavit dated 11.03.2020

- 11.4 Further, GVK vide its submissions dated 01.07.2020 stated that PSPCL is currently paying surface transportation costs to GVK at rate of Rs. 8 per ton per km irrespective of the source, quantum and distance transported, on the basis of BCCL Circular dated 29.03.2018. However, in case of NPL, PSPCL is paying surface transportation charges on actual, based on the rates charged by agencies appointed by NPL for surface transportation of coal and there is no benchmarking to CCL or BCCL surface transportation cost. **PSPCL's Submission**
- 11.5 PSPCL vide its Memo No. 6923/TR-5/886 dated 21.05.2018 submitted that as per Regulation 38 and 39 of the PSERS Tariff Regulations 2014, energy



6-3

charge for generating plants of the generating companies is to be determined on the basis of landed cost of primary as well as secondary fuel which is to be utilized for running the plant up-to its entire contracted capacity. In this regard, GVK has submitted that it has been successful in securing long term linkage of coal for the project, capable of operating the plant at 63% PLF; however, no material details of the same have been given. Further, GVK has stated that the balance coal for achieving the target availability is to be sourced from other sources, details of which have also not been provided in the present Petition. The fuel availability for GVK has been mired in uncertainty as placed before the Commission in earlier proceedings between the parties. Therefore, GVK without furnishing the particulars of required material but based on mere assumption that it would get full coal linkage before the end of March 2018, has prayed that the Commission may consider the price of coal arrived for the FY- 2018-19 and FY-2019-20 and that too, by factoring in a 5% escalation per annum.

- 11.6 As per clause 6.1.1 of the Amended and Restated PPA dated 26.5.2009, the tariff payable shall be a two-part tariff being capacity charges and energy charges and shall be payable based on the contracted capacity at normative availability. Without there being any firm coal linkage for the entire contracted capacity and there also being no material particularly available for the fuel actually being used for power generation, the landed fuel cost of primary fuel cannot be computed and thus tariff determination cannot take place. Even though GVK admits that the available fuel with them is sufficient only to run the plant at 63% PLF, the tariff proposal has been prepared on the basis of the plant achieving 83% PLF which is not only contradictory to GVK's own stand, but also factually incorrect.
 - 11.7 PSPCL vide Memo No. 5291/TR-5/886 dated 11.03.2020 has submitted the details of Energy Charges for FY 2017-18, FY 2018-19 and FY 2019-20 (till December 2019) as directed by the Commission vide Interim Order dated 07.02.2020.
 - 11.8 PSPCL vide Memo No. 5321/TR-5/886 dated 16.03.2020 submitted that details of energy charges given by GVK to the Commission vide affidavit dated 11.03.2020 have been tallied and are found in order.
 - 11.9 PSPCL vide letter 5533/TR 5/986 dated 26.06.2020, submitted the details of



1

Energy charges paid to GVK for the whole of FY 2019-20.

Commission's Analysis

- 11.10 The energy charges for FY 2017-18 to FY 2019-20 are payable by PSPCL to GVK in terms of the PPA, Order dated 01.02.2016 common to petition no. 65 of 2013 & 33 of 2015, Order dated 06.03.2019 in petition no. 68 of 2017 and Order dated 27.05.2019 in Petition No. 01 of 2018 as applicable.
- 12.0 Interest on under-recovered or over-recovered fixed charges:
- 12.1 The Commission notes that the applicability of Regulation 9 of PSERC Regulations, 2005 would be on the distribution companies or generating cum distribution companies and cannot be applied, as it is, to standalone generating companies.
- 12.2 The Regulation 8 (13) of CERC (Terms and Conditions of Tariff) Regulation, 2014 is re-produced below for reference: -

"The amount under-recovered or over-recovered, along with simple interest at the rate equal to the bank rate on 1st April of the respective year, shall be recovered or refunded by the generating company or the transmission licensee, as the case may be, in six equal monthly instalments starting within three months from the date of the tariff order issued by the Commission".

- The Commission observes that Regulation 8 (13) of CERC (Terms and Conditions of Tariff) Regulation, 2014 is squarely applicable to under recovery or over recovery of fixed charges in case of generating companies.
- 12.3 The Commission decides to adopt the CERC Regulations for determining interest equivalent to bank rate on under recovery or over recovery of fixed charges.

Commission's Analysis

Accordingly, interest shall be allowable or recoverable as per Regulation 8 (13) of CERC (Terms and Conditions of Tariff) Regulation, 2014 on under-recovered or over-recovered Annual Fixed Charges (AFC) determined by the Commission.

The Petition is disposed of accordingly. E(2020 marne (Kusumjit Sidhu) (S.S. Sarna) (Anjuli Chandra) Chairperson Member Member Authenticated ELECTIMENTY BEQUE AUO, Hilding Commission Chandigarh Dated: 05-08-2020 . Punic's Staty Elect 36 Cantan nia

ANNEXURE P-2

59



Å

Amended and Restated Power Purchase Agreement between

> Punjab State Electricity Board (the "Procurer")

> > And

GVK Power (Goindwal Sahib) Limited (the "Seller")

Date:26/05/2009

2x270MW COAL FIRED THERMAL POWER PLANT AT GOINDWAL SAHIB, DISTRICT TARN TARAN PUNJAB

AR



TABLE OF CONTENTS:-

 \bigcirc

•	ARTICLE 1: DEFINITIONS AND INTERPRETATION	
2	ARTICLE 2: TERM OF AGREEMENT	23
3	ARTICLE 3: CONDITIONS SUBSEQUENT TO BE SATISFIED BY THE SELLER AND THE PROCURERS.	
4	ARTCLE 4 : DEVELOPMENT OF THE PROJECT	
5	ARTICLE 5: CONSTRUCTION	36
6	ARTICLE 6: SYNCHRONISATION, COMMISSIONING AND COMMERCIAL OPERATION	
7	ARTICLE 7 : OPERATION AND MAINTENANCE	43
8	ARTICLE 8: CAPACITY, AVAILABILITY AND DISPATCH	44
9	ARTICLE 9: METERING AND ENERGY ACCOUNTING	49
1(ARTICLE 10: INSURANCES	
11	ARTICLE 11 : BILLING AND PAYMENT	51
12	ARTICLE 12 : FORCE MAJEURE	63
13	SARTICLE 13 : CHANGE IN LAW	70
14	ARTICLE 14 : EVENTS OF DEFAULT AND TERMINATION	
15	SARTICLE 15 : LIABILITY AND INDEMNIFICATION	
16	ARTICLE 16: ASSIGNMENTS AND CHARGES	82
17	ARTICLE 17: GOVERNING LAW AND DISPUTE RESOLUTION	
18	ARTICLE 18 : MISCELLANEOUS PROVISIONS	
1.	SCHEDULE 1: SITE	91
2	SCHEDULE 2: INITIAL CONSENTS	93
3.	SCHEDULE 3: FUNCTIONAL SPECIFICATION	94
4	SCHEDULE 4: COMMISSIONING AND TESTING	95
5.	SCHEDULE 5: AVAILABILITY FACTORS	97
6.	SCHEDULE 6: TARIFF	98
	W	~

h

2

麻

60



<u>ب</u>

1.00

7. SCHEDULE 7: DETAILS OF INTERCONNECTION POINT AND FACILITIES	103
8. SCHEDULE 8: REPRESENTATION & WARRANTIES.	.105
9. SCHEDULE 9: FORMAT OF THE PERFORMANCE GUARANTEE	.107
10. SCHEDULE 10: SUBSTITUTIONS RIGHTS OF LENDERS	. 109
11. SCHEDULE 11; ESTIMATED CAPITAL COST OF THE PROJECT	.116
12. SCHEDULE 12; CALCULATIONS OF 'X'-DAYS	.118
h	

3

T

AR.

in the रोक रताय एक सौ रुपये **Rs.**_100 ONE 友至1,00 HUNDRED RUPEES HIRGUNDIA NDIA NON JUDICIAL 318818 అంద్రప్రదేశ్ आन्ध्र प्रदेश ANDHRA PRADESH Saleana AA 241197 SALEEMA ANJUM Rs DD Ş.N STAMP VENDOR S.V.L.No.3/2009 Name H.No.13-4-661/A, KARWAN, HYDERABAD. X (Groindwal sahib) 1 Hd sec-bad Hor Whom This Amended and Restated Power Purchase Agreement (the "Agreement") is entered into on this 26 day of May 2009 between ^F1. Punjab State Electricity Board, a Board established under The Electricity (Supply) Act 1948, having its Head Office at "The Mall, Patiala - 147 001, Punjab, India" (hereinafter referred to as the "Procurer", which expression unless repugnant to the ā context or meaning thereof shall be deemed to include its successors, and permitted assigns) of the first part; and \$2. GVK Power (Goindwal Sahib) Limited, a Company incorporated under The Companies Act, 1956, having its registered office at "Paigah House", 156-159, Sardar Patel Road, Secunderabad- 500 003, Andhra Pradesh, India" (hereinafter referred to as the "Seller", which expression unless repugnant to the context or meaning thereof shall be deemed to include its successors, and permitted assigns) of the second ģ part. hereinafter referred to as "Party" or "Parties" as appropriate; â á 4<u>8</u>) ŝ S ġ

	¢	
	- भारतीय गेर	न्यायिक
	एक सौ रुपये	Rs. 100
		ONE HUNDRED RUPEES
	00100 100100100100 100100100100 10010010	DIA100100100 29129160100100 29129160186186
မ၀ဖ	5/0.W/0.D/0. Sheankax For Whom Gr.V.K. Power (Groindwal	Salasma AA 241198 SALEEMA ANJUM STAMP VENDOR S.V.L.NO.3/2009 H.NO.13-4-661/A, KARWAN, HYDERABAD. SONIDICA SEC-book
	ă	

192 - 192 - 192

Whereas:

12

3

48

R

gThe Government of Punjab and the Procurer had invited proposals during the year 1996 to establish a coal based power station with an Installed Capacity of approximately 462.5 MW at Goindwal Sahib, Amritsar District in the state of Punjab, India and the Seller having been selected by the Government of Punjab and the Procurer on the basis of International . Competitive Bidding has accepted the invitation of the Government of Punjab and the Procurer to Build, Own and Operate the coal based power station;

Whereas the Procurer and the Seller entered into a Power Purchase Agreement (PPA) on 17th April 2000.

Whereas the Parties have entered into a Memorandum of Understanding on 8th February 2006 and have agreed to modify the Power Purchase Agreement dated 17th April 2000 to gincorporate among other things the revised capacity 2 x 250 MW (+20%) of the Power Station agreed between the Parties, adoption of CERC norms for Tariff calculation and schedule for achieving commercial operation of Unit-1 & Unit-2 of the Power Station;

Whereas the Punjab State Electricity Regulatory Commission ("PSERC") (hereinafter geferred to as "Commission") was established after the signing of the Power Purchase



633

ġ

202

ğ

EAgreement (PPA) dated 17th April 2000 and PSERC is empowered to regulate electricity purchase and procurement process of the Procurer, the terms and conditions of this gAgreement including the price at which electricity shall be procured from the Seller

Whereas the Seller has agreed to sell the entire Contracted Capacity and energy generated by the Project to the Procurer and the Procurer has agreed to purchase the Contracted Capacity and scheduled energy from the Seller in accordance with the terms and conditions of this Amended and Restated Power Purchase Agreement

Whereas subsequently the Seller filed Petition for an amendment in the capacity of the Project to 2x270 MW from 2x300 MW indicated in the Amended & Restated PPA initialled between the Parties in January,2007, and accordingly the Contracted Capacity of the project shall now be 2x270 MW.

Whereas the Project has been since allotted the Tokisud North Sub Block as a captive coal mine and a part of Saregarha Block in the state of Jharkhand for supply of coal to the Project;

Whereas the PSERC had issued the order dated 29.04.08 in case of Petition No. 4 of 2007 gfiled by M/s GVK for approval of capital cost of the project and M/s GVK filed an Appeal



No. 104 of 2008 in the Appellate Tribunal for Electricity against this order to which the Tribunal has issued the order dated 08.04.2009 allowing additional costs to M/s GVK.

Whereas the estimated costs as approved by PSERC and additional costs allowed by Appellate Tribunal shall now be indicated in the estimated cost listed in the Schedule-11.

Whereas the Parties mutually desire to amend and restate the Power Purchase Agreement dated 17th April 2000 and enter into this Agreement, as per the order of the PSERC dated 06 March 2009 on Petition no. 3 of 2007 filed for approval of the initialed Amended and Restated PPA between the Parties;

Whereas M/s GVK Power (Goindwal Sahib) Ltd. filed the Appeal No. 70 of 2009 in the Appellate Tribunal for Electricity against the order of PSERC dated 06.03.09 which is yet to be decided.

Whereas M/s GVK Power also filed an Interim Application No. 167 of 2009 before the Appellate Tribunal to allow them to sign the Amended & Restated PPA as per the order of PSERC dated 6.3.2009 without prejudice to their right to appeal.

Whereas the Appellate Tribunal has issued the order dated 19.05.09 against I.A. No. 167 of 2009 allowing both the parties to sign the Amended & Restated PPA as per the order of PSERC dated 6.3.2009 and reserving the right to appeal of both the parties in any Court of Law.

Now therefore, in consideration of the premises and mutual agreements, covenants and conditions set forth herein, it is hereby agreed by and between the Parties, in supercession of all agreements, letters, communications and the like, anterior to this Agreement, as follows: \wedge





ARTICLE 1: DEFINITIONS AND INTERPRETATION

1.1 Definitions

The terms used in this Agreement, unless as defined below or repugnant to the context, shall have the same meaning as assigned to them by the Electricity Act, 2003 and the rules or regulations framed there under, including those issued/framed by Appropriate Commission (as defined hereunder), as amended or re-enacted from time to time.

The following terms when used in this Agreement shall have the respective meanings, as specified below:

into	
	8
	business in the State in which the Procurer's registered office is
"Business Day"	means with respect to the Seller and the Procurer, a day other than Sunday or a statutory holiday, on which the banks remain open for
"Bill Dispute Notice"	means the notice issued by a Party raising a Dispute regarding a Monthly Bill or a Supplementary Bill issued by the other Party;
"Auxiliary Energy Consumption or AUX"	means, in relation to a period in case of a generating station means, the quantum of energy consumed by auxiliary equipment of the generating station and transformer losses within the generating station expressed as a percentage of the sum of the gross energy generated at the generator terminals of all the Units of the generating station.
"Availability" "Available Capacity"	; shall have the meaning ascribed thereto in ABT;
"Availability Factor" or	as applicable. shall have the meaning ascribed thereto in ABT
"Availability Based Tariff" or "ABT"	Shall mean all the regulations contained in the Central Electricity Regulatory Commission (terms and conditions of Tariff) Regulations
or "PPA" "Appropriate Commission"	means the Punjab State Electricity Regulatory Commission (PSERC) or its successors;
"Agreement" or "Power Purchase Agreement"	most recently agreed to by the Parties and initialled by them for identification; means this document including its recitals and Schedules;
"Act" or "Electricity Act 2003" "Agreed Form"	means the Electricity Act 2003 or any amendments made to the same or any succeeding enactment thereof; means, in relation to any document, the form of the said document

In

头



located;

"Capacity Charge" or "Capacity Charges" "Capacity Notice"

'Capital Cost'

shall have meaning ascribed thereto in Schedule 6

shall have the meaning ascribed thereto under ABT or the Grid Code;

means

actual capital cost of the Project on a relevant date which shall not be later than the Commercial Operation Date of the Power Station, as certified by the auditors appointed jointly by the Seller and Procurer and as approved by PSERC.

provided that Capital Cost shall always exclude cost overruns arising due to a Seller Event of Default, or costs due to events for which compensation has been received by the Seller from the Procurer or Insurers or third parties;

Provided further that the Capital Cost in relation to a Unit shall be the Capital Cost allocated in proportion to the Contracted Capacity of the said Unit.

shall mean sources of finance used to finance the Capital Cost as 'Capital Structure Schedule' provided in the Financing Agreements; means the Tokisud North Sub Block" in the Hazaribagh District of "Captive Coal Mine(s)" Jharkhand State in India, and any other coal mine/block allocated by the Government of India for supplying coal to the Project and associated fuel transport system up to the Power Station; " "Central Transmission shall have the meaning ascribed thereto in the Electricity Act, 2003; Utility" or "CTU" "CERC" means the Central Electricity Regulatory Commission, as defined in the Electricity Act, 2003, or its successors; "CERC Tariff Means the regulations formed by CERC under section 178 of the Regulations" Electricity Act, 2003 and titled the Central Electricity Regulatory Commission (Terms and Conditions of Tariff) Regulations as applicable. "CERC Norms" means the norms for determination of tariff for the coal based power stations under the CERC (Terms and conditions of Tariff) Regulations as applicable. "Change in Law" shall have the meaning ascribed thereto in Article 13.1.1; "Commercial Operation means, in relation to a Unit, the date one day after the date when the



Date" or "COD"

"Commissioning" or "Commissioned" with its grammatical variations "Commissioning Tests" or "Commissioning Test" "Commissioned Unit" "Construction Contractor/s" "Construction Period"

"Consultation Period"

"Contract Year"

Procurer receives a Final Test Certificate of the Independent Engineer as per the provisions of Article 6.3.1 and in relation to the Power Station shall mean the date by which such Final Test Certificates as per Article 6.3.1 are received by the Procurer for all the Units:

68

oning" or means, in relation to a Unit, that the Unit or in relation to the Power oned" with Station, all the Units of the Power Station have passed the grammatical Commissioning Tests successfully;

means the tests provided in Schedule 4 herein;

means the Unit in respect of which COD has occurred; means one or more main contractors, appointed by the Seller to

design, engineer, supply, construct and commission the Project; means the period from (and including) the date upon which the Construction Contractor is instructed or required to commence work under the Construction Contract up to (but not including) the Commercial Operation Date of the Unit in relation to a Unit and of all the Units in relation to the Power Station;

means the period, commencing from the date of issue of a Seller Preliminary Default Notice or a Procurer Preliminary Default Notice as provided in Article 14 of this Agreement, for consultation between the Parties to mitigate the consequence of the relevant event having regard to all the circumstances

means the period beginning on the Effective Date (as defined hereunder) and ending on the immediately succeeding March 31 and thereafter each period of 12 months beginning on April I and ending on March 31 provided that :

(i) in the financial year in which Scheduled COD of the first Unit would have occurred, a Contract Year shall end on the date immediately before the Scheduled COD of the first Unit and a new Contract Year shall begin once again from the Scheduled Commercial Operation Date of the first Unit and end on immediately succeeding March 31 and provided further that

 the last Contract Year of this Agreement shall end on the last day of the term of this Agreement;

10



"Contracted Capacity"

means

a) (i) for the first Unit, [270] MW; (ii) for the second Unit, [270] MW; rated capacity and in relation to the Power Station as a whole means [540] MW rated capacity at the generator terminals of the units.

or such rated capacities as may be determined in accordance with Article 6.3.4 or Article 8.2 of this Agreement;

means the RLDC or SLDC or such other load control centre "Control Centre" or designated by the Procurer from time to time through which the "Nodal Agency" Procurer shall issue Dispatch Instructions to the Seller for the Power Station; "Debt Service" means the amounts which are due under the Financing Agreements by the Seller to the Lenders, expressed in Rupees (with all amounts denominated in currencies other than Rupees being converted to Rupees at the Reference Exchange Rate, the selling rate in Rupees for the Foreign Currency on the relevant day, as notified by the State Bank of India as its TT Rate at 12:00 noon on the Notice to Proceed.); "Declared Capacity" In relation to a Unit or the Power Station at any time means the net capacity of the Unit or the Power Station at the relevant time (expressed in MW at the Interconnection Point) as declared by the Seller in accordance with the Grid Code and dispatching procedures as per the Availability Based Tariff; " Delivery Point' " or means the points of delivery specified in Schedule 7 for fulfilling the "Interconnection Point" obligation of the Seller to deliver the Contracted Capacity less Auxiliary Energy Consumption to the Procurer; "Direct Non-Natural shall have the meaning ascribed thereto in Article 12.3(ii)(1). Force Majeure Event" "Dispute" means any dispute or difference of any kind between the Procurer and the Seller, in connection with or arising out of this Agreement including any issue on the interpretation and scope of the terms of this Agreement as provided in Article 17;

"Dispatch Instruction"

means any instruction issued by the Procurer through the respective SLDC and RLDC to the Seller, in accordance with applicable Grid

11



Code and this Agreement;

means the thirtieth (30th) day after a Monthly Bill or a Supplementary Bill is received and duly acknowledged by the Procurer (or, if such day is not a Business Day, the immediately succeeding Business Day) by which date such bill is payable by the Procurer;

means the date of signing of this Agreement by the Parties;

means the Electricity Act, 2003 and the rules and regulations made thereunder from time to time along with amendments thereto and replacements thereof and any other Law pertaining to electricity including regulations framed by the Appropriate Commission;

means the net electrical output of the Power Station at the Delivery Point, as expressed in kWh;

means a condition or situation that, in the opinion of the Procurer or RLDC or SLDC or the agency responsible for operating and maintaining the Interconnection and Transmission Facilities or the transmission company, as the case may be, poses a significant threat to the Procurer's or the said agency's or transmission company's ability to maintain safe, adequate and continuous electricity supply to its customers, or seriously endangers the security of persons, plant or equipment;

shall have the meaning ascribed to this term under Schedule - 6 means for any period, the Energy Units actually delivered by the Project, as metered at the Interconnection Point, reduced by the Energy Units supplied by the Procurer to the Seller at the Interconnection Point or to any Project facility for which the Seller is responsible located inside or outside the Project boundary such as

raw water or other pumping stations; means the 25th anniversary of the Commercial Operation Date of the Power Station. For the avoidance of doubt, in case the COD of the Power Station occurs on June 1, 2013, then the 25th anniversary of the Scheduled COD of the Power Station shall occur on June 1, 2038, i.e. in the Contract Year 2038-39;

means

12

a) a certificate of the Independent Engineer certifying and accepting the results of a Commissioning Test/s in accordance with Article 6.3.1 of this Agreement; or b) a certificate of the Independent Engineer certifying the result

of a Repeat Performance Tests in accordance with Article 8.2.1 of this Agreement;

Energy Charges "Energy Output"

"Due Date"

"Effective Date"

"Electricity Laws"

"Electrical Output"

"Emergency"

"Expiry Date'

"Final Test Certificate"



"Financial Closure" or "Financial Close"

"Financing Agreements" means the execution of all the Financing Agreements required for the Project and fulfilment of conditions precedents and waiver, if any, of any of the conditions precedent for the initial draw down of funds there under.

means all the loan agreements, notes, indentures, security agreements, letters of credit and other documents relating to the financing of the Project on or before the COD of the Power Station, as may be amended, modified, refinanced or replaced from time to time, but without in anyway increasing the liabilities of the Procurer therein;

means primary fuel used to generate electricity namely, domestic

means the agreement(s) entered into between the Seller and the Fuel

Supplier for the purchase, transportation and handling of the Fuel, required for the operation of the Power Station. In case the transportation of the Fuel is not the responsibility of the Fuel Supplier, the term shall also include the separate agreement between the Seller and the Fuel Transporter for the transportation of Fuel in addition to the agreement between the Seller and the Fuel Supplier

means the technical requirements and parameters described in

Schedule 3 of this Agreement and as provided in Grid Code relating to the operation, maintenance and dispatch of any Unit and the

means any set of regulations or codes issued by Appropriate Commission as amended and revised from time to time and legally binding on the Seller' and the Procurer' governing the operation of the Grid System or any succeeding set of regulations or code;

means the Interconnection and Transmission Pacilities and any other transmission or distribution facilities through which the Procurer's

shall have the meaning ascribed thereto in Article 12.3; shall have the meaning ascribed thereto in Grid Code;

"Force Majeure" "Forced Outage"

coal

for the supply of the Fuel;

Power Station;

and Article 8 herein.

13

"Fuel" .

"Fuel Supply Agreements⁶

"Functional Specifications"

"Grid Code" or "IEGC"

"Grid System"

"Independent Engineer"

SK

supply electricity to their customers or the transmission company transmits electricity to the Procurer; means an independent consulting engineering firm or group appointed jointly by the Procurer and the Seller to carry out the functions in accordance with Article 4.7.1 and Article 6, Article 12
functions in accordance with Article 4.7.1 and Article 6, Article 12 and Article 8 herein.

provided that separate Independent Engineer may be appointed for the purposes of Article 4.7.1, Article 6, Article 12 and Article 8; provided further that the separate Independent Engineer may be appointed for each financial year for the purposes of Article 8, and in such case, such Independent Engineer shall be appointed at least ninety (90) days prior to the beginning of the financial year. shall have the meaning ascribed thereto in Article 12.3(ii)(2).

"Indirect Non-Natural Force Majeure Event"

"Indian Governmental Instrumentality"

"Infirm Power"

"Initial Consents"

"Initial Performance Retest Period" "Interconnection Facilities" or "Interconnection and Transmission Facilities"

means the GOI, Government of Punjab, and any ministry or, department or board or agency other regulatory or quasi-judicial authority controlled by GOI or Government of Punjab where the Procurer and Project are located and includes the Appropriate Commission:

means energy produced by a generating unit and delivered to the Procurer prior to the Commercial Operation Date of such generating unit;

shall mean the consents listed in Schedule 2;

shall have the meaning ascribed thereto in Article 6.3.3 of this Agreement;

means the facilities on the Procurer's side of the Interconnection Point for receiving and metering Electrical Output in accordance with this Agreement and which shall include, without limitation, all other transmission lines and associated equipment, transformers and associated equipment, relay and switching equipment and protective devices, safety equipment and, subject to Article 9, the Metering. System required for the Project.

The Interconnection Facilities also include the facilities for receiving power at the Delivery Point where the transmission line from the Power Station Switchyard end is injecting power into the transmission network (including the dedicated transmission line connecting the Power Station with the STU /CTU transmission network);



"Invoice" or "Bill"

means either a Monthly Tariff Invoice, a Supplementary Invoice or a Procurer Invoice;

shall have the meaning ascribed thereto in Article 11.3.4;

"Late Payment Surcharge"

"Law"

"Lenders"

means, in relation to this Agreement, all laws including Electricity Laws in force in India and any statute, ordinance, regulation, notification or code, nule, or any interpretation of any of them by an Indian Governmental Instrumentality and having force of law and shall further include all applicable rules, regulations, orders, notifications by an Indian Governmental Instrumentality pursuant to, or under any of them and shall include all rules, regulations, decisions and orders of the Appropriate Commission;

means the banks, other financial institutions, multilateral agencies, RBI registered non banking financial companies, mutual funds and agents or trustees of debenture / bond holders, including their successors and assignees, who have agreed as on or befor COD of the power station to provide the Seller with the senior debt financing described in the Capital Structure Schedule, and any successor banks or financial institutions to whom their interests under the Financing Agreements may be transferred or assigned:

Provided that, such assignment or transfer shall not relieve the Seller of its obligations to the Procurer under this Agreement in any manner and shall also does not lead to an increase in the liability of the Procurer;

shall have the meaning ascribed thereto in Article 11.4.1;

"Meters" or "Metering means meters used for accounting and billing of electricity in accordance with Central Electricity Authority (Installation and Operations of Meters) Regulations, 2006, Grid Code and ABT, as amended from time to time; "Maintenance Outage" shall have the meaning as ascribed to this term as per the provisions of the Grid Code.

means gross Power Station or Unit Maximum Continuous Rating as defined in the Grid Code; Im Offtake

15

"Minimum Guarantee"

"MCR"

"Letter of Credit" or

"L/C"

means guaranteed offtake of sixty five per cent (65%) of the total



contracted capacity for the procurer during a contract year.

r uwgi stanun	Musico nic.
"Power Station"	means the:
Notice"	agreement:
"Performance Test " "Preliminary Default	means the test carried out in accordance with Article 1.2 of Schedule 4 of this Agreement; shall have the meaning ascribed thereto in Article 14 of this
· · ·	 in accordance with Article 3 of this Agreement and which shall include the additional bank guarantee furnished by the Seller under this Agreement;
	specified by the Procurer, in the form attached hereto as Schedule 9,
"Performance Guarantee"	Agreement; means the irrevocable unconditional bank guarantee, submitted and to be submitted by the Seller to the Procurer from a scheduled bank
"Party" and "Parties"	contract, if any; shall have the meaning ascribed thereto in the recital to this
"Operator" or "Operators"	means one or more contractors appointed as operator of power generation facilities of the Power Station pursuant to an O&M
"Operating Procedures	shall have the meaning ascribed thereto in Grid Code;
"O&M Contract" or "O&M Contracts"	until the expiry or earlier termination of this Agreement in accordance with Article 2 of this Agreement; means the contract/s entered into by the Seller with the Operator or • operators, if any;
Availability" "Operating Period"	Point on Contract Year basis ⁸ . in relation to the Unit means the period from its COD and in relation to the Power Station the date by which all the Units achieve COD,
Normative	means equal to eighty per cent (80%) Availability at the Delivery
" Non-Natural Force Majeure Event"	shall have the meaning ascribed thereto in Article 12.3(ii).
"Natural Force Majeure Event"	shall have the meaning ascribed thereto in Article 12.3(i).
"Monthly Bill" or "Monthly Invoice"	the event, where applicable, else a calendar month; means a monthly invoice comprising Capacity Charges (applicable after COD of the first unit) and Energy Charges, including incentive and penalty, as per Schedule 6 hereof;
"Month"	means a period of thirty (30) days from (and excluding) the date of

==.

私

، ^{معمر} ، ر

 \bigcirc

A.



- (a) Thermal power generation facility comprising of any or all the Units;
- (b) any associated fuel handling, treatment or storage facilities of the power generation facility referred to above;
- (c) any water supply, treatment or storage facilities required for the operation of the power generation facility referred to above;
- (d) the ash disposal system including ash dyke [as applicable];
- (e) township area for the staff colony; and
- (f) bay/s for transmission system in the switchyard of the power station,
- (g) all the other assets, buildings/structures, equipments, plant and machinery, facilities and related assets required for the efficient and economic operation of the power generation facility;

whether completed or at any stage of development and construction or intended to be developed and constructed as per the provisions of this Agreement.

means the coal based Thermal Power Station to be established at Goindwal Sahib undertaken for design, financing, engineering, procurement, construction, operation, maintenance, repair, refurbishment, development and insurance by the Seller in accordance with the terms and conditions of this Agreement; mean

- a) Construction Contracts;
- b) Fuel Supply Agreements including the Fuel Transportation Agreement,
- c) O&M contracts;

d) any other agreements designated in writing as such,

from time to time, jointly by the Procurer and the Seller;

means the practices, methods and standards that are generally accepted internationally from time to time by electric utilities for the purpose of ensuring the safe, efficient and economic design, construction, commissioning, operation and maintenance of power generation equipment of the type specified in this Agreement and which practices, methods and standards shall be adjusted as

17

"Project"

"Project Documents"

"Prudent Utility Practices"



necessary, to take account of:

and amendments thereof;

- a) operation and maintenance guidelines recommended by the manufacturers of the plant and equipment to be incorporated in the Project;
- b) the requirements of Indian Law; and
- c) the physical conditions at the Site;

shall have the meaning ascribed thereto in Article 8.1 of this agreement;

"Repeat Performance Test "

"RPC"

"RBI"

means the relevant Regional Power Committee established by the Government of India for a specific Region in accordance with the Electricity Act, 2003 for facilitating integrated operation of the power system in that Region; means Reserve Bank of India;

means as defined in the Grid Code and issued by the relevant RPC secretariat or other appropriate agency for each Week and for each Month (as per their prescribed methodology), including the revisions

"RLDC"

"Rupees" or "Rs." "SBAR"

"Regional Energy

Accounts" or "REA"

"Selectee"

means the relevant Regional Load Dispatch Centre as defined in the Electricity Act, 2003, in the region in which the Project is located; means the lawful currency of India;

means the prime lending rate per annum applicable for loans with one (1) year maturity as fixed from time to time by the State Bank of India. In the absence of such rate, any other arrangement that substitutes such prime lending rate as mutually agreed to by the Parties;

means a new company (i) proposed by the Lenders pursuant to Schedule 10 hereof and approved by the Procurer (ii) or proposed by the Procurer in accordance with Schedule 10 hereof and approved by the Lenders, for substituting the Seller for the residual period of the PPA by amendment of the PPA or by execution of a fresh PPA in accordance with the terms and conditions contained in the said Schedule.

means the State Electricity Regulatory Commission, as defined in the Electricity Act, 2003, or its successors;

means the date on or before which COD of a Generating Unit is required to occur, which shall be thirty six (36) Months from the date of Financial Closure for the First Generating Unit and six (6)

"SERC"

"Scheduled COD" or "Scheduled Commercial Operation



77

Date"	months from the COD of the First Generating Unit for the Second Generating Unit or such other dates from time to time, specified in accordance with the provisions of this Agreement;
"Scheduled Connection Date"	Shall mean the date falling 210 ¹⁰ days before the Scheduled COD of first Unit;
"Scheduled Energy" or "Scheduled Generation"	means scheduled generation as defined in the ABT;
"Scheduled Outage"	shall have the meaning ascribed to this term as per the provisions of
· · ·	the Grid Code;
"Scheduled Synchronisation Date"	means in relation to a Unit, the date, which shall be maximum of one hundred and eighty (180) days prior to the Scheduled COD of the
"Settlement Period"	respective Unit; • means the time block for issue of daily generation and drawal schedules as provided in ABT;
Site	means the land over which the Project will be developed as provided in Schedule 1;
"SLDC"	means the relevant State Load Dispatch Centre as defined in the Electricity Laws, in the State where the Procurer's registered office is located:
"State Transmission Utility" or "STU"	shall have the meaning ascribed thereto in the Electricity Act 2003;
"Supplementary Bill"	means a bill other than a Monthly Bill raised by any of the Parties in accordance with Article 11;
"Tariff Payment"	means the payments under Monthly Bills as referred to in Schedule 6 and the relevant Supplementary Bills;
"Tariff" ·	means the tariff as computed in accordance with Schedule 6 and approved by PSERC.
"Tested Capacity"	in relation to a Unit, or the Power Station as a whole (if all the Units of the Power Station have been Commissioned) means the results of the most recent Performance Test or Repeat Performance Test carried
	out in relation to the Power Station in accordance with Article 6, Article 8 and Schedule 4 of this Agreement;
"Termination Notice"	shall mean the notice given before termination of this Agreement in accordance with relevant clauses of this Agreement
"Term of Agreement"	shall have the meaning ascribed thereto in Article 2.1;
, 	- Him
·	
L	19

JAK.

~

.

J



"Total Debt Amount"

Means the sum of the following amounts, expressed in Rupces (with all amounts denominated in currencies other than Rupces being converted to Rupces at the Reference Exchange Rate, the selling rate in Rupces for the Foreign Currency on the relevant day, as notified by the State Bank of India as its TT Rate at 12:00 noon on the date of issuance of Substitution Notice by the Lenders

- (a) the principal amount of the senior debt incurred by the Seller (as per the terms of the Financing Agreements) to finance the Project according to the Capital Structure Schedule which remains outstanding on the date of issuance of Substitution Notice by the Lender after taking account of any senior debt repayments which could have been made out of the Monthly Tariff Payments received by the Seller on or before the date of issuance of Substitution Notice by the Lender as per the terms provided in the Financing Agreements; and
- (b) all accrued interest and financing fees payable under the Financing Agreements on the amounts referred to in (a) above from the date of the Capacity Charge payment (as specified in paragraph 1.2 of Schedule 6 hereof) immediately preceding the date of issuance of Substitution Notice by the Lender or, if the Capacity Charges have not yet fallen due to be paid, from the most recent date when interest and financing fees were capitalised, and
- (c) if this Agreement is terminated during the Construction Period, any amounts owed to the Construction Contractor for work performed but not paid for under the Construction Contract (other than amounts falling due by reason of the Seller's default):

Means one steam generator, steam turbine, generator and associated auxiliaries of the Power Station

shall have the meaning ascribed thereto in Rule 24 of the CERC (Terms and Conditions of Tariff) Regulations as applicable.

means a calendar week commencing from 00:00 hours of Monday, and ending at 24:00 hours of the following Sunday;

are the charges paid by the Procurer to the CTU or STU or any other agency for the transfer of power from the Power Station switchyard end to the Procurer's network.

"Unit"

"Unscheduled Interchange" or "UI" "Week"

"Wheeling Charges" or "Transmission Charges"



1.2 Interpretation

Save where the contrary is indicated, any reference in this Agreement to:

- 1.2.1 A "Recital", an "Article", a "Schedule" and a "paragraph/Clause" shall be construed as a reference to a Recital, an Article, a Schedule and a paragraph/clause respectively of this Agreement.
- 1.2.2 An "affiliate" of any party shall mean a company that either directly or indirectly controls or is controlled by or is under common control of the same person which controls the concerned party; and control means ownership by one company of at least twenty six percent (26%) of the voting rights of the other company.
- 1.2.3 A "crore" means a reference to ten million (10,000,000) and a "lakh" means a reference to one tenth of a million (1,00,000);
- 1.2.4 An "encumbrance" shall be construed as a reference to a mortgage, charge, pledge, lien or other encumbrance securing any obligation of any person or any other type of preferential arrangement (including, without limitation, title transfer and retention arrangements) having a similar effect.
- 1.2.5 "indebtedness" shall be construed so as to include any obligation (whether incurred as principal or surety) for the payment or repayment of money, whether present or future, actual or contingent;
- 1.2.6 A "person" shall be construed as a reference to any person, firm, company, corporation, society, trust, government, state or agency of a state or any association or partnership (whether or not having separate legal personality) of two or more of the above and a person shall be construed as including a reference to its successors, permitted transferees and permitted assigns in accordance with their respective interests.
- 1.2.7 The "winding-up", "dissolution", "insolvency", or "reorganization" of a company or corporation shall be construed so as to include any equivalent or analogous proceedings under the Law of the jurisdiction in which such company or corporation is incorporated or any jurisdiction in which such company or corporation carries on business including the seeking of liquidation, winding-up, reorganization, dissolution, arrangement, protection or relief of debtors.
- 1.2.8 Words importing the singular shall include the plural and vice versa.
- 1.2.9 This Agreement itself or any other agreement or document shall be construed as a reference to this or to such other agreement or document as it may have been, or may from time to time be, amended, varied, novated, replaced or supplemented.



- 1.2.10 A Law shall be construed as a reference to such Law including its amendments or re-enactments from time to time.
- 1.2.11 A time of day shall, save as otherwise provided in any agreement or document be construed as a reference to Indian Standard Time.
- 1.2.12 Different parts of this Agreement are to be taken as mutually explanatory and supplementary to each other and if there is any inconsistency between or among the parts of this Agreement, they shall be interpreted in a harmonious manner so as to give effect to each part.
- 1.2.13 The tables of contents and any headings or sub-headings in this Agreement have been inserted for ease of reference only and shall not affect the interpretation of this Agreement.
- 1.2.14 All interest payable under this Agreement shall accrue from day to day and be calculated on the basis of a year of three hundred and sixty five (365) days.
- 1.2.17 The words "hereof" or "herein", if and when used in this Agreement shall mean a reference to this Agreement.



2 ARTICLE 2: TERM OF AGREEMENT

2.1 Effective Date and Term of Agreement

This Agreement shall come into effect from the Effective Date. This Agreement shall be valid for a term commencing from the Effective Date until the Expiry Date ("Term of Agreement") unless terminated earlier pursuant to Article 2.2. Upon the occurrence of the Expiry Date, this Agreement shall, subject to Article 18.9, automatically terminate, unless mutually, extended by the Parties on mutually agreed terms and conditions, atleast one hundred and eighty (180) days prior to the Expiry Date, subject to approval of the Appropriate Commission, as necessary.

2.2 Early Termination

This Agreement shall terminate before the Expiry Date:

- i. if either the Procurer or Seller exercises a right to terminate, pursuant to Article 3.3.2, Article 3.3.3, Article 4.5.3, Article 14.4.5 or Schedule 8 of this Agreement or any other provision of this Agreement; or
- ii. in such other circumstances as the Seller and the Procurer may agree, in writing.

2.3 Survival

2.3.1 The expiry or termination of this Agreement shall not affect accrued rights and obligations of the Parties under this Agreement, including the right to receive Liquidated Damages as per the terms of this Agreement, nor shall it affect any continuing obligations for which this Agreement provides, either expressly or by necessary implication, the survival of, post its expiry or termination.

23



3 ARTICLE 3 : CONDITIONS SUBSEQUENTTO BE SATISFIED BY THE SELLER AND THE PROCURER

3.1 Satisfaction of conditions subsequent by the Seller and the Procurer

- 3.1.1 Prior to the Effective Date, the Seller have provided to the Procurer, the Performance Guarantee from any of the banks specified by the Procurer, of an aggregate amount of Rupees 40.5 crores. Subject to Article 3.4, the Performance Guarantee shall be initially valid till three (3) Months after the Scheduled COD of the Power Station and which shall be extended from time to time to be valid up to three (3) Months after the actual COD of the Power Station. In case the validity of Performance Guarantee is expiring before the validity period specified in this Article, the Seller shall at least thirty (30) days before the expiry of the Performance Guarantee or extend validity of existing Performance Guarantee which is valid and in force till the validity period specified in this Article.
- 3.1.2 The Seller agrees and undertakes to duly perform and complete the following activities within 12 (twelve) Months from the Effective Date unless such completion is affected due to any Force Majeure event or if any of the activities is specifically waived in writing by the Procurer :

i). the Seller shall have received the Initial Consents as mentioned in Schedule 2, either unconditionally or subject to conditions which do not materially prejudice its rights or the performance of its obligations under this Agreement;

ii) the Seller shall have executed Fuel Supply Agreement and provided the

. copies of the same to the Procurer;

iii)the Seller shall have

100

a) awarded the main plant contract for boiler, turbine and generator

("BTG"), for the Project and shall have given to such contractor an

__irrevocable notice to proceed; and

b) the Seller shall have achieved Financial Closure;

24



iv. the Seller shall have made available to the Procurer the data with respect to

the Project for design of Interconnection Facilities and Transmission Facilities, if required; 83

v. the Seller shall have finalised the specific delivery point for supply of power in consultation with the Procurer;

vi. the Seller shall have taken the possession of the land for the Power Station and have paid the Price of the Land, if any to the State Government authority acquiring the land,

vii. the Seller shall have provided an irrevocable letter to the Lenders duly accepting and acknowledging the rights provided to the Lenders under the terms of this Agreement and all other Project Documents.

3.1.3 Joint responsibilities of the Procurer and the Seller

The Procurer and Seller shall jointly appoint the Independent Engineer for the purposes of carrying out the functions as specified in Article 4.7.1, Article 6, Article 8 and Article 12, herein within a period of eight (8) months from the Effective Date.

3.2 Progress Reports

ARK

The Seller and the Procurer shall notify one another in writing at least once a Month on the progress made in satisfying the conditions in Articles 3.1.2, and 3.1.3.

3.3 Consequences of non-fulfillment of conditions under Article 3.1

3.3.1 If any of the conditions specified in Article 3.1.2 is not duly fulfilled by the Seller even within three (3) Months after the time specified under Article 3.1.2, then on and from the expiry of such period and until the Seller has satisfied all the conditions specified in Article 3.1.2, the Seller shall be liable to furnish to the Procurer additional weekly Performance Guarantee of Rs. 2.025 crores within two (2) business days of expiry of every such week. Such additional Performance Guarantee shall become part of the Performance Guarantee and all the provisions of this Agreement shall be construed accordingly. The Procurer shall be entitled to hold and/or invoke the Performance Guarantee, including such increased Performance Guarantee, in accordance with the provisions of this Agreement.



3.3.2 Subject to Article 3.3.3 if:

- (i). fulfilment of any of the conditions specified in Article 3.1.2 is delayed beyond the period of three (3) Months and the Seller fails to furnish any additional Performance Guarantee to the Procurer in accordance with Article 3.3.1 hereof; or
- (ii). the Seller furnishes additional Performance Guarantee to the Procurer in accordance with Article 3.3.1 hereof but fails to fulfil the conditions specified in Article 3.1.2 for a period of eight (8) months beyond the period specified therein,

the Procurer or Seller shall have the right to terminate this Agreement by giving a Termination Notice to the Seller / Procurer in writing of at least seven (7) days.

If the Procurer or the Seller elect to terminate this Agreement in the event specified in the preceding paragraph of this Article 3.3.2, the Seller shall be liable to pay to the Procurer an amount of Rupees 54 crores only as liquidated damages. The Procurer shall be entitled to recover this amount of damages by invoking the Performance Guarantee to the extent of Rupees 54 crores and shall then return the balance Performance Guarantee, if any, to the Seller. If the Procurer is unable to recover the said amount of Rupees 54 crores or any part thereof from the Performance Guarantee the amount not recovered from the Performance Guarantee, if any, shall be payable by the Seller to the Procurer within ten (10) days from the end of eight (8) Months period from the due date of completion of conditions subsequent.

It is clarified for removal of doubt that this Article shall survive the termination of this Agreement.

3.3.3 In case of inability of the Seller to fulfil the conditions specified in Article 3.1.2 due to any Force Majeure event, the time period for fulfilment of the Condition Subsequent as mentioned in Article 3.1.2 shall be extended for the period of such Force Majeure event, subject to a maximum extension period of ten (10) Months, continuous or non-continuous in aggregate. Thereafter, this Agreement may be terminated by either the Procurer or the Seller by giving a notice of at least seven (7) days, in writing to the other Party.

3.3.4Due to the provisions of Article 3.3.3 if there is any increase in the time period for completion of Conditions Subsequent mentioned under Article 3.1.2 there

26 X



shall be an equal increase in the time period for Scheduled COD and Scheduled Connection Date.

85

3.4 Reduction in the amount of Performance Guarantee

- 3.4.1 On the due fulfilment by the Seller of all the conditions specified under Article 3.1.2 and investment by the Seller of at least twenty five percent (25%) of the total equity required for the Project as certified by the lead lender of the Seller, the Performance Guarantee then existing shall be reduced by an aggregate amount of Rupees 13.5 crores. for the period specified in Article 3.4.2.
- 3.4.2 The Performance Guarantee specified in Article 3.4.1 hereof shall be in substitution of the earlier Performance Guarantee furnished under Article 3.1.1

The Performance Guarantee furnished under this Article shall be initially valid till three (3) Months after the Scheduled COD of the Power Station and which shall be extended from time to time to be valid up to three (3) Months after the actual COD of the Power Station.

- 3.4.3 The Performance Guarantee furnished under Article 3.1, 3.3 and 3.4 shall be for guaranteeing the due and timely completion of the Project and achievement of Scheduled Commercial Operation Date of each Unit within the time specified in this Agreement.
- 3.4.4 The failure on the part of the Seller to furnish and maintain the Performance Guarantee as mentioned above shall be a material breach of the term of this Agreement on the part of the Seller.
- 3.4.5 If the Seller fails to achieve COD of each of the Units on their respective Scheduled Commercial Operation Date specified in this Agreement, subject to conditions mentioned in Article 4.5.1, the Procurer shall have the right to encash the Performance Guarantee and appropriate in its favour as liquidated damages an amount specified in Article 4.6.1, without prejudice to the other rights of the Procurer under this Agreement.

3.5 Return of Performance Guarantee

叔

3.5.1 The Performance Guarantee as submitted by Seller in accordance with Article 3.4 shall be released by the Procurer within three (3) Months from the actual Commercial Operation Date of the Power Station. In the event of delay in achieving Scheduled COD of any of the Units by the Seller (otherwise than due to Force Majeure event) and consequent part invocation of the Performance Guarantee by the Procurer, the Procurer shall release the Performance Guarantee if any, remaining unadjusted under Article 3.4, after the satisfactory completion by the Seller of all the requirements regarding



achieving the Scheduled Commercial Operation Date of the remaining Units of the Power Station. It is clarified that the Procurer shall also return/release the Performance Guarantee in the event of (i) applicability of Article 3.3.2 to the extent the Performance Guarantee is valid for an amount in excess of Rupees 54 crores, or (ii) termination of this Agreement under Article 3.3.3.

3.5.2 The release of the Performance Guarantee shall be without prejudice to other rights of the Procurer under this Agreement.

28

122.27



ARTCLE 4 : DEVELOPMENT OF THE PROJECT

4

4.1 The Seller's obligation to build, own and operate the Project

- 4.1.1 Subject to the terms and conditions of this Agreement, the Seller undertakes to be responsible, at Seller's own cost and risk, for:
 - a) obtaining the Initial Consents and maintaining in full force and effect all Consents required by it pursuant to this Agreement and Indian Law;
 - b) executing the Project in a timely manner so as to enable each of the Units and the Power Station as a whole to be Commissioned no later than its Scheduled Commercial Operations Date and such that as much of the Contracted Capacity as can be made available through the use of Prudent Utility Practices will be made available reliably to meet the Procurer's scheduling and dispatch requirements throughout the term of this Agreement
 - c) owning the Project throughout the term of this Agreement free and clear of encumbrances, except those expressly permitted by Article 16;
 - d) procure the requirements of electricity at the Project (including construction, commissioning and start-up power) and to meet in a timely manner all formalities for getting such a supply of electricity;
 - e) provide on a timely basis relevant information on Power Station specifications which may be required for interconnecting system with the transmission system;
 - f) fulfilling all other obligations undertaken by him under this Agreement.
 - 4.2 Procurer's obligation
 - Subject to the terms and conditions of this Agreement, the Procurer:
 - a) shall be responsible for procuring the Interconnection and Transmission Facilities¹⁴-to enable the Power Station to be connected to the Grid System not later than the Scheduled Connection Date;
 - b) shall ensure that the Seller is provided an electrical connection for reasonable construction, commissioning and start up power at the Project as reasonably requisitioned by the Seller by written intimation to the Procurer, on the then prevalent terms and conditions as applicable to such consumers.;

20



shall be responsible for payment of the Transmission Charges and RLDC and SLDC charges;

- d) shall make all reasonable arrangements for the evacuation of the Infirm Power from the Power Station; subject to the availability of transmission
 ines and
- e) fulfilling obligations undertaken by them under this Agreement.

4.2 Purchase and sale of Available Capacity and Scheduled Energy

4.3.1 (a) Subject to the terms and conditions of this Agreement, the Seller undertakes to sell to the Procurer, and the Procurer undertakes to pay the Tariff for all of the Available Capacity up to the Contracted Capacity and Scheduled Energy of the Power Station, throughout the term of this Agreement.

b) Prior to the Commercial Operation Date of any generating Unit, the Seller shall sell and the Procurer shall purchase all the Infirm Power produced by that generating Unit for the consideration which shall be as per clause 1.2.9 of Schedule 6.

4.3.2Unless otherwise instructed by the Procurer, the Seller shall sell all the Available Capacity up to the Contracted Capacity of the Power Station pursuant to Dispatch Instructions.

4.4 Right to Available Capacity and ScheduledEnergy

1

4.4.1 Subject to other provisions of this Agreement, the entire Contracted Capacity of the Power Station and all the Units of the Power Station shall at all times be for the exclusive benefit of the Procurer and the Procurer shall have the exclusive right to purchase the entire Contracted Capacity from the Seller. The Seller shall not grant to any third party or allow any third party to obtain any entitlement to the Available Capacity and/or ScheduledEnergy

4.4.2 Notwithstanding Article 4.4.1, the Seller shall be permitted to sell power, being a part of the Available Capacity of the Power Station to third parties if:

 there is a part of Available Capacity which has not been Dispatched by the Procurer, ordinarily entitled to receive such part;

4.4.3 If the Procurer does not avail of power upto the Available Capacity provided by the Seller and the provisions of Article 4.4.2 have been complied with, the Seller shall be entitled to sell such Available Capacity not procured, to any person without losing the right to receive the Capacity Charges from the Procurer for such un-availed Available Capacity. In such a case, the sale realization in excess of Energy Charges,

30



shall be equally shared by the Seller with the Procurer. In the event, the Seller sells such Available Capacity to the shareholders of the Seller or any direct or indirect affiliate of the Seller/shareholders of the Seller without obtaining the prior written consent of the Procurer, the Seller shall be liable to sell such Available Capacity to such entity at tariffs being not less than the Tariff payable by the ProcurerIn such case, fifty percent (50%) of the excess over Energy Charges recovered by the Seller from sale to third party shall be retained by the Seller and the balance fifty percent (50%) shall be provided by the Seller to the Procurer/s During this period, the Seller will also continue to receive the Capacity Charges from the Procurer. Upon the Procurers intimating to the Seller of its intention and willingness to avail of the part of the Available Capacity not availed of and therefore sold to the third party, the Seller and said third party, commence supply of such capacity to the Procurer/s from the later of two (2) hours from receipt of notice in this regard from the Procurer/s or the time for commencement of supply specified in such notice.

4.4.4 The Seller shall not itself use any of the electricity generated by the Power Station during the term of this Agreement, except for the purpose of meeting the Power Station's auxiliary load requirements, as per the norms laid down by the Appropriate Commission and load requirements of the housing colony for the staff. 4.4.5 The sale under Unscheduled Interchange shall not be considered as sale to third party for the purposes of this Agreement.

4.5 Extensions of time

4.5.1 In the event that:

- (a) the Seller is prevented from performing its obligations under Article
 4.1.1(b) by the stipulated date, due to any Procurer Event of Default; or
- (b) a Unit cannot be Commissioned by its Scheduled Commercial Operations Date because of Force Majeure Events.

the Scheduled Commercial Operations Date, the Scheduled Connection Date and the Expiry Date shall be deferred, subject to the limit prescribed in Article 4.5.3, for a reasonable period but not less than 'day for day' basis, to permit the Seller through the use of due diligence, to overcome the effects of the Force Majeure Events affecting the Seller or in the case of the Procurer's Event of Default, till such time such default is rectified by the Procurer.

4.5.2 If the Parties have not agreed, within thirty (30) days after the affected Party's performance has ceased to be affected by the relevant circumstance, on how long the Scheduled Commercial Operation Date, the Scheduled Connection

[31	
ATA -		the
•		



Date or the Expiry Date should be deferred by, any Party may raise the Dispute to be resolved in accordance with Article 17.

4.5.3 In case of extension occurring due to reasons specified in Article 4.5.1(a), the original Scheduled Commercial Operation Date of any Unit or the original Scheduled Commercial Operations Date of the Power Station as a whole, would not be extended by more than two (2) years or the date on which the

Seller elects to terminate this Agreement, whichever is earlier.

As a result of such extension, the date newly determined shall be deemed to be the Scheduled Commercial Operations Date for the purposes of this Agreement.

4.6 Liquidated damages for delay in providing Contracted Capacity

4.6.1 If any Unit is not Commissioned by its Scheduled Commercial Operation Date other than for the reasons specified in Article 4.5.1, the Seller shall pay to the Procurer liquidated damages, for the delay in such Commissioning or making the Unit's Contracted Capacity available for dispatch by such date. The sum total of the liquidated damages payable by the Seller to the Procurer for such delayed Unit shall be calculated as follows:

SLDb = [CCun x dn x DR1], if dn <= 60

 $SLDb = [CCun \times 60 \times DR1] + [CCun \times (dn - 60) \times DR2], \text{ if } dn > 60$

Where:

a) "SLDb" are the liquidated damages payable by the Seller during the period beginning with the day from the Scheduled Commercial Operation Date of a Unit up to and including the day on which Unit is actually Commissioned;
b) "CCun" is the Contracted Capacity of Unit "n";

c) "d" is the number of days in the period beginning with the day after the Scheduled Commercial Operation Date of Unit "n" up to and including the day on which such Unit is actually Commissioned;

d) "DR1" is Rs. Ten Thousand (10,000) of damages per MW per day of delay in case "d" is less than 60 days and "DR2" is Rs. Fifteen Thousand (15,000) of damages per MW per day of delay in case "d" is equal to or more than 60 days

4.6.2 The Seller's maximum liability under this Article 4.6 shall be limited to the amount of liquidated damages calculated in accordance with Article 4.6.1 for

	32	99 99 - 199	- Hr
AF.			In
			· *



and upto twelve (12) Months of delay for the Unit. Provided that in case of failure of the Seller to Commission the Unit even after expiry of twelve (12) Months from its Scheduled Commercial Date, the provisions of Article 14 shall apply.

- 4.6.3 The Seller shall pay the amount calculated pursuant to Article 4.6.1 to the Procurer within ten (10) days of the earlier of:
 - (a) the date on which the Unit is actually Commissioned; or

(b) expiry of the twelve (12) month period mentioned in Article 4.6.2.

If the Seller fails to pay the amount of damages within the said period of ten (10) days, the Procurer shall be entitled to recover the said amount of the liquidated damages by invoking the Performance Guarantee. If the then existing Performance Guarantee is for an amount which is less than the amount of the liquidated damages payable by the Seller to the Procurer under this Article 4.6, then the Seller shall be liable to forthwith pay the balance amount.

4.6.4 The Parties agree that the formula specified in Article 4.6.1 for calculation of liquidated damages payable by the Seller under this Article 4.6, read with Article 14 is a genuine and accurate pre-estimation of the actual loss that will be suffered by the Procurer in the event of Seller's delay in achieving Commissioning of a Unit by its Scheduled COD.

4.7 Liquidated damages for delay due to Procurer Event of Default and Non Natural Force Majeure Events and Natural Force Majeure Event (affecting the Procurer)

4.7.1 If

- a Unit cannot be commissioned by its Scheduled Commercial Operations Date, due to a Procurer Event of Default or due to Non Natural Force Majeure Event (or Natural Force Majeure affecting the Procurer) provided such Non Natural Force Majeure Event (or Natural Force Majeure affecting the Procurer/s) has continued for a period of more than three (3) continuous or non-continuous Months; or
- b) a Unit is available for conducting Commissioning Tests and is anticipated to be capable of duly completing the Commissioning Tests as certified by the Independent Engineer, but the said Commissioning Tests are not undertaken or completed due to such Procurer Event of Default or due to Non Natural Force Majeure Event (or Natural Force



Majeure affecting the Procurer) provided such Non Natural Force Majeure Event (or Natural Force Majeure affecting the Procurer) has continued for a period of more than three (3) continuous or noncontinuous Months:

such Unit shall, until the effects of the Procurer Event of Default or of Non Natural Force Majeure Event (or Natural Force Majeure affecting the Procurer) no longer prevent the Seller from undertaking a Commissioning Test/s, be deemed to have a Tested Capacity equal to the Contracted Capacity and to this extent, be deemed to have been Commissioned with effect from the Scheduled COD without taking into account delay due to such Procurer Event of Default or Non Natural Force Majeure Event (or Natural Force Majeure affecting the Procurer) and shall be treated as follows.

a) In case of delay on account of the Procurer Event of Default, the Procurer shall make payment to the Seller of Capacity Charges calculated on Normative Availability of Contracted Capacity of such Unit for and during the period of such delay

b) In case of delay on account of Direct Non Natural Force Majeure Event, the Procurer shall make payment to the Seller of Capacity Charges calculated on Normative Availability of Contracted Capacity of such Unit for the period of such events in excess of three (3) continuous or non-continuous Months in the manner provided in (d) below.

c) In case of an Indirect Non Natural Force Majeure Event (or Natural Force Majeure affecting the Procurer), the Procurer shall make payments for amounts ("Debt Service") relatable to such Unit, which are due under the Financing Agreements, subject to a maximum of Capacity Charges based on Normative Availability, for the period of such events in excess of three (3) continuous or non-continuous Months in the manner provided in (d) below.

d) In case of delay due to Direct and Indirect Non Natural Force Majeure Events (or Natural Force Majeure affecting the Procurer), the Procurer shall be liable to make payments mentioned in (b) and (c) above, after Commissioning of the Unit, in the form of an increase in Capacity Charges. Provided such increase in Capacity Charges shall be determined by Appropriate Commissionon the basis of putting the Selfer in the same economic position as the Seller would have been in case the Seller had been paid amounts mentioned in (b) and (c) above in a situation where the Direct Non Natural Force Majeure or Indirect Non Natural Force Majeure Event, as the case may be, had not occurred.

34



For the avoidance of doubt, it is specified that the charges payable under this Article 4.7.1 shall be paid by the Procurer as per the Contracted Capacity.

4.7.2 In every case referred to in Article 4.7.1 hereinabove, the Seller shall undertake a Commissioning Test/s as soon as reasonably practicable [and in no event later than two (2) weeks or such longer period as mutually agreed between the Seller and the Procurer] after the point at which it is no longer prevented from doing so by the effects of Force Majeure Events or Procurer Event of Default (as appropriate) and if such Commissioning Test/s is not duly completed and / or demonstrate/s a Tested Capacity of a Unit which is less than ninety five (95) percent of its Contracted Capacity, then:

- a) The Unit which fails the Commissioning Tests, shall be deemed to have not been Commissioned from the deemed COD referred to in Article 4.7.1;
- b) The Seller shall repay to the Procurer, all sums received by way of Capacity Charge for such Unit along with interest at the same rate as Late Payment Surcharge; and
- c) If the Seller fails to achieve Commissioning by the Scheduled Commercial Operation Date, it shall also pay liquidated damages to the Procurer for such Unit calculated in accordance with Article 4.6.

4.8 Limit on amounts payable due to default

- 4.8.1 The Parties expressly agree that the Procurer's only liability for any loss of profits or any other loss of any other kind or description whatsoever (except claims for indemnity under Article 15), suffered by the Seller by reason of the Procurer's failure to meet its obligations under Article 4.2(a) to Article 4.2(d) shall be to pay the Seller the amounts specified in Article 4.7 and Article 14.
- 4.8.2 Similarly, Seller's only liability for any loss suffered by the Procurer of any kind or description whatsoever (except claims for indemnity under Article 15), by reason of the Seller's failure to meet its obligations of Commissioning the various Units on their Scheduled COD, shall be as per Article 4.6 and Article-14.

35



5 ARTICLE 5 CONSTRUCTION

5.1 Seller's Construction Responsibilities

The Seller shall be responsible for designing, constructing, erecting, commissioning, completing and testing the Power Station in accordance with the following, it being clearly understood that in the event of inconsistency between two or more of the following, the order of priority as between them shall be the order in which they are placed, with 'applicable law' being the first:

- a) applicable Law;
- b) the Grid Code;
- c) the terms and conditions of this Agreement;
- d) the Functional Specifications; and
- e) Prudent Utility Practices.

Notwithstanding anything to the contrary contained in this PPA, the Seller shall ensure that the technical parameters or equipment limits of the Project shall always be subject to the requirements as specified in points (a) to (e) above and under no event shall over-ride or contradict the provisions of this Agreement and shall not excuse the Seller from the performance of his obligations under this Agreement.

5.2 The Site

The Seller acknowledges that, before entering into this Agreement, it has had sufficient opportunity to investigate the Site and accepts full responsibility for its condition (including but not limited to its geological condition, on the Site, the adequacy of the road and rail links to the Site and the availability of adequate supplies of water) and agrees that it shall not be relieved from any of its obligations under this Agreement or be entitled to any extension of time or financial compensation by reason of the unsuitability of the Site for whatever reason.

5.3 Information Regarding Interconnection Facilities

The Procurer shall provide the Seller, on a timely basis, all information with regard to the Interconnection and Transmission Facilities as is reasonably necessary to enable the Seller to design, install and operate all interconnection plant and apparatus on the Seller's side of the Interconnection Point. A

36



95

5.4 Quality of Workmanship

The Seller shall ensure that the Power Station is designed, built and completed in a good workmanlike manner using sound engineering construction practices and using only materials and equipment that are new and of international utility grade quality such that, the useful life of the Power Station will be till the Expiry Date.

The Seller shall ensure that design, construction and testing of all equipment, facilities, components and systems of the Project shall be in accordance with Indian Standards and Codes issued by Bureau of Indian Standards and/or internationally recognised Standards and Codes, such as:

- i. American National Standards Institute (ANS)
- ii. American Society of Testing and Materials (ASTM)
- iii. American Society of Mechanical Engineers (ASME)
- iv. American Petroleum Institute (API)
- v. Standards of the Hydraulic Institute, USA
- vi. International Organization for Standardization (ISO)
- vii. Japanese Industrial Standards (JIS)
- viii. Tubular Exchanger Manufacturer's Association (TEMA)
- ix. American Welding Society (AWS)
- x. National Electrical Manufacturers Association (NEMA)

xi. National Fire Protection Association (NFPA)

- xii. International Electro-Technical Commission (IEC)
- xiii. Expansion Joint Manufacturers Association (EJMA)

xiv. Heat Exchange Institute (HEI)

- xv. American Water Works Association (AWWA)
- xvi. Deutsches Institut fur Normung (DIN)

Other international standards, established to be equivalent or superior to the above standards shall also be acceptable. However, in the event of any conflict between the requirements of the international codes and standards and the requirements of the Indian standards/regulations, the latter shall prevail.

5.5 Consents

The Seller shall be responsible for obtaining all Consents (other than those required for the Interconnection and Transmission Facilities) and the Initial Consents required for developing, financing, constructing, operating and maintenance of the Project and maintaining/ renewing all such Consents in



order to carry out its obligations under this Agreement in general and this Article 5 in particular and shall supply the Procurer promptly with copies of each application that it submits, and copy/ies of each consent/approval/license which it obtains. For the avoidance of doubt, it is clarified that the Seller shall also be responsible for maintaining/renewing the Initial Consents and for fulfilling all conditions specified therein.

5.6 Construction Documents

The Seller shall retain at the Site and make available for inspection to the Procurer at all reasonable times copies of the results of all tests specified in Schedule 4 hereof.

5.7 Co-ordination of Construction Activities

- 5.7.1 Before the tenth (10th) day of each Month, during the Construction Period:
 - (a) the Seller shall prepare and submit to the Procurer a monthly progress report, in the Agreed Form; and
 - (b) The Procurer shall prepare and submit to the Seller a monthly progress report, in the Agreed Form, regarding the Interconnection and Transmission Facilities.

The Seller and the Procurer shall designate from time to time, by giving a written notice to the other Party up to five (5) of its/their employees who shall be responsible for coordinating all construction activities relating to the Project and who shall have access at all reasonable times to the other Party's land for the purpose of apprising the progress of the work being carried on, subject to such designated persons or the Party appointing them giving reasonable notice to the other Party of such visit and subject to their complying with all reasonable safety procedures.

38

For the avoidance of doubt, it is clarified that the total number of the representatives of the Procurer shall not exceed five (5). \land



ARTICLE 6 : SYNCHRONISATION, COMMISSIONING AND COMMERCIAL OPERATION

6.1 Synchronization

6

6.1.1 The Seller shall give the Procurer and RLDC or SLDC at least sixty (60) days advance preliminary written notice and at least thirty (30) days advance final written notice, of the date on which it intends to synchronise a Unit to the Grid System.

6.1.2 Subject to Article 6.1.1, a Unit may be synchronised by the Seller to the Grid System when it meets all connection conditions prescribed in any Grid Code then in effect and otherwise meets all other Indian legal requirements for synchronisation to the Grid System

6.2 Commissioning

AR .

- 6.2.1 The Seller shall be responsible for ensuring that each Unit is Commissioned in accordance with Schedule 4 at its own cost, risk and expense.
- 6.2.2 The Seller shall give the Procurer and the Independent Engineer not less than ten (10) days prior written notice of Commissioning Test of each Unit.
- 6.2.3 The Seller, the Procurer and the Independent Engineer (individually) shall each designate qualified and authorised representatives to witness and monitor Commissioning Test of each Unit.
- 6.2.4 Testing and measuring procedures applied during each Commissioning Test shall be in accordance with the codes, practices and procedures mentioned in Schedule 4 of this Agreement.
- 6.2.5 Within five (5) days of a Commissioning Test, the Seller shall provide the Procurer and the Independent Engineer with copies of the detailed Commissioning Test results. Within five (5) days of receipt of the Commissioning Test results, the Independent Engineer shall provide to the Procurer and the Seller in writing, his findings from the evaluation of Commissioning Test results, either in the form of Final Test Certificate certifying the matters specified in Article 6.3.1 or the reasons for nonissuance of Final Test Certificate.

97



6.3 Commercial Operation

- 6.3.1 A Unit shall be Commissioned on the day after the date when the Procurer receives a Final Test Certificate of the Independent Engineer stating that:
 - a)* the Commissioning Tests have been carried out in accordance with Schedule -4 and are acceptable to him; and
 - b) the results of the Performance Test show that the Unit's Tested Capacity, is not less than ninety five (95) percent of its Contracted Capacity, as existing on the Effective Date.
- 6.3.2 If a Unit fails a Commissioning Test, the Seller may retake the relevant test, within a reasonable period after the end of the previous test, with three (3) day's prior written notice to the Procurer and the Independent Engineer. Provided however, the Procurer shall have a right to require deferment of any such re-tests for a period not exceeding fifteen (15) days, without incurring any liability for such deferment, if the Procurer are unable to provide evacuation of power to be generated, due to reasons outside the reasonable control of the Procurer or due to inadequate demand in the Grid.
- 6.3.3 The Seller may retake the Performance Test by giving at least fifteen (15) days advance notice in writing to the Procurer, up to eight (8) times, during a period of one hundred and eighty (180) days ("Initial Performance Retest Period") from a Unit's COD in order to demonstrate an increased Tested Capacity over and above as provided in Article 6.3.1 (b). Provided however, the Procurer shall have a right to require deferment of any such re-tests for a period not exceeding fifteen (15) days, without incurring any liability for such deferment, if the Procurer are unable to provide evacuation of power to be generated, due to reasons outside the reasonable control of the Procurer or due to inadequate demand in the Grid.
- 6.3.4 (i) If a Unit's Tested Capacity after the most recent Performance Test mentioned in Article 6.3.3 has been conducted, is less than its Contracted Capacityas existing on the Effective Date, the Unit shall be de-rated with the following consequences in each case with effect from the date of completion of such most recent test:
 - a) the Unit's Contracted Capacity shall be reduced to its Tested Capacity, as existing at the most recent Performance Test referred to in Article 6.3.3

40



and Capacity Charges shall be paid with respect to such reduced Contracted Capacity;

- b) the Seller shall not be permitted to declare the Available Capacity of the Unit at a level greater than its Tested Capacity;
- c) the Availability Factor of the derated Unit shall be calculated by reference to the reduced Contracted Capacity; and
- d) the Capital Cost and each element of the Capital Structure Schedule shall be reduced in proportion to the reduction in the Contracted Capacity of the Power Station as a result of that derating (taking into account the Contracted Capacity of any Unit which has yet to be Commissioned).

(ii) If at the end of Initial Performance Retest Period or the date of the eighth Performance Test mentioned in Article 6.3.3, whichever is earlier, , the Tested Capacity is less than the Contracted Capacity as existing on the Effective Datethe consequences mentioned in Article 8.2.2 shall apply for a period of one year. Provided that such consequences shall apply with respect to the Tested Capacity existing at the end of Initial Performance Retest Period or the date of the eighth Performance Test mentioned in Article 6.3.3, whichever is earlier

6.3.5 If a Unit's Tested Capacity as at the end of the Initial Performance Retest Period or the date of the eighth Performance Test mentioned in Article 6.3.3, whichever is earlier, is found to be more than it's Contracted Capacity as existing on the Effective Date, the Tested Capacity shall be deemed to be the Unit's Contracted Capacity if the Procurer agrees and intimates the same to the Seller within thirty (30) days of receipt of the results of the last Performance Test to purchase such excess Tested Capacity and also provide to the Seller additional Letter of Credit and Collateral Arrangement for payments in respect of such excess Tested Capacity agreed to be purchased by the Procurer. In case the Procurer decides not to purchase such excess Tested Capacity to any third party and the Unit's Contracted Capacity exceeded the Contracted Capacity.

Provided that in all the above events, the Seller shall be liable to obtain/maintain all the necessary consents (including Initial Consents), permits

41

" ARI



and approvals including those required under the environmental laws for generation of such excess Tested Capacity.

6.4 Costs Incurred

The Seller expressly agrees that all costs incurred by him in synchronising, connecting, Commissioning and / or Testing or Retesting a Unit shall be solely and completely to his account and the Procurer's liability shall not exceed the amount of the Energy Charges payable for such power output, as set out in Schedule 6.

42

AR



ARTICLE 7 : OPERATION AND MAINTENANCE

43

The Parties shall comply with the provisions of the applicable Law including, in particular, Grid Code as amended form time to time regarding operation and maintenance of the Power Station and all matters incidental thereto. Provided however the Seller shall not schedule the Maintenance Outage of a Unit when another Unit of the project is shut down or expected to be shut down except under Force Majeure or when the operation of Unit is not permissible due to technical considerations.

7



8 ARTICLE 8: CAPACITY, AVAILABILITY AND DISPATCH

8.1 Repeat Performance Tests

- 8.1.1 The Procurer may from time to time during the Operating Period, but only if the Available Capacity has not been one hundred per cent (100%) of the Contracted Capacity of the commissioned units (excluding the unit(s) under planned outage for capital maintenance in consultation with the Regional Power Committee, if any) even for one continuous period of at least three (3) hours during any three continuous months, require the Seller to demonstrate a Unit's or (if all the Units have been Commissioned, the Power Station's) Tested Capacity by carrying out a further Performance Test (a "Repeat Performance Test") in accordance with this Article 8.1. A Repeat Performance Test shall be carried out in accordance with Schedule 4, save that the test shall last twenty-four (24) hours instead of seventy two (72) hours. Provided that if the Tested Capacity after such test is less than one hundred percent (100%) of the Contracted Capacity as existing on the Effective Date of the Commissioned Units, the Seller shall also have a right to conduct not more than two (2) Repeat Performance Test within a period six (6) months, by giving a notice of not less than fifteen (15) days to the Procurer for each such test. Provided that the Procurer shall have a right to require deferment of each such re-tests for a period not exceeding five (5) days, without incurring any liability for such deferment, if the Procurer are unable to provide evacuation of power to be generated, due to reasons outside the reasonable control of the Procurer or due to inadequate demand in the Grid.
- 8.1.2 The Procurer shall give the Seller not less than seven (7) days' advance written notice of the time when a Repeat Performance Test of a Unit (or if all the Units have been Commissioned, of the Power Station's) is to begin. A Repeat Performance Test may not be scheduled for any period when the Unit to be tested is due to undergo a Scheduled Outage.
- 8.1.3 The Procurer and Seller shall jointly appoint the Independent Engineer to monitor the Repeat Performance Test and to certify the results in accordance with Article 8.2.
- 8.1.4 If the Seller wishes to take any Unit, out of service for repair before a Repeat Performance Test, it shall inform the Procurer in writing before its scheduled start of the repairs and the estimated time required to complete the repairs. The Parties shall then schedule a Maintenance Outage in accordance with the Grid

44



Code to enable the Seller to carry out those repairs and in such a case, the Procurer, , shall defer the Repeat Performance Test until such Unit is returned to service following that Maintenance Outage. Provided however the Seller shall not schedule the Maintenance Outage of a Unit when another Unit of the project is shut down or expected to be shut down except under Force Majeure or when the operation of Unit is not permissible due to technical considerations.

8.1.5 The Procurer, may for reasonable cause, defer any Repeat Performance Test for up to fifteen (15) days from the date originally notified to the Seller in accordance with Article 8.1.2 if the Procurer notifies the Seller in writing at least one (1) day before the Repeat Performance Test starts, the reason for the deferral and when the test is to be rescheduled.

Provided that, such deferment shall be permitted only once in respect of each of the Repeat Performance Tests.

- 8.1.6 The Seller, the Procurer and the Independent Engineer individually shall have the right to designate qualified and authorised representatives (but not more than three each) to monitor the Repeat Performance Test.
- 8.1.7 Testing and measurement procedures applied during the Repeat Performance Test shall be in accordance with the codes, practices of procedures as generally/normally applied for the Performance Tests.
- 8.1.8 Within five (5) days of a Repeat Performance Test, the Seller shall provide the Procurer and the Independent Engineer with copies of the detailed test results.
- 8.1.9 Within one (1) Month of the date by which all the Units have been Commissioned, the Seller shall conduct a Performance Test of the Power Station (hereinafter referred to as "Power Station Performance Test") whereafter the provisions of Article 8.2 shall apply. A Power Station Performance Test shall be carried out in accordance with Article 1.1 of Schedule 4, save that the test shall last twenty-four (24) hours instead of seventy two (72) hours.

8.2 Derating

8.2.1 A Repeat Performance Test shall be concluded when the Procurer receives the Final Test Certificate of the Independent Engineer stating that the Repeat Performance Test has been carried out satisfactorily in accordance with Schedule 4 and certified the Unit's (or if all the Units have been

45



8.2.2 (i) If a Unit's (or if all the Units have been Commissioned, of the Power Station's) then current Tested Capacity as established by the Repeat Performance Test and the Final Test Certificate issued by the Independent Engineer, is less than its Contracted Capacity as existing on the Effective Date, the Seller shall not be permitted to declare the Available Capacity of the Unit (or if all the Units have been Commissioned, of the Power Station's) at a level greater than its Tested Capacity, in which case:

- the Unit's (or if all the Units have been Commissioned, of the Power a) Station's) Contracted Capacity shall be reduced to its most recent Tested Capacity and Capacity Charges shall be paid with respect to such reduced Contracted Capacity.
- b) the Availability Factor of the derated Unit (or if all the Units have been Commissioned, of the Power Station's) shall be calculated by reference to the reduced Contracted Capacity, and:
 - c) the Capital Cost and each element of the Capital Structure Schedule shall be reduced in proportion to the reduction in the Contracted Capacity of the Power Station as a result of that derating (taking into account the Contracted Capacity of any Unit which has yet to be Commissioned);

(ii) The consequences mentioned in sub-Article (i) above shall apply from the completion date of each Repeat Performance Test.. If at the end of second Repeat Performance Test conducted by the Seller or the last date of the end of the six month period referred to in Article 8.1.1, whichever is earlier, the Tested Capacity is less than the Contracted Capacity as existing on the Effective Date the consequences mentioned in Article 8.2.2 shall apply for a period of at least one year after which the Seller shall have the right to undertake a Repeat Performance Test, Provided that such consequences shall apply with respect to the Tested Capacity existing at the end of second Repeat Performance Test conducted by the Seller or the last date of the end of the six month period referred to in Article 8.1.1, whichever is earlier

8.2.3 If the Independent Engineer certifies that it is unable to give a Final Test Certificate because events or circumstances beyond the Seller's reasonable control have prevented the Repeat Performance Test from being carried out in accordance with Schedule 4, the Procurer shall reschedule an Repeat Performance Test as soon as reasonably practicable.

46

122 (



8.2.4 If a Unit's or if all the Units have been Commissioned, of the Power Station's, Tested Capacity is found to be more than it's Contracted Capacity, the provisions of Article 6.3.5 shall apply mutatis mutandis.

47

AR

. ..



~



8.3 Availability

The Seller shall comply with the provisions of the applicable Law regarding Availability including, in particular, to the provisions of the ABT and Grid Code relating to intimation of Availability and the matters incidental thereto.

8.4 Dispatch

The Seller shall comply with the provisions of the applicable Law regarding Dispatch Instructions, in particular, to the provisions of the ABT and Grid Code relating to Dispatch and the matters incidental thereto.

48

jame. S

> s i Lini



9

ARTICLE 9: METERING AND ENERGY ACCOUNTING

9.1 Meters

For installation of Meters, Meter testing, Meter calibration and Meter reading and all matters incidental thereto, the Seller and the Procurer shall follow and be bound by the Central Electricity Authority (Installation and Operation of Meters) Regulations, 2006, the Grid Code and ABT as amended and revised from time to time. In addition, the Seller shall also allow and facilitate /STU in installation of one set of required main and standby special energy meters for accurate recording of energy supplied by Seller. For these /STU meters (110V, 1A, 4-wire), the Seller shall provide the required connection from EHV current transformers/ bushing CTs/ voltage transformers/ CVTs on EHV side of all generator-transformers, station transformers and outgoing lines, of meter accuracy of 0.2 class or better. The Seller may install any further meters for its own comfort at its own cost.

9.2 RLDC / SLDC Charges

All scheduling and RLDC / SLDC charges applicable shall be borne by the Procurer.


10 ARTICLE 10: INSURANCES

10.1 Insurance

The Seller shall effect and maintain or cause to be effected and maintained during the Construction Period and Operating Period, Insurances against such risks, with such deductibles and with such endorsements and co-insured(s), which the Prudent Utility Practices would ordinarily merit maintenance of and as required under the Financing Agreements.

10.2 Application of Insurance Proceeds

Save as expressly provided in this Agreement or the Insurances, the proceeds of any insurance claim made due to loss or damage to the Project or any part of the Project shall be first applied to reinstatement, replacement or renewal of such loss or damage.

If a Natural Force Majeure Event renders the Project no longer economically and technically viable and the insurers under the Insurances make payment on a "total loss" or equivalent basis, the Procurer shall have no claim on such proceeds of such Insurance

10.3 Effect on liability of the Procurer

50

Notwithstanding any liability or obligation that may arise under this Agreement, any loss, damage, liability, payment, obligation or expense which is insured or for which the Seller can claim compensation, under any Insurance shall not be charged to or payable by the Procurer,



11 ARTICLE 11 : BILLING AND PAYMENT

11.1 General

From the COD of the first Unit, the Procurer shall pay the Seller the Monthly Tariff Payment, on or before the Due Date, , determined in accordance with this Article 11 and Schedule 6. All Tariff payments by the Procurer shall be in Indian Rupees.

The Procurer shall pay the seller for any Electrical Output from the Seller prior to the Commercial Operation Date ("Infirm Power") of a Unit the Energy Charges for Infirm Power generated by such Unit. The quantum of Infirm Power generated by Units synchronized but not have been put on COD shall be computed from the energy accounting and audit meters installed at the Power Station as per Central Electricity Authority (installation and operation of meters) Regulations 2006 as amended from time to time.

11.2 Delivery and content of Monthly Bills

11.2.1 The Seller shall issue to the Procurer a signed Monthly Bill for the immediately preceding Month.

Provided that:

-

- i. if the COD of the first Unit falls during the period between the first (1st) day and up to and including the fifteenth (15th) day of a Month, the first Monthly Bill shall be issued for the period until the last day of such Month, or
- ii. if the COD of the first Unit falls after the fifteenth (15th) day of a Month, the first Monthly Bill shall be issued for the period commencing from the COD of the first Unit until the last day of the immediately following Month.

Provided further that if a Monthly Bill is received on or before the second (2nd) day of a Month, it shall be deemed to have been received on the second (2nd) Business Day of such Month.

11.2.2 Each Monthly Bill and Provisional Bill shall include:

51

i. Availability and energy account for the relevant Month as per REA for



Monthly Bill and RLDC's daily energy account for Provisional Bill;

- ii. the Seller's computation of the components of the Monthly Tariff Payment in accordance with Schedule 6 and
- iii. supporting data, documents and calculations in accordance with this Agreement.

11.3 Payment of Monthly Bills

11.3.1 The Procurer shall pay the amount payable under Monthly Bill on the Due Date to such account of the Seller, as shall have been previously notified by the Seller to the Procurer in accordance with Article 11.3.3 below.

All payments made by the Procurer shall be appropriated by the Seller in the following order of priority:

- 1. towards Late Payment Surcharge, payable by the Procurer, if any;
- 2. towards earlier unpaid Monthly Bill, if any; and
- 3. towards the then current Monthly Bill.
- 11.3.2 All payments required to be made under this Agreement shall only include any deduction or set off for:
 - i. deductions required by the Law; and
 - ii. amounts claimed by the Procurer from the Seller, through an invoice duly acknowledged by the Seller, to be payable by the Seller, and not disputed by the Seller within thirty (30) days of receipt of the said involce and such deduction or set-off shall be made to the extent of the amounts not disputed. It is clarified that the Procurer shall be entitled to claim any set off or deduction under this Article, after expiry of the said 30 day period.

Provided further, the maximum amounts that can be deducted or set-off by the Procurer under this Article in a Contract Year shall not exceed Rupees 13.5 crores only, except on account of payments under sub Article (i) above.

11.3.3 The Seller shall open a bank account at Patiala (the "Designated Account") for all Tariff Payments to be made by the Procurer to the Seller, and notify the Procurer of the details of such account at least ninety (90) days before the dispatch of the first Monthly Bill to the Procurer. The Procurer shall instruct its bankers to make all payments under this Agreement to the Designated Account and shall notify the Seller of such instructions on the same day. The Procurer shall also designate a bank account at Patiala for payments to be



made by the Seller (including Supplementary Bills) to the Procurer and notify the Seller of the details of such account ninety (90) days before the COD of the first Unit.

- 11.3.4 In the event of delay in payment of a Monthly Bill by the Procurer beyond its Due Date month billing, a Late Payment Surcharge shall be payable by the Procurer to the Seller at the rate of two (2) percent in excess of the applicable SBAR per annum, on the amount of outstanding payment, calculated on a day to day basis (and compounded with Monthly rest), for each day of the delay.
- 11.3.5 For payment of any Bill before Due Date, the following rebate shall be paid by the Seller to the Procurer in the following manner.
 - a) Provisional Bill will be raised by the Seller on the last Business day of the Month where the Capacity Charges shall be based on the Declared Capacity for the full Month and the Energy Charges shall be based on the final implemented Scheduled Energy upto 25th day of the Month. Rebate shall be payable at the rate of two point two five percent (2.25%) of the amount (which shall be the full amount due under the Provisional Bill) credited to Seller's account on first day of the Month and rebate amount shall reduce at the rate of zero point zero five percent (0.05%) for each day, upto fifth (5th) day of the Month.
 - b) Applicable rate of rebate at (a) above shall be based on the date on which payment has been actually credited to the Seller's account. Any delay in transfer of money to the Seller's account, on account of public holiday, bank holiday or any other reasons shall be to the account of the Procurers.
 - c) Two percent (2%) rebate for credit to Sellers account made within one (1) Day of the presentation of Monthly Bill for the Month for which the Provisional Bill was raised earlier.
 - d) For credit to Seller's account made on other days the rebate on Monthly Bill shall be as under:

Number of days before Due Date of Monthly Bill	Rates of Rebate applicable
29	Two percent (2.00%)
Each day thereafter upto the Due Date	2% less [0.033% x {29 less number of days before Due Date when the payment is made by the Procurers}]

e) Rebate of two point two five percent (2.25%) to two point zero five percent (2.05%) will be available only if the Procurer credits one hundred

53



percent (100%) of the Provisional Bill within first five (5) days of the Month to Seller's account/designated account and balance amount, if any, based on Monthly Bill (as per REA) within the Month.

- f) In the event only part amount of Provisional Bill is credited to Seller's
 account, within first five (5) days and the balance amount is credited to Sellers account during other days of the Month, rebate will be paid on such part amount, at the rate of two percent (2%) plus zero point zero three three percent (0.033%) per day for the number of days earlier than the 6th day when such part amount is credited to Sellers' account;
- g) The above rebate will be allowed only if the Procurer credits to Seller's account the full Monthly Bill.
- h) No rebate shall be payable on the bills raised on account of Change in Law relating to taxes, duties and cess;
- If the Provisional Bill has not been paid by the date of receipt of the Monthly Bill then such Provisional Bill shall not be payable, provided in case the Provisional Bill has already been paid, then only the difference between the Monthly Bill and Provisional Bill shall be payable.

11.4 Payment Mechanism

11.4.1 Letter of Credit:

The Procurer shall provide to the Seller, in respect of payment of its Monthly Bills, a monthly unconditional, revolving and irrevocable letter of credit ("Letter of Credit"), opened and maintained by the Procurer, which may be drawn upon by the Seller in accordance with Articles 11.4.1.1 through 11.4.1.5. The Letter of Credit shall be provided from the bank which is appointed as Default Escrow Agent under the Default Escrow Agreement.

11.4.1.1 Not later than one (1) Month prior to the the Scheduled COD `of the first Unit, the Procurer shall through a scheduled bank at Patiala open a Letter of Credit in favour of the Seller, to be made operative from a date prior to the Due Date of its first Monthly Bill under this Agreement. The Letter of Credit shall have a term of twelve (12) Months and shall be renewed annually, for an amount equal to:

54



- i) for the first Contract Year, equal to one point one (1.1) times the estimated average monthly billing based on Normative Availability;
- ii) for each subsequent Contract Year, equal to the one point one (1.1)
 times the average of the Monthly Tariff Payments of the previous Contract Year plus the estimated monthly billing during the current year from any additional Unit(s) expected to be put on COD during the current Contract Year based on Normative Availability.

Provided that the Seller shall not draw upon such Letter of Credit prior to the Due Date of the relevant Monthly Bill, and shall not make more than one drawal in a Month.

- Provided further that if at any time, such Letter of Credit amount falls short of the amount specified in Article 11.4.1.1 or 11.4.1.8 otherwise than by reason of drawal of such Letter of Credit by the Seller, the Procurer shall restore such shortfall within seven (7) days.
- 11.4.1.2 The Procurer shall cause the scheduled bank issuing the Letter of Credit to intimate the Seller, in writing regarding establishing of such irrevocable Letter of Credit.

In case of drawal of the Letter of Credit by the Seller in accordance with the terms of this Article 11.4.1, the amount of the Letter of credit shall be reinstated in the manner stated in Article 11.4.2.3 of this Agreement.

11.4.1.4

11.4.1.3

- If the Procurer fails to pay a Monthly Bill or part thereof within and including the Due Date, then, subject to Article 11.6.7, the Seller may draw upon the Letter of Credit, and accordingly the bank shall pay without any reference or instructions from the Procurer, an amount equal to such Monthly Bill or part thereof plus Late Payment Surcharge, if applicable, in accordance with Article 11.3.4 above, by presenting to the scheduled bank issuing the Letter of Credit, the following documents:
- i) a copy of the Monthly Bill which has remained unpaid by the Procurer;
- ii) a certificate from the Seller to the effect that the bill at item (i) above, or specified part thereof, is in accordance with the Agreement and has remained unpaid beyond the Due Date; and

55



iii) calculations of applicable Late Payment Surcharge, if any.

For the avoidance of doubt it is clarified that the Seller shall not be entitled to drawdown on the Letter of Credit for any failure of the Procurer to pay a Supplementary Bill.

- 11.4.1.5 The Procurer shall ensure that the Letter of Credit shall be renewed not later than forty five (45) days prior to its expiry.
- 11.4.1.6 All costs relating to opening and maintenance of the Letter of Credit shall be borne by the Procurer, however, Letter of Credit negotiation charges shall be borne and paid by the Seller.

11.4.1.7 Where necessary, the Letter of Credit may also be substituted by an unconditional and irrevocable bank guarantee or an equivalent instrument as mutually agreed by the Procurer and the Seller.

11.4.1.8 Upon fulfilment of the conditions mentioned under Article 11.4.2.2 the Letter of Credit amount as mentioned in Article 11.4.1.1 shall be changed to one (1) time the average of the Monthly Tariff Payments of the previous Contract Year instead of one point one (1.1) times the average of the Monthly Tariff Payments of the previous Contract Year.

11.4.2 Collateral Arrangement

11.4.2.1 As further support for the Procurer's obligations, on or prior to the Effective Date, the Procurer and the Seller shall execute Default Escrow Agreement (referred as "Default Escrow Agreement") for the establishment and operation of the Default Escrow Account in favour of the Seller, through which the revenues of the Procurer shall be routed and used as per the terms of the Default Escrow Agreement. The Procurer and the Seller shall contemporaneously with the execution of the Default Escrow Agreement enter into a separate Agreement to Hypothecate Cum Deed of Hypothecation, whereby the Procurer shall agree to hypothecate, to the Seller, effective from forty five (45) days prior to the Scheduled COD the amounts to the extent as required for the Letter of Credit as per Article 11.4.1.1 routed through the Default Escrow Account and the Receivables in accordance with the terms of the Agreement to Hypothecate Cum Deed of Hypothecation. The Default Escrow Agreements and the Agreement to Hypothecate Cum Deed of Hypothecation are collectively referred to as the "Collateral Arrangement". The minimum revenue flow in any Month in the Default Escrow Account shall be at least equal to the amount required for the Letter of Credit as per Article 11.4.1.1.





Provided that the Procurer shall ensure that the Seller has first ranking charge on the revenues routed through the Default Escrow Account and the 'Receivables' in accordance with the terms of the Agreement to Hypothecate Cum Deed of Hypothecation. However, such first ranking charge shall be on the amounts; in excess of amounts, which have already been charged or agreed to be charged prior to the date of the execution of the Default Escrow Agreement.

11.4.2.2 On the occurrence of all of the following events in respect of the Procurer:

(i) A period of not less than two (2) years from COD of Power Station, has elapsed; and

(ii) The Procurer has achieved, for its ability to honour its Tariff Payment obligations to the Seller under this Agreement, a credit rating of 'A' or better, from a SEBI registered Indian credit rating agency mutually agreed between Seller and the Lender/s, consistently for a period of at least three (3) years; and

(iii) Immediately prior to the three (3) year period mentioned in subclause (ii) above,, for a period of at least two (2) years there has been no Procurer Event of Default under Article 14 of the PPA, by the Procurer,

the Procurer shall intimate the Seller in writing of the occurrence of the same and its intention to discontinue the Collateral Arrangement. If the Seller desires to continue with the Collateral Arrangement, it shall intimate the same to the Procurer in writing within thirty (30) days of receipt of intimation from the Procurer and in such case the Seller shall be liable to bear the costs of continuation of the Collateral Arrangement with effect from such date. In case the Seller fails to respond or agrees to discontinue, the Collateral Arrangement shall forthwith cease and the Default Escrow Agreement and the routed through the Default Escrow Account and the 'Receivables' in accordance with the terms of the Agreement to Hypothecate Cum Deed of Hypothecation shall stand terminated as per terms thereof.

Provided that in case of any of conditions mentioned under (i), (ii) or (iii) in Article 11.4.2.2 ceases to be true, then within 90 days of the occurrence of such event, the Procurer shall reinstate the Collateral Arrangement, at its own cost.



11.4.2.3 If the Letter of Credit is insufficient to pay for the due payments to the Seller or is not replenished for the drawals made, then within a period of seven (7) days from the date such shortfall in the Letter of Credit occurs, the Letter of Credit shall be reinstated to the requisite amount specified in this Agreement, and in the manner specified in the Default Escrow Agreement.

11.5 Third Party Sales on default

- 11.5.1 Notwithstanding anything to the contrary contained in this Agreement, upon the occurrence of an event where the Procurer has not made payment by the Due Date of an Invoice through the payment mechanism provided in this Agreement, the Seller shall follow the steps as enumerated in Articles 11.5.2 and 11.5.3.
- 11.5.2 On the occurrence of the event mentioned in Article 11.5.1 and after giving a notice of at least seven (7) days to the Procurer, the Seller shall have the right to offer twenty five (25) per cent of the Contracted Capacity ("Default Electricity") to third parties.
- 11.5.3 On the occurrence of the event mentioned in Article 11.5.1 and after giving a
 notice of at least seven (7) days to the Procurer, the Seller shall have the right (but not the obligation) to make available and sell the Default Electricity or a part thereof to a third party, namely:
 - (a) any consumer, subject to applicable Law; or
 - (b) any licensee under the Electricity Act, 2003;
- 11.5.4 If the Collateral Arrangement is not fully restored by the Procurer within thirty (30) days of the non-payment of a Invoice by its Due Date, the provisions of Article 11.5.2 and Article 11.5.3 shall apply with respect to one hundred (100) per cent of the Contracted Capacity. Provided that in case the events mentioned in Article 11.4.2.2 (i), (ii) and (iii) are true, then this Article 11.5.4 shall be applicable as per Article 11.4.2.2.
- 11.5.5 Provided that, in the case of Article 11.5.3 or 11.5.4, the Seller shall-ensurethat sale of power to the shareholders of the Seller or any direct or indirect affiliate of the Seller/shareholders of the Seller, is not at a price less than the Energy Charges.
- 11.5.6 In case of third party sales as permitted by this Article 11.5, the adjustment of the surplus revenue over Energy Charge attributable to such electricity sold, shall be adjusted as under :

shall be adjusted as under : 58



(a) the surplus upto the Tariff shall be used towards the extinguishment of the subsisting payment liability of the Procurer towards the Seller; and(b) the surplus if any above the Tariff shall be retained by the Seller.

The liability of the Procurer towards making Capacity Charge payments to the Seller even for electricity sold to third parties or remaining unsold during such periods will remain unaffected. Provided such Capacity Charge payment liability shall cease on the date which occurs on the Expiry of a period of three (3) years and hundred (100) days from the date of occurrence of the Procurer Event of Default under Article 14.2 (i), provided if prior to such date, the Procurer Event of Default has not ceased and regular supply of electricity for a period of at least ninety (90) continuous days has not occurred.

- 11.5.7 Sales to any person or Party, under Article 11.5, shall cease and regular supply of electricity to the Procurer in accordance with all the provisions of this Agreement shall commence and be restored on the later of the two following dates or any date before this date at the option of Seller:
 - (a) the day on which the Procurer pays the amount due to the Seller and renews the Letter of Credit and restores Default Escrow Account (if applicable) as mentioned in Article 11.4.2.1; or
 - (b) the date being "x" days from the date on which the defaulting Procurer pays the amount due to the Seller, where "x" days shall be calculated in accordance with Schedule 12.

11.6 Disputed Bill

AR

- 11.6.1 If a Party does not dispute a Monthly Bill, Provisional Bill or a Supplementary Bill raised by the other Party within thirty (30) days of receiving it, such bill shall be taken as conclusive.
- 11.6.2 If a Party disputes the amount payable under a Monthly Bill, Provisional Bill or a Supplementary Bill, as the case may be, that Party shall, within thirty (30) days of receiving such bill, issue a notice (the "Bill Dispute Notice") to the invoicing Party setting out:
 - i) the details of the disputed amount;
 - ii) its estimate of what the correct amount should be; and
 - iii) all written material in support of its claim.
- 11.6.3 If the invoicing Party agrees to the claim raised in the Bill Dispute Notice issued pursuant to Article 11.6.2, the invoicing Party shall revise such bill within seven (7) days of receiving such notice and if the disputing Party has already made the excess payment, refund to the disputing Party such excess amount within fifteen (15) days of receiving such notice. In such agase excess

-



amount shall be refunded along with interest at the same rate as Late Payment Surcharge which shall be applied from the date on which such excess payment was made to the invoicing Party and upto and including the date on which such payment has been received.

18

11.6.4 If the invoicing Party does not agree to the claim raised in the Bill Dispute Notice issued pursuant to Article 11.6.2, it shall, within fifteen (15) days of receiving the Bill Dispute Notice, furnish a notice to the disputing Party providing:

i) reasons for its disagreement;

- ii) its estimate of what the correct amount should be; and
- iii) all written material in support of its counter-claim.
- 11.6.5 Upon receipt of notice of disagreement to the Bill Dispute Notice under Article 11.6.4, authorised representative(s) or a director of the board of directors/member of board of each Party shall meet and make best endeavours to amicably resolve such dispute within fifteen (15) days of receiving such notice of disagreement to the Bill Dispute Notice.
- 11.6.6 If the Parties do not amicably resolve the Dispute within fifteen (15) days of receipt of notice of disagreement to the Bill Dispute Notice pursuant to Article 11.6.4, the matter shall be referred to Dispute Resolution in accordance with Article 17.
- 11.6.7 In case of Disputed Bills, it shall be open to the aggrieved party to approach the Appropriate Commission for Dispute Resolution in accordance with Article 17 and also for interim orders protecting its interest including for orders for interim payment pending Dispute Resolution and the Parties shall be bound by the decision of the Appropriate Commission, including in regard to interest or Late Payment Surcharge, if any directed to be paid by the Appropriate Commission.
- 11.6.8 If a Dispute regarding a Monthly Bill, Provisional Bill or a Supplementary Bill is settled pursuant to Article 11.6 or by Dispute resolution mechanism provided in this Agreement in favour of the Party that issues a Bill Dispute Notice, the other Party shall refund the amount, if any incorrectly charged and collected from the disputing Party or pay as required, within five (5) days of the Dispute either being amicably resolved by the Parties pursuant to Article 11.6.5 or settled by Dispute resolution mechanism along with interest at the same rate as Late Payment Surcharge from the date on which such payment had been made to the invoicing Party or the date on which such payment was originally due, as may be applicable.

11.6.9 For the avoidance of doubt, it is clarified that despite a Dispute regarding an Invoice, the Procurer shall, without prejudice to its right to Dispute, be under an obligation to make payment, of the lower of (a) an amount equal to simple average of last three (3) Months invoices (being the undisputed portion of such



three Months invoices) and (b) Monthly Invoice which is being disputed, provided such Monthly Bill has been raised based on the REA and in accordance with this Agreement.

11.7 Quarterly and Annual Reconciliation

Both Parties acknowledge that all payments made against Monthly Bills, Provisional Bill and Supplementary Bills shall be subject to quarterly reconciliation at the beginning of the following quarter of each Contract Year and annual reconciliation at the end of each Contract Year to take into account REA, Tariff Adjustment Payments, Tariff Rebate Payments, Late Payment Surcharge, or any other reasonable circumstance provided under this Agreement. The Parties, therefore, agree that as soon as all such data in respect of any quarter of a Contract Year or a full Contract Year as the case may be has been finally verified and adjusted, the Seller and the Procurer shall jointly sign such reconciliation statement. Within fifteen (15) days of signing of a reconciliation statement, the Seller or the Procurer, as the case may be, shall raise a Supplementary Bill for the Tariff Adjustment Payments for the relevant quarter/ Contract Year and shall make payment of such Supplementary Bill for the Tariff Adjustment Payments for the relevant quarter/Contract Year. Late Payment Surcharge shall be payable in such a case from the date on which such payment had been made to the invoicing Party or the date on which any payment was originally due, as may be applicable. Any dispute with regard to the above reconciliation shall be dealt with in accordance with the provisions of Article 17.

11.8 Payment of Supplementary Bill

- 11.8.1 Either Party may raise a bill on the other Party ("Supplementary Bill") for payment on account of:
 - i) Adjustments required by the Regional Energy Account (if applicable);
 - ii) Tariff Payment for change in parameters, pursuant to provisions in Schedule 6; or
 - iii) Change in Law as provided in Article 13,

and such Bill shall be paid by the other Party.

11.8.2 The Procurer shall remit all amounts due under a Supplementary Bill raised by the Seller to the Seller's Designated Account by the Due Date and notify the Seller of such remittance on the same day. Similarly, the Seller shall pay all amounts due under a Supplementary Bill raised by the Procurer by the Due

61



Date to the Procurer's designated bank account and notify the Procurer of such payment on the same day. For such payments by the Procurer, rebates as applicable to Monthly Bills pursuant to Article 11.3.5 shall equally apply.

11.8.3 In the event of delay in payment of a Supplementary Bill by either Party beyond its Due Date, a Late Payment Surcharge shall be payable at the same terms applicable to the Monthly Bill in Article 11.3.4.

11.9 Payment for Start up Power

The Seller shall be liable to pay, for the power and energy consumed for startup of the Project and commissioning, to the Procurer at the then prevalent rates payable by such industrial consumers.

11.10 The copies of all notices/offers which are required to be sent as per the provisions of this Article 11, shall be sent by either Party, simultaneously to both Parties.

ંગ્ર



12 ARTICLE 12 FORCE MAJEURE

12.1 Definitions

In this Article 12, the following terms shall have the following meanings:

12.2 Affected Party

An affected Party means the Procurer or the Seller whose performance has been affected by an event of Force Majeure.

An event of Force Majeure affecting the CTU/STU or any other agent of the Procurer, which has affected the Interconnection Facilities, shall be deemed to be an event of Force Majeure affecting the Procurer.

Any event of Force Majeure affecting the performance of the Seller's contractors, shall be deemed to be an event of Force Majeure affecting Seller only if the Force Majeure event is affecting and resulting in:

- a. late delivery of plant, machinery, equipment, materials, spare parts, Fuel, water or consumables for the Project; or
- b. a delay in the performance of any of the Seller's contractors.

Similarly, any event of Force Majeure affecting the performance of the Procurer's contractor for the setting up or operating Interconnection Facilities shall be deemed to be an event of Force Majeure affecting Procurer only if the Force Majeure event is resulting in a delay in the Performance of Procurer's contractors.

12.3 Force Majeure

A 'Force Majeure' means any event or circumstance or combination of events and circumstances including those stated below that wholly or partly prevents or unavoidably delays an Affected Party in the performance of its obligations under this Agreement, but only if and to the extent that such events or circumstances are not within the reasonable control, directly or indirectly, of the Affected Party and could not have been avoided if the Affected Party had taken reasonable care or complied with Prudent Utility Practices:

63



act of God, including, but not limited to lightning, drought, fire and explosion (to the extent originating from a source external to the Site), earthquake, volcanic eruption, landslide, flood, cyclone, typhoon, tornado, or exceptionally adverse weather conditions which are in excess of the statistical measures for the last hundred (100) years,

ii. Non-Natural Force Majeure Events:

- 1. Direct Non Natural Force Majeure Events
- a) Nationalization or compulsory acquisition by any Indian Governmental Instrumentality of any material assets or rights of the Seller or the Seller's contractors; or
- b) the unlawful, unreasonable or discriminatory revocation of, or refusal to renew, any Consent required by the Seller or any of the Seller's contractors to perform their obligations under the Project Documents or any unlawful, unreasonable or discriminatory refusal to grant any other consent required for the development/ operation of the Project. Provided that an appropriate court of law declares the revocation or refusal to be unlawful, unreasonable and discriminatory and strikes the same down.
- c) any other unlawful, unreasonable or discriminatory action on the part of an Indian Government Instrumentality which is directed against the Project. Provided that an appropriate court of law declares the revocation or refusal to be unlawful, unreasonable and discriminatory and strikes the same down.
- 2. Indirect Non Natural Force Majeure Events
- a) any act of war (whether declared or undeclared), invasion, armed conflict or act of foreign enemy, blockade, embargo;, revolution, riot, insurrection, terrorist or military action; or
- b) Radio active contamination or ionising radiation originating from a source in India or resulting from another Indirect Non Natural Force Majeure Event excluding circumstances where the source or cause of contamination or radiation is brought or has been brought into or near the site by the Affected Party or those employed or engaged by the Affected Party.

c) Industry wide strikes and labor disturbances having a nationwide impact in India. \land

64



12.4 Force Majeure Exclusions

Force Majeure shall not include (i) any event or circumstance which is within the reasonable control of the Parties and (ii) the following conditions, except to the extent that they are consequences of an event of Force Majeure:

- a. Unavailability, late delivery, or changes in cost of the plant, machinery, equipment, materials, spare parts, Fuel or consumables for the Project;
- b. Delay in the performance of any contractor, sub-contractors or their agents excluding the conditions as mentioned in Article 12.2;
- c. Non-performance resulting from normal wear and tear typically experienced in power generation materials and equipment;
- d. Strikes or labour disturbance at the facilities of the Affected Party;
- e. Insufficiency of finances or funds or the agreement becoming onerous to perform; and
- f. Non-performance caused by, or connected with, the Affected Party's:
 - i) Negligent or intentional acts, errors or omissions;
 - ii) Failure to comply with an Indian Law; or
 - iii) Breach of, or default under this Agreement or any Project Documents.

12.5 Notification of Force Majeure Event

The Affected Party shall give notice to the other Party of any event of Force Majeure as soon as reasonably practicable, but not later than seven (7) days after the date on which such Party knew or should reasonably have known of the commencement of the event of Force Majeure. If an event of Force Majeure results in a breakdown of communications rendering it unreasonable to give notice within the applicable time limit specified herein, then the Party claiming Force Majeure shall give such notice as soon as reasonably practicable after reinstatement of communications, but not later than one (1) day after such reinstatement. Provided that such notice shall be a precondition to the Seller's entitlement to claim relief under this Agreement. Such notice shall include full particulars of the event of Force Majeure, its effects on the Party claiming relief and the remedial measures proposed. The Affected Party shall give the. other Party regular (and not less than monthly) reports on the progress of those



remedial measures and such other information as the other Party may reasonably request about the situation.

The Affected Party shall give notice to the other Party of (i) the cessation of the relevant event of Force Majeure; and (ii) the cessation of the effects of such event of Force Majeure on the performance of its rights or obligations under this Agreement, as soon as practicable after becoming aware of each of these cessations.

12.6 Duty to perform and duty to mitigate

To the extent not prevented by a Force Majeure event pursuant to Article 12.3, the Affected Party shall continue to perform its obligations pursuant to this Agreement. The Affected Party shall use its reasonable efforts to mitigate the effect of any event of Force Majeure as soon as practicable.

12.7 Available Relief for a Force Majeure Event

Subject to this Article 12:

(d)

- (a) no Party shall be in breach of its obligations pursuant to this Agreement to the extent that the performance of its obligations was prevented, hindered or delayed due to a Force Majeure Event;
- (b) both the Parties shall be entitled to claim relief in relation to a Force Majeure Event in regard to its obligations, including but not limited to those specified under Article 4.5.
- (c) For the avoidance of doubt, it is clarified that no Tariff shall be paid by the Procurer for the part of Contracted Capacity affected by a Natural Force Majeure Event affecting the Seller, for the duration of such Natural Force Majeure Event. For the balance part of the Contracted Capacity, the Procurer shall pay the Tariff to the Seller, provided during such period of Natural Force Majeure Event, the balance part of the Power Station is declared to be Available for scheduling and dispatch as per ABT for supply of power by the Seller to the Procurer.
 - If the average Availability of the Power Station is reduced below sixty (60) percent for over two (2) consecutive months or for any non consecutive period of four (4) months both within any continuous period of sixty (60) months, as a result of an Indirect Non Natural Force Majeure, then, with effect from the end of that period and for so long as the daily average Availability of the Power Station continues to be reduced below sixty (60) percent as a result of an Indirect Non Natural Force Majeure of any kind, the Procurer shall make payments

66



for Debt Service, subject to a maximum of Capacity Charges based on Normative Availability, relatable to such Unit, which are due under the Financing Agreements and these amounts shall be paid from the date, being the later of a) the date of cessation of such Indirect Non Natural Force Majeure Event and b) the completion of sixty (60) days from the receipt of the Financing Agreements by the Procurer from the Seller, in the form of an increase in Capacity Charge. Provided such Capacity Charge increase shall be determined by Appropriate Commissionon the basis of putting the Seller in the same economic position as the Seller would have been in case the Seller had been paid Debt Service in a situation where the Indirect Non Natural Force Majeure had not occurred.

Provided that the Procurer will have the above obligation to make payment for the Debt Service only (a) after the Unit(s) affected by such Indirect Non Natural Force Majeure Event has been Commissioned, and (b) only if in the absence of such Indirect Non Natural Force Majeure Event, the Availability of such Commissioned Unit(s) would have resulted in Capacity Charges equal to Debt Service.

e)

If the average Availability of the Power Station is reduced below eighty (80) percent for over two (2) consecutive months or for any non consecutive period of four (4) months both within any continuous period of sixty (60) months, as a result of a Direct Non Natural Force Majeure, then, with effect from the end of that period and for so long as the daily average Availability of the Power Station continues to be reduced below eighty (80) percent as a result of a Direct Non Natural Force Majeure of any kind, the Seller may elect in a written notice to the Procurer, to deem the Availability of the Power Station to be eighty (80) percent from the end of such period, regardless of its actual Available Capacity. In such a case, the Procurer shall be liable to make payment to the Seller of Capacity Charges calculated on such deemed Normative Availability, after the cessation of the effects of Direct Non Natural Direct Force Majeure in the form of an increase in Capacity Charge. Provided such Capacity Charge increase shall be determined by Appropriate Commission on the basis of putting the Seller in the same economic position as the Seller would have been in case the Seller had been paid Capacity Charges in a situation where the Direct Non Natural Force Majeure had not occurred.



For so long as the Seller is claiming relief due to any Non Natural Force Majeure Event (or Natural Force Majeure Event affecting the Procurer under this Agreement, the Procurer may from time to time on one (1) days notice inspect the Project and the Seller shall provide Procurer's personnel with access to the Project to carry out such inspections, subject to the Procurer's personnel complying with all reasonable safety precautions and standards. Provided further the Procurer shall be entitled at all times to request Repeat Performance Test, as per Article 8.1, of the Unit(s) Commissioned earlier and now affected by Direct or Indirect Non Natural Force Majeure Event (or Natural Force Majeure Event affecting the Procurer, where such Testing is possible to be undertaken in spite of the Direct or Indirect Non Natural Force Majeure Event (or Natural Force Majeure Event affecting the Procurer), and the Independent Engineer accepts and issues a Final Test Certificate certifying such Unit(s) being capable of delivering the Contracted Capacity and being Available, had there been no such Direct or Indirect Non Natural Force Majeure Event (or Natural Force Majeure Event affecting the Procurer). In case, the Available Capacity as established by the said Repeat Performance Test (provided that for such Repeat Performance Test, the limitation imposed by Article 8.1.1 shall not apply) and Final Test Certificate issued by the Independent Engineer is less than the Available Capacity corresponding to which the Seller would have been paid Capacity Charges equal to Debt Service in case of Indirect Non Natural Force Majeure Event (or Natural Force Majeure Event affecting the Procurer), then the Procurer shall make pro-rata payment of Debt Service but only with respect to such reduced Availability. For the avoidance of doubt, if Debt Service would have been payable at an Availability of 60% and pursuant to a Repeat Performance Test it is established that the Availability would have been 40%, then the Procurer shall make payment equal to Debt Service multiplied by 40% and divided by 60%. Similarly, the payments in case of Direct Non Natural Force Majeure Event (and Natural Force Majeure Event affecting the Procurer) shall also be adjusted pro-rata for reduction in Available Capacity.

In case of a Natural Force Majeure Event affecting the Procurer which adversely affects the performance obligations of the Seller under this - Agreement, the provisions of sub-proviso (d) and (f) shall apply.

68

126

g)

(f)



27

12.8 Additional Compensation and Procurer's Subrogation

69

If the Seller is entitled, whether actually or contingently, to be compensated by any person other than the Procurer as a result of the occurrence of a Non Natural Force' Majeure Event (or Natural Force Majeure Event affecting the Procurer) for which it has received compensation from the Procurer pursuant to this Article 12, including without limitation, payments made which payments would not have been made in the absence of Article 4.7.1, the Procurer shall be fully subrogated to the Seller's rights against that person to the extent of the compensation paid by the Procurer to the Seller. Provided that in case the Seller has actually received compensation from the any person other than the Procurer as well as the Procurer as a result of the occurrence of a Non Natural Force' Majeure Event (or Natural Force Majeure Event affecting the Procurer), then the Seller shall forthwith refund the compensation received by it from the Procurer but only to the extent of the compensation received by the Seller from any person other than the Procurer.



128

13 ARTICLE 13 : CHANGE IN LAW

13.1 Definitions

In this Article 13, the following terms shall have the following meanings:

13.1.1 "Change in Law" means the occurrence of any of the following events

(i) the enactment, bringing into effect, adoption, promulgation, amendment, modification or repeal, of any Law or (ii) a change in interpretation of any Law by a Competent Court of law, tribunal or Indian Governmental Instrumentality provided such Court of law, tribunal or Indian Governmental Instrumentality is final authority under law for such interpretation or (iii) change in any consents, approvals or licenses available or obtained for the Project, otherwise than for default of the Seller, which results in any change in any cost of or revenue from the business of selling electricity by the Seller to the Procurer under the terms of this Agreement,

but shall not include (i) any change in any withholding tax on income or dividends distributed to the shareholders of the Seller, or (ii) change in respect of UI Charges or frequency intervals by an Appropriate Commission.

13.1.2 "Competent Court" means:

The Supreme Court or any High Court, or any tribunal or any similar judicial or quasi-judicial body in India that has jurisdiction to adjudicate upon issues relating to the Project.

13.2 Application and Principles for computing impact of Change in Law

While determining the consequence of Change in Law under this Article 13, the Parties shall have due regard to the principle that the purpose of compensating the Party affected by such Change in Law, is to restore through Monthly Tariff payments, to the extent contemplated in this Article 13, the affected Party to the same economic position as if such Change in Law has not occurred.

70



a) Construction Period

As a result of any Change in Law, the impact of increase/decrease of Capital Cost of the Project in the Tariff shall be as approved by PSERC:

b)Operation Period

As a result of Change in Law, the compensation for any increase/decrease in revenues or cost to the Seller shall be determined and effective from such date, as decided by the Punjab State Electricity Regulatory Commission whose decision shall be final and binding on both the Parties, subject to rights of appeal provided under applicable Law.

Provided that the above mentioned compensation shall be payable only if and for increase/ decrease in revenues or cost to the Seller is in excess of an amount equivalent to 1% of the Letter of Credit in aggregate for a Contract Year.

13.3 Notification of Change in Law

- 13.3.1 If the Seller is affected by a Change in Law in accordance with Article 13.2 and wishes to claim a Change in Law under this Article, it shall give notice to the Procurer of such Change in Law as soon as reasonably practicable after becoming aware of the same or should reasonably have known of the Change in Law.
- 13.3.2 Notwithstanding Article 13.3.1, the Seller shall be obliged to serve a notice to the Procurer under this Article 13.3.2 if it is beneficially affected by a Change in Law. Without prejudice to the factor of materiality or other provisions contained in this Agreement, the obligation to inform the Procurer contained herein shall be material. Provided that in case the Seller has not provided such notice, the Procurer shall have the right to issue such notice to the Seller.

13.3.3 Any notice served pursuant to this Article 13.3.2 shall provide, amongst other things, precise details of:

- (a) the Change in Law; and
- (b) the effects on the Seller of the matters referred to in Article 13.2.



13.4 Tariff Adjustment Payment on account of Change in Law

72

13.4.1 Subject to Article 13.2, the adjustment in Monthly Tariff Payment shall be effective from:

(i) the date of adoption, promulgation, amendment, re-enactment or repeal of the Law or Change in Law; or

(ii) the date of order/judgment of the Competent Court or tribunal or Indian Governmental Instrumentality, if the Change in Law is on account of a change in interpretation of Law.

13.4.2 The payment for Changes in Law shall be through Supplementary Bill as mentioned in Article 11.8. However, in case of any change in Tariff by reason of Change in Law, as determined in accordance with this Agreement, the Monthly Invoice to be raised by the Seller after such change in Tariff shall appropriately reflect the changed Tariff.



14 ARTICLE 14 : EVENTS OF DEFAULT AND TERMINATION

14.1 Seller Event of Default

The occurrence and continuation of any of the following events, unless any such event occurs as a result of a Force Majeure Event or a breach by the Procurer of their obligations under this Agreement, shall constitute a Seller Event of Default:

- i) the failure to Commission any Unit by the date falling twelve (12) Months after its Scheduled Commercial Operation Date, or
- after the commencement of construction of the Project, the abandonment by the Seller or the Seller's Construction Contractors of the construction of the Project for a continuous period of two (2) Months and such default is not rectified within thirty (30) days from the receipt of first notice from the Procurer in this regard, or
- iii) if at any time following a Unit being Commissioned and during its rotest, as per Article 8, such Unit's Tested Capacity is less than ninety two (92) percent of its Contracted Capacity, as existing on the Effective Date, and such Tested Capacity remains below ninety two (92) percent even for a period of three (3) Months thereafter; or
- iv) after Commercial Operation Date of all the Units of the Power Station, the Seller fails to achieve Average Availability of sixty five percent (65%), for a period of twelve (12) consecutive Months or within a non-consecutive period of twelve (12) Months within any continuous aggregate period of thirty six (36) Months, or
- v) the Seller fails to make any payment (a) of an amount exceeding Rupees One
 (1) Crore required to be made to the Procurer under this Agreement, within three (3) Months after the Due Date of an undisputed invoice /demand raised by the Procurer on the Seller or (b) of an amount up to Rupees One (1) Crore required to be made to the Procurer under this Agreement within six (6) Months after the Due Date of an undisputed invoice/demand, or
- vi) any of the representations and warranties made by the Seller in Schedule 8 of this Agreement; being found to be untrue or inaccurate. Provided however, prior to considering any event specified under this sub-article to be an Event of Default, the Procurer shall give a notice to the Seller in writing of at least thirty (30) days, or

73

vii) if the Seller:

a) assigns or purports to assign any of its assets or rights in violation of this Agreement; or



b) transfers or novates any of its rights and/or obligations under this agreement, in violation of this Agreement; or

viii)

if (a) the Seller becomes voluntarily or involuntarily the subject of any bankruptcy or insolvency or winding up proceedings and such proceedings remain uncontested for a period of thirty (30) days, or (b) any winding up or bankruptcy or insolvency order is passed against the Seller, or (c) the Seller goes into liquidation or dissolution or has a receiver or any similar officer appointed over all or substantially all of its assets or official liquidator is appointed to manage its affairs, pursuant to Law,

Provided that a dissolution or liquidation of the Seller will not be an Event of Default if such dissolution or liquidation is for the purpose of a merger, consolidation or reorganization and where the resulting company continues to meet the financial and technical requirements of the seller as certified by the lenders till COD of the Power Station, and retains creditworthiness similar to the Seller and expressly assumes all obligations of the Seller under this Agreement and is in a position to perform them; or

- ix) the Seller repudiates this Agreement and does not rectify such breach even within a period of thirty (30) days from a notice from the Procurer in this regard; or
- x) except where due to any Procurer's failure to comply with its material obligations, the Seller is in breach of any of its material obligations pursuant to this Agreement, and such material breach is not rectified by the Seller within thirty (30) days of receipt of first notice in this regard given by the Procurer to the Seller.
- xi) the Seller fails to complete/fulfill the activities/conditions specified in Article
 3.1.2, beyond a period of 8 Months from the specified period in Article 3.1.2
 and the right of termination under Article 3.3.2 is invoked by the Procurer; or
- xii) The Seller fails to provide additional bank guarantee to the Procurer in accordance with Article 3.3 of this Agreement, or
- xiii) Occurrence of any other event which is specified in this Agreement to be a material breach/default of the Seller.

14.2 Procurer Event of Default

The occurrence and the continuation of any of the following events, unless any such event occurs as a result of a Force Majeure Event or a breach by the Seller of its obligations under this Agreement, shall constitute the Event of Default on the part of the Procurer:

i) the Procurer fails to pay (with respect to a Monthly Bill or a Supplementary Bill) an amount exceeding fifteen (15%) of the undisputed part of the most recent Monthly/Supplementary Bill for a period of ninety (90) days after the Due Date and the Seller is unable to

74





recover the amount outstanding to the Seller through the Collateral Arrangement and Letter of Credit; or

- ii) the Procurer repudiates this Agreement and does not rectify such breach even within a period of thirty (30) days from a notice from the Seller in this regard; or
- except where due to any Seller's failure to comply with its obligations, the Procureris in material breach of any of its obligations pursuant to this Agreement, and such material breach is not rectified by the Procurer within thirty (30) days of receipt of notice in this regard from the Seller to the Procurer;
- iv) any representation and warranties made by the Procurer in Schedule 8 of this Agreement, being found to be untrue or inaccurate. Provided however, prior to considering any event specified under this sub-article to be an Event of Default, the Seller shall give a notice to the Procurer in writing of at least thirty (30) days; or
- v) if (a) the Procurer becomes voluntarily or involuntarily the subject of any bankruptcy or insolvency or winding up proceedings and such proceedings remain uncontested for a period of thirty (30) days, or (b) any winding up or bankruptcy or insolvency order is passed against the Procurer, or (c) the Procurer goes into liquidation or dissolution or has a receiver or any similar officer appointed over all or substantially all of its assets or official liquidator is appointed to manage its affairs, pursuant to Law, except where such dissolution or liquidation of the Procurer is for the purpose of a merger, consolidation or reorganization and where the resulting entity has the financial standing to perform its obligations under this Agreement and has creditworthiness similar to the Procurer and expressly assumes all obligations of the Procurer under this Agreement and is in a position to perform them; or;
- vi) occurrence of any other event which is specified in this Agreement to be a material breach or default of the Procurer.

14.3 Procedure for cases of Seller Event of Default

- 14.3.1 Upon the occurrence and continuation of any Seller Event of Default under Article 14.1, the Procurer shall have the right to deliver to the Seller a Procurer Preliminary Default Notice, which shall specify in reasonable detail, the circumstances giving rise to the issue of such notice.
- 14.3.2 Following the issue of Procurer Preliminary Default Notice, the Consultation Period of ninety (90) days or such longer period as the Parties may agree, shall apply.



14.3.3 During the Consultation Period, the Parties shall, save as otherwise provided in this Agreement, continue to perform their respective obligations under this Agreement.

14.3.4 Within a period of seven (7) days following the expiry of the Consultation Period unless the Parties shall have otherwise agreed to the contrary or the Seller Event of Default giving rise to the Consultation Period shall have been remedied, the Lenders may exercise or the Procurer may require the Lenders to exercise their substitution rights and other rights provided to them, if any, under Financing Agreements and the Procurer would have no objection to the Lenders exercising their rights if it is in consonance with provisions of Schedule 10. Alternatively, in case the Lenders do not exercise their rights as mentioned herein above, the Capacity Charge of the Seller shall be reduced by 20% for the period of Seller Event of Default.

14.4 Termination for Procurer Events of Default

- 14.4.1 Upon the occurrence and continuation of any Procurer Event of Default pursuant to Article 14.2 (i), the Seller shall follow the remedies provided under Articles 11.5.2
- 14.4.2 Without in any manner affecting the rights of the Seller under Article 14.4.1, on the occurrence of any Procurer Event of Default specified in Article 14.2 the Seller shall have the right to deliver to the Procurer a Seller Preliminary Default Notice, which notice shall specify in reasonable detail the circumstances giving rise to its issue.
- 14.4.3 Following the issue of a Seller Preliminary Default Notice, the Consultation Period of ninety (90) days or such longer period as the Parties may agree, shall apply.
- 14.4.4 During the Consultation Period, the Parties shall continue to perform their respective obligations under this Agreement.
- 14.4.5 (i) After a period of seven (7) days following the expiry of the Consultation Period and unless the Parties shall have otherwise agreed to the contrary or the Procurer Event of Default giving rise to the Consultation Period shall have been remedied, the Seller shall be free to sell the then existing Contracted Capacity to any third party of his choice. Provided the Procurer shall have the liability to make payments for Capacity Charges based on Normative Availability to the Seller for the period three (3) years from the eighth day after the expiry of the Consultation Period. Provided further that in such three



year period, in case the Seller is able to sell electricity to any third party at a price which is in excess of the Energy Charges, then such excess realization will reduce the Capacity Charge payments due from the Procurer. For the avoidance of doubt, the above excess adjustment would be applied on a cumulative basis for the three year period. During such period, the Seller shall use its best effort to sell the Contracted Capacity of the Procurer generated or capable of being generated to such third parties at the most reasonable terms available in the market at such time, having due regard to the circumstances at such time and the pricing of electricity in the market at such time. Provided further, the Seller shall ensure that sale of power to the shareholders of the Seller or any direct or indirect affiliate of the Seller/shareholders of the Seller, is not at a price less than the Tariff, without obtaining the prior written consent of the Procurer. Such request for consent would be responded to within a maximum period of 3 days failing which it would be deemed that the Procurer has given his consent. Provided further that at the end of the three year period, this Agreement shall automatically terminate and thereafter, the Procurer shall have no further Capacity Charge liability towards the Seller. Provided further, the Seller shall have the right to terminate this Agreement with respect to the Procurer even before the expiry of such three year period provided on such termination, the future Capacity Charge liability of the Procurer shall cease immediately.



15 ARTICLE 15 LIABILITY AND INDEMNIFICATION

15.1 Indemnity

15.1.1 The Seller shall indemnify, defend and hold the Procurer harmless against:

- (a) any and all third party claims, actions, suits or proceedings against the Procurer for any loss of or damage to property of such third party, or death or injury to such third party, arising out of a breach by the Seller of any of its obligations under this Agreement, except to the extent that any such claim, action, suit or proceeding has arisen due to a negligent act or omission, breach of this Agreement or breach of statutory duty on the part of the Procurer, its contractors, servants or agents; and
- (b) any and all losses, damages, costs and expenses including legal costs, fines, penalties and interest actually suffered or incurred by the Procurer from third party claims arising by reason of (i) breach by the Seller of any of its obligations under this Agreement, (provided that this Article 15 shall not apply to such breaches by the Seller, for which specific remedies have been provided for under this Agreement) except to the extent that any such losses, damages, costs and expenses including legal costs, fines, penalties and interest (together to constitute "Indemnifiable Losses") has arisen due to a negligent act or omission, breach of this Agreement or breach of statutory duty on the part of the Procurer, its contractors, servants or agents or (ii) any of the representations or warranties of the Seller under this Agreement being found to be inaccurate or untrue.
- 15.1.2 Procurer shall indemnify, defend and hold the Seller harmless against:
 - (a) any and all third party claims, actions, suits or proceedings against the Seller, for any loss of or damage to property of such third party, or death or injury to such third party, arising out of a breach by the Procurer of any of its obligations under this Agreement except to the extent that any such claim, action, suit or proceeding has arisen due to a negligent act or omission, breach of this Agreement or breach of statutory duty on the part of the Seller, its contractors, servants or agents; and
 - (b) any and all losses, damages, costs and expenses including legal costs, fines, penalties and interest ('Indemnifiable Losses') actually suffered or incurred by the Seller from third party claims arising by reason of (i)



a breach by the Procurer of any of its obligations under this Agreement (Provided that this Article 15 shall not apply to such breaches by the Procurer, for which specific remedies have been provided for under this Agreement.), except to the extent that any such Indeminifiable Losses have arisen due to a negligent act or omission, breach of this Agreement or breach of statutory duty on the part of the Seller, its contractors, servants or agents or (ii) any of the representations or warranties of the Procurer under this Agreement being found to be inaccurate or untrue.

15.2 Monetary Limitation of liability

Each Party shall notify the other Party promptly of its entitlement, and intention, to make any claim for indemnification pursuant to this Article 15.

A Party ("Indemnifying Party") shall not be liable to indemnify the other Party ("Indemnified Party") under this Article 15 for any indemnity claims made in a Contract Year until the aggregate of all indemnity claims of the Indemnified Party in a given Contract Year exceeds half a percent (0.5%) of the average annual Tariff Payment for all the Contract Years up to the Contract Year in which the indemnity claim is made

15.3 Procedure for claiming indemnity

15.3.1 Third party claims

(a) Where the Indemnified Party is entitled to indemnification from the Indemnifying Party pursuant to Article 15.1.1(a) or 15.1.2(a), the Indemnified Party shall promptly notify the Indemnifying Party of such claim, proceeding, action or suit referred to in Article 15.1.1(a) or 15.1.2(a) in respect of which it is entitled to be indemnified. Such notice shall be given as soon as reasonably practicable after the Indemnified Party becomes aware of such claim, proceeding, action or suit. The Indemnifying Party shall be liable to settle the indemnification claim within thirty (30) days of receipt of the above notice. Provided however that, if:

(i) the Parties choose to contest, defend or litigate such claim, action, suit or proceedings in accordance with Article 15.3.1(b) below; and;



(ii) the claim amount is not required to be paid/deposited to such third party pending the resolution of the Dispute,

the Indemnifying Party shall become liable to pay the claim amount to the Indemnified Party or to the third party, as the case may be, promptly following the resolution of the Dispute, if such Dispute is not settled in favour of the Indemnified Party.

(b) The Indemnified Party may contest, defend and litigate a claim, action, suit or proceeding for which it is entitled to be indemnified under Article 15.1.1(a) or 15.1.2(a) and the indemnifying Party shall reimburse to the indemnified Party all reasonable costs and expenses incurred by the indemnified party. However, such indemnified Party shall not settle or compromise such claim, action, suit or proceedings without first getting the consent of the indemnifying Party, which consent shall not be unreasonably withheld or delayed.

An Indemnifying Party may, at its own expense, assume control of the defence of any proceedings brought against the Indemnified Party if it acknowledges its obligation to indemnify such Indemnified Party, gives such Indemnified Party prompt notice of its intention to assume control of the defence, and employs an independent legal counsel at its own cost that is reasonably satisfactory to the Indemnified Party.

15.4 Indemnifiable Losses

Where an Indemnified Party is entitled to Indemnifiable Losses from the Indemnifying Party pursuant to Article 15.1.1(b) or 15.1.2(b), the Indemnified Party shall promptly notify the Indemnifying Party of the Indemnifiable Losses actually incurred by the Indemnified Party. The Indemnifiable Losses shall be reimbursed by the Indemnifying Party within thirty (30) days of receipt of the notice seeking Indemnifiable Losses by the Indemnified Party. In case of non payment of such losses after a valid notice under this Article 15.4, such event shall constitute a payment default under Article 14.

15.5 Limitation on Liability

Except as expressly provided in this Agreement, neither the Seller nor the Procurer nor their respective officers, directors, agents, employees or Affiliates (or their officers, directors, agents or employees), shall be liable or responsible to the other Party or its Affiliates, officers, directors, agents,



employees, successors or permitted assigns (or their respective insurers) for incidental, indirect or consequential damages, connected with or resulting from performance or non-performance of this Agreement, or anything done in connection herewith, including claims in the nature of lost revenues, income or profits (other than payments expressly required and properly due under this Agreement), any increased expense of, reduction in or loss of power generation production or equipment used therefore, irrespective of whether such claims are based upon breach of warranty, tort (including negligence, whether of the Procurer, the Seller or others), strict liability, contract, breach of statutory duty, operation of law or otherwise. The Procurer shall have no recourse against any officer, director or shareholder of the Seller or any Affiliate of the Seller or any of its officers, directors or shareholders for such claims excluded under this Article. The Seller shall have no recourse against any officer, director or shareholder of the Procurer, or any affiliate of the Procurer or any of its officers, directors or shareholders for such claims excluded under this Article.

81



16 ARTICLE 16: ASSIGNMENTS AND CHARGES

16.1 Assignments

This Agreement shall be binding upon, and inure to the benefit of the Parties and their respective successors and permitted assigns. Subject to Article 16.2, this Agreement shall not be assigned by any Party (and no Party shall create or permit to subsist any encumbrance over all or any of its rights and benefits under this Agreement) other than by mutual consent between the Parties to be evidenced in writing:

Provided that, such consent shall not be withheld if the Procurer seeks to transfer to any transferee all of its rights and obligations under this Agreement; and

- (a) such transferee is either the owner or operator of all or substantially all of the distribution system of the Procurer and /or such transferee is a successor entity of the Procurer; and
- (b) this Agreement and the other Project Documents shall continue to remain valid and binding on such successor.

Seller shall be entitled to assign its rights and obligations under this Agreement in favor of the Selectee duly appointed pursuant to the terms of Schedule 10 of this Agreement.

16.2 Permitted Charges

- 16.2.1 Notwithstanding anything contained in Article 16.1, the Seller may create any encumbrance over all or part of the receivables, Payment Mechanism or the other assets of the Project or the Project Documents in favour of the Lenders or the Lender's Representative on their behalf, as security for:
 - (a) amounts payable under the Financing Agreements; and
 - (b) any other amounts agreed by the Parties,
 - Provided that:

Ĭ

- the Lenders or the Lender's Representative on their behalf shall have
 - agreed in writing to the provisions of Schedule 10 of this Agreement; and
- II any encumbrances granted by the Seller in accordance with this Article 16.2.1 shall contain provisions pursuant to which the Lenders or the Lender's Representative on their behalf agrees unconditionally with



the Seller acting for itself and as trustee of the Procurer to release from such encumbrances all of the right, title and interest to Additional Compensation so as to enable the Procurer to claim its right of subrogation. For the purposes of this Article, Additional Compensation shall mean the compensation that the Seller is entitled, whether actually or contingently, to receive from the Procurer as well as compensated by any person other than the Procurer for the same event.

16.2.2 Article 16.1 does not apply to:

£

- (a) liens arising by operation of law (or by an agreement evidencing the same) in the ordinary course of the Seller carrying out the Project;
- (b) pledges of goods, the related documents of title and / or other related documents, arising or created in the ordinary course of the Seller carrying out the Project; or
- (c) security arising out of retention of title provisions in relation to goods acquired in the ordinary course of the Seller carrying out the Project.



17 ARTICLE 17: GOVERNING LAW AND DISPUTE RESOLUTION

17.1 Governing Law

This Agreement shall be governed by and construed in accordance with the Laws of India.

17.2 Amicable Settlement

- 17.2.1 Either Party is entitled to raise any claim, dispute or difference of whatever nature arising under, out of or in connection with this Agreement including its existence or validity or termination (collectively "Dispute") by giving a written notice to the other Party, which shall contain:
 - (i) a description of the Dispute;
 - (ii) the grounds for such Dispute; and
 - (iii) all written material in support of its claim.
- 17.2.2 The other Party shall, within thirty (30) days of issue of dispute notice issued under Article 17.2.1, furnish:
 - (i) counter-claim and defences, if any, regarding the Dispute; and
 - (ii) all written material in support of its defences and counter-claim.
- 17.2.3 Within thirty (30) days of issue of notice by any Party pursuant to Article 17.2.1 or Article 17.2.2, both the Parties to the Dispute shall meet to settle such Dispute amicably. If the Parties fail to resolve the Dispute amicably within thirty (30) days of receipt of the notice referred to in the preceding sentence, the Dispute shall be referred to Dispute. Resolution in accordance with Article 17.3.

17.3 Dispute Resolution

17.3.1 Where any Dispute arises from a claim made by any Party for any change in or determination of the Tariff or any matter related to Tariff or claims made by any Party which partly or wholly relate to any change in the Tariff or determination of any of such claims could result in change in the Tariff or (ii) relates to any matter agreed to be referred to the Appropriate Commission under Articles 4.7.1, 13.2, 18.1 or clause 10.1.3 of Schedule 10 hereof, such

84



Dispute shall be submitted to adjudication by the Appropriate Commission. Appeal against the decisions of the Appropriate Commission shall be made only as per the provisions of the Electricity Act, 2003, as amended from time to time.

17.3.2 If the Dispute arises out of or in connection with any claims not covered in Article 17.3.1, such Dispute shall be resolved by arbitration under the Indian Arbitration and Conciliation Act, 1996 and the Rules of the Indian Council of Arbitration, in accordance with the process specified in this Article. In the event of such Dispute remaining unresolved as referred to in Article 17.2.3 hereof, any party to such Dispute may refer the matter to registrar under the Rules of the Indian Council of Arbitration.

- (i) The Arbitration tribunal shall consist of three (3) arbitrators to be appointed in accordance with the Indian Council of Arbitration Rules
- (ii) The place of arbitration shall be Chandigarh, India. The language of the arbitration shall be English.
- (iii) The arbitration tribunal's award shall be substantiated in writing. The arbitration tribunal shall also decide on the costs of the arbitration proceedings and the allocation thereof.
- (iv) The award shall be enforceable in any court having jurisdiction, subject to the applicable Laws.
- (v) The provisions of this Clause shall survive the termination of this PPA for any reason whatsoever.

17.4 Parties to Perform Obligations

á

Notwithstanding the existence of any Dispute and difference referred to the Appropriate Commission or the arbitral tribunal as provided in Article 17.3 and save as the Appropriate Commission or the arbitral tribunal may otherwise direct by a final or interim order, the Parties hereto shall continue to perform their respective obligations (which are not in dispute) under this Agreement.


18 ARTICLE 18 : MISCELLANEOUS PROVISIONS

18.1 Amendment

ź

This Agreement may only be amended or supplemented by a written agreement between the Parties and after duly obtaining the approval of the Appropriate Commission, where necessary.

18,2 Third Party Beneficiaries

This Agreement is solely for the benefit of the Parties and their respective successors and permitted assigns and shall not be construed as creating any duty, standard of care or any liability to, any person not a party to this Agreement.

18.3 No Waiver

A valid waiver by a Party shall be in writing and executed by an authorized representative of that Party. Neither the failure by any Party to insist on the performance of the terms, conditions, and provisions of this Agreement nor time or other indulgence granted by any Party to the other Party shall act as a waiver of such breach or acceptance of any variation or the relinquishment of any such right or any other right under this Agreement, which shall remain in full force and effect.

18.4 Entirety

- 18.4.1 This Agreement and the Schedules are intended by the Parties as the final expression of their agreement and are intended also as a complete and exclusive statement of the terms of their agreement.
- 18.4.2 Except as provided in this Agreement, all prior written or oral understandings, offers or other communications of every kind pertaining to this Agreement or the sale or purchase of Electrical Output and Contracted Capacity under this Agreement to the Procurer by the Seller shall stand superseded and abrogated.

18.5 Confidentiality

The Parties undertake to hold in confidence this Agreement and Project Documents and not to disclose the terms and conditions of the transaction contemplated hereby to third parties, except:



45

(a) to their professional advisors;

(b) to their officers, contractors, employees, agents or representatives, financiers, who need to have access to such information for the proper performance of their activities; or

(c) disclosures required under Law.

without the prior written consent of the other Party.

Provided that the Seller agrees and acknowledges that the Procurer may at any time, disclose the terms and conditions of the Agreement and the Project Documents to any person, to the extent stipulated under the Law.

18.6 Affirmation

The Seller and the Procurer, each affirm that:

- neither it nor its respective directors, employees, or agents has paid or undertaken to pay or shall in the future pay any unlawful commission, bribe, pay-off or kick-back; and
- (ii) it has not in any other manner paid any sums, whether in Indian currency or foreign currency and whether in India or abroad to the other Party to procure this Agreement, and the Seller and the Procurer hereby undertake not to engage in any similar acts during the Term of Agreement.

18.7 Severability

The invalidity or enforceability, for any reason, of any part of this Agreement shall not prejudice or affect the validity or enforceability of the remainder of this Agreement, unless the part held invalid or unenforceable is fundamental to this Agreement.

18.8 No Partnership

None of the provisions of this Agreement shall constitute a partnership or agency or any such similar relationship between the Seller and the Procurer.

18.9 Survival

Notwithstanding anything to the contrary herein, the provisions of this Agreement, including Article 3.3.2, Article 10.2 (Application of Insurance

 	87	 	1	7	
 XR			11/2 Z	- J~	
,					
				•	



Proceeds), Article 12 (Force Majeure), Article 14 (Events of Default and Termination), Article 15 (Liability and Indemnification), Article 17 including Article 17.3.2 (Governing Law and Dispute Resolution), Article 18 (Miscellaneous), and other Articles and Schedules of this Agreement which expressly or by their nature survive the term or termination of this Agreement shall-continue and survive any expiry or termination of this Agreement.

18.10 Notices

All notices to be given under this Agreement shall be in writing and in the English Language.

All notices must be delivered personally, by registered or certified mailpost or any method duly acknowledged or facsimile to the addresses below:

Seller : The Managing Director GVK Power (Goindwal Sahib) Limited Paigah House, 156-159, S. P. Road Secunderabad - 500 003 Fax : 040 - 27902665

88

Procurer : The Chairman Punjab State ElectricityBoard The Mall Patiala - 147 001

All notices or communications given by email or facsimile shall be confirmed by sending a copy of the same via post office in an envelope properly addressed to the appropriate Party for delivery by registered mail. All Notices shall be deemed validly delivered upon receipt evidenced by an acknowledgement of the recipient, unless the Party delivering the notice can prove in case of delivery through the registered post that the recipient refused to acknowledge the receipt of the notice despite efforts of the post authorities.

Any Party may by notice of at least fifteen (15) days to the other Parties change the address and / or addresses to which such notices and communications to it are to be delivered or mailed.



18.11 Language

The language of this Agreement and all written communication between the Parties relating to this Agreement shall be in English.

18.12 Breach of Obligations

The Parties acknowledge that a breach of any of the obligations contained herein would result in injuries. The Parties further acknowledge that the amount of the liquidated damages or the method of calculating the liquidated damages specified in this Agreement is a genuine and reasonable pre-estimate of the damages that may be suffered by the non-defaulting party in each case specified under this Agreement.

18.13 Nomination Restriction

Notwithstanding anything contained to the contrary in this Agreement, wherever a reference is made to the right of the Procurer to nominate a third Party to receive benefits under this Agreement, such Third Party shall have a financial standing comparable to that of the Procurer in question.

18.14 Commercial Acts

The Procurer and Seller unconditionally and irrevocably agree that the execution, delivery and performance by each of them of this Agreement and those agreements included in the Collateral Arrangement to which it is a Party constitute private and commercial acts rather than public or governmental acts;

18.15 Restriction of Shareholders/Owners Liability

Both Parties expressly agree and acknowledge that none of the shareholders of the Parties hereto shall be liable to the other Parties for any of the contractual obligations of the concerned party under this Agreement. Further, the financial liabilities of the shareholder/s of each Party to this Agreement, in such Party, shall be restricted to the extent provided in Section 426 of the Indian Companies Act, 1956.

The provisions of this Article shall supercede any other prior agreement or understanding, whether oral or written, that may be existing between the Procurer, Seller, shareholders/ owners of the Seller, shareholders/ owners of the Procurer before the date of this Agreement, regarding the subject matter of this Agreement.

18.16 No Consequential or Indirect Losses

The liability of the Seller and the Procurer shall be limited to that explicitly provided in this Agreement. Provided that notwithstanding anything contained in this Agreement, under no event shall the Procurer or the Seller claim from one another any indirect or consequential losses or damages.

IN WITNESS WHEREOF the Parties have executed these presents through their authorized representatives at Patiala.

For and on behalf of Punjab State Electricity Board

udre

(Er. H. S. Brar), Chairman Signature with seal CHAIRMAN P.S.E.B. PATIALÄ

Witness:

1. Er. G.S. Sra

2. Er. V. K. Singal

Member/Generation

Chief Engineer/Hydel Projects

For and on behalf of GVK Power (Goindwal Sahib) Limited

(S. Madhusudan), Director

Signature with seal Witness: 1. Debashis Ghosh

Géneral-Manager

2. Oliver Tyagi Dy. G.M. (PD)

90



1. SCHEDULE 1: SITE

2x270 MW Goindwal Sahib Thermal Power Project:

SITE DESCRIPTION

5.9

Sr.No.	Item	Details
1.	Location	About 45 Km from Amritsar, 24 km.
		from Tarn-Taran and 40 km from
	-	Jalandhar, Tarn Taran-Kapurthala Road
		about 3 km to Amrtisar, Punjab (India).
2.	Nearest Railway Station	Khadoor Sahib terminal station.
3.	Nearest air port	Amritsar
4.	Plant Elevation	222.5 m above MSL
5.	Ambient Air temperature	
	a) Maximum	46.7 Degree Centigrade
	b) Minimum	- 2.8 Degree Centigrade
	c) Reference Temp. for	50 Degree Centigrade
	design of Electricla/I&C	
L	equipment.	
6.	Relateive humidity	
	a) Maximum	74%
	b) Minimum	46%
7.	Rainfall	South Westerly summer monscons
	a) Annual average	649.1 mm
	b) Maximum for one day	169 mm
8.	Wind data	
ļ	a) Prevailing wind direction	North west to south east for nine months
		and reverse for 3 months.
	b) Mean wind speed	1.3 km/hr to 12.3km/hr
	c) Basic wind speed	47 meters/second
	d) Wind pressure	As per IS 875 (Part-3)-1987
9.	SEISMIC DATA	
	a) Zone	IV as per latest revision of IS:1893
10.	Climate	Generally dry except the brief south
	· ·	westerly monsoon season, a hot summer
		and a bracing winter.

91







2. SCHEDULE 2: INITIAL CONSENTS

Name of the Clearance
Approval from Government of Punjab
No Objection Certificate from Punjab State Pollution Control Board
Environmental and Forest Clearance from MOEF, GOI
Water Availability from State Government and Water Supply Agreement
Civil Aviation Clearance for Chimney height from National Airports Authority
NOC from Ministry of Defence
Permission to discharge effluents

wh



SCHEDULE 3: FUNCTIONAL SPECIFICATION 3.

S No. Description		Unit	Particulars
1.1 Grid Conditions	at Interconnection Point		
(i) Voltage	Nominal	KV	[220]
	Variation ±10%	Min 200KV	Max245 KV
(ii) Frequency :	Nominal	Hz. [50]
Variation		%	-3% to +1.5%
(iii) Combined Volta for Contracte	ge and Frequency variation d Capacity	%	[±5]
(iv) Power Factor	: Nominal	Ŭ	[.85] lag
	Variation	[0.8] to [0.95] lag lead
(v) Basic Impulse Le	evel (Peak)	kV	[1050]

1.2 Fault Levels:

(i) 3 Phase Minimum S.C Current rating and duration for Switchgear- 40kA for rhas lsecond (ii) Tar

11)	Target fault Clearance	time	Maximum	ms	100

1.3 Ramp Rates

All Units of the Power Station shall be capable of increasing or decreasing their output (generation level) by not less than one percent (1%) per minute. Such capability shall be demonstrated during the Unit load of more than 50%.

- -	An
94	
	A CONTRACTOR OF



SCHEDULE 4: COMMISSIONING AND TESTING

1.1 Performance Test

4.

- (i) (a) The Performance Test shall be conducted under any and all ambient conditions (temperature, humidity etc.) and any and all Fuel qualities that may exist during the time of the Performance Test and no corrections in final gross and net output of the Unit will be allowed as a result of prevailing ambient conditions or Fuel quality.
 - (b) The correction curves will only be used if the Grid System operation during the Performance Test exceeds electrical system limits.
 - (c) The Performance Test shall be deemed to have demonstrated the Contracted Capacity of the Unit under all designed conditions and therefore no adjustments shall be made on account of fuel quality or ambient conditions.
 - (d) The Seller shall perform in respect of each Unit a Performance Test, which such Unit shall be deemed to have passed if it operates continuously for seventy two consecutive hours at or above ninety five (95) percent of its Contracted Capacity, as existing on the Effective Date, and within the electrical system limits and the Functional Specifications.
- (ii) For the purposes of any Performance Test pursuant to this sub-article 1.1, the electrical system limits to be achieved shall be as follows:
 - (a) Voltage

The Unit must operate within the voltage levels described in the Functional Specification for the duration of the Performance Test. If, during the Performance Test, voltage tests cannot be performed due to Grid System, data supplied from tests of the generator step-up transformers and generators supplied by the manufacturers shall be used to establish the ability of the Unit to operate within the specified voltage limits.

(b) Grid System Frequency

The Unit shall operate within the Grid System frequency levels described in the Functional Specification for the duration of the Performance Test.

(c) Power Factor

The Unit shall operate within the power factor range described in the. Functional Specification for the duration of the Performance Test. If, during the Performance Test, power factor tests cannot be performed due to the Grid System, data supplied from tests of the generators and the generator step-up transformers supplied by the manufacturers shall



be used to establish the ability of the Unit to operate within the specified power factor range.

(d) Fuel quality and cooling water temperature

The Unit must operate to its Contracted Capacity with Fuel quality and water temperature available at the time of Testing and no adjustment shall be allowed for any variation in these parameters.

iii As a part of the Performance Test, the Seller shall demonstrate that the Unit meets the Functional Specifications for Ramping rate as mentioned in Schedule 3. For this purpose, representative samples of ramp rates shall be taken, by ramping up or down the gross turbine load while maintaining the required temperatures and temperature differences associated with each ramp rate within the turbine while maintaining all other operational parameters within equipment limits;

1.2 Testing and Measurement procedures applied during Performance Test shall be in accordance with codes, practices or procedures as generally/ normally applied for the Performance Tests

1.3 The Seller shall comply with the prevalent Laws, rules and regulations as applicable to the provisions contained in this Schedule from time to time.

96

10



5. SCHEDULE 5 : AVAILABILITY FACTORS

The following matters shall be determined as per the provisions of the Grid Code and ABT:

- a. Availability declaration and calculation of Availability or Availability Factor;
- b. Requirement for Spinning Reserves;
- c. Procedure for revision of Availability;

- d. Consequences of failure to demonstrate capacity or misdeclarations of capacity; and
- e. Other matters which may be related to Availability or Availability Factor.



SCHEDULE 6: TARIFF

1.1 General

6

i. The method of determination of Tariff Payments for any Contract Year during the Term of Agreement shall be in accordance with this Schedule. For the purpose of payments, the Tariff will be determined by the PSERC after the receipt of an application for determination of the Tariff from the Seller

ii. The Tariff shall be paid in two parts comprising of Capacity and Energy Charge.

iii. The full Capacity Charges shall be payable based on the Contracted Capacity at Normative Availability and Incentive shall be provided for Availability beyond 85% as provided in this Schedule. In case of Availability being lower than the Normative Availability, the Capacity Charges shall be payable on proportionate basis in addition to the penalty to be paid by Seller as provided in this Schedule.

1.2 Monthly Tariff Payment

1.2.1 Components of Monthly Tariff Payment

98

The Monthly Bill for any Month in a Contract Year shall consist of the following:

i. Monthly Capacity Charge Payment in accordance with Article 1.2.2 below;

ii. Monthly Energy Charge for Scheduled Energy in accordance with Article 1.2.3 below;

iii. Incentive Payment determined in accordance with Article 1.2.4 below (applicable on annual basis and included only in the Monthly Tariff Payment for the first month of the next Contract Year);

iv. Penalty Payment determined in accordance with Article 1.2.5 below (applicable on annual basis and included only in the Monthly Tariff Payment for the first month of the next Contract Year);

v. Penalty Payment determined in accordance with Article 1.2.8 below (applicable on annual basis and included only in the Monthly Tariff Payment for the first month of the next Contract Year)

1.2.2 Monthly Capacity Charge Payment - The Monthly Capacity Charges based on the Capital Cost as approved by PSERC shall be paid as per CERC (Terms & Conditions of tariff) Regulations as applicable.Provided, no Capacity Charges shall be paid for the Settlement Period during which the RLDC has not allowed the operation of the Power Station due to Seller's failure to operate the Power Station as per the provisions of Grid Code.



1.2.3 Monthly Energy Charges

1.2.3.1 The Energy Charges shall be calculated and paid as approved by PSERC as per CERC (Terms & Conditions of tariff) Regulations as applicable.

1.2.3.2 Source and Cost of Coal and Secondary Fuel

- 1.2.3.2.1. The fuel charges linked to Coal cost based on the quantity and quality of coal delivered at the Project Site (Goindwal Sahib) are not to exceed the cost as prevailing in the PSEB's existing Pachhawara Captive Coal Mine. Thus the coal cost of Pachhawara mine will be the maximum price at which coal would be supplied to GSTPS. The Seller in association with the procurer shall resort to a competitive bidding process, preferably international both for developing and operating the captive coal block at Tokisud North subblock and any other block allocated to the project and the lowest cost emanating as a result of this exercise shall not be more than the cost of the existing Pachhawara captive coal block which will form part of the mining agreement and be adopted for the purpose of working out the variable (fuel) charges. However the discount rate for 'F' grade coal will be decided by PSERC.
- 1.2.3.2.2 If the coal production rate from the Captive Coal Mine and the additional coal block partly allocated for the project is higher than the requirement of the Project such surplus coal production, if any, will be delivered to other thermal projects of the Procurer, as directed by the Procurer. In case the Project is shut down due to Force Majeure, the coal produced by the Captive Coal Mine and the additional coal block shall be supplied to any or all of the existing power projects of Procurer as directed by Procurer. Similarly in case Captive Coal Mine commences the production of coal ahead of the COD of the project, coal so produced shall be supplied to the existing thermal power stations of Procurer as directed by Procurer. The coal shall be supplied to the Procurer's other Projects at the same coal

157



58

price as mentioned in the Fuel Supply Agreement for the Project. Sale to the existing projects of the Procurer shall be subject to the approval of Ministry of Coal, Govt. of India. The delivery point for supply of coal to the Procurer's other Projects shall be the premises of the Captive Coal Mine. However the Seller shall arrange for transportation of coal to other projects by diverting the supplies under the fuel transportation agreement with the Railways for which freight charges shall be payable by PSEB.

1.2.3.2.3 The Seller shall purchase the Secondary Fuel from the Public Sector Undertakings like Indian Oil Corporation Ltd., Bharat Petroleum Corporation Ltd, Hindustan Petroleum Corporation Ltd etc. through Competitive Bidding.

1.2.4 Contract Year Energy Incentive Payment.

If and to the extent the Availability in a Contract Year exceeds eighty five percent (85%), an incentive at the rate of forty (40%) of the Capacity Charges (in Rs./kWh) for such Contract Year subject to a maximum of twenty five (25) paise /kwhr, shall be allowed on the energy (in kwh) corresponding to the Availability in excess of eighty five percent (85%).

1.2.5 Contract Year Penalty for Availability below 75% during the Contract Year

In case the Availability for a Contract Year is less than 75%, the Seller shall pay a penalty at the rate of twenty percent (20%) of the simple average Capacity Charge (in Rs./kWh) for all months in the Contract Year applied on the energy (in kwh) corresponding to the difference between 75% and Availability during such Contract Year.

1.2.6 Deviation from the schedule

Variation between Scheduled Energy and actual energy at the Delivery Point shall be accounted for through Unscheduled Interchange (UI) Charges as detailed in the Grid Code and ABT.

1.2.7 Transmission/Wheeling Charges and Scheduling Charges

100

The payment of transmission/wheeling charges shall be settled between the CTU/STU and the Procurer. The payment of scheduling charges to the respective nodal agency (RLDC or SLDC) shall be the responsibility of the Procurer.



1.2.8 Penalty and rights relating to minimum guaranteed quantity of fuel

In case Seller has to pay penalty to the Fuel supplier for not purchasing the minimum guaranteed quantity of Fuel mentioned in the Fuel Supply Agreement and if during that Contract Year Availability of the Commissioned Units is greater than the Minimum Offtake Guarantee but the Procurer has not Scheduled Energy corresponding to such Minimum Off-take Guarantee during that Contract Year, then Seller will raise an invoice for the lower of the following amount, (a) penalty paid to the Fuel supplier under the Fuel Supply Agreement in that Contract Year, along with documentary proof for payment of such penalty, or

(b) an amount corresponding to twenty percent (20%) of cumulative Monthly Capacity Charge Payment (in Rs.) by the Procurer made for all the months in that Contract Year multiplied by (1-x/y) where:

X is the Scheduled Energy during the Contract Year for the Procurer (in kwh); and Y is the Scheduled Energy corresponding to Minimum Offtake Guarantee for the Procurer during the Contract Year (in kwh).

Provided within ten (10) days of the end of each Month after the COD of the Ist Unit, the Seller shall provide a statement to the Procurer, providing a comparison of the cumulative despatch for all previous months during the Contract Year with the Minimum Offtake Guarantee of the Procurer. Further, such statement shall also list out the deficit, if any, in the Fuel Offtake under the Fuel Supply Agreement, due to Cumulative Despatch being less than the Minimum Offtake Guarantee. In case of a Fuel Offtake Deficit within a period of fifteen (15) days from the date of receipt of the above statement from the Seller and after giving a prior written notice of atleast seven (7) days to the Seller, the Procurer shall have the rights to avail himself such deficit at the same price at which such deficit fuel was available to the Seller under the Fuel Supply Agreement and to sell such deficit to third parties.

1.2.9 Sale of Infirm Power:

The procurer shall pay only the primary & secondary fuel charges for all infirm power delivered to it by the seller.



1.3 Settlement of Bills:

1. The penalty of actual Availability shortfall during the Contract Year, Deviation from the schedule, Transmission & Scheduling Charges, and Penalty to be paid to fuel supplier will be settled as detailed in Article 1.2.2, Article 1.2.5, Article 1.2.6, Article 1.2.7 and Article 1.2.8 of this Schedule.

2. Notwithstanding anything contained in this agreement, no separate reimbursement shall be allowed for the cost of the secondary fuel.



7 SCHEDULE 7: DETAILS OF INTERCONNECTION POINT AND FACILITIES

1. Supply of Information

The Seller shall within a period of two (2) Month after the date of Financial Closing (or such other date as is mutually agreed between the Parties), provide the Procurer with such information about the design of the Project as it may reasonably require to enable it to design the Interconnection Facilities.

2. Cost of Interconnection Facilities

The Procurer shall be responsible for the financing, design, construction, installation commissioning, operation and maintenance of the Interconnection Facilities and shall bear all costs associated with its rights and obligations under this Schedule 7 in accordance with the terms of this Agreement.

3. Ownership of Interconnection Facilities

The Procurer shall own the Interconnection Facilities.

4. Interconnection Equipment on Seller's side of Interconnection Point

The Seller shall be responsible for designing, constructing, installing and maintaining all auxiliary and interconnecting equipment on the Seller's side of the Interconnection Point and the Seller shall have ownership rights in all such auxiliary and Interconnection Equipment.

5. Protection Devices

Protection devices shall be approved by the Procurer (which devices shall conform to the Procurer's system requirements) on or prior to the date of Financial Closing. After the date of Financial Closing, subject to giving the Seller reasonable notice, the Procurer may require the Seller to modify or expand the requirements for protective devices and the Procurer shall reimburse the Seller for the reasonable costs of such modification or expansion.

6. Power Line Carrier Communication (PLCC)

The Seller and Procurer will liase with each other for design, installation of PLCC equipment on or prior to the date of Financial Closing. The Seller and

Δ

	an a	103		When
<u></u>	Att			L
-				

162



the Procurer will bear the cost of the equipment at their ends of the Interconnection Facility.

7. Changes Affecting Protective Devices

Each Party shall notify the other Party as soon as is reasonably practical any changes to either the Project's or the Procurer's transmission system that may affect the proper co-ordination of protective devices between the two systems.

8. Testing

The Parties shall co-operate in testing the Interconnection Facilities prior to Synchronisation of a Generating Unit and from time to time as either Party may reasonably require. All such testing shall be carried out on a timely basis.







63

8 SCHEDULE 8: REPRESENTATION AND WARRANTIES

1. Representations and Warranties by the Procurer

The Procurer hereby represents and warrants to and agrees with the Seller as follows and acknowledges and confirms that the Seller is relying on such representations and warranties in connection with the transactions described in this Agreement:

1.1 The said Procurer has all requisite powers authorising and has been duly authorised to execute and consummate this Agreement;

1.2 This Agreement is enforceable against the said Procurer in accordance with its terms;

1.3 The consummation of the transactions contemplated by this Agreement on the part of the said Procurer will not violate any provision of nor constitute a default under, nor give rise to a power to cancel any charter, mortgage, deed of trust or lien, lease, agreement, license, permit, evidence of indebtedness, restriction, or other contract to which the said Procurer is a party or to which said Procurer is bound, which violation, default or power has not been waived;

1.4The said Procurer is not insolvent and no insolvency proceedings have been instituted, nor threatened or pending by or against the said Procurer;

1.5There are no actions, suits, claims, proceedings or investigations pending or, to the best of the said Procurer's knowledge, threatened in writing against the said Procurer at law, in equity, or otherwise, and whether civil or criminal in nature, before or by, any court, commission, arbitrator or governmental agency or authority, and there are no outstanding judgements, decrees or orders of any such courts, commission, arbitrator or governmental agencies or authorities, which materially adversely affect its ability to comply with its obligations under this Agreement.

1.6 The Procurers makes all the representations and warranties above to be valid as on the date of this Agreement.

2. Representation and Warranties of the Seller

105

The Seller hereby represents and warrants to and agrees with the Procurer as follows and acknowledges and confirms that the Procurer is relying on such representations and warranties in connection with the transactions described in this Agreement:



2.1The Seller has all requisite power authorising and has been duly authorised to execute and consummate this Agreement;

2.2This Agreement is enforceable against the Seller in accordance with its terms;

2.3The consummation of the transactions contemplated by this Agreement on the part of the Seller will not violate any provision of nor constitute a default under, nor give rise to a power to cancel any charter, mortgage, deed of trust or lien, lease, agreement, license, permit, evidence of indebtedness, restriction, or other contract to which the Seller is a party or to which the Seller is bound which violation, default or power has not been waived;

2.4The Seller is not insolvent and no insolvency proceedings have been instituted, or not threatened or pending by or against the Seller;

2.5There are no actions, suits, claims, proceedings or investigations pending or, to the best of Seller's knowledge, threatened in writing against the Seller at law, in equity, or otherwise, and whether civil or criminal in nature, before or by, any court, commission, arbitrator or governmental agency or authority, and there are no outstanding judgements, decrees or orders of any such courts, commission, arbitrator or governmental agencies or authorities, which materially adversely affect its ability to execute the Project or to comply with its obligations under this Agreement.

2.6The Seller has neither made any statement nor provided any information, which was materially inaccurate or misleading at the time when such statement was made or information was provided. Further, all the confirmations, undertakings, declarations and representations made by the Seller are true and accurate and there is no breach of the same.

2.7 The Seller makes all the representations and warranties above to be valid as on the date of this Agreement,

106



9. SCHEDULE 9: FORMAT OF THE PERFORMANCE GUARANTEE

the respective Procurer as per the terms of PPA], on behalf of M/s. [Insert name of the Seller or the Selected Bidder on behalf of the Seller].

This guarantee shall be valid and binding on this Bank up to and including and shall not be terminable by notice or any change in the constitution of the Bank or the term of contract or by any other reasons whatsoever and our liability hereunder shall not be impaired or discharged by any extension of time or variations or alternations made, given, or agreed with or without our knowledge or consent, by or between parties to the respective agreement.

Our liability under this Guarantee is restricted to Rs. _____ (Rs. _____ only). Our Guarantee shall remain in force until ______. The Procurer shall be entitled to invoke this Guarantee till [Insert date which is 30 days after the date in the preceding sentence].

The Guarantor Bank hereby agrees and acknowledges that the Procurer shall have a right to invoke this BANK GUARANTE in part or in full, as it may deem fit.

The Guarantor Bank hereby expressly agrees that it shall not require any proof in addition to the written demand by the Procurer, made in any format, raised at the above mentioned address of the Guarantor Bank, in order to make the said payment to the Procurer.

The Guarantor Bank shall make payment hereunder on first demand without restriction or conditions and notwithstanding any objection by the Seller and/or any other person. The Guarantor Bank shall not require the Procurer to justify the invocation of this BANK GUARANTEE, nor shall the Guarantor Bank have any recourse against the Procurer in respect of any payment made hereunder Λ



66

This BANK GUARANTEE shall be interpreted in accordance with the laws of India.

The Guarantor Bank represents that this BANK GUARANTEE has been established in such form and with such content that it is fully enforceable in accordance with its terms as against the Guarantor Bank in the manner provided herein.

This BANK GUARANTEE shall not be affected in any manner by reason of merger, amalgamation, restructuring or any other change in the constitution of the Guarantor Bank.

This BANK GUARANTEE shall be a primary obligation of the Guarantor Bank and accordingly the Procurer shall not be obliged before enforcing this BANK GUARANTEE to take any action in any court or arbitral proceedings against the Seller, to make any claim against or any demand on the Seller or to give any notice to the Seller or to enforce any security held by the Procurer or to exercise, levy or enforce any distress, diligence or other process against the Seller.

The Guarantor Bank acknowledges that this BANK GUARANTEE is not personal to the Procurer and may be assigned, in whole or in part, (whether absolutely or by way of security) by Procurer to any entity to whom the Procurer is entitled to assign its rights and obligations under the PPA.

Notwithstanding anything contained hereinabove, our liability under this Guarantee is restricted to Rs. ______ (Rs. ______ 'only) and it shall remain in force until ______ [Date to be inserted on the basis of Article 3.1.1 of PPA] with an additional claim period of thirty (30) days thereafter. We are liable to pay the guaranteed amount or any part thereof under this Bank Guarantee only if the Procurer serves upon us a written claim or demand.

Signature		····	
Name			
Power of Attorney	No		
For	_		1

Banker's Stamp and Full Address. Dated this _____ day of ____, 20__



10. SCHEDULE 10: SUBSTITUTION RIGHTS OF THE LENDERS

1. Substitution of the Seller

a) Subject to the terms of the PPA, upon occurrence of a Seller Event of Default under the PPA, the Lenders shall, have the right to seek substitution of the Seller by a Selectee for the residual period of the PPA, for the purposes of securing the payments of the Total Debt Amount from the Seller and performing the obligations of the Seller, in accordance with the provisions of this Schedule.

b) The Lenders may seek to exercise right of substitution by an amendment or novation of the PPA and other Project Documents executed between the Procurer and the Seller in favour of the Selectee, the Procurer and the Seller shall cooperate with the Lenders to carry out such substitution.

2. The Procurer Notice of Default

The Procurer shall, simultaneously also issue a copy of the same to the Lenders.

3. Substitution Notice

In the event of failure of the Seller to rectify the Event Of Default giving rise to Preliminary Default Notice, the lenders, upon receipt of a written advice from the procurer confirming such failure, either on their own or through its representative ("the Lenders' Representative") shall be entitled to notify the Procurer and the Seller of the intention of the Lenders to substitute the Seller by the Selectee for the residual period of the PPA (the "Substitution Notice").

4. Omitted

5. Interim operation of Project

a) On receipt of a Substitution Notice, no further action shall be taken by any Party to terminate the PPA, except under and in accordance with the terms of this Schedule 10 of this Agreement.

b) On issue of a Substitution Notice, the Lenders shall have the right to request the Procurer to enter upon and takeover the Project for the interim and till the substitution of the Selectee is complete and to otherwise take all such steps as are necessary for the continued operation and maintenance of the Project, including levy, collection and appropriation of payments thereunder, subject to, the servicing of monies owed in respect of the Total Debt Amount as per the Financing Agreements and the Seller shall completely cooperate in any such takeover of the Project by the Procurer. If the Procurer, at his sole and exclusive discretion agrees to enter upon and takeover the Project, till substitution of the Selectec in accordance with this Agreement, the Procurer shall be compensated for rendering such services in accordance with clause10.1.3 herein.

c) If the Procurer refuses to takeover the Project on request by the Lenders in accordance with clause 5(b) above, the Seller shall have the duty and obligation to



continue to operate the Project in accordance with the PPA till such time as the Selectee is finally substituted under clause 8.8 hereof.

d) The Lenders and the Procurer shall, simultaneously have the right to commence the process of substitution of the Seller by the Selectee in accordance with these terms and the Seller hereby irrevocably consents to the same.

6. Process of Substitution of Seller

The Lenders' Representative may, on delivery of a Substitution Notice notify the Procurer and the Seller on behalf of all the Lenders about the Lenders' decision to invite and negotiate, at the cost of the Lenders, offers from third parties to act as Selectee, either through private negotiations or public auction and / or a tender process, for the residual period of the PPA. Subject to and upon approval of the Procurer referred to in clause 8.5, such Selectee shall be entitled to receive all the rights of the Seller and shall undertake all the obligations of the Seller under the PPA and other Project Documents executed between the Seller and the Procurers, in accordance with these terms of substitution.

The Lenders and the Seller shall ensure that, upon the Procurer approving the Selectee, the Seller shall transfer absolutely and irrevocably, the ownership of the Project to such Selectee simultaneously with the amendment or novation of the PPA and other Project Documents executed between the Seller and the Procurers in favour of the Selectee as mentioned in clause 1 (b).

7. Modality for Substitution

7.1 Criteria for selection of the Selectee.

The Lenders and / or the Lenders' Representative shall in addition to any other criteria that they may deem fit and necessary, apply the following criteria in the selection of the Selectee:

- (a) if the Seller is proposed to be substituted during the Construction Period, the Selectee shall possess the technical and financial capability comparable to that of the seller to perform and discharge all the residual duties, obligations and liabilities of the Seller under the PPA. If the Seller is proposed to be substituted during the Operation Period, this criteria shall not be applicable.
- (b) the Selectee shall have the capability and shall unconditionally consent to assume the liability for the payment and discharge of dues, if any, of the Seller to the Procurer under-and-in-accordance with the PPA and also payment of the Total Debt Amount to the Lenders upon terms and conditions as agreed to between the Selectee and the Lenders;

110



- (c) the Selectee shall have not been in breach of any agreement between the Selectee and any Bank or any Lender or between the Selectee and any of the Procurer, involving sums greater than Rupees twenty (20) crores at any time in the last two (2) years as on the date of the substitution of the Seller.
- (d) any other appropriate criteria, whereby continuity in the performance of the Selectee's obligations under the PPA is maintained and the security in favour of the Lenders under the Financing Agreements is preserved.
- 8. Modalities

The following modalities shall be applicable to any substitution of the Seller by the Selectee pursuant to this Agreement:

- 8.1 The Lenders' Representative shall on behalf of the Lenders propose to the Procurer (the "Proposal") pursuant to sub-clause 8.2 below, the name of the Selectee for acceptance, seeking:
- (a) grant of all the rights and obligations under the PPA and the other Project Documents executed between the Procurer and the Seller, to the Selectee (as substitute for the Seller);
- (b) amendment of the PPA and the other Project Documents executed between the Procurer and the Seller, to the effect that the aforementioned grant to the Selectee, shall be such that the rights and obligations assumed by the Selectee are on the same terms and conditions for the residual period of the PPA as existed in respect of the Seller under the original PPA and the other Project Documents executed between the Procurer and the Seller; and
- (c) the execution of new agreements as necessary, by the proposed Selectee for the residual period of the PPA on the same terms and conditions as are . included in this Agreement.
- 8.2 The Proposal shall contain the particulars and information in respect of the Selectee the data and information as the Procurer may reasonably require. The Procurer may intimate any additional requirement within thirty (30) days of the date of receipt of the Proposal.
- 8.3 The Proposal shall be accompanied by an unconditional undertaking by the . Selectee that it shall, upon approval by the Procurer of the Proposal:



- (a) observe, comply, perform and fulfil the terms, conditions and covenants of the PPA and all Project Documents executed between Seller and the Procurer or a new power purchase agreement or respective Project Document (in the case of the novation thereof), which according to the terms therein are required to be observed, complied with, performed and fulfilled by the Seller, as if such Selectee was the Seller originally named under the PPA; or the respective Project Document; and
- (b) be liable for and shall assume, discharge and pay the Total Debt Amount or then outstanding dues to the Lenders under and in accordance with the Financing Agreements or in any other manner agreed to by the Lenders and the Procurer as if such Selectee was the Seller originally named under such Financing Agreements.
- 8.4 At any time prior to taking a decision in respect of the Proposal received under clause 8.1, the Procurer may require the Lender / Lenders' Representative to satisfy it as to the eligibility of the Selectee. The decision of the Procurer as to acceptance or rejection of the Selectee, shall be made reasonably and when made shall be final, conclusive and binding on the Parties.
- 8.5 The Procurer shall convey his approval or disapproval of such Proposal, , to the Lender / Lender's Representative. Such decision shall be made by the Procurer at his reasonably exercised discretion within twenty one (21) days of:
 - (a) the date of receipt of the Proposal by the Procurer; or

(b) the date when the last of further and other information and clarifications in respect of any data, particulars or information included in the Proposal requested by the Procurer under clause 8.2 above is received; whichever is later.

Notwithstanding anything to the contrary mentioned in this Agreement, the approval of the Procurer for the Selectee shall not be withheld in case the Selectee meets the criteria mentioned in Clause 7.1.

- 8.6 Upon approval of the Proposal and the Selectee by the Procurer, the Selectee mentioned in the Proposal shall become the Selectee hereunder.
- 8.7 Following the rejection of a Proposal, the Lenders and/or the Lenders' Representative shall have the right to submit a fresh Proposal, proposing another Selectee (if the rejection was on the grounds of an inappropriate third party proposed as Selectee) within sixty (60) days of receipt of communication

112



8.8

regarding rejection of the Selectee previously proposed. The provisions of this clause shall apply mutatis mutandis to such fresh Proposal.

The substitution of the Seller by the Selectee shall be deemed to be complete upon the Selectee executing all necessary documents and writings with or in favour of the Seller, the Procurer and the Lenders so as to give full effect to the terms and conditions of the substitution, subject to which the Selectee has been accepted by the Lenders and the Procurer and upon transfer of ownership and complete possession of the Project by the Procurer or the Seller, as the case may be, to the Selectee. The Procurer shall novate all the Project Documents, which they had entered in to with the Seller in order to make the substitution of the Seller by the Selectee effective. The quantum and manner of payment of the consideration payable by the Selectee to the Seller towards purchase of the Project and assumption of all the rights and obligations of the Seller under the PPA and the Project Documents as mentioned in this Agreement shall be entirely between the Seller, Selectee and the Lenders and the Procurer shall in no way be responsible to bear the same.

8.9 Upon the substitution becoming effective pursuant to sub-clause 8.8 above, all the rights of the Seller under the PPA shall cease to exist:

Provided that, nothing contained in this sub-clause shall prejudice any pending / subsisting claims of the Seller against the Procurer or any claim of the Procurer against the erstwhile Seller or the Selectee.

- 8.10 The Selectee shall, subject to the terms and conditions of the substitution, have a period of ninety (90) days to rectify any breach and / or default of the Seller subsisting on the date of substitution and required to be rectified and shall incur the liability or consequence on account of any previous breach and / or default of the Seller.
- 8.11 The decision of the Lenders and the Procurers in the selection of the Selectee shall be final and binding on the Seller and shall be deemed to have been made with the concurrence of the Seller. The Seller expressly waives all rights to object or to challenge such selection and appointment of the Selectee on any ground whatsoever.
- 8.12 The Lenders shall be solely and exclusively responsible for obtaining any and all consents/approvals or cooperation, which may be required to be obtained from the Seller under this Agreement and the Procurer shall not be liable for the same.
- 8.13 All actions of the Lenders' Representative hereunder shall be deemed to be on behalf of the Lenders and shall be binding upon them. The Lenders' Representative shall be authorised to receive payment of compensation and any other payments, including the consideration for transfer, if any, in



accordance with the Proposal and the Financing Agreements and shall be

bound to give valid discharge on behalf of all the Lenders.

- 9. . Seller's Waiver
- 9.1 The Seller irrevocably agrees and consents (to the extent to which applicable law may require such consent) to any actions of the Lenders, the Lender's Representative and the Procurer or exercise of their rights under and in accordance with these terms.
- 9.2 The Seller irrevocably agrees and consents (to the extent to which applicable law may require such consents) that from the date specified in clause 8.9, it shall cease to have any rights under the PPA or the Financing Agreements other than those expressly stated therein.
- 9.3 The Seller warrants and covenants that any agreement entered into by the Seller, in relation to the Project, shall include a legally enforceable clause providing for automatic novation of such agreement in favour of the Selectee, at the option of the Lenders or the Procurer. The Seller further warrants and covenants that, in respect of any agreements which have already been executed in relation to the Project and which lack a legally enforceable clause providing for automatic novation of such agreement, the Seller shall procure an amendment in the concluded agreement to incorporate such clause.
- 10. Interim Protection Of Service And Preservation Of Security
- 10.1 Appointment of a Receiver
- 10.1.1 In every case of the Lenders issuing a Substitution Notice and the Procurer refusing to takeover the Project and the Seller failing to operate the Project in accordance with Clause 5(c) above and the Procurer not electing to act as Receiver as per sub-clause 10.1.1A hereof, the Lenders may institute protective legal proceedings for appointment of a receiver (the "Receiver") to maintain, preserve and protect the assets held as security by the Lenders if such right is granted under the terms of the Financing Agreements.
- 10.1.1A Provided that in event of the Procurer refusing to take over the Project and the Seller failing to operate the Project in accordance with Clause 5 (c) above, and if the assets of the Project are, in the opinion of the Procurer, necessary and required for the operation and maintenance of the Project, the Procurer shall be entitled to elect to act as the Receiver for the purposes of this Clause and be entitled to maintain, preserve and protect the said assets by engaging an operator/service provider to act on their behalf and the Lenders and Seller

114



hereby consent and agree to the same. Upon the Procurer so intimating the Seller and the Lender's representative their desire to act as Receiver, the Seller and the Lender's representative shall co-operate with the Procurer to facilitate the same.

173

- 10.1.2 Upon appointment of the Court appointed Receiver or the Procurer acting as Receiver, all the Receivables received by such Receiver shall be deposited by the Receiver in the bank account jointly designated by the Procurer and the Lenders. The Receiver shall be responsible for protecting the assets in receivership and shall render a true and proper account of the receivership to the lenders in accordance with the terms of its appointment.
- 10.1.3 When acting as a Receiver or operator in accordance with this clause 10 or clause 5(b), the Procurer shall be entitled to be remunerated for such services as may be determined by Punjab State Electricity Regulatory Commission. Furthermore, when acting as a Receiver, the Procurer shall not be liable to the Lenders, the Lenders' Representative, Seller or any third party for any default under the PPA, damage or loss to the Power Station or for any other reason whatsoever, except for wilful default of the Procurer.
- 11. Substitution Consideration
- 11.1 The Lenders and the Procurer shall be entitled to appropriate any consideration received for the substitution of the Seller as hereinabove provided, from the Selectee towards the payment of Lenders' and the Procurer's respective dues, to the exclusion of the Seller.
- 11.2 The Seller shall be deemed to have nominated, constitutes and appoints the Lenders' Representative as its constituted attorney for doing all acts, deeds and things as may be required to be done for the substitution of the Seller by the Selectee pursuant to these terms.
- 12. Change in Lenders:

The Parties hereto acknowledge that during the subsistence of the PPA, it is possible that any Lender may cease to remain as a Lender by reason of repayment of the debt or otherwise. Further it may possible that any Lender may be substituted or a new Lender may be added. In the event of any Lender ceasing to be a party to the Financing Agreement, the term and conditions as prescribed in this Schedule shall cease to automatically apply to such Lender. Further, upon any entity being added as a Lender and in the event such entity is given the right to substitute the Seller under the Financing Agreement and then the contents of this Schedule shall be applicable to the exercise of such right by the said new entity.



11. SCHEDULE 11: ESTIMATE OF CAPITAL COST AS APPROVED BY PSERC VIDE ORDERS DATED 29.04.2008 & 06.08.2008 AND MODIFIED AS PER APPELLATE TRIBUNAL ORDER DATED 08.04.2009)

174

Sr.	Item	Estimates of Capital Cost
No.		accepted by PSERC and
		modified as per Appellate Tribunal order dated
		08.04.2009 (in Rs. Crore)
1	Land	109.35
2	Rehabilitation & Resettlement	5.00
3	Preliminary Expenses	0.25
4	Boiler Turbine Generator Package	1070.58 (See Notes 5a & 5b)
5	Balance Of Plant	955.00 (See Notes 6a &
		6b)
6	Engineering, Erection, Civil Works	Included in BTG BOP contracts
7.	Taxes and Duties	Included in BTG BOP contracts
8	Recommended spares for BTG and BOP	39.65
	package	•
9	Non EPC	86.00
10	Site grading and Ash Pond	49.00
11	Start-up Expenses	15.00
12	Power and Water for Construction	12.00
13	Consultancy and Engineering	7.50
14	Pre-operative Expenses	50.00
15	Operator Training & Mobilisation	5.00
16	Insurance	11.44
17	Capital Cost excluding IDC, Financing Charges	2415.77
	& Contingency	
18	Interest During Construction (IDC)	365.19 (See Notes 3 & 4)
19	Financing Charges	70.00



20	Contingency	66.85
21	Estimated Capital Cost excluding WCM	2917.81 •
22	Working Capital Margin(WCM)	46.00
23	Estimated Capital Cost	2963.81

Notes :

1.Estimated Costs against Sr. No. 1,2, 3,4,5,9,13,15&16 are as per PSERC orders dated 29.04.2008 & 06.08.2008.

2.Estimated Costs against Sr. No.8,10,11,12,14,18,19,20&22 are as per PSERC orders dated 29.04.2008 & 06.08.2008 and modified as per the Appellate Tribunal order dated 08:04.2009.

3. Interest During Construction will be at actuals, based on phasing of expenditure consistent with the finalized packages with various contractors.

4. IDC at actuals as per Financing Agreements will be considered for completed Capital Cost determination.

5a. BTG contract price will be subject to price variation formulae for supply price component, freight charges and erection testing & commissioning as indicated in the LOI dated 12 November 2007 issued to M/s BHEL.

5b. Contract price in the order placed with BHEL for the BTG package is payable in Indian Rupee only and the exchange rates for the Euro and US Dollar will be applicable on the date of payment. The BHEL contract price is based on the following exchange rates :

- US \$ = Rs.41.00
- Euro =Rs.57.50
- 6a. The BOP contract price of Rs.955 Crore is for execution of the contract with 'Induced Draft Cooling Towers'. If Natural Draft Cooling Towers are provided instead of Induced Draft Cooling Towers the increase in the BOP contract price will be Rs 50 crore for both the Units.

6b. BOP contract price will be subject to change as per any variation in cement and steel prices at the time of supply, upto the estimated quantities, as indicated in the LOI dated 14 April 2008 issued to M/s Punj Lloyd Ltd.

15

1.1



SCHEDULE 12: CALCULATION OF 'X' DAYS

(Refer Article 11.5.7)

Percentage of Monthly Involge which is the objection default under Article II. 4: as not biodon	1 st time 2 nd time 3 rd time 4 th time				
The Nonce (ussued under Article N(25/2) relatable to the present occurrence.				and onwards	
	x = 20 days	x = 25 days	x = 40 days	x = 60 days	
	$x = 20^{\circ}$ days	x = 30 days	x = 45 days	x = 65 days	
	x = 20 days °	x = 35 days	x = 50 days	x = 70 days	
More than 52 or 61400.	x = 20 days	x = 40 days	x = 55 days	x = 75 days	
Moré thau-40° c	x = 20 days	x = 45 days	$\begin{array}{l} x = 60 \\ days \end{array}$	x =90 days	

118

1.275



REPORT ON :

GEOTECHNICAL INVESTIGATION FOR 2x270 MW GVK POWER GOINDWAL SAHIB THERMAL POWER PROJECT, NEAR GOINDWAL SAHIB VILLAGE, PUNJAB

- TRIAL AREA - BOILER UNIT 1 & 2 (POST COMPACTION)

Submitted to:

M/s. GVK Power (Goindwal Sahib) Ltd. Paigah House 156-159, Sardar Patel Road Secunderabad-500 003 (A.P.)

jeni S

CENGRS GEOTECHNICA PVT. LTD. Job No. 210025

Sheet No. ii

TABLE OF CONTENTS

٥

		Sheet No.
1.0	INTRODUCTION	1
	1.1 Project Description	1
	1.2 Purposes of Study	1
	1.3 Details of Field Work	2 2
	1.4 Related Reports	4
2.0	FIELD INVESTIGATION	4
	2.1 Soil Borings	4
	2.2 Plate Load Tests	5
•		F
3.0	LABORATORY TESTS	5
4.0	GENERAL SITE CONDITIONS	6
-	4.1 Site Stratigraphy	6
	4.1.1 Trial Area	6 7
	4.1.2 Boiler-I and II Area	7
	4.1.3 Presentation of Results 4.2 Groundwater	7
	4.2 Groundwater	ł
5.0	EVALUATION OF PLATE LOAD TESTS	. 8
	5.1 Load-Settlement Data	8
	5.2 Analysis of Test Results	
	5.3 Discussion of Results	9
	5.4 Limitation of Plate Load Tests	9
6.0	RECOMMENDATIONS	10
7.0	CLOSURE	10
		¥ -
		M-
		$\mathcal{I}_{\mathcal{V}}$

· .

179 =

CENGRS GEOTECHNICA PVT. LTD.

ŵ

Table No.

Fig. No.

TABLES

Soil / Rock Profiles	1 to 4
Engineering Description of Soils	5
Engineering Description of Rock	6
Uncertainty in Laboratory Measurements	7

ILLUSTRATIONS

Plan of field investigation1a & 1bSummary of Borehole Profile2Standard Penetration Test Results3a & 3bPlate Load Test Results4a to 5bGrain Size Analysis6 to 9

Typical Calculations : Sheet Nos. 1 and 2

· •
CENGRS GEOTECHNICA PVT. LTD.

Job No. 210025

Sheet No: 1

INTRODUCTION 1.0

1.1 **Project Description**

M/s. GVK Power Ltd. as owner is setting up a 2x270 MW Coal Based Power Plant at Goindwal Sahib, Tarn Taran Distt., Punjab. M/s. TCE Consulting Engineers Limited is the project consultant from the owner side. M/s. Punj Lloyd Ltd. (PLL) is constructing the proposed Coal-Based Thermal Power Plant. M/s Fichtner Consulting Engineers is the Engineering Consultant for PLL. M/s Cengrs Geotechnica Pvt. Ltd. is the Geotechnical Engineering consultant for PLL/GVK.

We had earlier conducted a detailed geotechnical investigation at the site (Refer to our Report No. 209049). Based on the detailed investigations and analysis, it was concluded that the project site is susceptible to liquefaction to about 10.0 m below EGL · (RL208.5~209.0 m) in the event of an earthquake.

The clients decided to adopt vibro-compaction technique for the ground improvement at the site for all planned structures to mitigate the liquefaction potential. As per the information provided to us, the vibro-compaction has been carried out on 3x3 m grid pattern for various structures/units.

To confirm the safe bearing capacity for open foundations after ground improvement by vibro compaction; two plate load tests and two boreholes (BH-1&2) were conducted in the trial area. Two boreholes were also conducted in Boiler-1 and Boiler-2 areas. This report presents the results of four (4) boreholes and two (2) plate load tests and our engineering recommendations.

According to the information provided to us, ground levels at the site vary from RL 219.3 m to RL 219.6 m. A layout plan showing the locations of our field investigation is illustrated on Fig. No.'s 1a &1b.

1.2 Purposes of Study

The overall purpose of this study is to assess the safe bearing capacity for shallow open foundations on the treated ground, based on the results of the plate load tests and boreholes data. To accomplish these purposes, the study was conducted in the following phases:



Sheet No. 2

18

- a) drilling four (4) boreholes to 10 m depth and conducting standard penetration tests (SPT) at 1.5 m depth intervals, in order to investigate the site stratigraphy and to collect disturbed soil samples for laboratory testing;
- b) performing two (2) plate load tests in the trial area using 30 cm x 30 cm plate at a depth of 1.5 m below EGL so as to assess the load-settlement behavior of the underlying soil after vibro compaction; and
- c) analyzing the plate load test results and borehole data to assess the safe bearing capacity.
- 1.3 Details of Field Work

and the second se			
Test Location	Test Designation	Coordinates*	Existing* Ground Level, m
	BH-1	S:255.748- W:265.819	219.61
Trial Area	BH-2	S:279.756- W:297.448	219.61
IIIai Alea	PLT-1	S:277.485- W:263.099	219.61
	PLT-2	S:256:672- W:301.219	219.59
Boiler-1	BH-3	N:33.981-W:62.717	219.35
Boiler-2	BH-4	N:135.252- W:66.019	219.35

The test details are summarized below:

* as per the information given to us on site

1.4 Related Reports

Details of past investigations carried out in the areas under consideration are presented below for reference:

CENGRS GEOTECHNICA PVT. LTD.

182

Trial Area (Refer to Fig. No. 1a):

		\$			-
Test Designation	Co-ordinates*	Surface Elevation (RL), m	Termination Depth, m	Remarks	Report Reference
SCPT-1	244.144 S, 270.181W	219.480	15.0		
SCPT-2	244.144 S, 300.181W	219.485	15.0		
SCPT-3	274.144 S, 300.181W	219.475	15.0	Pre-	Report No.
SCPT-4	274.144 S, 270.181W	219.445	15.0 •	compaction	209188
SCPT-5	259.144 S, 285.181W	219.505	15.0		
SCPT-1	244.144 S, 270.181W	219.480	15.0		
SCPT-1	244.144 S,· 220.181W	219.586	11.0		
SCPT-2	244.144 S, 300.181W	219.591	7.4		
SCPT-3	274.144 S, 300.181W	219.641	15.0	Post-	Report No.
SCPT-4	274.144 S, 270.181W	219.616	11.0	compaction	209188-A
SCPT-5	259.144 S, 285.181W	219.641	9.4		
SCPT-6	281.644 S, 262.681W	219.700	13.6		

Boiler-I and II (Refer to Fig. No. 1b):

Location	Test Desig- nation	Co- ordinates	Surface Elevation (RL); m	Termina- tion Depth, m	Remarks	Report Reference
Boiler	SCPT-B1	60.238 N, 48.489 W	219.400	15.0		
Area-1	SCPT-B2	4.649 N, 74.815 W	219.400	15.0	Pre-	Report No.
Boiler	SCPT-B3	130.000 N, 50.000 W	219.450	15.0	Compaction	209188-B
Area-2	SCPT-B4	160.000 N, 60.000 W	219.450	15.0		

<u>,</u> ,



CENGRS GEOTECHNICA PVT. LTD.

025 Shee

Sheet No. 4

-183

Location	Test Desig- nation	Co- ordinates	Surface Elevation (RL), m	Termina- tion Depth. m	Remarks	Report Reference
ESP-1	SCPT-B5	51.583 N, 148.262 W	219.226	15.0	Pre-	Report No.
ESP-2	SCPT-B6	163.962 N; 154.699 W	219.260	15.0	Compaction	209188-B
Boiler	SCPT-A1	60.238 N, 48.489 W	219.400	14.2		
Area-1	SCPT-A2	18.439 N, 63.588 W	219.500	15.0		
Boiler	SCPT-A3	130.000 N, 50.000 W	219.500	15.0	- -	
Area-2	SCPT-A4	160.000 N, 60.000 W	219.500.	14.6		
ESP-1	SCPT-A5	33.50 N, 130.60 W	219.530	15.0		
ESP-2	SCPT-A6	163.962 N, 154.699 W	219.410	15.0		

2.0 FIELD INVESTIGATION

2.1 Soil Borings

The borehole was progressed using a mechanized shell and auger to the specified depth or refusal, whichever is encountered earlier. The borehole diameter was 150 mm. Where caving of the borehole occurred, 150 mm diameter casing was used to keep the borehole stable. The work was in general accordance with IS: 1892-1979.

Standard Penetration Tests (SPT) were conducted in the boreholes at 1.5 m depth intervals upto 10.0 m depth. The test was performed by connecting a split spoon sampler to 'A' rods and driving it by 45 cm using a 63.5 kg hammer falling freely from a height of 75 cm. The tests were conducted in accordance with IS: 2131-1981.

The number of blows for each 15 cm of penetration of the split spoon sampler was recorded. The blows required to penetrate the initial 15 cm of the split spoon for seating the sampler is ignored due to the possible presence of loose materials or cuttings from the drilling operation. The cumulative number of blows required to penetrate the balance 30 cm of the 45 cm of split spoon sampler is termed the SPT value or the 'N' value. CENGRS GEOTECHNICA PVT, LTD, Job No. 210025

Sheet No. 5

The SPT N-values (field and corrected) are presented in a graphical format on Fig. No's. 3a and 3b.

Disturbed samples were collected from the split spoon after conducting SPT. The samples were preserved in transparent polythene bags. All samples were transported to our NABL accredited laboratory at Delhi for further examination and testing.

2.2 Plate Load Tests

Plate load tests were performed at the site at specified level using a 30 x 30 cm size test plate. The test procedure is in general accordance with IS: 1888-1982.

An anchoring system by earth augers was used to provide the reaction. The plate was loaded by pushing up against the reaction anchors using a 50-ton capacity hydraulic jack.

Three dial gauges measured the plate settlement with reference to a stable reference bar. The load is applied in small increments upto a maximum loading intensity of 6.0 kg/cm² or 40 mm settlement of the plate, whichever occurs first.

Each load was held until the time rate of settlement became negligible (less than 0.02 mm per minute). The load was applied in stages about till the maximum applied pressure of 6.0 kg/cm² load was reached or until the total settlement exceeded 40 mm.

Test results are presented on Fig. Nos.4 and 5 as curves of bearing pressure on plate versus the measured settlement on liner and log-log scales.

3.0 LABORATORY TESTS

The laboratory testing has been carried out in our NABL accredited laboratory. The quality procedures in our laboratory conform to ISO/IEC-17025-2005.

CENGRS GEOTECHNICA PVT. LTD. Job N

Sheet No. 6

185

The laboratory testing programme was aimed at verifying the field classifications and developing parameters for engineering analysis. All testing was performed in accordance with the current applicable IS specifications.

The following tests were conducted on selected soil samples recovered from the boreholes:

Laboratory Tests	IS : Code Referred
Bulk density	By calculation
Natural moisture content	IS : 2720 (Part-2) -1973
Grain size analysis	IS : 2720 (Part-4) -1985
Atterberg Limits	IS : 2720 (Part-5) -1985

4.0 **GENERAL SITE CONDITIONS**

4.1 <u>Site Stratigraphy</u>

4.1.1 Trial Area

The surficial soils in the trial area generally consist of fine sand/silty sand from the ground surface to the final explored depth of 10.45 m (RL 208.9 m).

Field SPT values range from 31 to 39 to 3.0 m depth and 18 to 24 to about 4.0 m depth. Below this, field SPT values range from 43 to 50 to 5.0~6.0 m depth, 16 to 32 to 9.0 m depth, and 32 to 42 to the final explored depth of 10.45 m.

Corrected SPT values range from 43 to 54 to 3.0 m depth and 21 to 29 to about 4.0 m depth. Below this, corrected SPT values range from 32 to 36 to 5.0~6.0 m depth, 16 to 18 to 9.0 m depth, and 23 to 28 to the final explored depth of 10.45 m.

There is significant variation in the recorded SPT N-values at BH-1 and BH-2 locations with depth as well as laterally across the site at the same depth.

-186

4.1.2 Boiler-I and II Area

The surficial soils in the Boiler-I and II area generally consist of fine sand/silty sand from the ground surface to the final explored depth of 10,45 m (RL 208.9 m). A discontinuous layer of sandy silt is encountered at BH-3 between 0.5 m and 2.0 m depths.

Field SPT values range from 12 to 15 to 3.0 m depth. Below this, field SPT values at BH-3 range from 21 to 25 to about 5.0 m depth, whereas field SPT values at BH-4 are relatively higher and range from 39 to 62 to about 5.0 m depth. Below this, field SPT values range from 20 to 25 to about 8.0 m depth, about 45 to 10.0 m depth, and 32 to 34 to the final explored depth of 10.45 m.

Corrected SPT values range from 17 to 21 to 3.0 m depth. Below this, corrected SPT values at BH-3 range from 20 to 28 to about 5.0 m depth, whereas corrected SPT values at BH-4 are relatively higher and range from 31 to 43 to about 5.0 m depth. Below this, corrected SPT values range from 18 to 23 to about 8.0 m depth, about 30 to 10.0 m depth, and 23 to 24 to the final explored depth of 10.45 m.

4.1.3 <u>Presentation of Results</u>

The soil profiles (BH-1 to BH-4) with all necessary laboratory results are given on the Tables 1 to 4. Summary of the borehole profiles is illustrated on Fig. No.2. Field and corrected SPT results are plotted on Fig. No's. 3a and 3b.

Plate load test results are plotted on Fig No's 4 and 5. Grain size analysis curves are presented on Fig No's. 6 to 9.

4.2 Groundwater

Groundwater level was measured in the boreholes 24 hours after drilling and sampling was completed. Based on our measurements in the completed boreholes, groundwater was met at 3.2~3.6 m depth (RL 216.0-216.1 m) below EGL at the time of our field investigation (March, 2010).



CENGRS GEOTECHNICA PVT. LTD.

187

5.0 EVALUATION OF PLATE LOAD TESTS

5.1 Load-Settlement Data

Two plate load tests have been conducted at specified locations over the treated ground (i.e by vibro compaction). The tests were carried out upto a maximum load intensity of 6.0 kg/cm².

The following table presents the measured settlements of the two plate load tests carried out in the trial area:

Test No.	Test Level, m		ired Settle blied Bearin			Ultimate Bearing
INO.		10 T/m ²	12 T/m ²	15 T/m ²	20 T/m ²	Capacity, T/m²
PLT-1	RL 218.116 m	4.4	4.5	4.8	5.2	38.0
PLT-2	RL 218.091 m	5.3	5.8	6:2	7.8	30.0

5.2 Analysis of Test Results

The test results have been extrapolated for open foundations (2.5-3.0 m width) using the following equation:

$$\frac{S_{f}}{S_{p}} = \left[\frac{B_{f}(B_{p} + 0.3)}{B_{p}(B_{f} + 0.3)} \right]^{2}$$

where

 $S_f = Settlement of footing with width of footing as <math>B_f$ $S_p = Settlement of plate with width of plate as <math>B_o$

Settlement of isolated square foundations for width of footing 2.5~3.0 m has been extrapolated using the above equation. A suitable multiplying factor has been applied to account for saturation of soils as well as variations in the soil conditions.

Analyzing the test results, the following table presents the extrapolated settlement of the 2.5 - 3.0 m wide square footing:

CENGRS GEOTECHNICA PVT. LTD.

Sheet No. 9

Location	Test No.	of 2.5	m wide	ttlemen footing aring Pr of	under	of 3.0	ated Se m wide olied Be	footing	under
(an and a second se		10 T/m ²	12 T/m ²	15 T/m ²	20 T/m ²	10 T/m ²	12 T/m²	15 T/m²	20 T/m²
Trial;	PLT-1	44.0	45.0	48.0	52.0	42.1	43.0	45.9	49.7
area	PLT-2	53.0	58.0	62.0	78.0	50.7	55.5	59.3	74.6

5.3 Discussion of Results

- 1. The ultimate capacity of 30 cm square plates bearing at RL 218.1 m in the trial area (*Refer to Fig. No. 1a*) is in the range of 30-38 T/m². Applying a factor of safety of 2.5, the safe bearing capacity is in the range of 12-15 T/m² for 30 cm plate size.
- 2. The initial settlement of the plate upon application of the first load increment (corresponding to an applied bearing pressure of about 1.1 kg/cm²) was 4.4 mm and 5.3 mm at PLT-1 and PLT-2 locations, respectively. This indicates that the surficial soils below the test level are loose in condition. We suggest that the top layers be properly compacted prior to construction in these areas.
- 3. From the analysis of plate load test results, the settlement of 2.5~3.0 m wide footings bearing at RL 218.1 m for a net allowable bearing pressure of 10 T/m² works out to be about 44~53 mm. However, we expect that the actual foundation settlements may be lower if the top layers are properly compacted prior to foundation construction.
- 4. We suggest that the plate load test results be reviewed in conjunction with borehole and SCPT data in the area.
- 5.4 Limitations of Plate Load Tests

The analysis presented in this report is governed by the inherent limitations of plate load test, some of which are listed below:

(a) The analysis is applicable only for uniform Isotropic formations. Stratified deposits are not modelled effectively by the test.

CENGRS GEOTECHNICA PVT. LTD. Job No. 210025

- The test stresses the soils only to a depth of "2 B_p" below test (b) level (B_0 = plate width). Large size foundations will stress the deeper soils also. However, the behavior of the deeper soils cannot be evaluated by the test.
- The settlement measured during the test is primarily immediate (c) settlement. Consolidation or long-term settlement cannot be determined by the test.

RECOMMENDATIONS 6.0

- 1) **Trial Area** : Based on the results of plate load tests, boreholes and static cone penetration tests (SCPT's) carried out in the Trial area (Refer to Fig. No. 1a); we suggest a net allowable bearing capacity of 10 T/m² for open foundations of minor or lightly-loaded structures bearing on the improved ground at RL 218.0 m. The total settlement of the foundations is expected to be about 40~50 mm.¹ Due to the presence of loose soils at this level in some areas, we suggest that the soil at foundation level be thoroughly compacted using vibratory rollers, prior to. laying the foundation.
- 2) Boiler-I and Boiler-II: Based on the results of boreholes and SCPT's carried out in this area; we suggest a net allowable bearing pressure of 10 T/m² for the design of open foundations of minor or lightly-loaded structures bearing on the improved ground at RL 218.0 m. The total settlement of the foundations is expected to be about 40~50 mm. Due to the presence of loose soils at this level in some areas, we suggest that the soil at foundation level be thoroughly compacted using vibratory rollers, prior to laying the foundation.

7.0 **CLOSURE**

We appreciate the opportunity to perform this investigation for you and have pleasure in submitting this report. Please contact us when we can be of further service to you.

for CENGR'S GEOTECHNICA PRIVATE LIMITED

(RAVI SUNDARAM) DIRECTOR

(SANJAY GUPTA) MANAGING DIRECTOR

	-							•					•			·				19	0
\prec			9011	DC	ROFILE	Project : 2X270 Sahib, Punjab (F	MW Thermal Post Vibro Co	Power Pro	ject G	oindwa	al	BH.N	0; 1			RMINA DEPT		Table	No: 1a		(ili
				11			p-ordinates 8 S, 265.819	10/		ce Ele 19.61		TAW I	ER TA		1	0.45		JOB			
	ε	•				Location : Tr				in Siz		l Ilysis	3.6 m Atter		L Limits	<u> </u>			21	0025	Certificate
N-Value*	Reduced Level,	Depth (m)	Sample No.	Symbol		SOIL DESCR	RIPTION		Gravel %	Sand %	Silt %	Clay %	Liquid %	Plastic %	Piasticity Index %	Specific Gravity	Natural Density gms/cm ³	Dry Density gms/cm ³	Content	Confining Pressure Kalorr ²	g/cm²
	219.61 219.11		DS1		Dense grey f	fine sand (SP)				0)			<u></u>	<u> </u>	08	- 0	2.0		≥		<u>10 e</u>
	219.11 218.61		DS2					2	0	97	3	0									
39	218.11 217.66	1.50 1,95.	SPT1	T	Dense brown	n silly fine sand (S	SM)	(1.5m)													
	217.36 217.06	2.25 2.55	UDS1				o	(3.0m)	0	81	19	0					1.82	1.56	16,8		
18	216,61 216.16	3.45	SPT2		Medium dens - medium	se to dense grey f dense, 3.0 to 4.5	fine sand (SP m	(<u>3.011)</u> -SM)													
43	215.11 214.66	4.95	SPT3		- dense, 4	4.5 to 7.5 m															
	214.36	5.55	DS3						0	94	6	a				-		-			
32	213.61 213.16	6.45						. An approximation of party													
16	212.11 211.66		SPT5		- medium	dense, 7.5 to 9.0	m	And and the second of													
	211.36	8.55	DS4).0 to 10.45 m		an a													
39	210.61 210.16		SPT6		- with trac	es of gravel, 9.0 t	o 10.45 m	and the second second	2	93	5	0				·					

				•		() - () - ()- ()													19 ⁻		
					OFILE	Project : 2X270 MW Sahib, Punjab (Post	Thermal Power Proj Vibro Compaction)	ect Ge	aindwa	1	BH.No	»: 1			MINA		Table	No: 1b		60	
Ň	\mathbf{v}			11		Co-ord			ce Ele		WAT	ER TAI	BLE :				JOBI	NO.			Ŋ
	1		******	T	1	255.748 S, :		2	19,61	m		3.6 m			0.451	m)			0025	Certificate	<u>_</u> No.T-0809
	E					Location : Trial A	Area ·	Gra	in Siz	e Anal	ysis	Atter	perg t	imits				%	. 1	riaxial Te	
N-Value*	Reduced Level.	Depth (m)	Sample No.	Symbol		SOIL DESCRIPT	¢.	Gravel %	Sand %	% WS	Clay %	Liquid %	Plastic %	Plasticity Index %	Specific Gravity	Natural Density gms/cm ³	Dry Density gms/cm ³	Moisture Content %	Confining Pressure Kg/cm²	Cohesion Intercept Kg/cm²	Angle of Internal Friction
32	209.61	10.00	SPT7		Dense grey	fine sand with traces o	t gravel (SP-SM) (10.45m)						2 2								· · · · ·
I				t	<u></u>		-								فىنتىقىتىتىيەر		l		· c	Jutside NA	BL scc

.

×			SUI	pp	OFILE	Project : 2X270 MW Thermal Power Sahib, Punjab (Post Vibro Compactic	Project G n)	oindwa	al	BH.N	o: 2			MINA DEPTI		Table	No: 2a	i	
	Nor we have a second se		VVIL	. []]		Co-ordinates 279.756 S, 297.448 W	1	ace Ele 219.61		1	ER TA 3.55 m		1	0.45		JOB			
	E					Location : Trial Area		ain Siz					imits				21	0025 T	Certific
.Value*		Depth (m)	Sample No.	Symbol		SOIL DESCRIPTION	Gravel %	Sand %	Sill %	Clay %	Liquid %	Plastic %	Plasticity Index	Specific Gravity	√atural Density gms/cm³	Dry Density gms/cm ⁵		Confining Pressure Kgrom ²	on Malan ²
	219.61 219.11		DS1		Dense grey	fine sand (SP-SM)						<u>ц.,</u>	<u>0. 6</u>	0)	<u>د ن</u>	LI DI	-4	<u>O a, x</u>	01
	219.11 218.61	0.50 1.00	DS2				0	93	7	0									
31	218.11 217.66	1.50 1.95	SPT1			(2.0m) Medium dense brown silty fine sand (SM) (3.0m)						. *							
	217.36 217.06	2.25 2.55	UDS1		iviearum aen				14	0					1.86	1.76	5.63		
24	216.61 216.16	3.00 3.45	SPT2		Medium den - medium	(3,0 se to dense grey fine sand (SP-SM) dense, 3,0 to 4,5 m	<u>m)</u>												
50	215.11 214.66		SPT3		- dense, 4	4.5 to 6.0 m													
	214.36 214.06		UDS2				0	92	8	0					1,92	1.55	23.9		
17	213.61 213.16	6.45	SPT4		- medium	dense, 6.0 to 9.0 m													
20	211.66	7.95	SPT5			М.									•				
	211.36		DS3		- dense. 9	9.0 to 10.45 m	0	95	5	0			:						
42	210.61	9.00	SPT6		- with trac	es of gravel, 9.0 to 10.0 m	1		1	1									

1999 - Contractor States - Sta

	Γ							Project : 2X'	270 MW Thermal áb (Post Vibro Co	Power Proj	ject Gc	bindwa	ıl B	3H.No: 2		TEF	RMINA		Table	No: 2b		(All and a second	
	_				SOIL	. PK	ROFILE		Co-ordinates 9.756 S, 297.448	1	2	19.61	vation V m	3.55	5 m		DEPTH 10.45 /	н	JOB N		0025		
		ł	ε		Γ				: Trial Area		Gra	in Siz	e Analys	sis At	terberc	J Limits		<u> </u>	 	T		Certificate I Triaxial Te	est
		N-Value*	Reduced Level,	Depth (m)	Sample No.	Symbol			SCRIPTION		Gravej %	Sand %	Sit %	Clay % Liouid %	Liquid 76 Prestic %	Plasticity Index %	Specific Gravity	Natural Density gms/cm ³	Dry Density gms/cm³	Moisture Content %	Canfining Pressure Kg/am ²	Cohesion Intercept Kg/cm ²	Angle of Internal
۰.		40	209.61 209.16	10.00	SPT7		Dense grey f	fine sand (SP-	-SM)	(10.45m)						- Un u -							4
.,																							
																							1
•			-																				
		. 1	1							,									!	!			

			SUI	pp	OFILE	Project : 2X270 MW Thermal Power Pro Sahib, Punjab (Post Vibro Compaction)	ject Go	oindwa	31	BH.N	o: 3			MINA		Table I	No: 3a		. Car	Landa Landa
			VVIL	11		Co-ordinates	Surfa	ce Ele	vation	WAT	ER TA		1			JOB			166	I))
	·		<u></u>			33,981 N, 62,717 W		19,35			3.2 m		1	0.45	m	50037		0025	\square	Ľ
	ε			ļ	-	Location : Boiler-1	Gra	in Siz	e Ana	lysis	Atter	bergl	imits					T*****	Conflicate i Triaxial Te	
N-Value*	Reduced Level.	Depth (m)	Sample No.	Symbol		SOIL DESCRIPTION	Gravel %	Sand %	Silt %	Clay %	Liquid %	Plastic %	Plasticity Index %	Specific Gravity	latural Density ms/cm ³	Dry Density gms/cm ³	Moisture Content %	Confining Pressure Ka/am ²	n t Kg/cm²	Angle of Internal Friction
	219.35 218.85	0.00	DS1		Medium dens	e grey fine sand (SP-SM)	0	90	10	0		<u>a</u> .	0.8	<u></u>	2 0	<u>0</u> 6	~	063	ÖS	A P
	1 1	0.50	DS2		Medium dens	(0.5m) e brown sandy silt, low plastic (CL)	0	33	56	11										
15	217.85 217.40	1.60 1.95	SPT1		Modium down	(2.0m)					32.5	20.2	12.3							
	217.10 216.80	2.25 2.55	UDS1		- medium densi	e to dense grey fine sand (SP) lense, 2.0 to 4.5 m	0	97	3	0					1.74	1.62	7.38			
21	216.35 215.90		SP12																	
35	214.85 214.40		SPT3		- dense, 4.	5 to 6.0 m														
	214.10 213.80		UDS2				0	96	4	0					1.66	1.51	9.85			
29	213.35 212.90	6,00 6.45	SPT4		- medium d	lense, 6.0 to 8.0 m														
20	211.85 211.40	7.50 7.95	SPT5		0	(8.0m)											···			
	211.10 210,80		DS3		wense grey fir	ie sand (SP-SM)	0	91	9	0										
45	210.35 209.90		SPT6		Cont'd on Table															

$\langle $			SOI	PR	OFILE	Project : 2X270 MW Thermal F Sahib, Punjab (Post Vibro Con	paction)				8H.No			TER	MINA' DEPTI	TION 1	Table I	No: 3b		
	\sim			, 1 1 1 		Co-ordinates 33.981 N, 62.717 W		2	19,35	m		ER TAB 3.2 m	1		0.45 i	m	JOB I		0025	Certificate
	E					Location : Boiler-1		Gra	in Siz	e Ana	ysis	Atterb	erg L	imits			1			riaxial T
N-Value*	Reduced Level,	Depth (m)	Sample No.	Symbol		SOIL DESCRIPTION		Gravel %	Sand % .	Sit %	Clay %	Liquid %	Plastic %	Plasticity Index %	Specific Gravity	Natural Density gms/om ³	Dry Density gms/cm ³	Moisture Content %	Confining Pressure Ka/cm ²	Cohesion Intercept Ka/cm ²
34	209.35 208.90	10.00 10.45	SPT7		Dense grey f	ine sand (SP-SM)	(10.45m)						<u> </u>	0.8	<u></u>	2.0		~	OQX	<u> </u>
			i				e:													
					а -															
														{						

							ÿ									л.,			
	Y AL		<u>e</u> nii	DD	OFILE	Project : 2X270 MW Thermal Power Sahib, Punjab (Post Vibro Compacti	Project G	oindwa	31	BH.No	o: 4			MINA		Table	No: 4a		6
	20031 T			ГЛ /	VFILE	Co-ordinates 135,252 N, 66,019 W		ice Ele 19.35	evation m		ER TA 3.25 m		1	0.45		JOB		0025	
	3					Location : Boiler-2	Gra	in Siz	e Ana	lysis	Atter	berg I	limits		<u> </u>		8		i cerri riaxia
N-Value.	Reduced Level,	Depth (m)	Sample No.	Symbol		SOIL DESCRIPTION	Gravel %	Sand %	Sit %	Clay %	Liquid %	Plastic %	Plasticity Index %	Specific Gravity	Natural Density gms/cm ³	Dry. Density gras/cm ³	Moisture Contênt	Confining Pressure Ka/cm ²	Conesion
	219.35 218.85 218.85 218.35	0.50	DS1 DS2		Medium den:	se grey fine sand (SP-SM)	0	90	10	0			u o.		2 01	<u> </u>			
12	217.85 217.40 217.10 216.80	2.25	SPT1 UDS1		(SM)	(1. se to very dense brown silty fine sand dense, 1.5 to 3.0 m	<u>5m)</u> 0	79	21	0				40	1.61	1.41	14.3		
39	216.35 215.90 214.85	3.45	SPT2			3.0 to 4.5 m						•							
62	214.40 214.10 213.80	4.95 5.25	SPT3 UDS2		Medium dens	ise, 4.5 to 5.0 m (5. se to dense grey fine sand (SP-SM) dense, 5.0 to 9.0 m	0m) 0	94	6	0.					2.01	1.61	25.2		
22	213,35 212,90	6.00	SPT4																
25	211.85 211.40		SPT5																

			SUI	DD	OFILE	Project : 2X270 MW Thern Sahib, Punjab (Post Vibro	nal Power Pro Compaction)	oject Go	oindwa	il	BH.No	b: 4			MINA		Table I	Vo: 4b			2
	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~			.1 11		Co-ordinates 135.252 N, 66.01			ce Ele 19.35			ER 1AB 3.25 m	BLE :	1	0.45		JOBI		0025		
	Ε					Location : Boiler-2				e Ana		Atterb	bergl	imits	[				*****	Certificato riaxial T.e	
N-Value*	Reduced Level, m	Depth (m)	Sample No.	Symbol	-	SOIL DESCRIPTION	、 ·	Gravel %	© Sand %	Sift %	Clay %	Liquid %	Plastic %	Plasticity Index %	specific Gravity	Natural Density gms/cm ³	Dry Density gms/cm³	Moisture Content %	Confining Pressure Ka/cm ²	g/cm²	Angle of Internal Friction
32	209.35	10.00	SPT7		Denše grey	fine sand with gravel SP)	<u>(10,45m</u>	9	89	<u>0</u> 2	0		<u>.</u>	<u> </u>	ω	Zõ	G <u>B</u>			ŎĔ	9-1 170

* Outside NABL scope

197



1

## PLASTICITY OF CLAY

Plasticity	Liquid Limit
Low Plastic	< 35
Medium Plastic	35 to 50
High Plastic	> 50



## CONSISTENCY OF COHESIVE SOILS

Consistency	Cohesion Intercept, kg/sq.cm	SPT (N) Value
Very Soft	< 0,1	0 to 2
Soft	0.1 to 0.25	2 to 4
Firm/Medium	0.25 to 0.5	4 to 8
Stiff	0.5 to 1.0	8 to 15
Very Stiff	1.0 to 2.0	15 to 30
Hard	> 2.0	> 30

## DENSITY CONDITION OF GRANULAR SOILS

Density Descriptor	SPT (N) Value ∘	Static Cone Tip Resistance kg/sq.cm
Very Loose	0 to 4	< 20
Loose	4 to 10	20 to 40
Medium Dense	10 to 30	40 to 120
Dense	30 to 50	120 to 200
Very Dense	> 50	> 200

## DEGREE OF EXPANSION OF FINE GRAINED SOILS

Liquid Limit	Plasticity Index	Shrinkage Index	Free Swell Percent	Degree of Expansion	Degree of Severity
20 – 35	< 12	< 15	· < 50	Low	Non-critical
.35 – 50	12 – 23	15 – 30	50 - 100	Medium	Marginal
50 - 70	23 – 32	30 – 60	100 - 200	High	Critical
. 70 – 90	> 32	> 60	> 200	Very High	Severe

## ENGINEERING DESCRIPTION OF SOILS

198

CENGRS GEOTECHNICA PVT. LTD.

199

### NABL ACCREDITED LABORATORY

Our laboratory is accredited to National Accreditation Board for Laboratories (NABL), New Delhi. The quality procedures in our laboratory conform to the International Standard ISO/IEC: 17025-2005.

The accreditation assures our clients of work quality in conformance with international norms and practices. It authorizes us to use the NABL logo on test results.

To maintain the necessary level of quality and reliability in all measurements on a continual basis, we indulge in the following:

- Use of calibrated equipment, regular maintenance and good housekeeping are a part of our work culture.
- > Inter-laboratory comparison, proficiency testing and replicate testing, continuing education ensure uniform quality of results.
- Internal Audit of quality procedures is done by our qualified ISO 17025 auditors to maintain the requisite standards. NABL conducts external audit.

### UNCERTAINTY

Every measurement entails an uncertainty. It is well known that no measuring instrument can determine the true value of any measurement. The cumulative effect of factors such as sensitivity of equipment, accuracy in calibration, human factors and environmental conditions will determine the overall uncertainty in the parameter determined from these measurements.

As a part of our commitment to our clients, we have worked out the uncertainty in the parameters reported by our laboratory. Although this does not form a part of our contract agreement, we present below our statistical estimate of uncertainty of various parameters based on our most recent evaluation (March 2009).

Test / Paran	neter	Uncertainty*	Test / P	Param	eter	Uncert	ainty*
Moisture Cont	ent, %	± 0.22 %	Free Swe	ell Ind	ex, %	± 0.20	0 %
Bulk & Dry D	ensity	± 0.003 g/cc	Swell	Press	ure	± 0.43 ł	(g/cm ²
Specific Gra	avity	± 0.02	1		C _{c1}	± 0.0	003
Liquid Lin	nit -	± 0.22 %	Consolidat	ion	C _{c2}	± 0.0	003
Plastic Lir	nit	± 0.22 %	Gunsuluar	300	mv	± 0.0003	cm²/kg
Shrinkage L	.imit	± 0.12 %	]		Pc	± 0.15 k	kg/cm ²
Unconfined Compression	с	0.01 kg/cm ²	CD Direct S	hear	φ	± 0.2	25°
UU Triaxial	с	$\pm$ 0.07 kg/cm ²	Soil	1 -	ands & y sands	± 0.5% of p	article size
Test	Φ	± 0.2°	Gradation	1	ndy silt / yey silt	± 1.5% of p	article size
Std/Mod Proctor	MDD	= 0.03 g/cc	Coeff Perm			± 1.3 % c	of value
Compaction	омс	± 0.22 %				Crushing strength	± 0.80% of value
Laboratory (	BR	± 0.07 %	R	ock		Point Load strength index	± 2.40% of value

* at 95 percent confidence level for coverage factor of 2

### UNCERTAINTY IN LABORATORY MEASUREMENTS



Fig N200a



쪻





PLAN OF FIELD INVESTIGATION (Boiler Area- -1&2)
2 x 270 MW Thermal Power Project Goindwal Sahib, Punjab

Not to Scale





2X270 MW Thermal Power Project Goindwal Sahib, Punjab (Post Vibro Compaction)

Jh-



		LEGEND	
Symbol	BH.No.	Location	Reduced Level,m
-0-	219.61	Trial Area	219.610
-0	219.61	Trial Area	219.610
	219.35	Boiler 1	219.350
	219.35	Boiler 2	219.350

2X270 MW Thermal Power Project Goindwal Sahib, Punjab (Post Vibro Compaction)







(Post Vibro Compaction)

1



-

 $\widehat{\mathbb{Q}}$ 

(Post Vibro Compaction)



LABORATORY: CENGRS GEOTECHNICA PVT. LTD. An ISO /IEC: 17025-2005 accredited Laboratory



Certificate No.T-0809





.

LABORATORY: CENGRS GEOTECHNICA PVT. LTD. An ISO /IEC: 17025-2005 accredited Laboratory

Job No.: 210025



## Grain Size Analysis

		SAMPL	E DETAILS		]			IES	T RESU	LIS		
	orehole umber	Sample Depth, m	Sample De	escription		% Gravel	% Sand	% Silt	% Clay	D ₆₀	D ₁₀	C,
	BH-2	0.50	Fine sand	(SP-SM)		0	93	7	0	0.269	0.098	2.8
	BH-2	2.25	Silty fine s	and (SM)		0	86	14	0	0.220		
	BH-2	5.25	Fine sand	(SP-SM)		0	92	8	0	0.239	0.105	2.3
	BH-2	8.25	Fine sand	(SP-SM)		0	95	- 5	0	0.263	0.117	2.3
								,				<del>, , , , , , , , , , , , , , , , , , , </del>
	CLAY		SILT			FINE	r		D DIUM	COAR	GRAV	/EL
100 ·	·	· · · · · · · · · · · · · · · · · · ·					<u>l</u>			- D	······	 
	1					* * *	1					
90 -	5 2 8 2			: ۲۰۰۰ مور میسیسی میشویت ۲۰۱۰ مور میسیسی : : :		3			1		a * * *	
80 -	ء ج ب ب	6 1 5 4 6 8 2 1 		a a a a a a a a a a a a a a a a a a a		5 5 1 1			· · · · · · · · · · · · · · · · · · ·	3 4 - 2 		
:70 -	1 3 1	, , , , , , , , , , , , , , , , , , ,	<pre></pre>	2		,     			-		· · · · · · · · · · · · · · · · · · ·	
50 -	1 7 8 9			·	10 11 10 10 10 10 10 10 10 10 10 10 10 1				an or an or an an	2 2 2 2 3 1		
60 -	···					1			· ·	4 i 4 i 5 -	-	5 1 1 2 . 2
50 -	2 	E 4 8 8 3 5 8 5 5 8 2 4 8 3 8 2		3		[]]		2 2 4 1 1 3 2 2 2 2 2 3 5 2 4 7 5 5 5 5			1 1 4 4 4 - 4 - 4 - 4 - 4 - 4 - 4 - 5 - 4 - 5 - 5 - 5	
40 -	\$			и Х. и . и 8 - Т. и . и 2 - К. и			4 8 3 3			t 1 t 4 2 2	۲۰۰۰ ۲۰۰۰ ۱۹۹۰ - ۲۰۰۰ ۲۰۰۰ ۱۹۹۰ - ۲۰۰۰ - ۲۰۰۰ ۲۰۰۰ ۱۹۹۰ - ۲۰۰۰ ۲۰۰۰	
30 +	2 2 2							1 2 8 8 1 3 3 8 3 - 1 8 8 4 5 8 8 8 7 8 8 8 8 7 8 8 8 8 8 7 8 8 8 8 8 7 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8				
	1 2 3 6 4	1 2 1 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7			And A Destination		a á tr		orehole No:	2 Depih: 0	5 m	5 i 2 i 1 t - a
20 +	5 3 4				/	A	1		orehole No:			
10 +	3 1			· · · · · · · · · · · · · · · · · · ·			2 2 3	<del>. 4</del> B	orehole No:	2 , Depth: 5	25 m -	
0 +	2 3 1	3 1 1 2 1 , 1 3 3 . 1 3 4 5 1 				3 5 5	·		orehole No:	2 , Depth: 8	25 m	1 · 4 1 · 2 3 · 3
0.00	1		0.010	Particle	0.100 Size			1.	000			10.00
				Locatio	on: Tr	ial Area						
,	2X27	70 MW Therr	mal Power Proj	ect Goind	wal :	Sahib, F	ounjab	(Post V	ibro Co	mpacti	on)	



10

 1

0.010

" s }







		SAMPL	E DETAILS				TES	T RESU	LTS		
	orehole umber	Sample Depth, m	Sample D	escription	% Gravel	% Sand	% Silt	% Clay	D ₆₀	D ₁₀	C,
	BH-3	0.00	Fine sand	J (SP-SM)	O	90	10	0	0.272		
1	BH-3	0.50	Sandy	silt (CL)	. 0	33	56	11	0.017		
f	ЗН-3	2.25	Fine sa	nd (SP)	0	97	3	0	0.208	0.090	2.3
E	3H-3	5.25	Fine sa	nd (SP)	0	96	4	0	0.247	0.125	2.(
E	3H-3	8.25	Fine sand	(SP-SM)	0	91	9	0	0.233	0.080	2.9
	······	•									
	CLAY		SILT	-	FINE		SANI MEE		COARS	SE GRAN	/EL
00 -	·					- 9 - 24				~	
30 -		· · · ·							0	1 1 1 1 1	
	,					111		1	1	4 2 1 5	1.1
				2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2		14 19			- I - I - I - I - I - I - I - I - I - I		
	- 								1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -	4 2 1 5   6 2 2 2   1 5 1 7   1 3 2 1   1 3 2 1   1 3 2 1   1 3 2 7   2 3 7   5 2 3 7   6 2 3 7   1 2 3 7   2 3 7 7   5 3 5 7   8 5 7 7   8 5 7 7	2 3   3 4   3 3   4 1   5 4   4 1   4 1   4 1
						0			1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -	4     2     1     5       4     5     4     5     6       6     5     7     7     7       1     2     2     1     7       1     2     2     1     7       1     2     7     7     7       2     2     7     7     7       2     2     7     7     7       2     2     7     7     7       3     4     7     7     7       4     1     7     7     7       5     4     7     7     7       4     1     7     7     7       5     2     7     7     7       6     3     7     7     7       6     3     7     7     7       6     3     7     7     7       7     4     7     7     7	
·0									I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I	4     3     1     5       4     4     4     4       4     5     4     5       4     5     6     5       4     5     6     5       4     5     7     7       1     2     2     7       1     3     2     7       2     3     1     7       2     3     1     1       2     3     1     1       2     3     1     1       2     3     1     1       2     3     1     1       2     3     1     1       2     3     1     1       2     2     1     1       3     2     2     1       3     2     2     1       4     2     1     1       4     2     2     3       5     1	2 3   3 4   2 3   4 3   5 4   7 4   7 4   8 5   1 4   1 5   1 4   1 4   1 4   1 4   1 4   1 4   1 4   1 4   1 4   1 4   1 4   1 4   1 1   1 1   1 1
0 +									I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I       .     I	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	2 3   3 3   3 3   3 3   4 4   5 3   4 4   4 4   5 3   7 4   4 4   5 3   7 4   4 4   5 3   7 4   8 5   1 1   1 1   1 1   1 1   1 1   2 3   1 1   1 1   2 3   2 3   3 1   2 3   3 1   3 1   2 3   3 1   3 1   4 2   3 1   4 3   5 1   5 2   4 2
										$\begin{array}{cccccccccccccccccccccccccccccccccccc$	

0.100

Location: Boiler-1 2X270 MW Thermal Power Project Goindwal Sahib, Punjab (Post Vibro Compaction)

Particle Size, mm

1.000

- Borehole No: 3 , Depth: 0.5 m

Borehole No: 3 , Depth: 5.25 m - Borehole No: 3 , Depth: 8.25 m

-Borehole No: 3 . Depth: 2.25 m



10.000









~

**ANNEXURE P-3** 

# PUNJAB STATE ELECTRICITY REGULATORY COMMISSION

Site No.3, Block B, Madhya Marg, Sector 18-A, CHANDIGARH

# Petition No. 32 of 2019 Date of Order:<u>07.09.2020</u>

Petition for True up of Tariff for FY 2016-17 under Sections 62 and 86 of the Electricity Act, 2003 read with (a) Punjab State Electricity Regulatory Commission (Terms and Conditions for Determination of Tariff) Regulations, 2005; (b) the Central Electricity Regulatory Commission (Terms and Conditions for Determination of Tariff) Regulations, 2014; and (C) Amended and Restated Power Purchase Agreement dated 26.05.2009 executed between Petitioner (GVK) Limited and Punjab State Power Corporation (formerly Limited known Punjab State as Electricity Board)

# AND

In the matter of:

GVK Power (Goindwal Sahib) Limited, Paigah House,156-159, Sardar Patel Road,Secundrabad-540003.

....Petitioner

## Versus

Punjab State Power Corporation Limited, The Mall Patiala

....Respondent

## Present:

Ms. Kusumjit Sidhu, Chairperson Shri S.S.Sarna, Member Ms. Anjuli Chandra, Member

# **ORDER**

1.1 GVK Power (Goindwal Sahib) Limited (GVK) has filed the present petition for True Up of Tariff for FY 2016-17 for its 540

MW Coal Based Thermal Power Station located at Goindwal Sahib in the State of Punjab. GVK had submitted estimated and actual Annual Revenue Requirement of Rs. 2192.63 Crore and Rs. 1056.81 Crore respectively under different heads considering a capital cost of Rs 4267.30 Crore. GVK has prayed to:

- Admit the present petition seeking approval of True up for FY 2016-17.
- b) Approve the True-up petition for 2016-17.
- c) To pass such order(s) as the Commission may deem fit in the circumstances and facts of the present petition.
- 1.2 The deficiencies observed by the Commission were conveyed to GVK vide letter no. PSERC/Reg/2618-2619 dated 21.01.2020, directing GVK to file its reply to the deficiencies before 29.01.2020. Notice was also issued to PSPCL. The petition was taken up for hearing on admission on 06.02.2020. Some more deficiencies were observed in the petition and the same were conveyed to GVK alongwith the Order dated 07.02.2020.
- 1.3 GVK replied to the deficiencies vide affidavit dated 11.03.2020. The petition was admitted vide order dated 13.03.2020 further directing PSPCL to file its reply to the petition by 26.03.2020 and rejoinder, if any, by GVK by 03.04.2020.
- 1.4 PSPCL filed its reply to the petition vide memo No. 5415 dated 27.05.2020. The Commission vide order dated 16.06.2020

directed GVK public to issue notice inviting а objections/suggestions from the general public/stakeholders within 10 days of the publication of the notice. PSPCL filed its reply vide memo No. 5516 dated 23.06.2020. The Commission vide order dated 26.06.2020 allowed GVK to file the revised figures according to the Order of the Commission in petition No. 54 of 2017. PSPCL was allowed to submit its sur-rejoinder, if any, to the rejoinder filed by GVK. The Commission directed GVK to submit the draft public notice by 03.07.2020 failing which GVK will be liable to action under Section 142 of the Electricity Act, 2003 and the petition was fixed for hearing on 29.07.2020.

- 1.5 The public notice was published in Times of India (English), Daily Ajit (Punjabi) and Ajit Samachar (Hindi) on 10.07.2020. GVK, in pursuance to the Order dated 26.06.2020, submitted the revised figures vide affidavit dated 16.07.2020 alongwith the reply to deficiencies raised by the Commission in its notice dated 21.01.2020. PSPCL vide memo no. 5466 dated 28.07.2020 filed its amended reply to the revised true-up proposal submitted by GVK. The petition was taken up for hearing as well as public hearing on 29.07.2020, however, nobody appeared from the public in public hearing. After hearing the parties, the Order was reserved vide Order dated 30.07.2020 while allowing the parties to file their written submissions as per their request.
- 1.6 GVK Power (Goindwal Sahib) Limited (GVK), the Petitioner, filed the instant Petition for True-up of FY 2016-17 of its 2 x 270 MW Coal Based Thermal Power Project under Section 63

3
of the Electricity Act, 2003 read with Regulation 62 of Punjab State Electricity Regulatory Commission (Terms and conditions for determination of Generation, Transmission, wheeling and retail supply Tariff) Regulations, 2014.

1.7 The commercial operation of this project was declared on 16.04.2016. The Commission vide its order dated 17.01.2020 in Petition No.54 of 2017, approved the capital cost of the project as Rs 3058.37 Crore and determined the provisional Annual Fixed Cost of Rs 545.42 Crore for FY 2016-17. GVK in its original petition has claimed Annual Revenue Requirement of Rs. 1056.81 Crore which was further revised on 21.07.2020 and 21.08.2020 to Rs. 649.44 Crore as given below:

Table No.1: Annual Revenue Requirement submitted by GVK for FY 2016-17

				(Rs. Crores)
Sr. No.	Particulars	Estimated	Approved by the Commission	Actual
1	Fuel Cost	and		
a)	Primary Fuel Cost	1180.95	52.07	65.34
b)	Secondary Fuel Cost	5.00		1.12
2	SLDC Fees & Charges			181
3	O&M expenses	139.81	30.41	30.78
	a) R&M Expenses			
	b) Employee Expenses			00
	c) A&G Expenses			
4	Depreciation	202.59	141.27	139.41
5	Interest on Loans	404.72	225.51	261.81
6	Interest on Working Capital	73.62	12.25	15.50
7	Prior Period Expense	Y.		
8	Extraordinary Items			
9	Other Debts and Write-offs			
10	Income Tax	0.00		0.00
11	Less: Expenses capitalised			
	a) Interest Charges			
	Capitalized			
	b) R&M Expenses Capitalized			
	c) A&G Expense Capitalized			
	d) Employee Expenses			
	Capitalized			

4

Sr. No.	Particulars	Estimated	Approved by the Commission	Actual
	Subtotal (a+b+c+d)	0.00		0.00
	Subtotal Expenditure (1+2+3+4+5+6+7+8+9+10-11)	2006.69	461.51	513.96
С	Return on Equity	185.94	135.98	135.98
D	Non Tariff and other Income	0.00	0.00	0.50
Е	Annual Revenue Requirement (B+C-D)	2192.63	597.49	649.44
Noto				

Notes:

- 1. Depreciation, Interest on Loan, Interest on working Capital and Return on Equity have been claimed in accordance with PSERC Tariff Regulations 2005.
- 2. O&M Expenses have been claimed as actual O&M expenses as per books of accounts

#### 2.0 Capital Cost for FY 2016-17

## GVK's submission:

- 2.1 GVK in the petition has claimed capital expenditure of Rs. 4383.46 Crore as on the Commercial Operation Date i.e. 16.04.2016 (including un-discharged liabilities of Rs. 116.16 Crore). The Un-discharged liabilities of the project, as submitted by GVK are as follows:
  - (a) Retention money payable to BHEL of Rs. 111.11 Crore.
  - (b) Bills outstanding against BTG Contracts of Rs. 3.54 Crore.

(c) Amount outstanding against spares of Rs. 1.51 Crore.

The capital expenditure (excluding un-discharged liabilities) is Rs. 4267.30 Crore. Further, GVK has not claimed any additional capital cost during the period from COD to 31.03.2017.

2.2 Further, GVK in its revised submission received on 21.07.2020 submitted that Capital Cost incurred till COD of the Project i.e. 16.04.2016 is Rs. 4376.41 Crore. The financial closure of the Project was achieved in 2010 considering a capital cost of Rs. 3200 Crores and a debt equity ratio of 75:25. However, due to

events of Force Majeure and Change in Law, i.e. cancellation of the Captive Coal Blocks, actual COD took place on 16.04.2016, resulting in time overrun of approximately 39 months from SCOD as per financial closure and cost overrun of Rs. 1176.41 Crore. The overall debt to equity ratio at the stage of project completion stood at 73:27. However, the Commission vide order dated 17.01.2020 in Petition no. 54 of 2017 approved a capital cost of Rs.3058.37 Crore.

## **PSPCL's submission:**

2.3 PSPCL vide its submissions dated 27.05.2020 stated that the Commission vide its Order dated 17.01.2020 passed in the petition no. 54 of 2017 has determined the capital cost of the GVK's project as Rs. 3058.37 Crore as on the date of commissioning of the project.

## **Commission's Analysis**

2.4 The Commission vide its Order dated 17.01.2020 in petition no. 54 of 2017 has determined the capital cost of the project after prudence check at Rs. 3058.37 Crore as on the date of commissioning of the GVK Power Plant i.e. 16.04.2016.

As, GVK has not claimed any additional capital expenditure from 17.04.2016 to 31.03.2017 in the true-up petition for FY 2016-17, thus, the Commission approves capital cost of Rs. 3058.37 Crore for the FY 2016-17 as on 31.03.2017.

#### 3.0 Operation and Maintenance Expenses

## **GVK's Submission**

3.1 GVK submitted that the Project being a coal-based power plant

is governed by Regulation 29 of the CERC Tariff Regulations.

The CERC Tariff Regulations prescribes Operation and Maintenance expenses for 200/210/250 MW sets for FY 2016-2017 at the rate of Rs. 27.00 Lakhs per MW. The O&M expenses has been considered at Rs. 139.81 Crores on normative basis.

3.2 GVK in its revised submission received on 21.07.2020 and 21.08.2020, submitted that in terms of Regulation 28(2) of the PSERC Tariff Regulation 2005, O&M Expenses allowable to GVK is to be computed as under:

## "28. 2 [OPERATION AND MAINTENANCE EXPENSES

- (1) 'Operation & Maintenance expenses' or 'O&M expenses' shall mean repair and maintenance (R&M) expenses, employee expenses and administrative & general expenses (A&G) including insurance.
- (2) O&M expenses for distribution licensee (s) shall be determined by the Commission as follows :
- (a) O&M expenses as approved by the Commission for the year 2011-12 (true up) shall be considered as base O&M expenses for determination of O&M expenses for subsequent years.
- (b) Base O&M expenses (except employee cost) as above shall be adjusted according to variation in the average rate (on monthly basis) of Wholesale Price Index (all commodities) over the year to determine the O&M expenses for subsequent years.
- (c) In case of a new distribution licensee (s), the Commission shall make suitable assessment of base O&M expenses of the new licensee (s) and allow O&M expenses for subsequent years for the new licensee (s) on the basis of such estimation and principle as given in clause (b) above. However, for employee cost the principle specified in clause (3) below will be followed.

- (3) The employee cost for a distribution licensee (s) shall be determined as follows:
- (a) The employee cost as claimed by the distribution licensee (s) shall be considered in two parts:
  - (i) Terminal benefits such as Death-cum-Retirement Gratuity, Pension, Commuted Pension, Leave Encashment, LTC, Medical reimbursement including fixed medical allowance in respect of pensioners and share of BBMB employee expenses and
  - (ii) all other expenses accounted for under different sub-heads of employee cost taken together.

The cost component of terminal benefits and BBMB expenses shall be allowed on actual basis and increase in all other expenses under different subheads shall be limited to the increase in Wholesale Price Index (all commodities) as per clause (2) (b) above.

- (b) Exceptional increase in employee cost on account of pay revision etc. shall be considered separately by the Commission.
- (c) The additional employee cost in case of New installations/Network for the year of installation shall be considered separately by the Commission on case to case basis keeping in view the principles and methodologies enunciated in these regulations.
- (5)(a) For the determination of O&M expenses (except employee cost) for generating company, the Commission shall allow O & M expenses (except employee cost) in accordance with Clause (2). The employee cost will, however, be determined keeping in view the provisions contained in Clause (3).
- (b) In case of a new generating company (s), the Commission shall make suitable assessment of base O&M expenses of the new licensee (s) and allow O&M expenses for subsequent years for the

new licensee (s) on the basis of such estimation and principle as given in Clause (2)(b) above. However, for employee cost the principle specified in Clause (3) above will be followed."

3.3 The actual O&M Expenditure incurred by GVK in FY 2016-17 is as under:

Table No. 2: O& M Expenses claimed by GVK for FY 2016-17

(Rs. Crore)

Sr. No	Particulars	Estimated	Approved by the Commission	Actuals
1	Plant Capacity	540MW	540 MW	540 MW
2	O&M Expenses	145.80	31.80	32.08
3	Pro rata O&M Expenses for FY 2016-17	139.41	30.41	30.78

3.4 In reply to the deficiencies raised by the Commission, GVK submitted proof of payment of employee contributions alongwith the requisite details of employees for FY 2016-17.

## **PSPCL's Submission**

3.5 PSPCL vide Memo No. 5516 TR-5/952 dated 23.06.2020 submitted that the O&M expenses are required to be recomputed by the Petitioner strictly in accordance with the methodology prescribed under Regulations 28 of the PSERC Tariff Regulations 2014.

3.6 PSPCL vide its submission dated 28.07.2020 and 10.08.2020 stated that this Commissions in its Order dated 17.01.2020 while approving O&M expenses for the Petitioner at Rs.30.41. Crores has held as under:

#### "26.1.1

. . .

As per the above sub regulation(5)(b), the Commission has to make suitable assessment of base O&M expenses in case of new generating company. This exercise will be taken up during true-up of FY 2016-17.

..... An expenditure Rs. 0.28 Crore depicted in 'other operating expenses" in Annual Audited Accounts relates to provision for diminution in value of investment and the same is not allowable as 0&M expenses. Thus, 0&M expenses are allowed at Rs.31.80 (32.08-0.28) Crore for FY 2016-17."

Thus, while agreeing to take up the true up exercise for O&M expenses at the appropriate stage, this Commission had declined to allow Rs. 0.28 Crore shown as other operating expenses" by GVK in its accounts, GVK is now seeking the said amount in the present true-up Petition. However, yet again no details of the same have been furnished by GVK. As such the same is liable to strict scrutiny and prudence check by the Commission.

#### **Commission's Analysis**

- 3.7 The Commission had provisionally approved O& M expenses of Rs. 30.41 Crore for GVK for the period 17.04.2016 to 31.03.2017 in the order dated 17.01.2020 in Petition no 54 of 2017.
- 3.8 The O&M expenses are to be determined as per provisions of Regulation 28 of PSERC (Terms and Conditions for determination of Tariff), Regulation 2005 as mentioned in para 3.2 above.
- 3.9 As per the above sub regulation(5)(b), the Commission has to make a reasonable assessment of the base O&M expenses in case of a new generating company. The Annual Audited Accounts of GVK for FY 2016-17 reveals Employee benefit expenses of Rs 8.66 Crores which includes Terminal Benefits of Rs 0.65 Crore. The Commission allows the terminal benefits for FY 2016-17 as claimed by GVK in the petition. The Commission

allows the claim of the Employee cost as under:

## Table No.3: Employee cost allowed by the Commission for the<br/>period 17.04.2016 to 31.03.2017.

		(Rs. Crore)
Sr.no	Particulars	Amount
1	Terminal benefits	0.65
2	Other Employee cost	8.01
3	Total Employee Cost for FY 2016-17	8.66
4	Total Employee Cost (8.66*349/365)	8.28

3.10 The other operating expenses (A&G and R&M expenses) are Rs.23.42 (10.48+12.94) Crore as per the Annual Audited Accounts of GVK for FY 2016-17. An expenditure of Rs.0.28 Crore depicted as "other operating expenses" in the Annual Audited Accounts relates to provision for diminution in value of investment and the same is not allowable as O&M expenses. Thus, the Commission approves R&M and A&G expenses as per under:

Table No.4: R&M and A&G expenses approves by the Commission for the period 17.04.2016 to 31.03.2017.

		(RS. Crore)
Sr. No.	Particulars	Amount
1.	R & M expenses	12.94
2.	A & G expenses	10.48
3.	R&M and A&G expenses	23.42
4.	Less Diminution in value of investment	0.28
5.	R & M and A & G expenses for FY 2016-17	23.14
6.	R&M and A&G expenses (23.14*349/365)	22.13

3.11 Accordingly, the Commission approves O&M expenses for the period 17.04.2016 to 31.03.2017 as under:

Table No. 5: O&M expenses approved by the Commissionfor the period17.04.2016 to 31.03.2017.

(Rs. Crore)

Sr. No	Particulars	FY 2016-17	17.04.2016 to 31.03.2017
1.	Employee Cost	8.66	8.28
2.	R&M expenses	23.14	22.13
3.	O & M expenses	31.80	30.41

## 4.0 Depreciation

## GVK's Submission:

4.1 GVK submitted that depreciation has been computed for FY 2016-17 in accordance with Regulation 27 of the CERC Tariff Regulations.

## "27. Depreciation:

(1) Depreciation shall be computed from the date of commercial operation of a generating station or unit thereof or a transmission system including communication system or element thereof. In case of the tariff of all the units of a generating station or all elements of a transmission system including communication system for which a single tariff needs to be determined, the depreciation shall be computed from the effective date of commercial operation of the generating station or the transmission system taking into consideration the depreciation of individual units or elements thereof.

Provided that effective date of commercial operation shall be worked out by considering the actual date of commercial operation and installed capacity of all the units of the generating station or capital cost of all elements of the transmission system, for which single tariff needs to be determined.

- The value base for the purpose of depreciation shall (2) be the capital cost of the asset admitted by the of multiple units of a Commission. In case or multiple elements generating station of transmission system, weighted average life for the generating station of the transmission system shall be applied. Depreciation shall be chargeable from the first year of commercial operation. In case of commercial operation of the asset for part of the year, depreciation shall be charged on pro rata basis.
- (3) The salvage value of the asset shall be considered as 10% and depreciation shall be allowed up to maximum of 90% of the capital cost of the asset:

Provided that in case of hydro generating station, the salvage value shall be as provided in the agreement signed by the developers with the State Government for development of the Plant:

Provided further that the capital cost of the assets of the hydro generating station for the purpose of computation of depreciated value shall correspond to the percentage of sale of electricity under longterm power purchase agreement at regulated tariff:

Provided also that any depreciation disallowed on account of lower availability of the generating station or generating unit or transmission system as the case may be, shall not be allowed to be recovered at a later stage during the useful life and the extended life.

- (4) Land other than the land held under lease and the land for reservoir in case of hydro generating station shall not be a depreciable asset and its cost shall be excluded from the capital cost while computing depreciable value of the asset.
- (5) Depreciation shall be calculated annually based on Straight Line Method and at rates specified in Appendix-II to these regulations for the assets of the generating station and transmission system:

Provided that the remaining depreciable value as on 31st March of the year closing after a period of 12 years from the effective date of commercial operation of the station shall be spread over the balance useful life of the assets."

4.2 GVK in its petition has submitted that the weighted average rate of depreciation of 4.80% based on the actual capital cost as on Station COD and the applicable depreciation rates for the different categories of assets as per Appendix-II (Depreciation Schedule) of the CERC Tariff Regulations. Further, GVK has worked out depreciation of Rs. 196.24 Crore at the weighted average rate of depreciation of 4.80% based on the Capital cost of Rs 4267.30 Crore.

- 4.3 GVK in its revised submission received on 21.07.2020 and 21.08.2020 has submitted that in terms of Regulation 27(1) (d) of the PSERC Tariff Regulations 2005 depreciation up to 90% of capital cost of the power plant using the straight line method is allowed over the useful life of the Project at the rate of depreciation specified by the Central Electricity Regulatory Commission.
- 4.4 GVK submitted that the gross fixed asset of the Project for the purpose of depreciation is Rs. 3058.37 Crore in its revised submission dated 21.07.2020. The category-wise gross fixed asset value and the corresponding depreciation rates as submitted by GVK in accordance to the Appendix-II (Depreciation schedule) of the CERC Tariff Regulations, 2014 is as under:

Sr. No	Particulars	Asset value as on Station COD	Rate of depreciation (as per CERC 2014- 19 Tariff Regulations)
1.	Land and land rights	96.75	0.00%
2.	Buildings	486.29	3.34%
3.	Plant and machinery	2447.66	5.28%
4.	Vehicles	0.63	9.50%
5.	Furniture and fixtures	1.23	6.33%
6.	Office equipment	1.12	6.33%
7.	Computers	0.63	15.00%
8.	Computer software	0.12	15.00%
9.	Right to Use Railway Line	23.94	0.00%
10.	Total	3058.37	Weighted avg. rate of depreciation: 4.77%

# Table No. 6: Rates of Depreciation submitted by GVK for FY 2016-17.

4.5 GVK has further submitted that the depreciation for the year has been computed based on the Capital cost approved by the

Commission in Order dated 17.01.2020 in Petition no 54 of 2017 and the rate of depreciation derived in the above table. Depreciation thus derived has been pro-rated based on Project COD i.e. 16.04.2016 as shown under:

Table No. 7: Depreciation expenses	submitted by GVK
------------------------------------	------------------

(Rs.	Crore)
------	--------

				,
Particulars	МТЛ	Estimated	Approved by the Commission	Truing Up
Opening Capital Cost as on Station COD	А		3058.37	3058.37
Less: Undischarged liabilities included in above	В		1.02	
Opening Capital Cost excluding undischarged liabilities	c = a <b>-</b> b		3058. <mark>37</mark>	3058.37
Additional capitalization during the year	D			1
Less: Undischarged liabilities included in additional capitalization	E		$\langle \rangle \rangle$	M
Add: Liabilities discharged during the year	5.817		IA	S
Closing Capital Cost	g = d - e + f	11-11	3058.37	<mark>3058</mark> .37
Average Capital Cost	h = (c + g)/2	111	3058.37	3058.37
Freehold land	HEATHE AND		96.75	96.75
Weighted Average Rate of depreciation	J	2	4.77%	4.77%
Remaining depreciable value	k = 90% x (h - i)	10-10-	10	2665.46
Depreciation (annualised)	l = Min (j x h, k)	202.59	141.27	145.80
Depreciation (for the period)	m = l x (349/365)	202.59	141.27	139.41
Cumulative depreciation at the beginning of the period	N	ant	SP/	-
Cumulative depreciation at the end of the period	0 = M + N	202.59	141.27	139.41

4.6 In reply to the deficiencies raised by the Commission, GVK submitted the fixed asset register prepared for the assets created on 16.04.2016 and as on 31.03.2017 for which depreciation has been claimed during FY 2016-17. GVK further submitted that the fixed asset register has been prepared

as per the actual cost incurred by GVK for each asset.

#### **PSPCL's Submission**

- 4.7 PSPCL vide Memo No. 5516 dated 23.06.2020 submitted that since the base figures for calculation of depreciation based on the capital cost of the project, the same has been considered as the amounts claimed by GVK and not the amounts approved by the Commission in its Final Tariff Order, the same are inadmissible.
- 4.8 PSPCL vide its submission dated 28.07.2020 and 10.8.2020 submitted that the Commission in its Order dated 17.01.2020 had considered a weighted average rate of 4.77% based upon the depreciation of Rs. 201.37 Crore for FY 2016-17 depicted by GVK in its annual audited accounts and had determined sum of Rs.141.27 Crore as permissible depreciation for GVK's project. However, GVK in the present Petition has submitted a sum of Rs.139.41 Crore as depreciation of its gross fixed assets, excluding freehold land as described under Format 12 annexed to the present petition. Since, the determined deprecation amount of the gross fixed assets forms the basis of calculation of interest on loan capital which is also the provision under Regulation 26(5) of the PSERC Tariff Regulations, 2005, the Commission may subject the same to strict scrutiny and prudence check before allowing true-up of the same.
- 4.9 PSPCL further submitted that it has observed certain abnormal expenditures while checking the said fixed asset register furnished by GVK and requested the Commission for prudence check of the same.

#### **Commission's Analysis**

- 4.10 The Commission has in Petition no.54 of 2017 already carried out a detailed prudence check of the Capital Cost of the project.Depreciation is allowable on the approved Capital cost.
- 4.11 The Commission had provisionally approved Depreciation of Rs. 141.27 Crore for GVK for the period 17.04.2016 to 31.03.2017 in the order dated 17.01.2020.
- 4.12 Depreciation is to be determined as per provisions of Regulations 27 of PSERC (Terms and Conditions for determination of Tariff), Regulation 2005 which is as under:

#### 27. DEPRECIATION

(1) For the purpose of tariff, depreciation shall be computed in the following manner:

(a) The value base for the purpose of depreciation shall be the capital cost of the assets admitted by the Commission: Provided that land is not a depreciable asset and its cost shall be excluded from the capital cost while computing depreciation: Provided further that depreciation has been calculated after deduction of consumer contributions, capital subsidies/Government grants.

(b) The historical cost of the asset shall include additional capitalisation.

(c) The historical cost shall include foreign currency funding converted to equivalent rupees at the exchange rate prevalent on the date when foreign currency was actually availed but not later than the date of commercial operation.

(d) Depreciation for generation and transmission assets shall be calculated annually as per straight-line method over the useful life of the asset at the rate of depreciation specified by the Central Electricity Regulatory Commission from time to time. Provided that the total depreciation during the life of the asset shall not exceed 90% of the original cost.

(e) Depreciation for distribution assets and other assets not specified by CERC shall be at the rates notified by the Commission. Provided that the total depreciation during the life of the asset shall not exceed 90% of the original cost.

(2) The generating company and the licensees shall make all efforts for aligning the tenure of long term debt with permissible rate of depreciation to reduce front loading of tariffs. In case of inadequacy of cash for repayment of debt, only in extreme cases, the Commission may allow Advance Against Depreciation (AAD) in addition to the allowable Depreciation in the following manner:

AAD = Loan repayment amount as per the schedule of repayment subject to a ceiling of 1/10th of loan amount as per Regulation 24 minus depreciation as per schedule Provided that Advance Against Depreciation shall be permitted only if the cumulative repayment up to a particular year exceeds the cumulative depreciation up to that year, Provided further that Advance Against Depreciation in a year shall be restricted to the extent of difference between cumulative repayment and cumulative depreciation up to that year.

(3) On repayment of entire loan, the remaining depreciable value shall be spread over the balance useful life of the asset.

(4) Depreciation shall be chargeable from the first year of operation. In case of operation of the asset for part of the year, depreciation shall be charged on pro-rata basis.

4.13 As per the Annual Audited Accounts of GVK for FY 2016-17, depreciation of Rs. 201.37 Crore has been shown on Gross Fixed Asset of Rs. 4224.02 Crore (excluding land value). There has been no addition of assets from 17.04.2016 to 31.03.2017. As per table no 6 GVK has claimed weighted average rate of depreciation as 4.77% which is considered by the Commission for computing the depreciation for the period from 17.04.2016 to 31.03.2017.

4.14 The Commission has determined Gross Fixed Assets (GFA) of Rs. 3058.37 Crore vide its Order dated 17.01.2020. The Commission allows the depreciation for the period from 17.04.2016 to 31.03.2017 as under:

Table No.8: Depreciation allowed by the Commission forthe period 17.04.2016 to 31.03.2017.

		(Rs. Crore)
Sr. No.	Particulars	Amount
1.	Opening value of GFA as on 17.04.2016	3058.37
2.	Value of Land	96.75
3.	Opening value of GFA (net of Land)	2961.62
4.	Addition /disposal during the year	0.00
5.	Closing value of GFA	2961.62
6.	Average value of GFA	2961.62
7.	Weighted Average Rate of Depreciation	4.77%
8.	Depreciation for FY 2016-17 (6 x 7)	141.27
9.	Depreciation (141.27 x 349/365)	135.08

Accordingly, the Commission allows the depreciation of Rs. 135.08 Crore for the period 17.04.2016 to 31.03.2017.

#### 5.0 Return on equity

## **GVK's Submission**

5.1 GVK had originally computed the Return on Equity for 2016-17 as Rs.171.25 Crore in accordance with Regulation 24 of the CERC Tariff Regulations, 2014, as under:

## "24. Return on Equity:

(1) Return on equity shall be computed in rupee terms, on the equity base determined in accordance with regulation 19.

(2) Return on equity shall be computed at the base rate of 15.50% for thermal generating stations, transmission system including communication system and run of the river hydro generating station, and at the base rate of 16.50% for the storage type hydro generating stations including pumped storage hydro generating stations and run of river generating station with pondage..."

- 5.2 GVK has submitted that for computation of the base equity, the opening debt-equity ratio as on Station COD has been considered at 73:27 as stated earlier in this petition, and the opening capital cost as on Station COD has been considered net of undischarged liabilities. ROE thus derived has been prorated based on Station COD.
- 5.3 GVK in its revised submission on 21.08.2020 has submitted that it has funded the claimed capital cost of Rs 4267.38 Crore in the ratio of 73.8 :26.2 and in terms of Regulation 24 of PSERC Tariff Regulations 2005,the total hard cost of the project ought to be considered as per actual debt equity ratio. However, in Order dated 17.01.2020,this Commission subtracted the entire equity infused by GVK from the hard cost(~50%) and considered the balance amount as outstanding loan amounts of GVK.
- 5.4 GVK in its revised submission (received on 21.07.2020) has calculated Return on Equity @ 15.50% as per Regulation 25 of PSERC Tariff Regulations, 2005 on the paid up equity capital determined in terms of Order dated 17.01.2020. GVK has submitted that for the computation of the base equity, the opening debt-equity ratio as on Station COD has been considered at 70:30 as stated earlier in this petition, and the opening capital cost as on Station COD has been considered. Return on Equity thus derived has been pro-rated based on Station COD. Computation of ROE is shown in the following table:

# Table No. 9: Return on Equity submitted by GVK for FY 2016-17(Rs. Crore)

Particulars		Estimated	Approved by the Commission	Truing Up
Opening Capital Cost as on Station COD	(a)		3058.37	3058.37
Equity percentage as on COD	(b)		30%	30%
Normative Opening Equity considered for ROE computation (= a x b)	(c)	HEGI	917.51	917.51
Equity addition due to additional capitalization during the year	(d)		14	
Normative Closing Equity (= c + d)	(e)		917.51	917.51
Average Equity (Average of 'c' and 'e')	(f)		917.51	917.51
Normative Rate of ROE as per PSERC Tariff Regulations 2005	(g)		15.50%	15.50%
Return on Equity - Annualized (= f x g)	(h)	1	142.21	142.21
Return on Equity - Pro-rata (= h x 349 / 365) )(w.e.f 16.04.2016)	(i)	185.94	13 <mark>5.98</mark>	135.98

## **PSPCL's Submission**

- 5.5 PSPCL vide Memo No. 5516 dated 23.06.2020 submitted that since the base figures for calculation of Return on Equity, being the capital cost of the project, the equity infused by GVK has been considered as the amounts claimed by GVK and not the amounts approved by this the Commission in its Final Tariff Order, the same are inadmissible.
- 5.6 PSPCL vide its submission dated 28.07.2020 and 10.08.2020 submitted that based on Regulation 24 of the PSERC Tariff Regulation, 2005, GVK has submitted that since the actual equity invested by it is in excess of 30% of the capital cost, the equity component for the purpose of tariff is considered as 30% of the approved capital cost of Rs. 3058.57 Crore. As per Regulation 24, in case equity invested by a generating station is in excess of 30% of the completed capital cost, the said equity in excess of 30% is to be considered as normative loan. In its

Order dated 17.01.2020, the Commission has observed that since the actual equity infused by GVK in the project was Rs.1118.06 Crore and 30% of the completed capital cost of Rs. 3058.37 Crore comes to Rs. 917.51 Crore, the remaining Rs. 200.55 Crore may be deemed as normative loan. As such, the total loan component for GVK's project as approved by the Commission is Rs.1940.31 Crore (Rs. 3058.37 Crore - Rs. 1118.06 Crore) and in addition thereto Rs. 200.55 Crore has been approved as normative loan. The debt-equity ratio for GVK's project for FY 2016-17 is liable to be viewed accordingly by the Commission.

5.7 PSPCL further submitted that as per the submissions of GVK,
there is no element of any true-up under the present head. As
such, no response to the same is required from the PSPCL.

#### **Commission's Analysis**

- 5.8 The Commission had provisionally approved Return on equity of Rs.135.98 Crore for GVK for the period 17.04.2016 to 31.03.2017 in the order dated 17.01.2020.
- 5.9 As per Regulation 24 of PSERC (Terms and Conditions for determination of Tariff), Regulation 2005 which is as under:

#### 24. DEBT-EQUITY RATIO

(1) For the purpose of determination of tariff, debt-equity ratio in case of a new project commencing after the date of notification of these Regulations shall be 70:30. Where equity employed is more than 30%, the amount of equity for the purpose of tariff shall be limited to 30% and the balance amount shall be considered as loan. Where actual equity employed is less than 30%, the actual debt and equity shall be considered for determination of tariff.

Provided that the Commission may, in appropriate cases, consider equity higher than 30% for the purpose of

determination of tariff, where the generating company or the licensee is able to establish to the satisfaction of the Commission that deployment of equity more than 30% is in the interest of the general public;

(2) In case of existing and ongoing projects, the actual debtequity ratio shall be considered for determination of tariff. However, any expansion shall be governed as per clause (1) above.

(3) The debt and equity amounts arrived at in accordance with clause (1) and (2) shall be used for all purposes including for determining interest on loan, return on equity, Advance against Depreciation and Foreign Exchange Rate Variation.

- 5.10 The Regulations state that Debt-equity ratio in case of new projects shall be 70:30 for the purpose of determination of tariff. Where equity employed is more than 30%, the amount of equity for the purpose of tariff shall be limited to 30% and the balance amount shall be considered as loan. The Commission has considered the equity of Rs. 1118.06 Crore in the capital cost of Rs. 3058.37 Crore which is more than the 30% of the approved capital cost. Accordingly, equity works out to be Rs. 917.51 (30% of 3058.37) Crore and the balance of Rs. 200.55 (1118.06-917.51) Crore is considered as normative loan as per the above referred Regulation.
- 5.11 Return on equity shall be computed @15.5% as per Regulation 25 of PSERC (Terms and Conditions for determination of Tariff), Regulation 2005 on equity determined in accordance with Regulation 24. The relevant Sub Regulation 25 of PSERC (Terms and Conditions for determination of Tariff), Regulation 2005 is reproduced below:

(1) Return on Equity shall be computed @ 15.5% on the paid up

equity capital determined in accordance with Regulation 24.

5.12 Thus, Commission works out return on equity for FY 2016-17 as under:-

## Table No.10: Return on equity allowed by the Commission for the period 17.04.2016 to 31.03.2017.

(Rs. Crore)

Sr. No.	Particulars	Amount
1.	Capital Cost as on 17.04.2016	3058.37
2.	Equity as on 17.4.2016	1118.06
3.	Equity - 30% of Capital Cost	917.51
4.	Normative Loan (2-3)	200.55
5.	Addition during the year	0.00
6.	Closing Equity as on 31.03.2017	917.51
7.	Average Equity	917.51
8.	Rate of Return on Equity	15.50%
9.	Return on Equity for FY 2016-17 (7 x 8)	142.21
10.	Return on Equity (142.21x349/365)	135.98

Accordingly, the Commission allows the return on equity of Rs. 135.98 Crore for the period 17.04.2016 to 31.03.2017.

6.0 Interest & Finance charges

## GVK's Submission

6.1 GVK has computed Interest on Loan Capital for FY 2016-2017 as per Regulation 26 of CERC Tariff Regulations, 2014, as under:

#### "26. Interest on loan capital:

- (1) The loans arrived at in the manner indicated in regulation 19 shall be considered as gross normative loan for calculation of interest on loan.
- (2) The normative loan outstanding as on 1.4.2014 shall be worked out by deducting the cumulative repayment as admitted by the Commission up to 31.3.2014 from the gross normative loan.
- (3) The repayment for each of the year of the tariff period 2014-19 shall be deemed to be equal to the depreciation allowed for the corresponding

year/period. In case of decapitalization of assets, the repayment shall be adjusted by taking into account cumulative repayment on a pro rata basis and the adjustment should not exceed cumulative depreciation recovered upto the date of decapitalisation of such asset.

- (4) Notwithstanding any moratorium period availed by the generating company or the transmission licensee, as the case may be, the repayment of loan shall be considered from the first year of commercial operation of the project and shall be equal to the depreciation allowed for the year or part of the year.
- (5) The rate of interest shall be the weighted average rate of interest calculated on the basis of the actual loan portfolio after providing appropriate accounting adjustment for interest capitalized:

Provided that if there is no actual loan for a particular year but normative loan is still outstanding, the last available weighted average rate of interest shall be considered:

Provided further that if the generating station or the transmission system, as the case may be, does not have actual loan, then the weighted average rate of interest of the generating company or the transmission licensee as a whole shall be considered.

- (6) The interest on loan shall be calculated on the normative average loan of the year by applying the weighted average rate of interest.
- (7) The generating company or the transmission licensee, as the case may be, shall make every effort to re-finance the loan as long as it results in net savings on interest and in that event the costs associated with such re-financing shall be borne by the beneficiaries and the net savings shall be shared between the beneficiaries and the generating company or the transmission licensee, as the case may be, in the ratio of 2:1.
- (8) The changes to the terms and conditions of the loans shall be reflected from the date of such re-

financing.

- (9) In case of dispute, any of the parties may make an application in accordance with the Central Electricity Regulatory Commission (Conduct of Business) Regulations, 1999, as amended from time to time, statutory re-enactment thereof including for settlement of the dispute: Provided that the beneficiaries or the long transmission term customers /DICs shall not withhold any payment on account of the interest claimed by the generating company or the transmission licensee during the pendency of any dispute arising out of re-financing of loan.
- 6.2 In terms of the foregoing, the interest on loan capital for FY 2016-17 has been computed as follows:
  - (a) Gross normative opening loan as on Station COD has been computed as the difference between the Opening capital cost as on Station COD (excluding undischarged liabilities) and the Normative opening equity considered for ROE computation.
  - (b) Loan repayment has been considered at normative basis being equal to depreciation.
  - (c) Actual Weighted Average Interest rate has been computed as ratio of the actual interest paid to the actual average loan as per annual accounts for FY 2016-17.
- 6.3 GVK in its petition has stated that the weighted average rate of interest has been calculated on the basis of average balance of actual individual loans at 13.96% (Including penal interest). Based on the same GVK had worked out interest on normative loan (pro-rata) of Rs. 403.92 Crore on average normative loan of Rs. 3017.01 Crore.
- 6.4 GVK in its revised submission received on 21.07.2020 has computed Interest on Ioan capital as per Regulation 26 of PSERC Regulations, 2005 and has submitted that the weighted

average rate of interest has been calculated on the basis of average balance of actual individual loans at 13.22% (Including penal interest).

6.5 GVK in reply to the deficiencies, submitted the loan wise details of actual term loans availed from the commercial Banks/ Financial institutions, depicting opening balance of loan as on 17.04.2016, addition during the year, repayment of loan, closing balance as on 31.03.2017, interest rate, interest due and paid there- against on long term loans and working capital loans.

Accordingly, the revised computation of interest on loan for FY 2016-17 is as under:

Table No. 11: Interest on Loan submitted by GVK for F	Y 2016-17
	(Rs. Crore)

Sr. No	Particulars	Estimated	Approved by the Commission	True-up
1.	Normative Loan – Opening	10	2140.86	2140.86
2.	Addition of loan due to Additional Capitalization during the period from COD to year end			*/-
3.	Less: Repayment of Loan during the Period (Normative; equal to Depreciation claimed for the year)	F	141.27	139.41
4.	Net Closing loan		1999.59	2001.45
5.	Average Normative Loan		2070.22	2071.15
6.	Actual Weighted Average Rate of Interest on Loan	2	13.25%	13.22%
7.	Interest on Normative Loan (Annualized)	fam	274.30	273.82
8.	Interest on Normative Loan (pro rata)	404.72	225.51	261.81

6.6 GVK in its revised submission dated 21.08.2020 has submitted that in the petition no. 54 of 2017 it had originally claimed interest of Rs 451.91 Crore on actual average loan of Rs 3307.63 Crore as per Annual Audited Accounts. However, this Commission in Order dated 17.01.2020 has worked out average long term loan as Rs. 2070.22 Crore based on normative parameters in terms of PSERC Tariff Regulations 2005.GVK further submitted that once normative approach has been adopted by the Commission there cannot be any further adjustments on the basis of actual expenses on account of it being less, the said position of law has been upheld by the Hon'ble Tribunal in Haryana Power Generation Co. Ltd v. Haryana Electricity Regulatory Commission judgement dated 31.07.2009 in Appeal no.42 & 43 of 2008.

6.7 GVK prayed that in the present case also, this Commission may be allow interest on normative loan as claimed (Rs.261.81 Crore) in terms of the PSERC Tariff Regulations 2005.

#### **PSPCL's Submission**

- 6.8 PSPCL vide Memo No. 5516 dated 23.06.2020 submitted that since the base figures for calculation of Interest on Ioan, being the capital cost of the project, the term Ioans for the project has been considered as the amounts claimed by GVK and not the amounts approved by this the Commission in its Final Tariff Order, the same are inadmissible.
- 6.9 PSPCL vide its submission dated 28.07.2020 and 10.08.2020 stated that GVK under the present head has submitted actual figures of interest on loan to be Rs. 273.82 Crore as against Rs. 274.30 Crore allowed by the Commission in its order dated 17.01.2020. However, GVK has also submitted an amount of Rs. 261.81 Crore as 'Interest on normative loan (Pro rata)' as against Rs. 225.51 Crore as approved by this Commission. As regards this difference of Rs. 36.3 Crore the Commission in its

Order dated 17.01.2020 has held as under:

"As per the Annual Audited Accounts of GVK the average loans for FY 2016-17 have been worked out to Rs.3307.63 Crore and the interest on this has been depicted as Rs. 451.91 Crore. However, GVK has a liability of Rs. 226.42 Crore towards interest accrued. The interest actually paid on long term loans is Rs. 225.49 (451.91-226.42) Crore.

Accordingly, the Commission considers the interest of Rs. 225.49 Crore which has been Actually paid by GVK during this year. The balance amount of interest i.e. Rs. 36.79 (262.28-225.49) Crore will be considered in the year in which year it will be paid by GVK."

Thus, the Commission held that this sum of Rs. 36.79 Crore (inclusive of finance charges) would be considered in the year in which the same is actually paid by GVK. However, in the present Petition, GVK has not provided any details whether the said amount has actually been paid or not in FY 2016-17. In the absence of such information, GVK's true-up claim under this head is inadmissible to that extent.

#### **Commission's Analysis**

- 6.10 The Commission had provisionally approved Interest and finance charges of Rs. 225.51 Crore for GVK for the period 17.04.2016 to 31.03.2017 in the order dated 17.01.2020.
- 6.11 Interest on long term loans is to be allowed as per Regulation26 of PSERC (Terms and Conditions for determination of Tariff),Regulation 2005, which is as under:

26. INTEREST AND FINANCE CHARGES ON LOAN

(1) For Existing Loan Capital, Interest & finance charges shall be computed on the outstanding Loans, duly taking into account the rate of interest & schedule of repayment as per the Terms & Conditions of relevant agreements. The rate of interest shall be the actual rate of interest paid/payable by the Licensee(s) or the State Bank of India Advance rate as on April, 1 of the relevant year, wherever is lower.

(2) For New investments, Interest & finance charges shall be computed on the loans, duly taking into account the rate of interest & schedule of repayment as per the Terms & Conditions of relevant agreements. The rate of interest shall be the actual rate of interest paid/payable by the Licensee(s) or the State Bank of India Advance rate as on April, 1 of the relevant year, wherever is lower.

(3) The interest rate on the amount of equity above 30% treated as loan shall be the weighted average rate of interest on loan capital of the generating company / licensee

Provided that interest and finance charges of renegotiated loan agreements shall not be considered, if they result in higher charges. Provided further that interest and finance charges on works in progress shall be excluded and shall be considered as part of the capital cost.

(4) Interest charges on security deposits, if any, made by the consumers with a generating company/licensee, shall be considered at the rate specified by the Commission from time to time.

(5) In case any moratorium period is availed of, depreciation provided for in the tariff during the years of moratorium shall be treated as repayment during those years and interest on loan capital shall be calculated accordingly.

(6) The Commission shall allow obligatory taxes on interest, commitment charges, finance charges (including guarantee fee payable to the Govt.) and any exchange

rate difference arising from foreign currency borrowings, as finance cost.

(7) Any saving in costs on account of subsequent restructuring of debt shall be shared between the consumers and the generating company / licensee in such ratio as may be decided by the Commission.

- 6.12 The Commission in its order dated 17.01.2020 has determined capital cost of the GVK's project as Rs. 3058.37 Crore and equity as Rs. 1118.06 Crore. Long term loans of Rs.1940.31 (3058.37–1118.06) Crore have been considered. The Commission has further considered additional loan of Rs. 200.55 Crore (equity in excess of 30%) as part of normative long term loans as per Regulation 24 of PSERC (Terms and Conditions for determination of Tariff), Regulation 2005 for the purpose of allowing interest. As such, the long term loans allowed for interest purpose workout to Rs. 2140.86 (1940.31+200.55) Crore. The depreciation for FY 2016-17 which has been worked out as Rs. 141.27 Crore in this Order is considered as re-payment of the long term loans as per Regulation 26(5) considering that there would be moratorium period in the 1st year of operation.
- 6.13 In the Annual Audited Accounts of GVK for FY 2016-17 interest charges were shown as Rs. 451.91 Crore. But in the Annual Audited Accounts of FY 2017-18, the previous year figures of interest charges for FY 2016-17 has been rearranged/regrouped as Rs. 448.37 Crore and Rs 3.54 (451.91 448.37) Crore has been shown as Other Finance Charges.
- 6.14 As per the Regulation, 26 (2) of PSERC (Terms and Conditions for determination of Tariff), Regulation 2005, the rate of interest

is to be considered at the actual rate of interest paid /payable by the licensee or the State bank of India advance rate as on April, 01 on the relevant year, whichever is lower. The State Bank of India Advance Rate (SBAR) as on 01.04.2016 was 14.05%. GVK in its revised submission dated 21.07.2020 has claimed weighted average rate of interest of 13.22% (including penal interest). During hearing on 29.07.2020, GVK was asked to submit the rate of penal interest included in 13.22%.GVK stated that weighted average rate of interest is only 13.22% and is exclusive of penal interest. The Commission considers the weighted average rate of interest of 13.22% for long term loans as claimed by the GVK in the revised submission dated 21.07.2020. Therefore, the Commission determines the interest on long term loans as under:

 
 Table No.12: Interest on Long Term Loans determined by the Commission for the period 17.04.2016 to 31.03.2017

The set of the state of the set o		(Rs. Crore)	
Sr. No.	Particulars	Amount	
1.	Opening Long term Loans as on 17.04.2016	1940.31	
2.	Add: Normative Loan	200.55	
3.	Opening Long Term Loans	2140.86	
4.	Addition during the year	0.00	
5.	Re-payment of Loans (equal to depreciation)	141.27	
6.	Closing Long Term Loans	1999.59	
7.	Average Long Term Loans	2070.22	
8.	Rate of Interest	13.22%	
9.	Interest (7 x 8) for FY 2016-17	273.68	
10.	Interest (273.68 x 349/365)	261.69	

6.15 As per the Annual Audited Accounts of GVK for FY 2016-17 interest has been worked out to Rs. 448.37 Crore but GVK has a liability of Rs. 226.42 Crore towards interest accrued. The interest actually paid on long term loans is only Rs. 221.95 (448.37-226.42) Crore. Accordingly, the Commission approves the interest of Rs. 221.95 Crore which has been actually paid by GVK during FY 2016-17. The balance amount of interest i.e. Rs. 39.74 (261.69 – 221.95) Crore will be considered in the year in which it will be paid by GVK.

#### 6.16 Finance Charges:

GVK has not claimed any finance charges for FY 2016-17 but Rs. 3.54 Crore has been shown as other finance charges in the Annual Audited Accounts for FY 2016-17 on loan amount of Rs. 3366.49 Crore. Finance charges works out as Rs. 2.05 Crore for FY 2016-17 on a loan of Rs. 1940.31 Crore and Rs 1.96 (2.05x349/365) Crore is allowed proportionately for the period 17.04.2016 to 31.03.2017. Thus, the total interest and finance charges for FY 2016-17 work out to Rs. 224.00(221.95+2.05) Crore.

Accordingly, the Commission allows interest and finance charges of Rs. 223.91(221.95+1.96) Crore for the period 17.04.2016 to 31.03.2017.

## 7.0 Interest on Working Capital GVK's Submission

- 7.1 GVK has computed Interest on Working Capital in terms of Regulation 28 of CERC Tariff Regulations, 2014. GVK submitted that in terms of Regulation 28 of the CERC Tariff Regulations, 2014 the Working Capital Interest rate applicable to GVK is 12.80% based on which Interest on Working Capital works out to Rs. 79.63 Crore.
- 7.2 In its revised submission dated 21.07.2020 and 21.08.2020GVK has computed Working Capital and Interest on WorkingCapital as per Regulation 30(3) of PSERC Tariff Regulations

2005. GVK has submitted computation of receivables for working capital requirement as under:

Table No.13: Computation of receivables for working<br/>capital requirement by GVK

Sr.No	Particular	Amount
1	Fuel Cost	66.45
2	O&M Expenses	30.78
3	Depreciation	139.41
4	Return on Equity	135.98
5	Interest of finance charges/Loan	261.81
6	Interest on working capital loan	15.50
7	Total (Receivables)	649.93

7.3 GVK in its revised submission on 21.08.2020 has submitted the computation of Interest on Working Capital for true-up of FY 2016-17 as shown in the following table:

# Table No.14: Interest on Working Capital claimed byGVK for FY 2016-17

## (Rs Crore)

Sr. No.	Particulars	Approved by the Commission	Amount
1	Maintenance Spares @ 15% of O&M Expenses	4.56	4.62
2	Fuel Cost for two months	8.68	11.08
3	O&M Expenses for one month	2.53	2.57
4	Receivable for two months	99.58	108.24
5	Total	115.35	126.50
6	Rate of Interest	10.62%	12.25%
7	Interest on Working Capital	12.25	15.50

#### **PSPCL's Submission**

7.4 PSPCL vide its submission dated 28.07.2020 and 10.08.2020 submitted that as per Regulation 30 of the PSERC Tariff Regulations, 2005 working capital for a generating company and interest on working capital is required to be calculated as under:

*"30. WORKING CAPITAL AND INTEREST RATE ON WORKING CAPITAL* 

(3) working capital for a generating company shall be the sum of the following:

(i) Fuel Cost for two months

(ii) operation and Maintenance expenses for one month

(iii) Receivables equivalent to two months

(iv) Maintenance spares @15% of O&M expenses,

In its order dated 17.01.2020, the Commission has observed that as per the Annual Audited Accounts of GVK for FY 2016-17, the fuel cost is Rs. 66.45 Crore. However, as per the information available with the Commission in Petition No.68/2017 filed by GVK regarding billing disputes, the fuel cost for the period 16.04.2016 to 31.03.2017 given by PSPCL to GVK is Rs. 52.07 Crore. As such, fuel cost of Rs. 52.07 Crore have been considered by the Commission for determination of receivables. As regards the O&M Expenses, of the Commission has approved a pro-rata sum of Rs. 30.41 Crore for the period 17.4.2016 to 31.03.2017, depreciation was considered as Rs.141.27 Crore, return on equity at Rs.135.98 Crore and interest and finance charge at Rs.225.51 Crore. PSPCL submitted that GVK's claim in the true up petition is subject to the Commission allowing the said true-up figures after consideration of the objections of the Respondent in the present Reply and after due prudence check. As regards the cost of fuel for 2 months, GVK has claimed a revision of Rs. 2.40 Crore without providing any justification thereto. As per the Annual Audited Accounts for FY 2016-17 and GVK's own admission during the proceedings in Petition No.68/2017, the annual fuel cost of GVK is Rs. 52.07 Crore, based on which fuel cost for 2 months works out to Rs. 8.68 Crore. Thus, GVK cannot be

permitted to deviate from the said settled and accepted fuel cost. It may also be mentioned here that billing of the energy charges (which includes in itself the fuel costs) for **GVK's** project is being done on the basis of energy scheduled by PSPCL. The Energy charge rate is calculated on the basis of normative parameters instead of actual and in accordance with directions of the Commission in its the Order dated 6.03.2019.

## **Commission's Analysis**

- 7.5 The Commission had provisionally approved Interest on working capital of Rs.12.25 Crore for GVK for the period 17.04.2016 to 31.03.2017 in the order dated 17.01.2020.
- 7.6 As per the Annual Audited Accounts of GVK for FY 2016-17 the fuel cost was Rs 66.45 Crore. The energy charges paid by PSPCL during FY 2016-17 were originally Rs. 52.07 Crore which were later revised in terms of the Commission's Order dated 06.03.2019 to Rs. 55.12 Crore including Rs. 3.05 Crore as arrears. Thus, fuel cost of Rs. 55.12 Crore has been considered for determination of receivables for FY 2016-17.
- 7.7 The Working Capital and Interest rate on Working capital has been determined as per Regulation 30 of PSERC (Terms and Conditions for determination of Tariff), Regulation 2005, which is as under:

30. WORKING CAPITAL AND INTEREST RATE ON WORKING CAPITAL

(3) Working capital for a generating company shall be the sum of the following:(i) Fuel Cost for two months

- (ii) Operation and Maintenance expenses for one month
- (iii) Receivables equivalent to two months

(iv) Maintenance spares @15% of O&M expenses

(5) The rate of interest on working capital shall be equal to the actual rate of interest paid/ payable on loans by the licensee (s) or the State Bank of India Advance Rate as on April 1 of the relevant year, whichever is lower. The interest on working capital shall be payable on normative basis notwithstanding that the licensee (s) has not taken working capital loan from any outside agency or has exceeded the working capital loan amount worked out on the normative figures.

- 7.8 The Commission considers the weighted average rate of interest of 12.25% for working capital loans as claimed by the GVK in the revised submission dated 21.07.2020. The State Bank of India advance rate as on 01.04.2016 was 14.05%.
- 7.9 The details of Working Capital requirement and allowable interest thereon is depicted in the following table:
- Table No.15: Working Capital and interest thereon allowed by the Commission for the period 17.04.2016 to 31.03.2017.

	E W R LEICHE Anth	RS. Crore)
Sr. No.	Particulars	Amount
1.	Maintenance spares @ 15% of O&M (15% of 30.41)	4.56
2.	Fuel Cost for 2 months (55.12*60/349)	9.48
3.	O & M expenses for 1 month (30.41*30/349)	2.61
4.	Receivables for 2 months	102.24
5.	Total Working capital	118.89
6.	Weighted Average Rate of Interest	12.25%
7.	Interest on working capital for 17.04.2016 to 31.03.2017	7 14.56

Accordingly, the Commission allows interest on Working Capital of Rs. 14.56 Crore for the period 17.04.2016 to 31.03.2017.

## 8.0 Tax on Income: GVK's Submission

8.1 GVK submitted that Regulation 32 of the PSERC Tariff Regulations provides that taxes on the income of a generating company shall be computed as an expense and shall be recovered from consumers.

8.2 GVK submitted that tax on income liable to be paid for FY 2016-2017 is Nil.

250

## **PSPCL's Submission:**

8.3 PSPCL vide its submission dated 10.08.2020 submitted that the claim of GVK as regards Tax on Income may be allowed by the Commission in accordance with applicable Regulations framed by the Commission after due prudence check.

## **Commission's Analysis:**

8.4 Tax on income is allowable as per Regulation 32 of PSERC (Terms and Condition for Determination of Tariff Regulations, 2005) (amended from time to time). It is stipulated that tax on income shall be limited to tax on return on equity allowed, as under:

## "32. TAX ON INCOME

(1) Obligatory taxes, if any, on the income of the generating company or the licensee from its core/licensed business shall be computed as an expense and shall be recovered from the customers/consumers.

Provided that tax on any income other than the core / licensed business shall not constitute a pass through component in tariff and tax on such other income shall be payable by the generating company or the licensee.

(2) Tax on income, if actually liable to be paid, shall be limited to tax on return on equity allowed, excluding incentives.

(3) The Tax on income shall be considered at income tax rate including surcharge, cess, etc as applicable during the relevant year in accordance with the provisions of Income Tax Act, 1961 duly amended from time to time.

(4) The benefits of tax holiday and the credit for carrying forward losses applicable as per the provisions of the Income Tax Act, 1961 shall be fully passed on to the customers/consumers."

The Commission allows income tax paid as Nil for the period 17.04.2016 to 31.03.2017 based on the Annual Audited Accounts .

# 9.0 Non-Tariff Income GVK's Submission

- 9.1 GVK submitted that Regulation 37 of the PSERC Tariff Regulations provides components of income, which shall be treated as non-tariff income. In terms thereof the following components of income shall be treated as non-tariff income:
- 9.2 The non-tariff income of GVK for FY 2016-2017 is as under:

## Table No. 16: Non-tariff income claimed by GVK. (Rs. Crore)

Sr. No.	Particulars	Amount
1.	Interest on FDR	0.18
2.	Sale of Fly Ash	0.32
3.	Total	0.50

## **PSPCL's Submission**

9.3 PSPCL vide its submission dated 10.08.2020 submitted that the claim of GVK as regards Non-Tariff Income may be allowed by the Commission in accordance with applicable Regulations framed by the Commission after due prudence check.

## **Commission's Analysis**

- 9.4 The Non-Tariff Income has been determined as per Regulation-34 of PSERC (Terms and Condition for Determination of Tariff Regulations, 2005) (amended from time to time).
- 9.5 The Commission notes that Audited Annual Accounts of GVK for FY 2016-17 has shown "Other income" which includes income from bank deposits and fly ash of Rs 0.18 Crore and Rs. 0.32 Crore respectively. As per para 20.3.4 of the Commission's Order dated 17.01.2020 in Petition no. 54 of
2017, GVK had diverted the funds meant for capital expenditure out of the loan taken from financial institutions by investment in Mutual Funds. The Commission has neither considered interest paid on loans used for such other investments nor income earned from such other business/ investment as part of the capital expenditure. Accordingly, the Commissions determines Non-Tariff Income as Rs. 0.32 Crore (sale of fly ash) for FY2016-17 and allows Non-tariff income as Rs. 0.32 Crore.

# Accordingly, the Commission approves Non-Tariff Income as Rs. 0.32 Crore for the period 17.04.2016 to 31.03.2017.

#### 10.0 Capacity charges for FY 2016-17

- 10.1 The Commission had provisionally approved Capacity charges of Rs. 545.42 Crore for GVK for the period 17.04.2016 to 31.03.2017 in the order dated 17.01.2020.
- 10.2 The Annual Fixed Charges (AFC) for the period 17.04.2016 to 31.03.2017, as projected by GVK in the Petition, revised in its additional submission received on 21.07.2020 & 21.08.2020 and approved by the Commission is summarized in the following table:-

## Table No.17: Annual fixed charges approved by the Commissionfor the period 17.04.2016 to 31.03.2017

					(RS Crore)
Sr. No.	Particulars	Allowed by Order dated 17.1.2020	Claimed by GVK in original Petition	Claimed by GVK in revised submission received on 21.08.2020	Approved by the Commission
1.	O&M Expenses	30.41	139.81	30.78	30.41
2.	Depreciation	141.27	196.24	139.41	135.08
3.	Interest charges	225.51	403.92	261.81	223.91
4.	Return on Equity	135.98	171.25	135.98	135.98
5.	Interest on Working Capital	12.25	79.63	15.50	14.56
6.	Income Tax	0.00	0.00	0.00	0.00
7.	Total Expenses	545.42	990.85	583.48	539.94
8.	Less: Non-Tariff Income	0.00	0.50	0.50	0.32
9.	Annual Fixed Cost	545.42	990.35	582.98	539.62

10.3 GVK shall be entitled for payment of capacity charges in accordance with Regulation 30 of the Central Electricity Regulatory Commission (Terms and Conditions of Tariff) Regulations, 2005 (as amended) where the same is not specified in the PSERC Tariff Regulations.

#### 11.0 Generation for FY 2016-17

#### **GVK's submission:**

11.1 GVK has submitted the generation data for FY 2016-17 in the petition and its submissions dated 12.03.2020 in response to the Commission's interim Order dated 07.02.2020 as hereunder:

Table No. 18:	Generation data submitted by GVK
	for FY 2016-17

Sr. No.	Description	April 2016	July 2016	August 2016	Total for FY 2016-17
1.	Gross Generation in MU	31.05	83.68	82.36	197.09
2.	Actual Generation in MU	26	79	87	192
3.	Plant Availability Factor (%)	13.17	22.89	22.53	4.71 (Cumulative)
4.	Declared Capacity in MU	31	84	82	197
5.	Scheduled Generation in MU	23.96	71.12	80.24	175.32

11.2 GVK vide its revised submissions received on 21.07.2020 submitted that the project is operating as part of the interconnected grid and abides by the Rules and Regulations framed by the CERC and the Commission to ensure grid safety and interconnected grid operations. However, GVK suffered loss of generation on account of backing down instructions received from Punjab State Load Despatch Centre (SLDC), even though the Project was available for generation. The revised Gross Generation (MU) and Scheduled Generation (MU), submitted by GVK is as under:

Sr. No.	No. Description		July 2016	August 2016	Total for FY 2016-17
1.	Gross Generation in MU	26	79	87	192
2.	Scheduled Generation in MU	23.96	71.12	80.24	175.32

#### Table No. 19: Generation data submitted by GVK for FY 2016-17.

11.3 GVK further submitted that Regulation 36 (A) of CERC Tariff Regulations provide the norms of operation and in turn provide that Normative Annual Plant Availability Factor (NAPAF) for thermal generating stations shall be 85% and 83% in case of shortage of coal and uncertainty of coal supply on sustained basis. While normative PAF is 83%, the PAF has been impacted on account of force majeure event being cancellation of the captive coal block. In terms of Article 12.3(ii) read with Article 12.7 (a) and (b) of the Amended and Restated PPA, GVK is excused from performance and is entitled to compensation on account of unavailability of coal. In terms of the foregoing, PAF ought to be considered corresponding to actual coal available. Actual cumulative PAF for FY 2016-17 was 4.707%.

GVK vide its submission dated 21.08.2020 in reply to PSPCL's averment that the Gross Generation stated by GVK is in fact the Declared Capacity of the project as per the State Energy Accounts (SEA) issued by SLDC submitted that for the purposes of payment of tariff, capacity charges are payable as per the declared capacity of GVK and the said charges are to be paid irrespective of scheduling/dispatch by PSPCL. Further in terms of Article 4.3.1 of the Amended and Restated PPA, PSPCL is bound to pay capacity charges based on capacity declared available by GVK.

#### **PSPCL's submission:**

11.4 PSPCL vide its submissions dated 27.05.2020, 23.06.2020, 28.07.2020 and 10.08.2020 submitted as under:

The payment of fixed charges is based on the quantum of energy declared available as certified by SLDC. The quantum must be certified by the SLDC, which is a statutory authority. The certification of SLDC is on ex-bus basis, and on this GVK is entitled to the fixed charges at tariff determined by the Commission. As per the provisions in the Regulations, the payment of monthly fixed charges is based on the AFC, the Cumulative Plant Availability Factor and Normative PAF. The cumulative PAF and Normative PAF must be certified by the SLDC which is a statutory authority.

Under the head 'Installed Capacity & Generation', GVK has submitted gross generation during FY 2016-17 as 197.09 MU which has been revised to 192 MU. However, this is not the Gross Generation but is rather the Declared Capacity (DC) of the project, which is as per the State Energy Accounts (SEA) issued by SLDC. The Commission was requested to rely upon month-wise DC of GVK's project on which the billing had been done in respect of capacity charges for FY 2016-17. The payment of energy charges to GVK is being done as per the month wise energy scheduled by PSPCL and the billing had been done as per State Energy Accounts issued by the SLDC for FY 2016-17.

#### **Commission's Analysis**

11.5 The issue raised by GVK on the force majeure due to non availability of coal has already been decided by the

Commission in its Order dated 01.02.2016 common to Petition no. 65 of 2013 & 33 of 2015, Order dated 21.05.2018 in Petition no. 45 of 2017 and Order dated 06.03.2019 in Petition no. 68 of 2017.

256

11.6 As checked up from the SLDC's website, in terms of the Final State Energy Accounts for FY 2016-17 containing the data on the declared capacity, plant availability factor and the scheduled energy for FY 2016-17, the details are as under:

Table No. 20: Declared Capacity and Scheduled Generation as per SLDC for FY 2016-17

Sr. No.	Description	April 2016	July 2016	August 2016	Total for FY 2016-17
1.	Declared Capacity in MU	31.059	83.686	82.360	197.105
2.	Scheduled Generation in MU	23.955	71.124	80.241	17 <mark>5.32</mark> 0

#### 12.0 Energy Charges for FY 2016-17

#### GVK's submission:

12.1 GVK has submitted the details of components of the Energy Charge in the petition as hereunder:

#### Table No. 21: Details of components of the Energy Charge submitted by GVK

Sr. No	Description	April 2016	July 2016	August 2016
1.	Aux Consumption (%)	8.02	8.46	8.57
2.	Station Heat Rate (kCal/kWh)	2715	2687	2521
3.	Fuel Oil Consumption (HDO/LDO) (KL)	159.31	429.34	422.57
4.	GCV of Coal (kCal/kg)	3686	4537	4006
5.	Landed Price of Coal (Rs./MT)	6211	5701	5429

12.2 Further, the fuel cost details as furnished by GVK in response to the information sought by the Commission vide its Order dated 07.02.2020 are as under:

Sr. No.	ltem	Unit	Actual
1.	Generation	MU	192
2.	Auxiliary Consumption	%	8.85
3.	Net Generation	MU	175
4.	Station Heat Rate	Kcal/kwh	2621
5.	Specific Oil Consumption	ml/kwh	3.21
6.	Calorific Value of Oil	Kcal/Litre	10410
7.	Calorific Value of Coal (Weighted Average)	Kcal/Kg	3944
8.	Overall Heat (1*4)	Gcal	503232
9.	Heat from Oil (1*5*6/1000)	Gcal	6416
10.	Heat From Coal (8-9)	Gcal	496816
11.	Total Oil Consumption	KL	615.40
12.	Transit Loss of Coal	%	5.4
13.	Total Coal Consumption excluding transit and other losses	Tonnes	1,11,084
14.	Quantity of imported / captive coal priced on FOR basis	Tonnes	2
15.	Quantity of coal not priced on FOR basis	Tonnes	1
16.	Total Quantity of Coal required (Including Transit Loss and other losses) - As per Books of Account	Tonnes	1,17,083
17.	Price of Oil	Rs/ KL	18050
18.	Price of Coal	Rs. / Tonne	5581
19.	Total Cost of Oil (As per Audited Financial)	Rs. Crores	-1.11
20.	Total Coal Cost (As per Audited Financial)	Rs. Crores	65.34
21.	Total Fuel Cost	Rs. Crores	66.45

Table No. 22: Details of fuel cost submitted by GVK for FY 2016-17

12.3 GVK vide its revised submissions received on 21.07.2020 and 21.08.2020 stated that since, GVK's Project was commissioned on 16.04.2016 and accordingly the CERC Tariff Regulations 2014 would be applicable to GVK's Project for determination of energy charges. GVK further submitted that in terms of the formula mentioned in Regulation 30(6)(a) of the CERC Tariff Regulations, 2014, energy charges are to be determined taking into account the weighted average landed cost of fuel. GVK further submitted details of components of the Energy Charge as hereunder:

### 258

#### i) Auxiliary Consumption:

The normative auxiliary energy consumption for Coal-based generating stations as per Regulation 36(E)(a) of the CERC Tariff Regulations, 2014 is 8.5% for a unit of 200 MW series. Further, for thermal generating stations with induced draft cooling towers, additional 0.5% is allowed. Thus, the normative auxiliary consumption has been considered at 9% for the computation of energy charge. On 21.08.2020, while reiterating its earlier submissions, GVK submitted that While PSPCL has agreed that for the purposes of billing, normative auxiliary consumption of 9% shall be considered, PSPCL is yet to make payments towards the same. It is submitted that from FY 2016-17 to January 2020, PSPCL has not paid any amounts towards auxiliary consumption claimed by GVK.

GVK further submitted that the Declared Capacity data on exbus basis considered by PSPCL as energy declared during bill period ought to be grossed up by 9% in order to arrive at the total available capacity at the generator terminal so as to include auxiliary consumption in the total amounts payable to GVK. Auxiliary consumption is a legitimate head which ought to be included in the available energy for payment of capacity charges. In this context, it is pertinent to note that in terms of Regulation 3(44) and 3(45) read with 30(3) and 30 (6) of CERC Tariff Regulations 2014, 'Normative Annual Plant Availability' and 'Plant Availability Factor' are to be calculated by grossing up the Declared Capacity with the normative auxiliary energy consumption. It is submitted that capacity of the Project utilized for Auxiliary Consumption is critical for operations of the Project and GVK ought to be paid fixed charges for the same so as to continue to operate the Project. It is pertinent to note that for the month of January2020, PSPCL has paid GVK auxiliary consumption at 9% as part of the fixed charges. Accordingly, PSPCL ought to pay fixed charges taking into account auxiliary consumption for FY 2016-17.

The Commission in Order dated 06.03.2019 in Petition No. 68 of 2017 has held that once the capital cost of the Project is determined in petition No. 54 of 2017, PAFM/PAFY would be worked out to determine capacity charges payable by it. However, in Order dated 17.01.2020 in Petition No. 54 of 2017, the Commission did not return any findings on auxiliary consumption. GVK has filed Appeal No. 189 of 2019 against Order dated 06.03.2019 and Appeal No. 41 of 2020 against Order dated 17.01.2020 and the same are pending. GVK reserves its rights in this regard

ii) Design Heat Rate:

The Design Heat Rate for the Project under the EPC contract is 2221 Kcal/kWh. Regulation 36(C)(b) of the CERC Tariff Regulations specifies that for tariff calculation, the design station heat rate is to be multiplied by a factor of 1.045 for a new coal based thermal power plant. Accordingly, the gross station heat rate of the Project for tariff calculation purpose comes out to be 2321 kcal/kWh, which has been considered for Energy Charge calculation. iii) Specific Fuel Oil Consumption:

Regulation 36 (D) of the CERC Tariff Regulations, 2014 stipulates specific oil consumption of 0.5 ml/kWh for a coal-based generating station.

iv) GCV of Coal:

For Energy Charge Rate calculation, the GCV of coal has been computed as the weighted average GCV of coal on as received basis.

v) Landed Price of Coal:

In terms of the Regulation 23 of the CERC Tariff Regulations, GVK is entitled to the landed fuel cost of the primary fuel. The landed price of coal is computed as the weighted average landed price of coal, blending ratio and calorific value for the purposes of computation of energy charges. The landed price of coal also includes the price incurred by GVK towards surface transportation from mining point/extraction point till point of delivery within the mine (internal surface transport) and price of surface transportation from delivery point in the mine till the railway siding (external surface transportation).

GVK vide its submission dated 21.08.2020 submitted that Regulations framed under Section 178 or 181 of the Electricity Act override power purchase agreements as held by Constitutional Bench of the Hon'ble Supreme Court in the PTC Judgment. The said position has been upheld by the Hon'ble Tribunal in Odisha Power Generation Corporation Ltd. vs Odisha Electricity Regulatory Commission: (2017 ELR (APTEL) 0538) (Para 28, 30 & 32) wherein it was held that PPA, even though approved by the regulatory commission, cannot override the tariff regulations and tariff for supply of power has to be determined in terms of applicable regulations. GVK has further submitted that the present case is a cost plus project, in terms of Section 62 of Electricity Act, hence the issue of capping transportation charges does not arise. As per submissions of GVK, the cancellation of the Captive Coal Blocks is an event of force majeure and change in law event and therefore, GVK is entitled to compensation for the consequential impact for the same.

GVK has submitted that as regards surface transportation costs, PSPCL has contended that the same is to be limited to the rates prescribed by BCCL for upto 20 km and GVK is required to submit separate bills in order to claim in consonance with rates prescribed by BCCL, as decided by the Commission in Order dated 06.03.2020 in Petition No. 68 of 2017. GVK submitted that it has challenged the directions of the Commission in Order dated 06.03.2020 including limiting surface transportation charges claimed by GVK to the rates prescribed by BCCL vide Appeal no. 189 of 2019 and the same is pending adjudication. GVK has further submitted that PSPCL's contention that the Commission has already granted relief to GVK for change in law and force majeure events i.e. cancellation of the captive coal blocks by allowing pass through of coal cost procured under Shakti Scheme is misplaced and denied. GVK has submitted that coal supply under Shakti Scheme is not subject matter of the present petition as Shakti Scheme supply commenced in March 2018 whereas the

present Petition pertains to FY 2016-17. Order dated 06.03.2019 has been challenged by GVK by way of Appeal No. 189 of 2019 before the Hon'ble Appellate Tribunal for Electricity and GVK reserves its rights in this regard.

GVK has submitted that in this context, it may be noted that for FY 2016-17, GVK has procured coal from e-auction sources and imported coal and did not source any coal from BCCL. Therefore, comparison of GVK's surface transportation costs with that of BCCL is not tenable. The rates paid by GVK for surface transportation have to be discovered by way of a competitive bidding process and ought to be allowed on the basis of actuals. Furthermore, the surface transportation charges paid by GVK does not depend solely on the distance between the railway siding and the mine. GVK pays different rates to the contractors for transportation of coal by road which is dependent on uncontrollable factors such as:

- Geographical location of the mine and accessibility to the mine and railway siding.
- Law and order problems and restrictions on movement of trucks

The above factors have a bearing on the rates charged by contractors (discovered by way of transparent bidding process) operating in that area, which are beyond the control of GVK. Therefore, the normative approach of limiting surface transportation charges to BCCL rates result in under-recovery of costs incurred by GVK, contrary to the PSERC and CERC Tariff Regulations.

The Hon'ble Supreme Court in the Nabha Power Ltd vs PSPCL: (2018) 11 SCC 508 (Para 63, 64) had held that:

- The mode of transportation of coal is irrelevant and the cost incurred from point to point transportation of coal has to be reimbursed to the generator.
- All cost of coal incurred up to the Project site must be reimbursed to the generator on actual.

Further, the said directions were reiterated by the Hon'ble Supreme Court by Order dated 07.08.2019 in a contempt Petition filed by Nabha Power Ltd. against PSPCL for noncompliance of directions and directed PSPCL to pay the same on actuals. In the present case PSERC Tariff Regulations and CERC Tariff Regulations allow landed cost of fuel on actuals, which includes recovery of cost of coal, cost of surface transportation and cost of railway transportation as incurred by GVK on actuals. However, the Commission has limited surface transportation charges payable to GVK to BCCL rates, contrary to the extant Regulations. While doing so the Commission has neither held that GVK has acted imprudently, nor is there any allegation of profiteering or related party transaction on GVK. More importantly, neither GVK nor PSPCL asked for limiting surface transportation charges payable to GVK.

GVK has submitted that it has come to the knowledge of GVK that PSPCL is paying Nabha Power Ltd. Surface transportation charges on actuals based on the rates being charged by the agencies appointed by Nabha Power for surface transportation. These charges vary by location and distance of the coal mine and PSPCL pays Nabha Power based on actual distance travelled and rates charged. In case of Nabha Power there is no benchmarking to CCL or BCCL surface transportation cost. As per information available with GVK, the distance and the cost of surface transportation varies from Rs 445 per MT to Rs 1400 per MT. However, PSPCL has not raised any objection to such distances (to the best of GVK's knowledge) and pays Nabha Power based on distances and rates actually charged. In fact, PSPCL, in Appeal No. 192 of 2019 (filed by PSPCL challenging Order dated 06.03.2019), has admitted that it is paying Nabha Power Ltd. at actual rates and in any event following a methodology different from that of GVK. It is submitted that PSPCL is paying Nabha Power Ltd. tariff which has been discovered by way of competitive bidding process in terms of Section 63 of the Electricity Act, 2003. In GVK's case, the tariff is to be determined on the basis of the PSERC Tariff Regulations which allow for recovery of energy charges on actuals.

GVK submitted that if benchmarking of the surface transportation charge was permissible, Nabha Power Ltd should also have been limited to actuals surface transportation charges or rates charged by BCCL. However, Nabha Power Ltd. is able to recover the cost on actuals whereas surface transportation charges payable to GVK are limited the rates charged by BCCL.

12.4 The revised parameters submitted by GVK on actual basis are summarized as under:

Sr. No.	Month	Apr-16	Jul-16	Aug-16
1.	Coal Consumption in MT	20,378.70	51,633.98	45071*
2.	Auxiliary Consumption in MU	2	8	7
3.	Heat Rate (Kcal/Kwh)	2,715	2,687	2,521
4.	HFO/LDO Consumption(KL)	83	253	279
5.	GCV (Kcal/Kg)	3,686	4,537	4,006
6.	Landed price of Coal (Rs./MT)	5836	5670	5363

Table No. 23: Revised parameters	on actual basis submitted by GVK
----------------------------------	----------------------------------

*On pointing out by PSPCL, GVK in the revised submission received on 21.07.2020 has included the coal consumption of 3163 MT originally shown for the month March 2017, in the coal consumption for August, 2016, which originally was submitted as 41908 MT.

Based on the above aspects, the energy charge computed for the tariff period of FY 2016-17, on actual costs incurred by GVK is as under:

Table No.24: Energy charges computed for FY 2016-17 on actual cost by GVK

Sr. No.	Month	Apr-16	Jul-16	Aug-16
1.	Energy Charges (Rs./Unit)	4.46	3.36	3.47

GVK submitted that one of the submissions of GVK before the Commission in Petition No. 65 of 2013 and Petition No.33 of 2015 was that it was not possible to compute energy charges due to cancellation of the Tokisud and Pachwara captive coal blocks. The Arbitral Tribunal having taken note of the same, held that the Commission will determine the tariff payable to GVK. This understanding is also consistent with Order dated 11.06.2018 passed by the Commission in Petition No. 68 of 2017. In view of the foregoing, it is submitted that Order dated 01.02.2016 was an interim order, which ceased to operate as Arbitral Tribunal held that the cancellation of the captive coal blocks is an event of force majeure and change in law. The assertion that GVK has accepted the Order dated 01.02.2016 as final determination of fuel charges is incorrect. In the proceedings before the Arbitral Tribunal, GVK has specifically sought relief against the capping of the coal as evident from prayer (e) below:

(a) Declare that the Cancellation of the Coal Blocks pursuant to the judgment of the Hon'ble Supreme Court dated 25.08.2014 and Order dated 24.09.2014 is Change in Law Event in terms of Article 13 of the PPA.

- (b) Declare that the Promulgation of the Ordinance is a Change in Law event in terms of Article 13 of the PPA.
- (c) Declare that the Cancellation of the Coal Blocks pursuant to the judgment of the Hon'ble Supreme Court dated 25.08.2014 and Order dated 24.09.2014 is Force Majeure Event in terms of Article 12 of the PPA.
- (d) Declare that the Promulgation of the Ordinance is a Force
   Majeure Event in terms of Article 12 of the PPA.
- (e) Devise an alternate mechanism for the sourcing of Fuel in terms of the suggestions provided by the Claimant in Paragraph 109 to 115 of the Statement of Claim including necessary amendments to the Amended and Restated PPA;
- (f) Grant consequential extension of SCOD till the issue of procurement of fuel is decided by this Hon'ble Commission.

Therefore, GVK has always proceeded on the basis that Order dated 01.02.2016 was an interim order and fuel cost has to be on the basis of actuals since cancellation of the captive coal block is an event of change in law and force majeure.

GVK further submitted that the Hon'ble Tribunal's judgment dated 21.12.2017 in Appeal No. 193 of 2017 titled GMR Kamalanga Energy Limited v CERC (Para 59, 62 – 64), has upheld that cancellation of captive coal blocks is a change in law event and accordingly GMR Kamalanga Energy Ltd was entitled to compensation for expenditure incurred in procuring coal from alternate sources to meet the shortfall of coal from its assured sources. GVK is entitled to compensation for the change in law event i.e. cancellation of the captive coal blocks such that GVK is restored to the same economic position as if change in law event did not take place in terms of Article 13 of the Amended and Restated PPA. The said compensation would include recovery of cost of coal procured from sources such as e-auction of coal and imported coal on actuals including reimbursing GVK, for the cost of fuel including transportation charges. Further, energy charges are payable on actuals.

#### **PSPCL's submission:**

- 12.5 PSPCL vide letter dated 27.05.2020, 23.06.2020, 28.07.2020 and 10.08.2020 submitted as under:
- PSPCL submitted that Regulation 36 of the CERC Tariff a) Regulations, 2014 provide for a normative PAF of 85% for all thermal generating plants and 83% in case of any shortage of fuel. GVK's has contented that it is entitled to compensation due to unavailability of coal thereby entitling its plant to have PAF corresponding to actual coal availability. The actual cumulative PAF of GVK's plant for FY 2016-17 has been 4.707%. CERC Tariff Regulations while making the normative PAF of 83% instead of 85% itself contemplate a relief when there is shortage of supply of coal. Thus, GVK cannot be allowed to seek a dispensation more that what has already been provided under the Regulations. Secondly, at the time of declaration of COD, GVK had undertaken that it had sufficient fuel to run the plant for 2 to 2¹/₂ years at its full capacity i.e. GVK had sufficient fuel to deliver the entire contracted capacity in FY 2016-17. However, GVK could achieve a PAF of only 4.707% during FY 2016-17. Therefore, GVK is not entitled for any

further relief on its alleged claim of force-majeure as regards PAF when the Regulations are clear. The Commission while passing its Order dated 27.05.2019 in Petition No. 01/ 2018, specifically took strict note of the fact that despite directions from the Commission to procure the balance coal at the earliest, GVK has still not been able to procure the same. As such, GVK has not taken any reasonable steps to mitigate the consequential impact of the force majeure on its coal availability, thereby dis-entitling it from any dispensation as regards change in PAF and the same ought to be as prescribed under the Regulations which is 85%.

- b) With regard to Design Heat Rate and specific oil consumption, PSPCL submitted that computation of energy charges is required to be done strictly in terms of the formula as per Regulation 30(6) of the CERC Tariff Regulations, 2014 as per Order dated 06.03.2019 passed in Petition No. 68/2017. Thus, the norms as prescribed under the said Regulations may be applied to GVK's project as well.
- c) With regard to surface transportation cost in the landed price of coal, PSPCL submitted that GVK has considered weighted average landed cost of coal inclusive of the price of surface transportation from the delivery point in the mine till the railway siding (external STC). The Commission in its Order dated 06.03.2019, has categorically held that in case of any surface transportation cost incurred by GVK beyond the pick-up point within the mine (external STC), the payment of the same would be limited to the cost incurred to carry the coal to the nearest railway siding and would be as per the rates prescribed by M/s

Bharat Coking Coal Ltd. (BCCL) upto 20KM and the said charges would be charged by the Coal Company with separate bills for cost of coal and STC. Since, no stay has been granted by the Hon'ble APTEL in favour of GVK in Appeal No.189/2019, as such, the above findings of the Commission are binding upon GVK. Therefore, in order to claim charges for any external STC, GVK is required to submit separate bills in consonance with the rates prescribed by M/s BCCL as decided by the Commission in the above said Order.

- d) PSPCL has pointed out that GVK has submitted that total coal consumed by GVK during the month of March,2017 was 3163.04 MT of coal. However, as per the state energy account issued by SLDC for the month of March-2017, both the Declared Capacity (DC) of the plant and scheduled energy (SE) by PSPCL were NIL. PSPCL submitted that it is incomprehensible as to how coal can be consumed by GVK without there being any generation during the month of March 2017. GVK is required to provide justification for the same.
- e) Responding to GVK submission that normative auxiliary consumption for billing purpose by PSPCL has been considered as 9% in line with Regulation 36(E)(a) of CERC Tariff Regulations, 2014 for FY 2016-17 instead of taking actual auxiliary consumption, PSPCL submitted that the computation of energy charges is required to be done strictly in terms of the formula prescribed in Regulation 30(6) of the CERC Tariff Regulations, 2014 in terms of the Commission's Order dated 06.03.2019 in Petition No. 68/2017. In the said formula, the auxiliary consumption is to be considered in the denominator,

which has in fact been considered. The declaration of availability by the generator is on ex-bus basis. The ex-bus energy declared to be available is the quantum of energy for which the fixed charges are payable. Thus, the plea of GVK in the said Petition that PAF should be determined by grossing up the declared capacity with the normative auxiliary consumption of 9% has been considered and rejected by the Commission in its Order dated 06.03.2019. GVK has challenged the said Order before the Hon'ble Tribunal in Appeal No.189/2019 which is presently pending adjudication wherein no stay has been granted to GVK. As such, the findings of the Commission are binding on GVK.

- f) PSPCL has stated that GVK has submitted that the Station heat rate for FY 2016-17 has been calculated on the basis of CERC Tariff Regulation 36(C)(b) 2014. The station heat rate has been calculated by multiplying the Design Heat Rate (2221Kcal/kwh) with a factor 1.045. Accordingly, the station heat rate (SHR) of the plant comes out 2321 kcal/kWh which has also been considered for the purpose of billing by PSPCL during FY 2016-17 instead of the actual Design Heat Rate. Similarly, for billing purposes, the specific fuel oil consumption considered by PSPCL is based on normative basis i.e. 0.5 ml/kwh for FY 2016-17 which is as per CERC Tariff Regulations 2014.
- g) PSPCL submitted that GVK has considered the GCV for the purpose of billing in compliance of the Commission's Order dated 06.03.2019 in Petition No. 68/2017. In compliance of the above Order, the GCV considered by PSPCL for the purpose of billing is based on a comparison of coal cost per GCV of its own

thermal power plants i.e. GNDTP/GHTP/GGSSTP with that of the GVK whichever is lower. However, findings of the Commission as regards GCV in the said Order have been challenged by PSPCL before the Hon'ble Appellate Tribunal in Appeal No. 192/2019 and the same is pending adjudication. As such, any true up on the said aspect would be subject to the outcome of the said Appeal.

- h) PSPCL stated that GVK has claimed an amount of Rs. 6211/MT for April 2016, Rs. 5701/MT for July 2016 and Rs. 5429/MT for the month of August 2016. However, the landed price of coal considered by PSPCL for billing purpose for FY 2016-17 is based on the landed price of coal procured by GVK, Railway Transportation charges thereof and Road Transportation charges based on the Commission's Order dated 06.03.2019 in Petition No. 68/2017. The bills submitted by GVK differ from the Landed Price of Coal submitted by GVK. PSPCL requested the Commission to refer to and rely upon the details provided by PSPCL.
- While working out the energy charges, fuel transit & handling loss has been considered by PSPCL as per the Commission's Order dated 06.03.2019 in Petition no.68/2017 i.e. 1% or actual transit loss whichever is less.
- j) PSPCL stated that GVK has submitted the total fuel cost of Rs. 66.45 Crore for FY 2016-17. It is submitted that the billing of the energy charges for the GVK's project is done on the basis of energy scheduled by PSPCL. The energy charge rate is calculated on the basis of normative parameters instead of actual. PSPCL made the payment of energy bills submitted by

GVK to PSPCL for the months of April-2016, July-2016 & Aug-2016. The energy charges paid during FY 2016-17 were originally Rs. 52.07 Crore which were later revised in terms of the Commission's Order dated 06.03.2019 to Rs. 55.12 Crore including Rs. 3.05 Crore as arrears. In addition Rs. 0.51 Crore was paid as interest.

#### **Commission's Analysis**

12.6 As GVK has referred to the previous Orders of the Commission in its various submissions, it is pertinent to bring out the following:

GVK had filed petition no. 65 of 2013 and 33 of 2015 with the following prayers:

#### Petition no. 65 of 2013

- (a) direct the extension of the SCOD for completion and commercial operation of the project for a period of 9 months in the case of unit #1 and for a further period of 6 months for unit #2 to be calculated from the closure of the Force Majeure events namely, approval of railway drawings both in regard to Power project and in regard to coal mine siding and the availability of the land to enter upon and commence mining operations; and
- (b) pass any such further order or orders as this Commission may deem just and proper in the circumstances of the case.

#### Petition no. 33 of 2015

(a) Declare that the Cancellation of the Coal Blocks pursuant to the judgment of the Hon'ble Supreme Court dated 25.08.2014 and Order dated 24.09.2014 is Change in Law Event in terms of Article 13 of the PPA.

- (b) Declare that the Promulgation of the Ordinance is a Change in Law event in terms of Article 13 of the PPA.
- (c) Declare that the Cancellation of the Coal Blocks pursuant to the judgment of the Hon'ble Supreme Court dated 25.08.2014 and Order dated 24.09.2014 is Force Majeure Event in terms of Article 12 of the PPA.
- (d) Declare that the Promulgation of the Ordinance is a Force
   Majeure Event in terms of Article 12 of the PPA.
- (e) Devise an alternate mechanism for the sourcing of Fuel including necessary amendments to the Amended and Restated PPA;
- (f) Grant consequential extension of SCOD till the issue of procurement of fuel is decided by this Hon'ble Commission.

Vide interim Order dated 12.08.2015, the Commission decided to refer all issues for arbitration except prayer (e) of the petition no. 33 of 2015, considering it was urgent in nature and needed to be decided by the Commission forthwith in the interest of the project. The Commission decided the said issue vide final Order dated 01.02.2016 common to petition no. 65 of 2013 & 33 of 2015. The common Order dated 01.02.2016 was challenged by PSPCL before Hon'ble Appellate Tribunal for Electricity in Appeal No. 68 of 2016 (petition no. 33 of 2015) and Appeal No. 69 of 2016 (petition no. 65 of 2013).

Further, the findings of the Commission in Order dated 06.03.2019 in petition no. 68 of 2017 filed by GVK on the various issues i.e auxiliary consumption, cost of coal, surface transport at mine end and handling charges, GCV, transit and handling loss besides some other issues has been challenged by GVK in Appeal No. 189 of 2019.

The Order of the Commission dated 21.05.2018 in petition no. 45 of 2017, where the availability of plant in FY 2016-17 was an issue, has also been challenged by GVK in Appeal No. 218 of 2018.

Also the Order of the Commission dated 17.01.2020 in petition no. 54 of 2017, wherein the Commission has determined the capital cost of the project and determined the AFC/capacity charges and the energy charges for FY 2016-17, has been challenged by GVK in Appeal No. 41 of 2020. In this petition, GVK has raised some of the issues already decided in the aforementioned Orders e.g. coal cost, surface transportation, auxiliary consumption etc. This is not permissible as the issues already decided in these Orders cannot be taken again wherever GVK has tried to do so. Furthermore, there is no stay on these Orders by Hon'ble APTEL. Accordingly, the energy charges for FY 2016-17 are payable/paid by PSPCL to GVK in terms of the PPA, Order dated 01.02.2016 common to petition no. 65 of 2013 & 33 of 2015 and Order dated 06.03.2019 in petition no. 68 of 2017 filed by GVK.

13.0 Interest on under-recovered or over-recovered fixed charges

62

- 13.1 The Commission notes that the applicability of Regulation 9 of PSERC Regulations, 2005 would be on the distribution companies or generating cum distribution companies and cannot be applied as it is to the standalone generating companies. The Commission observes that Regulation 8 (13) of CERC (Terms and Conditions of Tariff) Regulation, 2014 are squarely applicable to under recovery or over recovery of fixed charges in case of generating companies.
- 13.2 The Regulation 8 (13) of CERC (Terms and Conditions of Tariff) Regulation, 2014 is re-produced below for reference: -

"The amount under-recovered or over-recovered, along with simple interest at the rate equal to the bank rate on 1st April of the respective year, shall be recovered or refunded by the generating company or the transmission licensee, as the case may be, in six equal monthly instalments starting within three months from the date of the tariff order issued by the Commission".

13.3 The Commission decides to adopt the CERC Regulations for determining interest equivalent to bank rate on under recovery or over recovery of fixed charges.

Accordingly, interest shall be allowable or recoverable as per Regulation 8 (13) of CERC (Terms and Conditions of Tariff) Regulation, 2014 on under-recovered or overrecovered Annual Fixed Charges (AFC) determined by the Commission.

This Petition stands disposed off accordingly.

Sd/-

Sd/-

Sd/-

(Anjuli Chandra) Member (S.S. Sarna) Member (Kusumjit Sidhu) Chairperson

Chandigarh Dated: 07.09.2020

## **ANNEXURE P-4**

#### PUNJAB STATE ELECTRICITY REGULATORY COMMISSION SITE NO. 3, BLOCK B, SECTOR 18-A, MADYA MARG, CHANDIGARH

#### Petition No.34 of 2019 Date of Order:17.09.2020

Present :

Ms. Kusumjit Sidhu, Chairperson Shri S.S.Sarna, Member Ms. Anjuli Chandra, Member

In the matter of : Petition for True up of Tariff for the control period (FY 2017-18 and 2018-19) under section 62 and 86 of the Electricity Act 2003 read with (a) Punjab State Electricity Regulatory Commission (Terms and conditions for determination of Generation, Transmission, wheeling and Retail Supply Tariff) Regulations, 2014; and (b) Amended and Restated Power Purchase Agreement dated 26.05.2009 executed between GVK Power (Goindwal Sahib) Limited and Punjab State Power Corporation Limited (formerly known as Punjab State Electricity Board)

#### AND

In the matter of: GVK Power (Goindwal Sahib) Limited, Paigarh House,156-159, Sardar Patel Road, Secundrabad-540003.

-----Petitioner

Versus

Punjab State Power Corporation Limited. -----Respondent

276

#### <u>ORDER</u>

- 1.1 GVK Power (Goindwal Sahib) Limited (GVK) has filed the present petition for True Up of the Annual Fixed Cost for FY 2017-18 and FY 2018-19 of its 2x270 MW Coal Based Thermal Power Project at Goindwal Sahib in the State of Punjab. The present Petition for True up for FY 2017-18 and FY 2018-19 has been prepared on the basis of audited financial statements for FY 2017-18 and FY 2017-18 and FY 2018-19 and PSERC (Terms & Conditions for Determination of Tariff) Regulations, 2014.GVK has prayed as follows:
  - a) Admit the present petition.
  - b) Allow the Annual Fixed Cost for the Control Period as per the Table No. 12 of the petition.
  - c) Allow the Annual Energy Charge for the Control period as per Table No. 17 of the petition.
  - d) To pass any other Order as the Commission may deem fit and appropriate under the circumstance of the case and in the interest of justice.
- 1.2 The discrepancies observed in the petition by the Commission were conveyed to GVK on 21.02.2020. After the Lockdown, the Petition was admitted on 03.06.2020 and GVK was asked to issue a public notice. The matter was delayed by GVK and the Commission directed GVK to file its reply to the deficiencies and revise its submission as per the capital cost approved by the Commission in its Order dated 17.01.2020 in Petition no 54 of 2017. The petition was taken up for hearing on 24.06.2020.

GVK requested to file the revised figure(s) according to its petition no. 54 of 2017. The Commission vide order dated 08.07.2020 directed GVK to file the same along with the rejoinder to the reply filed by PSPCL. GVK was also directed to reply to the deficiencies observed by the Commission vide notice no. PSERC/Reg/2618-2619 dated 21.02.2020 and submit the information mentioned in Annexure-A enclosed with the Order within 3 weeks. PSPCL was allowed to submit their sur-rejoinder filed by GVK as well as information mentioned in Annexure-A within a further one week. The petition was fixed for hearing as well as public hearing on 05.08.2020. Public notice was issued on 10.07.2020 in three newspapers. GVK responded to the deficiencies and filed its reply alongwith revised submissions on 31.07.2020 through email and hard copy on 04.08.2020. The Public hearing was held on 05.08.2020. No one from the public appeared in the hearing. PSPCL and GVK were allowed to file rejoinder and surrejoinder. The final hearing took place on 09.09.2020 and the Order was reserved. The Commission directed GVK to submit its rejoinder on the same day of hearing i.e. on 09.09.2020 itself. GVK submitted rejoinder on 09.09.2020 through email and hard copy on 10.09.2020.

1.3 The commercial operation of this project was declared on 16.04.2020. The Commission vide its Order dated 17.01.2020 in Petition no. 54 of 2017 approved the capital cost of the project and determined the Annual Fixed Cost for the energy supplied by the to PSPCL. The true up for FY 2016-17 was approved vide Order dated 07.09.2020 in Petition No. 32 of 2019. The Business Plan and Capital investment Plan for the First MYT Period was approved vide Order dated 30.07.2020 in Petition no 70 of 2017. The Commission vide its order dated 05.08.2020 in Petition no 69 of 2017 allowed Rs.605.64 Crore and Rs 471.61 Crore as provisional AFC for FY 2017-18 and FY 2018-19 respectively.

#### 2.0 Capital Cost for FY 2017-18 and FY 2018-19

#### **GVK's submission**

- 2.1 GVK has claimed capital expenditure of Rs. 4383.54 Crore as on 01.04.2017. GVK submitted that at the time of filing MYT petition for the control period of FY 2017-18 to FY 2019-20 (petition no. 69 of 2017), an additional capital expenditure of Rs. 230 Crore during FY 2017-18 was envisaged against which it has incurred an amount of Rs. 13.74 Crore only in FY 2017-18 and FY 2018-19. GVK submitted that it was unable to carry out several pending works within the original scope due to financial constraints. The additional capital expenditure of Rs. 12.90 Crore incurred by it in FY 2017-18 and Rs. 0.84 Crore in FY 2018-19 was towards the Initial capital spares (BTG) which was contemplated under the original scope of project works and towards the Coal Testing Laboratory respectively.
- 2.2 GVK craved liberty to claim the balance amount of Rs. 216.26 Crore in the remaining MYT Control Period for FY 2019-20. GVK further requested the Commission to allow the actual

additional capitalization of Rs. 13.74 Crore as incurred by it in FY 2017-18 and FY 2018-19.

- 2.3 GVK vide its revised submissions on 31.07.2020, submitted that the actual expenditure incurred by it as on CoD was Rs.4267 Crore. However, vide Order dated 17.01.2020 in petition no. 54 of 2017, the Commission has approved the capital cost of Rs. 3058.37 Crore, and the same has been considered for the purpose of the revised submissions.
- 2.4 In its reply to deficiencies, GVK stated that Rs. 0.84 Crore is the amount incurred by GVK towards the coal testing laboratory for FY 2018-19. The total amount incurred towards coal testing lab is Rs. 1.54 Crore. The balance amount of Rs. 0.70 Crore was incurred in FY 2019-20 and the same has been claimed as Additional Capitalization under Petition No. 70 of 2017.

#### PSPCL's submission

2.5 PSPCL vide submission dated 23.06.2020 stated that GVK's submissions under this head are based on the underlying assumption that since the tariff for the base year i.e. FY 2016-2017 is yet to be determined by the Commission, GVK is entitled to true-up of its tariff for FY 2017-18 based on its claimed capital cost (revised) i.e. Rs. 4267.40 Crore. It was reiterated by PSPCL that the figures submitted by GVK no longer hold good in the light of the Final Tariff Order passed by the Commission wherein the completed capital cost has been approved at Rs. 3058.57 Crore. Thus, GVK is required to revise the same in accordance with the approved capital cost of the

project. As regards the claim of additional capital expenditure of Rs.13.84 Crore in FY 2017-18 and FY 2018-19, it was reiterated that the same is first to be adjudicated in the pending proceedings in Petition No.70/2017 before any true-up in that behalf can be claimed from the Commission.

2.6 PSPCL in its submission received on 25.08.2020 stated that the capital cost as claimed by GVK for FY 2017-18 and 2018-19 in the revised submission is in line with the approved capital cost of the project as approved in the Tariff Order for FY 2017-18 to FY 2019-20 and thus may be accepted by the Commission.

#### **Commission's Analysis**

2.7 The Commission has approved the capital cost of Rs. 3058.37 Crore as on 31.03.2017 in its Order dated 07.09.2020 in petition no. 32 of 2019 for True-up of FY 2016-17. Capital Investment Plan for the first MYT period has also been approved vide Order dated 30.07.2020 in Petition no. 70 of 2017. In the present petition GVK has claimed additional capital expenditure of Rs. 12.90 Crore towards the Initial capital spares (BTG). Further Rs. 0.84 Crore have been claimed towards the coal testing laboratory set up by GVK as per directions of the Commission in Order dated 06.03.2019 in petition no. 68 of 2017.

Regulation 3.19 and 18.2 of the PSERC Tariff Regulations, 2014 specifies as under:

"3.19. "Cut-off Date" means 31st March of the year closing after two years of the year of commercial operation of the project, and in case the project is declared under commercial operation in the last quarter of a year, the cut-off date shall be 31st March of the year closing after three years of the year of commercial operation;

. . . . . .

18.2. The Capital Expenditure of the following nature actually incurred after the cut-off date may be admitted by the Commission subject to prudence check:

a. Un-discharged/Deferred liabilities relating to works/services within the original scope of work;

b. Liabilities to meet award of arbitration or for compliance of the order or decree of a court;

c. On account of change of law;

d. Any additional works/services which have become necessary for efficient and successful operation of the project, but not included in the original project cost; and

e. ....."

In view of the above, the Commission allows the additional capital expenditure of Rs. 12.90 Crore in FY 2017-18 and Rs. 0.84 Crore in FY 2018-19. Considering the capital cost of Rs. 3058.37 Crore as on 31.03.2017 and the additional capital expenditure of Rs. 12.90 Crore during FY 2017-18, the capital cost as on 31.03.2018 works out Rs. 3071.27 Crore for FY 2017-18. Further, considering the capital cost of the project as Rs. 3071.27 Crore as on 01.04.2018 and additional capital expenditure of Rs. 0.84 Crore during FY 2018-19, the capital cost as on 31.03.2019 works out Rs. 3072.11 Crore for FY 2018-19.

#### 3.0 Operation & Maintenance Expenses GVK's Submission

3.1 GVK submitted that Regulation 26 of the PSERC Tariff

Regulations, 2014 provides for Operations & Maintenance expenses along with its subsequent amendment vide 03.02.2016.

3.2 GVK further submitted that Regulations 8.1 and 8.2 of PSERC Tariff Regulations, 2014 provide for determination of Baseline values by this Commission based on estimates of the expected figures for the relevant year, industry bench mark/ norms and other factors etc. However, since no baseline values were approved by the Commission, GVK has adopted the baseline values for the O& M expenses as provided in the CERC Regulations.

#### Employee Cost:

- 3.3 The Employee cost includes salaries payable to employees, allowances and other terminal benefits.
- 3.4 Inflation factor: The escalation index has been computed in terms of the PSERC MYT Regulations,2014 considering the WPI (available till March 2019) and CPI index (available till March ,2019) as under:-

Period	FY 2016-17	FY 2017-18	FY 2018-19
CPI Index (April-March)	275.92	284.41	299.92
Increase/Decrease (%)		3.08	5.45
WPI Index (April-March)	116.60	114.90	119.80
Increase/Decrease (%)		2.96	4.27
CPI:WPI Index (50:50)(%)	ਰਤਾ, ਪੀ	3.02	4.86

 Table No. 1: Computation of Escalation Index

3.5 In view of the above, the inflation factor for the period 2017-18 and 2018-19 as per PSERC Tariff Regulation, 2014 has been computed as 3.02% and 4.86%. GVK in its revised submission dated 31.07.2020 and 09.09.2020 stated that the inflation factor

is on the lower side considering the average rise in salaries and other expenses and in order to retain employees with experience, it has to pay competitive remuneration to match industry standards. Considering the fact that Employee Expenses is the most critical component in the overall operational expenditure, a 3.906% hike in Employee Expenses is insufficient to maintain salaries even at industry average for the Control Period.

3.6 GVK has incurred an expenditure of Rs. 9.99 Crore for FY 2017-18 and Rs 12.67 Crore for FY 2018-19 on employee cost. In its submission dated 31.07.2020 GVK stated that the increase in cost is due to increase in number of employees in FY 2018-19 from 114 to 143 which was in turn due to increase in operation of the plant as compared to previous years. Accordingly, the Employee Cost for the Period 2017-18 and FY 2018-19 is as under:-

Table No 2 : Employee Cost submitted by GVK for FY 2017-18 and FY 2018-19

			(Rs.Crore)
Sr.No	Particular	FY2017-18	FY 2018-19
1 0	Employee Cost	9.99	12.67

#### **PSPCL's Submission**

3.7 PSPCL vide memo no. 5612 dated 21.08.2020 submitted that GVK in the present Petition has claimed a sum of Rs. 154.98 Crore for FY 2017-18 and Rs. 164.10 Crore for FY 2018-19 as Operations & Maintenance (O&M) expenses in accordance with the formula prescribed under Regulation 26.1 of the PSERC Tariff Regulations. The inflation factor is to be used for indexing the employee cost which is to be a combination of the Consumer Price Index (CPI) and Wholesale Price Index (WPI) of nth year and is to be calculated as 0.50xCPIn+0.50xWPIn.

PSPCL submitted that under the PSERC Tariff Regulations, the Commission has prescribed definitive formulae based on realistic indexes and GVK cannot be permitted to seek deviation from the same. Any additional employee cost beyond what is permissible under the Regulations, is to be borne by GVK itself and cannot be permitted to be passed on to the consumers in the State.

#### **Commission's Analysis**

- 3.8 GVK in Petition no 69 of 2017 claimed an employee cost of Rs 15.58 Crore and Rs 17.92 Crore for FY 2017-18 and FY 2018-19 respectively. The Commission in its Order dated 05.08.2020 had approved employee cost of Rs 9.99 Crore and Rs 10.70 Crore for FY 2017-18 and FY 2018-19 respectively.
- 3.9 The baseline values of O&M expenses are to be determined as per Regulations 8(1) of PSERC MYT Regulation-2014, which states as

#### **"8.1 Baseline values**

- a) The baseline values for the control period shall be determined by Commission and the projections for the control period shall be based on these figures.
- b) The baseline values shall be inter-alia based on figures approved by the Commission in the past, latest audited accounts, estimates of the expected figures for the relevant year, industry benchmarks/norms and other factors considered appropriate by the Commission."

3.10 The employee cost is considered in two parts -Terminal benefits and other employee cost. Terminal benefits are to be allowed on actual basis. GVK has claimed Rs. 9.99 Crore and Rs. 12.67 Crore as employee cost for FY 2017-18 and FY 2018-19 respectively as per Annual Audited Accounts which includes Terminal benefit of Rs 0.73 Crore and Rs.0.99 Crore respectively. The Commission allows the actual Terminal benefit of Rs 0.73 Crore for FY 2017-18 and FY 2018-19 respectively. Accordingly, terminal benefits allowed by the Commission for FY 2017-18 and FY 2018-19 respectively.

Table No. 3 : Terminal benefits allowed by the Commission

(Rs. Crore)

Sr.No	Particulars	FY 2017-18	FY 2018- 19
1	Terminal benefits	0.73	0.99

- 3.11 The 'Other Employee Cost' of Rs. 9.26 Crore was provisionally approved by the Commission in Order dated 05.08.2020 in Petition no 69 of 2017 for FY 2017-18. Since, the plant has run for 349 days in that year, therefore taking FY 2016-17 as the baseline year is not feasible. The Commission considers Rs. 9.26 Crore as baseline for calculating 'Other Employee Cost' for subsequent years.
- 3.12 The Employee Costs are to be determined as per Regulation 26.1 of PSERC MYT Regulations, 2014 (as amended from time to time). Relevant sections of Regulation 26 of MYT Regulations, 2014 are reproduced below for reference:

#### (ii) EMPn = (EMPn-1)*(INDEX n/INDEX n-1)

- **INDEXn** Inflation Factor to be used for indexing the Employee Cost.
- This will be a combination of the Consumer Price Index (CPI) and the Wholesale Price Index (WPI) of nth year and shall be calculated as under:-

#### INDEXn = 0.50*CPIn + 0.50*WPIn

**'WPIn**' means the average rate (on monthly basis) of Wholesale Price Index (all commodities) over the year for the nth year.

**'CPIn'** means the average rate (on monthly basis) of Consumer Price Index (Industrial workers) over the year for the nth year.

3.13 Accordingly, the Commission has calculated the INDEXn for FY 2017-18 and FY 2018-19 as under:

Table No. 4: WPI and CPI Increase considered for FY 2017-18

Sr. No.	Particulars 1 -	FY 2016-17	FY 2017-18	Increase (%) over FY2016-17
1	CPI	275.92	284.42	3.08
2	WPI	111.60	114.90	2.96

INDEX n/INDEX n-1 = (0.5*3.08) +(0.5*2.96) = 3.02%

#### Table No. 5: WPI and CPI Increase considered for FY 2018-19

Sr. No.	Particulars	FY 2017-18	FY 2018-19	Increase (%)
1	CPI	284.42	299.92	5.45
2	WPI	114.88	119.79	4.28

#### INDEX n/INDEX n-1 = (0.5*5.45) + (0.5*4.28) = 4.86%

3.14 The Commission considers the escalation of 3.02% and 4.86% to determine other employee cost for FY 2017-18 and FY 2018-
19 respectively. The other employee cost and employee cost approved by the Commission for FY 2017-18 and FY 2018-19 is as under:

# Table No.6 : Employee Cost approved by the Commission for FY2017-18 and FY 2018-19

(Rs. Cro			(Rs. Crore)
Sr. No.	Particulars	FY 2017-18	FY 2018-19
1.	Other Employee Cost	9.26	9.26
2.	Escalation Factor	3.02%	<mark>4.86</mark> %
3.	Other Employee Cost	9.26	9.71
4.	Terminal Benefits	0.73	0.99
5.	Total Employee Cost	9.99	10.70

Note: Since the Commission has not taken FY 2016-17 as base year for calculation of employee cost, the question of escalation over FY 2016-17 has not been considered.

## Repair and Maintenance and A&G Expenses GVK's Submission:

3.15 GVK in its revised submission dated 31.07.2020 stated that the R&M and A&G expenses are linked to K factor and WPI Index, "K" being the constant (expressed in percentage) that governs the relationship between R&M and A&G expenses and Gross Fixed Assets. For computing K factor, GVK has estimated the R&M expenses and A&G expenses by deducting the projected Employee Cost for the FY 2017-18 from the total O&M expenses as per the CERC Regulations for the FY 2017-18. The details are as under:-

Table No 7 : Computation of K by GVK for the Control period

(Rs. Crore)

Particulars	FY 2017-18
Total O&M expenses for the FY 2017-18 as	154.98
per CERC Regulations	
Less : Employee Cost for the FY 2017-18	9.99
R&M and A&G costs	144.99
Opening GFA	3058.37
Closing GFA	3071.27
Average GFA	3064.82
R&M and A&G Exp. as % of GFA	4.73
K factor	4.73

- 3.16 In view of the above, GVK prayed to approve the K Factor as mentioned above. GVK has considered the said K factor (as 4.73%) for the FY 2017-18.
- 3.17 GVK in its revised submission dated 31.07.2020 and 09.09.2020 submitted that the increase in WPI Index works out to 4.27% for FY 2018-19. Hence, for the purpose of R&M and A&G Expenses, the Petitioner has considered the escalation Index of 4.27% (i.e., average of increase in WPI for FY 2018-19 as per latest data available). Accordingly, GVK has projected combined R&M and A&G expenses as under:

Table no. 8: O&M Expenses for FY 2017-18 and FY 2018-19

(Rs. Crore)

Sr.No	Particulars	FY 2017-18	FY 2018-19
1	Opening FGA	3058.37	3071.27
2	Additional Capitalization	12.90	0.84
3	Closing GFA	3071.27	3072.11
4	Average GFA	3064.82	3071.49
5	K Factor(adjusted for WPI index)	4.73%	4.9%
6	Escalation factor		4.27
7	R & M and A & G Expenses	144.99	151.51
8	Employee Cost	9.99	12.67
9	Total O&M Expenses	154.98	164.18

3.18 GVK in its revised submission dated 31.07.2020 and 09.09.2020 has stated that since the actual O&M expenses FY 2017-18 and FY 2018-19 incurred are lower than the normative O&M expenses hence, it prays that the Commission may allow the normative O&M expenses for the purpose of the present true up for FY 2017-18 and FY 2018-19.

Table No.9: Actual O&M Expenses claimed by GVK

(Rs. Crore)

Sr.No	Particulars	FY 2017- 18	FY 2018- 19
1	R&M Expenses	19.15	25.86
2	A & G Expenses	37.85	27.08
3	R & M and A& G Expenses	56.99	52.94
4	Employee Cost	9.99	12.67
5	Total O & M Expenses	66.98	65.61

3.19 GVK prayed to allow O & M expenses as detailed above in line with provisions of PSERC Tariff Regulations,2014.

#### **PSPCL's submission:**

3.20 PSPCL vide memo no. 5612 dated 21.08.2020 submitted that as regards calculation of Repair and Maintenance Costs (R&M) and Administrative and General Costs(A&G) is concerned, the Regulations provide the same to be calculated as R&Mn + A&Gn = K*GFA*(WPIn/WPIn-1), wherein 'K factor' is a constant (expressed in %) governing the relationship between R&M and A&G expenses and gross fixed assets (GFA). Thus, for the purpose of calculation of the same, the gross fixed assets, as approved by the Commission are required to be taken as a base of which % value of the summation of R&M and A&G expenses is considered to the 'K factor'. The Commission while arriving at the above figures, has also held that while scrutinising the books of accounts of GVK, it has been observed that certain expenses under this head are abnormally high for FY 2017-18 as compared to FY 2016-17 and FY 2018-19. As such, it is clear that the figures submitted by GVK in the present true-up Petition are highly inflated and are inadmissible in view of the already determined O&M Expenses by the Commission in the Tariff Order for FY 2017-19.

#### **Commissions' Analysis**

- 3.21 R&M and A&G expenses are a part of O&M expenses. The Commission had provisionally approved O&M expenses for FY 2017-18 and FY 2018-19 as Rs. 27.92 Crore and Rs. 29.20 Crore respectively vide its Order dated 05.08.2020 in Petition no 69 of 2017. The opening value of approved Gross Fixed Assets as on 1.04.2017 is Rs. 3058.37 Crore as determined in the para 25 of Petition no 54 of 2017.
- 3.22 GVK in their annual audited accounts for FY 2016-17, FY 2017-18 and FY 2018-19 have shown other expenses including R&M and A&G expenses as Rs. 23.42 Crore, Rs. 232.63 Crore and Rs.130.67 Crore respectively. However, these included liquidated damages of Rs. 167.67 Crore and Rs.17.37 Crore for FY 2017-18 and FY 2018-19 respectively. The liquidated damages cannot be considered as a part of R&M and A&G expenses. Similarly, expenses of Rs.58.12 Crore on account of Bad debts written off and provision for doubtful debts cannot be

considered as R&M and A&G expenses. Therefore, the R&M and A&G expenses works out to be Rs. 23.42 Crore, Rs.64.96 Crore and Rs 55.18 Crore for FY 2016-17, FY 2017-18 and FY 2018-19 respectively. On scrutiny, it has been found that certain A&G expenses are abnormally high for FY 2017-18 as compared to FY 2016-17 and FY 2018-19. These are:

- i) Insurance: GVK has incurred expenses of Rs. 6.02 Crore in FY 2017-18 as compared to Rs. 2.68 Crore, and Rs. 2.84 Crore on Insurance in FY 2016-17 and FY 2018-19 respectively.
- ii) Rates and Taxes: GVK has incurred expenses of Rs. 3.63
   Crore in FY 2017-18 as compared to Rs. 0.51 Crore and Rs. 1.12 Crore on Rates and taxes in FY 2016-17 and FY 2018-19 respectively.
- iii) Power and Fuel: GVK has incurred expenses of Rs. 13.33
   Crore in FY 2017-18 as compared to Rs. 6.93
   Crore and Rs. 4.36
   Crore on power and fuel in FY 2016-17 and FY 2018-19 respectively.
- iv) Provision for Diminution in value of Investment: In the order dated 17.1.2020 in Petition 54 of 2017, the Commission has noted that GVK had diverted the funds meant for capital expenditure out of the loan taken from financial institutions by investment in Mutual Funds. The Commission has neither considered interest paid on loans used for such other investments nor income earned from such other business/investment as part of the capital expenditure.

- v) Miscellaneous Expenses: GVK has incurred expenditure of Rs. 5.69 Crore in FY 2017-18 as compared to Rs. 1.54 Crore and Rs. 2.12 Crore on miscellaneous expenses in FY 2016-17 and FY 2018-19 respectively.
- vi) Legal and Professional charges: GVK has incurred legal and professional charges of Rs 4.67 Crore and Rs.9.34 Crore during FY 2017-18 and FY 2018-19 respectively as compared to Rs 3.41 Crore in FY 2016-17.
- vii)GVK has achieved PLF of approximate 4%, 32% and 52% during FY 2016-17, FY 2017-18 and Rs 2018-19 respectively. In this regard the following has been observed:
  - Contract Manpower: GVK has shown expenses on contract manpower as Rs 5.15 Crore, Rs. 13.48 Crore and Rs 16.63 Crore for FY 2016-17, FY 2017-18 and FY 2018-19 respectively as against employee cost of Rs 8.66 Crore, Rs 9.99 Crore and Rs 12.67 Crore for FY 2016-17, FY 2017-18 and Rs 2018-19 respectively.
  - Consumption of stores: GVK has incurred Rs 10.07
     Crore on consumption of stores and spares in FY 2017-18 as against Rs. 1.19
     Crore in FY 2016-17
     which is abnormally high considering the fact that the plant is new and has only been operational since 16.4.2016.
  - Ash Handling Charges: GVK has claimed ash handling charges of Rs 0.77 Crore, Rs.3.12 Crore

and Rs.3.10 Crore for FY 2016-17, FY 2017-18 and FY 2018-19 respectively, but has not shown any income from sale of ash. These charges cannot be allowed considering labour, power and water charges are already being allowed separately.

3.23 The Commission also observes that, the A&G and R&M expenses of GHTP and GGSSTP (Thermal Plants) of PSPCL are lower than GVK in spite of being older and of higher capacity as shown in the table below:

 Table No. 10: Comparison of A&G and R&M expenses of PSPCL

 owned Thermal Plants and GVK for FY 2017-18 and FY 2018-19

(Rs. Crore)

Sr.	Particulars		FY 2017-18			FY 2018-19			
No.	1 1.			GHTP	GGSSTP	GVK	GHTP	GGSSTP	GVK
1	Capaci	ty (MW)	No.	920	840	540	920	840	540
2	A&G expens	and es	R&M	48.59	61.06	64.96	49.80	62.58	55.18

3.24 The Commission has taken into consideration R&M and A&G expenses based on audited accounts of GVK for FY 2016-17, FY 2017-18 & FY 2018-19 and the industry benchmark for determining the baseline values of R&M and A&G expenses for FY 2017-18 as under:

Table No. 11: Determination of Baseline value of the R&M and A&G expenses for FY 2017-18 based on the Annual Audited Accounts for FY 2016-17, 2017-18 and FY 2018-19

(Rs. Crore)

Sr. No	Particulars	FY 2016-17	FY 2017-18	FY 2018-19	Base line value for FY 2017-18
I	A&G Expenses				
1.	Insurance	2.68	6.02	2.84	2.76

Sr. No	Particulars	FY 2016-17	FY 2017-18	FY 2018-19	Base line value for FY 2017-18
2.	Rent	0.03	0.03	0.06	0.05
3.	Rates & Taxes	0.51	3.63	1.12	0.82
4.	Legal & Professional Charges	3.41	4.67	9.34	3.00
5.	Auditor's Remunerations: Statutory Audit	0.02	0.03	0.07	0.03
	Tax Audit	0	0.02	0	0.02
	Other Services	0	0.01	0.01	0.01
6.	Provision for Diminution in value of Investment	0.28	0.12	0.00	0.00
7.	Communication cost	0.22	0.21	0.08	0.08
8.	Travelling expenses	0.28	0.75	0.98	0.63
9.	Miscellaneous expenses	1.54	5.69	2.12	2.12
10.	Inventory Written off	0	0.53	0.00	0.00
11.	Contract Manpower	5.15	13.48	16.63	8.00
12	Ash Handling Charges	0.77	3.12	3.10	0.00
13.	Water drawl charges	0	1.21	0.00	0.00
14.	Power & Fuel	6.93	13.33	4.36	4.36
15.	Total	21.82	52.85	40.71	21.88
Ш	R & M Expenses	LASL	181	11	1 19
1.	Consumption of Stores & Spares	1.19	10.07	11.64	4.00
2.	Repair: Buildings	0	0.27	0.29	0.27
3.	Repair: Plant & Machinery	0.12	1.17	1.48	1.17
4.	Repair: Other Assets	0.29	0.6	1.06	0.60
5.	Total	1.60	12.11	14.47	6.04
	R&M and A&G Expenses	23.42	64.96	55.18	27.92

3.25 The Commission has already approved investment/expenditure of Rs. 12.90 Crore and Rs. 0.84 Crore for FY 2017-18 and FY 2018-19 respectively as per the Capital Investment Plan. Capitalization of assets is considered as an addition to Gross Fixed Assets based on the nature of the capital expenditure. The additional capitalization/GFA during FY 2017-18 and FY 2018-19 is being considered as Rs. 12.90 Crore and Rs. 0.84 Crore respectively. 3.26 The Commission considers Rs. 27.92 Crore as base R&M and A&G expenses for FY 2017-18 and determines the K factor for the 1st Control Period as under:

Table No.12: R&M and A&G expenses based on K factor and indexation for FY 2017-18 and FY 2018-19 determined by the Commission

(Rs. Crore)

Sr. No.	Particulars	Amount
1.	Opening GFA as on 1.4.2017	3058.37
2	Addition during the year	12.90
3.	Closing GFA as on 31.3.2018	3071.27
4.	Average GFA	3064.82
5.	R&M and A&G expenses	27.92
6	Less: Audit Fee	0.06
7	Base R&M and A& G expenses	27.86
8.	K factor	0.909%

3.27 The Commission determines the R&M and A&G expenses for FY 2017-18 and FY 2018-19 based on K factor calculated above as under:

# Table No. 13: R&M and A&G determined by the Commission for FY 2017-18 and FY 2018-19

		(Rs.	Crore)
Sr. No.	Particulars	FY 2017-18	FY 2018-19
1	Opening GFA	3058.37	3071.27
2	Addition during the year	12.90	0.84
3	Closing GFA	3071.27	3072.11
4	Average GFA	3064.82	3071.69
5	K factor	0.909%	0.909%
6	WPI Index (as per para 3.13)		4.28%
7	R&M and A&G expenses after adjusting WPI increase and K factor	27.86	29.12
8	Audit Fee	0.06	0.08
9	Total R&M and A&G expenses	27.92	29.20

3.28 The O&M expenses consisting of employee cost and R&M and A&G expenses as determined in para 3.14 and 3.27 are approved as under:

Table No.14: O&M expenses for FY 2017-18 and FY 2018-19

			(Rs. Crore)
Sr. No	Particulars	FY 2017-18	FY 2018-19
1	Employee Cost	9.99	10.70
2	R&M and A&G Expenses	27.92	29.20
3	O&M Expenses	37.91	39.90

#### 4.0 Depreciation

### **GVK's Submission:**

- 4.1 GVK submitted that Regulation 21 of the PSERC Tariff Regulations, 2014, as amended by Notification dated 03.02.2016, provides for calculation of depreciation in respect of Coal Based Thermal Generating Plants.
- 4.2 GVK further submitted that depreciation for Generation Assets shall be calculated annually as per the straight-line method over the useful life of the Asset at the rate of depreciation specified by the Central Electricity Regulatory Commission ("CERC") from time to time. Accordingly, GVK has computed the depreciation on the Gross Fixed Assets excluding land. GVK in its revised submission dated 31.07.2020 has submitted the depreciation charge for the Control Period as under:

# Table No.15 : Depreciation Charges for the Control Period(Rs. Crore)

S	r.No	Particulars	FY 2017-18	FY 2018-19
1		Opening Gross Fixed Assets	3058.37	3071.27

2	Less: Undischarged liabilities included in	0.00	0.00
	the above		
3	Opening Capital Cost excluding	3058.37	3071.27
	undischarged liabilities (1-2)		
4	Additional Capitalization	12.90	0.84
5	Closing Capital Cost (3+5)	3071.27	3072.11
6	Average Capital Cost (Average of 3 and 5)	3064.82	3071.69
7	Freehold Land	96.75	96.75
8	Rate of Depreciation	4.77%	4.77%
9	Remaining depreciable value{90%x(6-7)}	2671.26	2677.44
10	Depreciation (annualised) (Minimum of	146.17	146.51
	(8x6) and 9}		
11	Depreciation ( for the period) (=10)	146.17	146.51
12	Cumulative depreciation at the end of the period	285.98	432.49

4.3 GVK prayed the Commission to allow depreciation as detailed above in line with the provisions of the PSERC Tariff Regulations, 2014. GVK in its submission on 09.09.2020 submitted that the difference in the annual depreciation is due to the reason that while calculating the weighted average depreciation rate, rate of depreciation of land has been considered at 0%. The value of land had not been deducted from the Gross Fixed Assets, while computing depreciation. Deducting the value of land from the Gross Fixed Assets, the rate of depreciation comes upto 4.77% as claimed in the Petition. Accordingly, GVK prayed to the Commission to approve the depreciation rate at 4.77%.

#### **PSPCL's Submission**

4.4 PSPCL vide memo no. 5612 dated 21.08.2020 submitted that in terms of Regulation 21 of the PSERC Tariff Regulation 2014, GVK has calculated depreciation for generation assets as per the straight-line method over the useful life of the Asset at a weighted average rate of 4.77% and claimed a depreciation of Rs.146.17 Crore and Rs.146.51 Crore in FY 2017-18 and 2018-19. However, in the original Petition filed by it, depreciation has been claimed at a weighted average rate of 4.80%. Since, depreciation approved by the Commission in the Tariff Order for FY 2017-18 is based upon the annual audited accounts of GVK for the said FY, the same may be approved as against the claimed figures by GVK in the present Petition

#### **Commission's Analysis**

4.5 As regards the Depreciation, Regulation 21 of PSERC MYT Regulations, 2014 has been amended vide notification dated 03.02.2016 as under:

**"21.1.** The value base for the purpose of depreciation shall be the capital cost of the assets admitted by the Commission:

Provided that land other than the land held under lease and the land for reservoir in case of hydro generating station shall not be a depreciable asset and its cost shall be excluded from the capital cost while computing depreciable value of the asset;

Provided further that depreciation shall be calculated after deduction of consumer contributions, capital subsidies/ Government grants.

**21.2.** The cost of the asset shall include additional capitalization.

**21.3.** The cost shall include foreign currency funding converted to equivalent rupees at the exchange rate prevalent on the date when foreign currency shall actually be availed but not later than the date of commercial operation.

**21.4.** Depreciation for generation and transmission assets shall be calculated annually as per straight line method over the useful life of the asset at the rate of depreciation specified by the Central Electricity Regulatory Commission from time to time.

21.5. Depreciation for distribution assets and other assets not

specified by CERC shall be at the rates notified by the Commission:

Provided that the total depreciation during the life of the asset shall not exceed 90% of the original cost;

Provided that the remaining depreciable value as on 31st March of the year closing after a period of 12 years from date of commercial operation/ put in use of the asset shall be spread over the balance useful life of the assets;

Provided further that in case of hydro generating stations, the salvage value shall be as provided in the agreement signed by the developers with the State Government for creation of the site.

**21.6.** Depreciation shall be chargeable from the first year of commercial operation/asset is put in use. In case of commercial operation of the asset/put in use of asset for part of the year, depreciation shall be charged on pro rata basis."

4.6 The Commission considers the rate of depreciation as 4.77%

claimed by GVK in its revised submission dated 31.07.2020 for FY 2017-18 and FY 2018-19 and determines Gross Fixed Assets (net of land and land rights) as under:-

Table No. 16: Gross Fixed Assets (net of Land and land Rights) determined by the Commission for FY 2017-18 and FY 2018-19

			(Rs Crore)	
Sr.N	Particulars	FY 2017-18	FY 2018-19	
0		Ch .		
1	Opening GFA	3058.37	3071.27	
2	Addition during the year	12.90	0.84	
3	Closing GFA	3071.27	3072.11	
4	Average GFA	3064.82	3071.69	
5	Land and Land rights	96.75	96.75	
6	Average GFA (net of land and land rights)	2968.07	2974.94	
7	Rate of Depreciation	4.77%	4.77%	
8	Depreciation	141.58	141.90	

Accordingly, the Commission approves depreciation Charges of Rs.141.58 Crore and Rs.141.90 Crore for FY 2017-18 and FY 2018-19 respectively.

## 5.0 Return on Equity GVK's Submission

- 5.1 GVK submitted that Regulations 19 & 20 of PSERC Tariff Regulations, 2014 provides for Debt Equity Ratio and Return on Equity.
- 5.2 GVK in its revised submission dated 31.07.2020 and 09.09.2020 has stated that in the Order dated 17.01.2020 this Commission has approved a capital cost of Rs.3058.37 Crore. GVK submitted that the actual equity invested amounts to more than 30% of the capital cost. Hence, the normative equity has been considered as 30% of the capital cost. Balance equity shall be considered under normative loan.

		1. 19. 10	(Rs	(Rs. Crore)	
Sr.No	Particulars	27	FY 2017-18	FY 2018-19	
1	Opening Gross Block excluding undischarged liabilities)	(a)	3058.37	3071.27	
2	Equity percentage in Opening Gross Block	(b)	30%	30%	
3	Normative Opening Equity	(C)	917.51	921.38	
4	Normative Equity addition @30% due to additional Capitalisation during the year	(d)	3.87	0.25	
5	Actual Equity addition during the year(as per Annual Accounts)	(e)	13.41	-	
6	Normative Equity addition considered for ROE computation(Minimum of d and e)	( f)	3.87	-	
7	Normative Closing Equity (c + f)	(g)	921.38	921.63	
8	Average Equity(Average of c and g)	(h)	919.45	921.51	
9	Normative Rate of ROE as per PSERC 2014 Tariff Regulations	(i)	15.50%	15.50%	
10	Return on Equity (h x i)		142.51	142.83	

Table No.17: Return on Equity submitted by the GVK

(Rs. Crore)

5.3 GVK prayed this Commission to allow the Return on Equity at

the rate of 15.50% in accordance with the PSERC Tariff Regulations, 2014.

#### **PSPCL's Submission**

5.4 PSPCL vide memo no. 5612 dated 21.08.2020 that based on Regulation 19 and 20 of the PSERC Tariff Regulations, 2014, GVK has submitted a break up of debt and equity and calculation of return on equity thereon based upon the normative rate of 15.50%. The debt equity ratio and return of equity as claimed by GVK for FY 2017-18 and 2018-19 in the present Petition is in line with the approved capital cost of the project as approved in the Tariff Order for FY 2017-19 and thus may be accepted by the Commission.

#### **Commission's Analysis:**

5.5 Regulation 20 of PSERC MYT Regulations, 2014 provides for recovery of Return on Equity which is reproduced hereunder: *"20. RETURN ON EQUITY* 

> Return on Equity shall be computed at the rate of 15.5% on the paid up equity capital determined in accordance with regulation

19: Provided that assets funded by consumer contributions, capital subsidies/Govt. grants shall not form part of the capital base for the purpose of calculation of Return on Equity."

5.6 Regulation 19 of PSERC MYT Regulations, 2014 provides for Debt-Equity Ratio which is reproduced hereunder:

### *"19. DEBT EQUITY RATIO*

19.1. Existing Projects - In case of the capital expenditure projects having Commercial Operation Date prior to the effective date, the debt-equity ratio shall be as allowed by the Commission for determination of tariff for the period prior to the effective date:

Provided that the Commission shall not consider the increase in equity as a result of revaluation of assets (including land) for the purpose of computing return on equity.

19.2. New Projects – For capital expenditure projects declared under commercial operation on or after the effective date:

a. A Normative debt-equity ratio of 70:30 shall be considered for the purpose of determination of Tariff;

b. In case the actual equity employed is in excess of 30%, the amount of equity for the purpose of tariff determination shall be limited to 30%, and the balance amount shall be considered as normative loan;

c. In case, the actual equity employed is less than 30%, the actual debt-equity ratio shall be considered;

d. The premium, if any raised by the Applicant while issuing share capital and investment of internal accruals created out of free reserve, shall also be reckoned as paid up capital for the purpose of computing return on equity subject to the normative debt-equity ratio of 70:30, provided such premium amount and

internal accruals are actually utilized for meeting capital expenditure of the Applicant's business.

19.3. Renovation and Modernization: Any approved capital expenditure incurred on Renovation and Modernization including the approval in the Capital Investment plan shall be considered to be financed at normative debt-equity ratio of 70:30. If the actual equity employed is less than 30% then the actual debt equity ratio shall be considered."

- 5.7 The Commission in its Order dated 05.08.2020 in Petition no 69 of 2017 approved provisional return on equity of Rs.142.51 Crore and Rs.142.83 Crore for FY 2017-18 and FY 2018-19. Further, the Commission in its Order dated in Petition no 32 of 2020 (GVK's True Up for FY 2016-17) approved equity of Rs. 917.51 Crore as on 31.03.2017 which is considered as opening balance for FY 2017- 18. As per Regulation 19 of PSERC MYT Regulations, 2014, Debt Equity ratio of 70:30 has to be considered. Equity of Rs. 3.87(30% of Rs.12.90 Crore of assets addition of FY 2017-18) Crore and Rs 0.25 (30% of Rs. 0.84 Crore of asset addition of FY 2018-19) Crore have been considered for determining return on equity for FY 2017-18 and FY 2018-19 respectively.
- 5.8 The Commission worked out Return on Equity @ 15.50 % on the average paid up equity capital as under:

## Table No.18: Return on Equity for FY 2017-18 and FY2019-20 determined by the Commission

			(Rs Crore)
Sr. No	Particulars	FY 2017-18	FY 2018- 19
1	Opening Equity for the year	917.51	921.38
2	Addition of Equity during the year	3.87	0.25
3	Closing Equity for the year	921.38	921.63
4	Average Equity for the year	919.45	921.51
5	Rate of Return on Equity (%)	15.50%	15.50%
6	Return on Equity	142.51	142.83

Accordingly, the Commission approves Return on Equity of Rs.142.51 Crore for FY 2017-18 and Rs.142.83 Crore for FY 2018-19.

#### 6.0 INTEREST AND FINANCE CHARGES

#### **GVK's Submissions**

- 6.1 GVK submitted that Regulation 24 of the PSERC Tariff Regulations, 2014 provides for Interest and Finance Charges on Loan Capital.
- 6.2 GVK in its revised submission dated 31.07.2020 and 09.09.2020 stated that the interest payable towards Long Term Loans has been calculated on the outstanding loan amounts and prevailing interest rates on the said amounts on the basis of the Completed Capital Cost of the Project as determined by this Commission vide Order dated 17.01.2020. The interest expenses have been computed taking into account repayment towards outstanding loan amounts and applicable interest rates in line with the PSERC Tariff Regulations, 2014.
- 6.3 GVK further submitted that in terms of the PSERC Tariff

Regulations, 2014, the computation of interest on term loans is based on the following:

- (a) The opening gross normative loan on the Completed Capital Cost as approved by this Commission.
- (b) The rate of interest has been considered at the actual applicable advance rate of State Bank of India as on 01.04.2017 and 01.04.2018, which was at 13.22% p.a. for FY 2017-18 and 13.22% p.a. for FY 2018-19.
- (c) The repayment for the Control Period i.e., FY 2017-18 to FY 2018-19 has been considered equal to the depreciation allowed for that year.

6.4 The details of interest on long term loan as submitted by GVK is as under:

Table No. 19: Interest on Long	Term Loans for FY	2017-18 and
FY 2018-19		

.7		(Rs	. Crore)
Sr.	100112		FY 2018-
No			19
1	Gross Normative Loan-Opening	2140.86	2149.89
2	Cumulative Repayment upto Previous year (Cumulative depreciation upto Previous year	139.81	285.98
3	Net Loan Opening	2001.05	1863.91
4	Less: Repayment during the year (Considering depreciation as Principal Repayment)	146.17	146.51
5	Loan Addition due to additional Capitalization during the year (Actual additional capitalisation-Normative Equity Addition considered for ROE computation)	9.03	0.59
6	Net Loan Closing	1863.91	1717.99
7	Average Loan	1932.48	1790.95

Sr. No	Particulars	FY 2017-18FY 2018- 19	
8	Weighted Average Rate of Interest on Loan (Normative)	13.22%	13.22%
9	Interest on Loan	255.53	236.81

6.5 GVK prayed this Commission to allow the interest on Loan Capital as detailed above in accordance with the provisions of the PSERC Tariff Regulations, 2014.

#### **PSPCL's Submission**

6.6 PSPCL vide memo no. 5612 dated 21.08.2020 submitted that GVK has claimed a sum of Rs. 255.53 Crore for FY 2017-18 and Rs. 236.81 Crore for FY 2018-19 as interest on financing charges on long-term loan capital calculated on the normative loan amounts and prevailing interest rates on the said amounts on the basis of the completed capital cost of the project as determined by the Commission vide Order dated 17.01.2020. Since the interest and finance charges on long-term loan capital approved by the Commission in the Tariff Order for FY 2017-19 is based upon Order dated 17.01.2020 passed in petition No.54/2017 and the annual audited accounts of the Petitioner for the said Financial Year, the same may be approved as against the claimed figures by GVK in the present Petition.

#### **Commission's Analysis**

6.7 Regulation 24 of PSERC MYT Regulations, 2014 provides for Interest on Loan Capital which is reproduced hereunder:

#### "24. INTEREST ON LOAN CAPITAL

24.1. For existing loan capital, interest and finance charges on loan capital shall be computed on the outstanding loans, duly taking into account the actual rate of interest and the schedule of repayment as per the terms and conditions of relevant agreements. The rate of interest shall be the actual rate of interest paid/payable (other than working capital loans) on loans by the licensee or the State Bank of India Advance Rate as on April 1 of the relevant year, whichever is less.

24.2. Interest and finance charges on the actual loan capital for new investments shall be computed on the loans, duly taking into account the actual rate of interest and the schedule of repayment as per the terms and conditions of relevant agreements. The rate of interest shall be the actual rate of interest paid/payable (other than working capital loans) on loans by the licensee or the State Bank of India Advance Rate as on April 1 of the relevant year, whichever is less.

24.3. The repayment for each year of the tariff period shall be deemed to be equal to the depreciation allowed for the corresponding year. In case of de-capitalisation of assets, the repayment shall be adjusted by taking into account cumulative depreciation made to the extent of de-capitalisation. 24.4. The Commission shall allow obligatory taxes on interest, finance charges (including guarantee fee payable to the Government) and any exchange rate difference arising from foreign currency borrowings, as finance cost.

24.5. The interest on excess equity treated as loan shall be serviced at the weighted average interest rate of actual loan taken from the lenders."

6.8 The Closing balance of loan amounting to Rs.1999.59 Crore as on 31.3.2017 determined by the Commission in FY 2016-17 in Petition no 32 of 2020, has been considered as the opening loan balance for FY 2017-18. Asset addition of Rs.12.90 Crore for FY 2017-18 and Rs.0.84 Crore for FY 2018-19 has been approved in this Order. 70% of asset addition has been considered to be sourced from debt i.e. Rs. 9.03 (12.90*70%) Crore for FY 2017-18 and Rs. 0.59 (0.84*70%) Crore for FY 2018-19 as normative loan. Repayment of loan has been considered equal to depreciation allowed as per Regulation 24.3 of PSERC MYT Regulations, 2014. GVK claimed the weighted average rate of interest 13.22% for FY 2017-18 and FY 2018-19 which has been considered for calculating interest on long term loan and since rates at which loan has been taken by GVK are less than SBI advance rate, the same have been considered to calculate the interest. The interest on long term loans is calculated as under :

TableNo.20:InterestchargesonLongTermLoansdetermined by the Commission for FY 2017-18 and FY 2018-19

( Rs. Cro			s. Crore)
Sr.No	Particulars	FY 2017-18	FY 2018-19
1	Opening balance of loan	1999.59	1867.04
2	Add: Receipt of loan during the year	9.03	0.59
3	Less: Repayment of loan during the year	141.58	141.90
4	Closing balance of loan	1867.04	1725.73
5	Average Loan	1933.32	1796.39
6	Rate of interest	13.22%	13.22%
7	Interest Charges	255.58	237.48

6.9 As per the Annual Audited Accounts of GVK for FY 2017-18 and FY 2018-19 interest has been worked out to Rs.573.57 Crore and Rs.636.29 Crore respectively on average Loans of Rs.3567.86 Crore and Rs 3758.93 Crore for FY 2017-18 and FY 2018-19 respectively. However, GVK has a liability of Rs. 313.91 Crore and Rs 852.24 Crore towards interest accrued for FY 2017-18 and FY 2018-19 respectively. The interest actually paid on long term loans is as under:

# Table No.21: Interest actually paid by GVK for FY 2017-18 and FY 2018-19

(Rs. Crore)

Sr.No	Particulars	FY 2016-17	FY2017-18	FY 2018-
		250		19
1	Interest charges as per	448.37*	573.57	636.29
	annual audited accounts			
2	Interest paid:			
	Current year	221.95	259.66	-
	Previous year	-	226.42	97.96
3	Closing balance of interest	226.42	313.91	852.24
	due but not paid as per			
	annual audited accounts			

* In the Annual Audited Accounts of GVK for FY 2016-17 interest charges were shown as Rs 451.91 Crore. But in the Annual Audited Accounts of FY 2017-18, the previous year figures of interest charges for FY 2016-17 has been rearranged/regrouped as Rs. 448.37 Crore and Rs 3.54 (451.91 - 448.37) Crore has been shown as Other Finance Charges.

6.10 The Commission determined interest charges for FY 2017-18

and FY 2018-19 as under:

## Table No.22: Interest charges allowed by the Commission for FY2017-18 and FY 2018-19

			s. Crore)
Sr.No	Particulars	Particulars FY 2017-18	
		A A	19
1	Interest determined as per Table 20	255.58	237.48
2	Interest actually paid by GVK	259.66	97.96
3	Interest allowed	255.58	97.96

The balance amount of interest i.e. Rs.139.52 (237.48 - 97.96) Crore of FY 2018-19 will be considered in the year in which they will actually be paid by GVK.

6.11 As per the Annual Audited Accounts for FY 2017-18 and FY 2018-19, finance charges amount to Rs. 3.44 Crore and 0.28 Crore on the average loan amount of Rs. 3567.86 Crore and Rs.3758.93 Crore respectively. Finance charges proportionately work out as Rs.1.86 Crore and Rs 0.13 Crore on the average loan of Rs 1930.65 Crore and Rs.1788.66 Crore for FY 2017-18 and FY 2018-19 respectively and the same are allowed. Thus, total interest and finance charges for FY 2017-18 and FY 2018-19 work out to Rs. 257.44 (255.58+1.86) Crore

#### and Rs. 98.09 (97.96+ 0.13) Crore respectively.

Accordingly, the Commission allows interest and finance charges of Rs.257.44 Crore and Rs. 98.09 Crore for FY 2017-18 and FY 2018-19 respectively.

#### 7.0 INTEREST ON WORKING CAPITAL:

#### **GVK's Submissions**

- 7.1 GVK submitted that Regulation 34 of the PSERC Tariff Regulations, 2014 provides for components of Interest on Working Capital in respect of Coal Based Thermal Generating Plants.
- 7.2 The weighted average rate of interest is computed at 12.25 % p.a. for FY 2017-18 and 12.25 % p.a. for FY 2018-19. Interest on working capital is computed on normative basis as per PSERC Tariff Regulations,2014
- 7.3 GVK in its revised submission dated 31.07.2020 and 09.09.2020 has calculated the interest on working capital for MYT Control Period (i.e. FY 2017-18 to FY 2018-19) as per PSERC Tariff Regulations, 2014. Interest on working capital is projected for the Control Period by applying the rates as mentioned above on the components of Working Capital as given in the table below:

# Table No. 23: Interest on Working Capital claimed by GVK (Rs.Crore)

Sr. No.	Particulars	FY 2017- 18	FY 2018-19
1	Fuel Cost – Primary Fuel & Secondary Fuel	250.75	301.60

	(for 2 months)		
2	O&M expenses for one month	5.58	5.47
3	Maintenance spares (15% of O&M expenses)	10.05	9.84
4	Receivables (2 months of Fixed and Variable	365.61	415.22
	Cost based on Normative Annual Plant		
	Availability Factor)		
5	Total working capital	631.99	732.13
6	Rate of interest (p.a.)	12.25%	12.25%
7	Interest on working capital	77.42	89.69

7.4 Accordingly, GVK prayed to allow the interest on Working Capital as detailed above in line with the provisions of the PSERC Tariff Regulations, 2014.

#### **PSPCL's submission:**

- 7.5 PSPCL vide memo no. 5612 dated 21.08.2020 GVK has submitted that it has calculated interest on working capital on the basis of computed working capital as provided under Regulation 34 of the PSERC Tariff Regulations and claimed a sum of Rs. 77.42 Crore and Rs. 89.69 Crore for FY 2017-18 and 2018-19.
- 7.6. PSPCL further submitted that the Commission had approved the working capital requirement based upon the audited annual statement of accounts and actual fuel cost and receivables received by the Petitioner during the relevant FY. The amounts claimed by GVK in the present Petition appear to be highly inflated as against the approved interest on working capital allowed by the Commission. As such, GVK is entitled only to a working capital requirement of Rs. 268.15 Crore for FY 2017-18, Rs.392.17 Crore for FY 2018-19 for which interest thereon of Rs. 32.85 Crore, Rs.48.04 Crore and Rs. 26.39 Crore only for FY 2017-18 and FY 2018-19 respectively which is

permissible under the applicable Regulations.

#### **Commission's Analysis**

7.7 Regulation 34 of PSERC MYT Regulations, 2014 provides for Interest on Working Capital which is reproduced hereunder:

#### "34.1. Components of Working Capital

- a. Coal-based Thermal Generating Plants: The Working Capital shall cover the following:
- *i.* Fuel Cost for 2 months corresponding to the normative annual plant availability factor;
- ii. Operation and maintenance (O&M) Expenses for 1 month;
- iii. Maintenance spares @ 15% of the O&M expenses;
- iv. Receivables equivalent to two (2) months of fixed and variable charges for sale of electricity calculated on the normative annual plant availability factor.

C. ....."

**34.2** Rate of Interest on Working Capital shall be as per Regulation 25.1 which is reproduced hereunder:

**25.1** The rate of interest on working capital shall be equal to the actual rate of interest paid on working capital loans by the licensee/generating company/SLDC or the State Bank of India Advance Rate as on April 1 of the relevant year, whichever is less. The interest on working capital shall be payable on normative basis notwithstanding that the licensee/generating company/SLDC has not taken working capital loan from any outside agency or has exceeded the working capital loan amount worked out on the normative figures. In case, the licensee/generating company/SLDC has not availed any working capital loan during the year, the rate of interest shall be weighted average of the interest rates on working capital loan of the

b. .....

#### previous three years."

- 7.8 As per PSERC Regulations, the rate of interest on working capital shall be equal to the weighted average rate of interest paid/ payable on loans by the generating company or the State Bank of India Advance rate as on April 1 of the relevant year, whichever is less. The interest on working capital is payable on normative basis notwithstanding that the generating company has not taken working capital loan from any outside agency or has exceeded the working capital loan amount worked out on normative basis.
- 7.9 The Weighted Average Rate of Interest has been considered for FY 2017-18 and FY 2018-19 @12.25% p.a as claimed by GVK in this petition. The State Bank of India Advance Rate as on 01.04.2017 and 01.04.2018 is 13.85 % p.a and 13.45% p.a respectively.
- 7.10 As per the Annual Audited Accounts of GVK for FY 2017-18 and FY 2018-19 the fuel cost was Rs 635.58 Crore and Rs. 1112.36 Crore respectively. The energy charges paid by PSPCL during FY 2017-18 and FY 2018-19 were originally Rs. 416.74 Crore and Rs. 724.38 Crore which were later revised in terms of the Commission's Order dated 06.03.2019 to Rs. 470.51 Crore including Rs. 53.77 Crore as arrears for FY 2017-18. The fuel cost for FY 2018-19 was revised to Rs. 833.90 Crore including arrears of Rs. 109.22 Crore. Thus, fuel cost of Rs. 470.51 Crore and Rs. 833.90 Crore has been considered for the determination of receivables for FY 2017-18 and FY 2018-19 respectively.

7.11 Interest on working capital for MYT Control Period has been calculated as per PSERC MYT Regulations 2014. Interest on Working capital has been calculated for FY 2017-18 and FY 2018-19 by applying the rate of interest of 12.25 % p.a. on components of Working capital i.e. (maintenance spares @ 15% of O&M expenses, O&M expenses for one month and Receivables @ 2 month Annual Fixed Cost) as given below:

TableNo.24: Interest on Working Capital approved by theCommission for FY 2017-18 and FY 2018-19.

(Rs. Crore)

Sr. No.	Particulars	FY 2017-18	FY 2018-19
1	Fuel Cost for two months	78.42	138.98
2	Maintenance spares @15% of O&M	5.69	5.98
3	O&M Expenses for one month	3.16	3.32
4	Receivables for two months	180.43	216.89
5	Total Working Capital	267.70	365.18
6	Rate of Interest (%)	12.25%	12.25%
7	Interest on Working Capital	32.79	44.73

Thus, the Commission approves working capital requirement of Rs.267.70 Crore for FY 2017-18 and Rs. 365.18 Crore for FY 2018-19 and interest thereon of Rs. 32.79 Crore and Rs. 44.73 Crore for FY 2017-18 and FY 2018-19 respectively.

#### 8.0 STATUTORY LEVIES AND TAXES

8.1 GVK submitted that Regulation 23 of the PSERC Tariff Regulations, 2014 provides for Income Tax which is reproduced hereunder: **"23.1** Obligatory taxes, if any, on the income of the generating company or the licensee or the SLDC from its core/licensed business shall be computed as an expense and shall be recovered from the customers/consumers:

Provided that tax on any income other than the core/licensed business shall not constitute a pass through component in tariff and tax on such other income shall be payable by the generating company or the licensee or the SLDC.

**23.2** Tax on income, if actually liable to be paid, shall be limited to tax on return on equity allowed, excluding incentives.

**23.3** Tax on income shall be considered at income tax rate including surcharge, cess etc. as applicable during the relevant year in accordance with the provisions of Income Tax Act, 1961 duly amended from time to time.

**23.4** The benefits of tax holiday and the credit for carrying forward losses applicable as per the provisions of the Income Tax Act, 1961 shall be fully passed on to the customers/consumers.

**23.5** The penalty, if any, arising on account of delay in deposit of tax or short deposit of tax amount shall not be claimed by the generating company or the licensee or the SLDC, as the case may be."

8.2 GVK submitted that it is entitled at actuals, as pass through, any cess, duty, tax, government levy and royalty etc., payable for generation and supply of power to the PSPCL from time to time. GVK in its submission dated 31.07.2020 stated that no claims have been made for FY 2017-18 and FY 2018-19.

#### **PSPCL's submission:**

8.3 GVK has submitted that as per Regulation 23 of the Tariff Regulations, it is entitled a pass through of actual taxes paid by it during the FY 2017-2019. It is submitted that per Regulation 23.2 of the PSERC Tariff Regulation 2014, tax on income, if actually liable to be paid, is to be limited to tax on return on equity allowed, excluding incentives. As such, it is submitted that since the Regulations provide a cap on pass through of tax on income only to the extent of tax paid on return of equity, the return on equity is liable to be first determined for GVK's project for FY 2017-2019. As stated hereinabove, the Commission has approved a sum of Rs.142.51 Crore and Rs.142.83 Crore as the return on equity for FY 2017-18 and 2018-19. As such, GVK is only entitled to the tax paid on the said amount. The Commission while allowing any pass through of tax in the tariff may kindly take note of the above.

#### **Commission's Analysis**

8.4 The benefits of any tax Holiday have to be passed on to the consumer/customer as per PSERC MYT Regulations 2014. GVK has claimed Nil on account of taxes for FY 2017-18 and FY 2018-19.

The Commission does not approve any amount on account of taxes and duties for FY 2017-18 and FY 2018-19.

#### 9.0 Non-Tariff Income

9.1 GVK has submitted actual Non tariff income for FY 2017-18 is Rs 0.56 Crore and for FY 2018-19 is Rs. 0.27 Crore as per audited annual accounts. The breakup for the same is as below:

(a) Interest on Fixed Deposits – Rs.0.47 Crore and Rs. 0.22 Crore in FY 2017-18 and FY 2018-19 respectively. (b) Sale of Scrap – Rs.0.09 Crore and 0.04 Crore in FY 2017-18 and FY 2018-19 respectively.

9.2 GVK further submitted that in accordance with Regulation 28 of the PSERC Tariff Regulations,2014, the Non-Tariff Income has been deducted to arrive at the Net Annual Fixed Cost for FY 2017-18 and FY 2018-19.

#### **PSPCL's submission**

9.3 PSPCL submitted that GVK has stated that non-tariff income is required to be deducted while arriving at the AFC for a relevant FY as per Regulation 28 of the PSERC Tariff Regulations. It is submitted that since in Petition No.69/2017, GVK has not submitted any claim of non-tariff income, the AFC of GVK as determined by the Commission in the Tariff Order for FY 2017-19 is liable to be revised by deducting the above said amounts as submitted by GVK in the present Petition.

#### **Commission's Analysis**

- 9.4 The Non-Tariff Income has been determined as per Regulation-28 of PSERC MYT Regulations-2014(amended from time to time).
- 9.5 The Commission notes that Audited Annual Accounts of GVK for FY 2017-18 and FY 2018-19 show "Other income" as Rs.0.61 Crore and Rs. 0.27 Crore respectively which includes income from bank deposits of Rs. 0.47 Crore and Rs 0.22 Crore respectively. As per para 20.3.4 of the Order dated 17.01.2020 in Petition no 54 of 2017, the Commission had observed that

GVK had diverted funds meant for capital expenditure out of the loan taken from financial institutions by investing the same in Mutual Funds. The Commission has neither considered interest paid on loans used for such other investments nor income earned from such other business/investment as part of the capital expenditure. Other income includes non-operating income of Rs. 0.14 Crore and Rs. 0.05 Crore for FY 2017-18 and FY 2018-19 respectively on account of credit balance written off and sale of scrap. The Commission notes that Rs. 3.12 Crore and Rs. 3.10 Crore have been booked under Ash Handling charges during FY 2017-18 and FY 2018-19 respectively but no income from Ash has been booked during FY 2017-18. Accordingly, the Commissions determines Non-Tariff Income as Rs. 0.14 Crore and Rs. 0.05 Crore as per Audited Annual Accounts of FY 2017-18 and FY 2018-19 respectively.

Accordingly, the Commission approves Non-Tariff Income as Rs. 0.14 Crore and Rs. 0.05 Crore for FY 2017-18 and FY 2018-19 respectively.

#### 10.0 Capacity charges for FY 2017-18 to 2019-20

10.1 The Capacity charges from FY 2017-18 to FY 2019-20, as projected by GVK and approved by the Commission are summarized in the following table:-

Table No.25: Annual fixed charges for FY 2017-18 and 2018-19

(Rs Crore)

Sr.	Particulars	Approved in	Submitted by	Approved by the
No.	Farticulars	Petition no. 69	GVK	Commission

		of 2017					
		FY FY		FY FY		FY	FY
		2017-18	2018-19	2017-18	2018-19	2017-18	2018-19
1	O&M Expenses	37.91	39.90	66.98	65.61	37.91	39.90
2	Depreciation	142.47	142.80	146.17	146.51	141.58	141.90
3	Interest charges	250.04	98.09	255.53	236.81	257.44	98.09
4	Return on Equity	142.51	142.83	142.51	142.83	142.51	142.83
5	Interest on Working Capital	32.85	48.04	77.42	89.69	32.79	44.73
6	Income tax	0.00	0.00	0.00	0.00	0.00	0.00
7	Total Expenses	605.78	471.66	688.61	681.45	612.23	467.45
8	Less:Non-Tariff	0.14	0.05	0.56	0.27	0.14	0.05
9	Annual Fixed Charges	605.64	471.61	688.05	681.18	612.09	467.40

10.2 GVK shall be entitled for payment of capacity charges in accordance with Regulation 30 of the Central Electricity Regulatory Commission (Terms and Conditions of Tariff) Regulations, 2014 (as amended) where the same is not specified in the PSERC Tariff Regulations.

#### 11.0 Generation for FY 2017-18 and FY 2018-19

#### GVK's submission

11.1 GVK in its petition has submitted the month-wise details of annual availability of the Project for FY 2017-18and FY 201819. The gross generation, auxiliary consumption and Scheduled generation for FY 2017-18 and FY 2018-19 are as under:

# Table No. 26: Gross Generation, Auxiliary Consumption and Scheduled generation for FY 2017-18 and FY 2018-19

Sr. No.	Year	Gross Generation (MU)	Auxiliary Consumption (MU)	Scheduled Generation (MU)	Aux Consumption (%)
1.	FY 2017-18	1,542.44	155.32	1,387.12	10.07%
2.	FY 2018-19	2,447.47	244.22	2,203.24	9.98%

11.2 GVK submitted that it could not achieve the target PAF during FY 2017-18 and FY 2018-19 on account of non-availability of coal due to cancellation of the captive coal block allocated to GVK pursuant to judgment of Hon'ble Supreme Court and consequent Cancellation Order dated 24.09.2014. GVK submitted that cancellation of coal blocks has been held to be a Change in Law and Force Majeure event by the Arbitral Tribunal vide its award dated 10.04.2017. Since force majeure was because of factors beyond the control of GVK, it is excused from performance and target PAF should be considered in relation to actual coal available.

In its revised submission on 31.07.2020, GVK submitted the month-wise generation details for FY 2017-18 and FY 2018-19 as under:

Table No.27: Month-wise details of annual generation forFY 2017-18 and FY 2018-19 submitted by GVK FY 2017-18

Month	PAF (%)	PLF (%)	Gross Generation (MU)	Auxiliary Consumptio n (MU)	Scheduled Generatio n (MU)	Aux Consumptio n (%)
Apr-17	8	-	>		1h	1
May-17		54	-		Mar -	-
Jun-17	21.56	16	61.40	6.74	54.66	10.98
Jul-17	49.12	39	154.37	14.89	139.48	9.65
Aug-17	55.10	44	175.39	17.22	158.17	9.82
Sep-17	81.05	64	252.35	23.70	228.65	9.39
Oct-17	56.28	49	195.38	18.76	176.62	9.60
Nov-17	69.65	39	151.95	15.14	136.81	9.96
Dec-17	63.78	52	204.79	18.93	185.86	9.24

Month	PAF (%)	PLF (%)	Gross Generation (MU)	Auxiliary Consumptio n (MU)	Scheduled Generatio n (MU)	Aux Consumptio n (%)
Jan-18	29.95	23	93.03	9.06	83.97	9.74
Feb-18	43.93	35	127.85	12.89	114.96	10.08
Mar-18	35.65	27	119.94	12.01	107.93	10.01
			1,536.46	149.34	1,387.12	9.72

Month	PAF (%)	PLF (%)	Gross Generation (MU)	Auxiliary Consumption (MU)	Scheduled Generation (MU)	Aux Consumption (%)
Apr-18	34.85	26	99.56	10.46	89.09	10.51
May-18	86.70	62	247.11	24.67	222.44	9.98
Jun-18	46.68	42	164.08	16.99	147.09	10.35
Jul-18	41.20	38	152.79	15.18	137.61	9.94
Aug-18	49.14	36	147.31	15.78	131.53	10.71
Sep-18	59.41	37	145.43	15.09	130.34	10.38
Oct-18	52.05	51	205.62	19.25	1 <mark>86</mark> .37	9.36
Nov-18	58.53	54	210.10	18.87	<mark>1</mark> 91.23	8.98
Dec-18	81.90	66	263.63	23.82	2 <mark>39.8</mark> 1	9.04
Jan-19	99.98	64	257.09	23.59	233.50	9.18
Feb-19	85.30	64	232.84	20.97	211.87	9.01
Mar-19	98.67	78	310.39	28.02	282.37	9.03
100	2		2,435.94	244.22	2,203.24	9.55

### **PSPCL's submission**

11.3 PSPCL vide submissions received on 25.08.2020 submitted the details of declared capacity, scheduled energy, PAF and PLF of the project for FY 2017-18 and FY 2018-19.

## **Commission's Analysis**
11.4 As checked from the SLDC's website, in terms of the Final State Energy Accounts for FY 2017-18 and FY 2018-19 containing the data on the declared capacity and the scheduled energy for FY 2017-18 and FY 2018-19, are as under:

Table No. 28:Declared Capacity and Scheduled Generation ofGVK's project for FY 2017-18 & FY 2018-19

Sr. No	Description	H	FY 2017-18	FY 2018-19
1.	Declared Ca	pacity (MU)	1814.339	2850.617
2.	Scheduled (MU)	Generation	1387.124	2203.245

# 12.0 Energy Charges for FY 2017-18 and FY 2018-19 GVK's submission

- 12.1 GVK in the present petition and revised submission dated 31.07.2020 and 09.09.2020 has submitted the details of components of the Energy Charge as follows:
  - i) Landed Cost of Primary Fuel

In compliance of the Commission's Order dated 01.02.2016 in Petition 33 of 2013 and Petition 65 of 2013, GVK was successful in securing long term coal linkage under the SHAKTI Scheme for the Project to the extent of 1.706 MTPA from February 2018 onwards. However, the quantity of coal allocated was sufficient to achieve only 63% PLF. The balance coal was being sourced from other sources, including through e-auction and imported coal, to achieve target availability. GVK submitted that road transportation charges, handling charges related to

loading the coal in to the trucks, unloading of the same at the railway siding and loading the same to the railway wagons, coal sampling cost for analysis to obtain the GCV of coal received and the railway freight, all form part of the Landed Cost of Coal. Thus, the actual landed cost of coal is to be considered for computation of fuel cost/coal cost. Consequently, GVK filed Petition No. 68 of 2017 invoking the dispute resolution procedure envisaged under the PPA. The Commission disposed of the said Petition by its order dated 06.03.2019 which was challenged by both GVK and PSPCL before the Hon'ble Tribunal. Since the matter is sub-judice before the Hon'ble Tribunal, GVK craves leave of the Commission to approach the Commission for modification of the variable cost sought in this Petition, as and when Hon'ble Tribunal decides the matter.

GVK procured coal from e-auction/imported coal and under Shakti scheme and did not source any coal from BCCL mines. Therefore, comparison of GVK's surface transportation cost with BCCL is not tenable. The surface transportation charges paid by GVK do not depend solely on the distance between the railway siding and the mine. The rates paid by GVK for surface transportation should be allowed.

The Regulations under Section 178 or 181 of the Electricity Act override PPAs as held by Hon'ble Supreme

Court in PTC judgment. The said position has been upheld by the Hon'ble Tribunal in the case of OPGCL Vs OERC wherein it was held that PPA, even though approved by the Commission, cannot override regulations and tariff for supply of power has to be determined in terms of applicable regulations.

ii) Landed cost of secondary fuel

The Secondary fuel for the Plant is LDO / HFO. This Secondary Fuel is being procured from PSU Oil Marketing companies. This Secondary Fuel is delivered at site and the weighted average cost incurred on the same may be allowed by the Commission. The energy charge is to be determined as per Regulation 39.4 of the PSERC Tariff Regulations.

iii)

### Normative Auxiliary energy consumption

As per Regulation 36 of the PSERC Tariff Regulations, 2014, the norms for performance parameters which includes normative auxiliary energy consumption for Coalbased generating stations shall be as per the CERC norms. Accordingly, as per Regulation 36(E)(a) of the CERC Tariff Regulations, 2014, the normative auxiliary energy consumption is 8.5% for a unit of 200 MW series. Further, for thermal generating stations with induced draft cooling towers, additional 0.5% is allowed. Thus, normative auxiliary consumption has been considered at 9% for the computation of energy charge. In its revised submissions on 31.07.2020 and 09.09.2020, GVK has claimed actual auxiliary consumption of 9.55% for FY 2017-18 and 9.72% for FY 2018-19 explaining that the auxiliary consumption has been on higher side because of part load operation of the plant during FY 2017-18 and FY 2018-19.

iv) Fuel Transit & Handling Losses

Normative transit and handling loss are considered in accordance with Regulation 40 of the PSERC Tariff Regulations 2014. Further, in its revised submission on 31.07.2020 and 09.09.2020, GVK stated that normative transit and handling loss are 1% in accordance with Regulation 40 of the PSERC Tariff Regulations 2014. However, actual transit and handling losses have been considered for calculation of Energy Charges. Fuel Transit & Handling Losses are 4.20% and 2.43% for FY 2017-18 and FY 2018-19 respectively. Pursuant to the cancellation of the captive coal blocks, coal has been procured in FY 2017-18 under e-auction mode. In FY 2018-19, coal was procured under Shakti Scheme (supply commenced on March 2018). The coal is being transported to the Project on road/rail mode as coal is allocated from various mines which are not under the control of GVK. Furthermore, the railway sidings at the mines are not available to GVK since preference is given to public sector companies/pendency of rakes. Therefore, GVK is required to transport coal over longer distances by road/rail mode resulting in higher Fuel Transit & Handling Losses. The same are due to reasons beyond control of GVK and the Commission may allow the same.

v) Station Heat Rate

As per Regulation 36 of the PSERC Tariff Regulations, 2014, the norms for performance parameters, which includes Station Heat Rate for Coal-based generating stations shall be as per the CERC norms. Regulation 36(C)(b) of the CERC Tariff Regulations 2014, specifies that for tariff calculation, the design station heat rate is to be multiplied by a factor of 1.045 for a new coal based thermal power plant. The Guaranteed Heat Rate for the power plant under the EPC contract is 2221 Kcal/kWh. Accordingly, the Station Heat Rate of GVK for tariff calculation purpose comes out to be 2321 kcal/kWh, which been considered for Energy has Charge calculation. In the revised submission on 31.07.2020 and 09.09.2020, GVK stated that the actual station heat rate as achieved by GVK for FY 2017-18 and FY 2018-19 is 2475 kcal/kWh and 2315 kcal/kWh respectively. Increase in Station Heat Rate for FY 2017-18 is due to part load operations. GVK further requested the Commission to approve Gross Station Heat Rate of 2475 kcal/kWh for FY 2017-18. GVK has submitted that the Station Heat Rate of 2315 kCal/kWh has been achieved in FY 2018-19 and the Commission may allow it to retain the efficiency gains

in terms of Regulation 30 of PSERC Tariff Regulations, 2014.

12.2 Accordingly, GVK has computed the actual Energy Charges for FY 2017-18 and FY 2018-19 as under:

Table No.29 : Computation	of actual Energy Charges by
GVK for FY 2017-18 and FY 20	)18-19

Sr.	Particulars <b>Particulars</b>	FY 2017-18		FY 2018-19	
No.	CRIFE.	Submitted in the Petition	Revised submission received on 31.07.2020		Revised submission received on 31.07.2020
1	Variable Charge per Unit (Rs. / kWh)	4.58	4.65	5.05	5.10
2	Scheduled Energy (in MUs)	1387.12	1387.12	2203.24	2203.24
3	Annual Energy Charge (Rs. in Crore)	635.30	645.01	1112.64	1123.65

## **PSPCL's submission**

12.3 PSPCL vide its submissions dated 23.06.2020 and 21.08.2020 submitted as under:

i) PSPCL submitted that GVK has stated that apart from the approved coal being procured by it through SHAKTI Scheme 2017, it is procuring domestic coal from eauctions conducted by Coal India Limited which is allocated from mines of Central Coal Fields and Northern Coal fields. This necessitates GVK to transport the coal by road, unload the same at railway siding and then load the coal to the railway wagons whenever it is allotted by the Indian Railways for onward transportation to the Project and this process includes road transportation charges and handling charges related to loading the coal in to the trucks, unloading the same at the railway siding and loading the same into the railway wagons. Further, at the project site, the coal sample is taken from the railway wagons and sent for analysis to obtain the GCV of coal received. All these costs are stated to form part of the landed cost of coal including the railway freight. GVK has also submitted that the outcome of Appeal No.189/2019 and Appeal No.192/2019 pending adjudication before the Hon'ble Appellate Tribunal would also have a material bearing on the Variable Cost as sought by GVK.

ii)

PSPCL submitted that the Commission in its Order dated 06.03.2018 passed in Petition No.68/2019 has settled the issue of cost of coal after taking into consideration all relevant facts and circumstances and the regulatory position and has held that in case of any surface transportation cost incurred by GVK beyond the pick-up point within the mine (external STC), payment of same would be limited to the cost incurred to carry the coal to the nearest railway siding and would be as per the rates prescribed by M/s Bharat Coking Coal Ltd.(BCCL) upto 20KM and the said charges would be charged with separate bills for cost of coal and STC.As regards the costs claimed by GVK towards expenditure incurred on account of testing of coal for GCV, the Commission in the above said Order has clarified that GVK is not to be paid any testing charges and is required to construct its own

testing facility. GVK has challenged the above said findings of the Commission before the Appellate Tribunal in Appeal No.189/2019. However, no stay has been granted by the Appellate Tribunal in favour of GVK and as such, the above findings of the Commission are binding upon GVK. Therefore, in order to claim charges for any external STC, GVK is required to submit separate bills in consonance with the rates prescribed by M/s BCCL as decided by the Commission in the above said Order and GVK is not entitled to any testing charges till it establishes a coal testing facility of its own.

iii) PSPCL submitted that GVK has stated that in Petition No.68/2017, GVK had raised disputes regarding reimbursements of costs incurred by it in procuring coal for its project and after final adjudication of the said Petition vide the Commission's Order dated 06.03.2019, PSPCL has paid a sum total of Rs.470.08 Crore for claims of GVK in FY 2017-18 and Rs.892.23 Crore for FY 2018-19. However, PSPCL is yet to pay balance sums of Rs.106.11 Crore towards claims of FY 2017-18 and Rs.218.27 Crore towards claims of FY 2018-19 which amounts are subject matter of the cross Appeals [being Appeal Nos.189 and 192 of 2019 pending adjudication before the Hon'ble Appellate Tribunal. PSPCL further submitted that GVK has stated that it would approach the Commission to claim modifications in its variable cost as

and when the said Appeals are adjudicated by the Hon'ble Appellate Tribunal.

As no stay has been granted by the Appellate Tribunal in favour of GVK or PSPCL in the above Appeals and as such, presently the above findings of the Commission and amounts already paid by PSPCL are binding upon the parties. PSPCL further submitted that actual energy charges of Rs. 470.51 Crore and Rs.833.80 Crore paid to GVK for FY 2017-18 and FY 2018-19 respectively are in compliance of Order dated 06.03.2019 passed in Petition No.68/2017 and Order dated 27.05.2019 passed in Petition No.1/2018 by the Hon'ble Commission.

PSPCL submitted that GVK has prayed that it be allowed the weighted average cost incurred by it on procurement of secondary fuel being procured from PSU Oil Marketing Companies. In this regards it is submitted that Regulation 37.2 of the Tariff Regulations provides as under:

iv)

"37.2 Initially, the LFC of primary fuel and secondary fuel for tariff determination shall be based on actual weighted average cost of primary fuel and secondary fuel for the preceding three months, and in the absence of landed costs for the preceding three months, LFC shall be based on the latest procurement price of primary fuel and secondary fuel for the generating station." In view of the above, it is submitted that GVK may be allowed the cost of secondary fuel strictly in accordance with the above Regulation.

v) PSPCL submitted that as regards the claim of GVK towards normative auxiliary energy consumption and station heat rate, the same are to be allowed by the Commission in accordance with the norms prescribed under the Regulation 36 of the CERC Tariff Regulations 2014.

vi)

- PSPCL further submitted that as regards the claim for transit and handling charges, the Commission in its Order dated 06.03.2019 has clarified that transit and handling charges for the coal are to be paid to GVK only if it shows actual loss after proper checking and weighment at both loading and project end and the bills included proof of actual loss and states whether actual loss was being billed or the normative loss of 1.0% as per Regulation 40 of the PSERC Tariff Regulations, 2014. The present Petition being completely silent on the actual losses having been faced by it, GVK is only entitled to transit and handling losses as percentage of the quantity of domestic coal being procured by it from Coal India limited as 1.0% percent. However, GVK is required to provide complete details of same. In view thereof, PSPCL submitted that:
  - The normative auxiliary consumption has been considered at 9% by GVK as per Regulation 36(E)(a) of CERC Tariff Regulations 2014, for 58

computation of energy for FY 2017-18 and FY 2018-19;

- The fuel transit & handling loss has been considered by PSPCL as per the Commission's Order dated 06.03.2019 passed in Petition No.68/2017 in the following manner:
  - a) For FY 2017-18 & for FY 2018-2019 (01.04.2018 to 30.09.2018) as 1% or actual transit loss whichever is less.
  - b) For FY 2018-2019 (01.10.2018 to 31.03.2019) as
    0.8 % as per applicable CERC tariff regulation or actual transit loss whichever is less; and

3) The station heat rate for FY 2017-18 and FY 2018-19 has been calculated on the basis of Regulation 36 (c) and (b) of the CERC Tariff Regulation, 2014. The design station heat rate has been multiplied by a factor 1.045 which comes out to be 2321 Kcal/Kwh.

vii) PSPCL submitted that GVK has submitted that GVK and PSPCL have undertaken a reconciliation process pursuant to the Commission's Order dated 06.03.2019 in Petition No.68 of 2019 against which cross Appeals being Nos.189 and 192 of 2019 are pending adjudication before the Hon'ble Appellate Tribunal. GVK has challenged the above said Order dated 06.03.2019 before the Hon'ble Appellate Tribunal in Appeal No.189/2019 on the findings with respect to following issues:

- Calculation of capacity charges after inclusion of 9% normative auxiliary consumption
- Testing charges of coal
- IEGC compensation for backing down power and surface transportation charges
- viii) PSPCL submitted that it has also filed an Appeal against the said Order before the Hon'ble APTEL in Appeal No.
   192/2019 on the following issues:
  - Calculation of GCV on ARB total moisture and

Rebate reversal rate and interest on rebate
 Both the above said Appeals are pending adjudication
 before the Hon'ble Appellate Tribunal where no stay of
 operation of Order dated 06.03.2019 has been granted
 and as such the said Order is binding upon both parties.
 The other issues as decided by the Commission have
 remained unchallenged by either parties and thus the said
 Order to that extent has attained finality and is binding on
 both the parties.

ix) PSPCL submitted that GVK is being paid the Landed Fuel Cost (LFC) based upon the actual procurement price of the fuel based upon original bills submitted by it to PSPCL under the orders of the Commission. As such, there is no occasion for the Commission to determine an energy charge rate for GVK in terms of the formula provided under Regulation 39.4 of the PSERC Tariff Regulations 2014.

- X) PSPCL submitted that GVK in its revised submissions has claimed actual transit and handling losses for calculation of energy charges at the rate of 4.20% and 2.43% for FY 2017-18 and FY 2018-19 respectively. GVK has sought the same on grounds that pursuant to cancellation of the captive coal blocks, coal has been procured by GVK in FY 2017-18 under e-auction mode and in FY 2018-19 coal has been procured under Shakti scheme. Thus, coal has been transported to project on road/rail mode over longer distances by road/rail mode resulting in higher fuel transit and handling losses. The Commission in its Order dated 06.03.2019 in petition 68 of 2017 has already clarified that transit and handling charges for coal would be capped at 1% upto 30.09.2018 and at 0.8% from 01.10.2018 onwards as per applicable PSERC Tariff Regulations amended from time to time. Therefore, GVK has to show the actual transit and handling loss including proof of actual loss which would be capped as per order dated 06.03.2019 in petition 68 of 2017.
- xi) PSPCL vide Memo No. 5458/TR-5/963 dated 24.07.2020 submitted the details of payments made to GVK on monthly basis with regards to energy charges including details of fuel cost for true-up of FY 2017-18 and FY 2018-19 with segregation of the payments made to GVK on the basis of weighted average cost of coal to PSPCL's

generating stations and payments on the basis of Shakti coal as well as details of arrears.

### **Commission's Analysis**

- 12.4 The energy charges for FY 2017-18 and FY 2018-19 are payable by PSPCL to GVK in terms of the PPA, Order dated 01.02.2016 common to petition no. 65 of 2013 & 33 of 2015, Order dated 06.03.2019 in petition no. 68 of 2017 and Order dated 27.05.2019 in petition no. 01 of 2018 as applicable. The said Orders are under challenge in Hon'ble APTEL either by GVK or by PSPCL or by both, however there is no stay on the operation of these Orders.
- 13.0 Interest on under-recovered or over-recovered fixed charges:
- 13.1 The Commission notes that the applicability of Regulation 9 of PSERC Regulations, 2005 would be on the distribution companies or generating cum distribution companies and cannot be applied as it is to the standalone generating companies. The Commission observes that Regulation 8 (13) of CERC (Terms and Conditions of Tariff) Regulation, 2014 are squarely applicable to under recovery or over recovery of fixed charges in case of generating companies.
- 13.2 The Regulation 8 (13) of CERC (Terms and Conditions of Tariff) Regulation, 2014 is re-produced below for reference: -

"The amount under-recovered or over-recovered, along with simple interest at the rate equal to the bank rate on 1st April of the respective year, shall be recovered or refunded by the generating company or the transmission licensee, as the case may be, in six equal monthly installments starting within three months from the date of the tariff order issued by the Commission".

13.3 The Commission decides to adopt the CERC Regulations for determining interest equivalent to bank rate on under recovery or over recovery of fixed charges.

Accordingly, interest shall be allowable or recoverable as per Regulation 8 (13) of CERC (Terms and Conditions of Tariff) Regulation, 2014 on under-recovered or overrecovered Annual Fixed Charges (AFC) determined by the Commission.

This Petition stands disposed off accordingly.

	a land a sub-	55
Sd/-	Sd/-	Sd/
(Anjuli Chandra)	(S.S. Sarna)	(Kusumji <mark>t Sid</mark> hu)
Member	Member	Chairperson
Chandigarh Dated:17.09.2020		

# ANNEXURE P-5

Fransaction Details	
Reference ID:	77888492
Transaction Type:	IMPS P2A Payment
Initiator Account:	10912011013160
Transaction Date(dd/MM/yyyy):	05/10/2020
Beneficiary Name:	The Secretary, Punjab State Electricity Regulatory Commission, Chandigarh
Beneficiary Nickname:	PSERC Chandigarh
Beneficiary Account Number:	14300100000034
Beneficiary Bank Name:	INDIAN OVERSEAS BANK
Beneficiary Branch:	CHANDIGARH - SECTOR 32-D
Beneficiary Bank Identifier:	IOBA0001430
Transaction Amount:	INR 10,000.00
Total Charge Amount:	INR 0.00
Total Amount:	INR 10,000.00
Remark:	Court Fee for Review Petition seeking PSERC to review its orders in Petition No 69 of 2017 Dt 05082020
Transaction Status:	Success
External Reference Number:	027914633752